LARGE FILING SEPARATOR SHEET

CASE NUMBER: 06-1363-GA-AGG

FILE DATE: 12/12/2008

SECTION: (Part 2 of 2)

NUMBER OF PAGES: 181

DESCRIPTION OF DOCUMENT: Renewal Application

Item 2. Properties

We do not own any real property. We lease the business premises in the following locations for the stated principal uses:

Location	Approximate Floor Space (Square Feet)	Principal Use
446 Main Street, Worcester, Massachusetts (1)	7,458	Executive office and general administration
1215 19th Street NW, Washington, District of Columbia (2)	400	Branch office
1035 Pearl Street, Boulder, CO (3)	322	Branch office
10001 Woodloch Forest Drive, The Woodlands, TX (4)	2,027	Branch office
208 West 10th Street, Georgetown, TX (5)	400	Branch office
770 East Market Street, West Chester, PA (6)	575	Branch office
4495 Bradenton Avenue, Dublin OH (7)	4,500	Branch office

Note:

(1)	Pursuant to two lease agreements with Sovereign Bank expiring December 31, 2009, at a monthly rate escalating to \$11,931 in 2009, plus operating expenses and taxes.
(2)	Pursuant to a five-year lease agreement with Roosevelt Land, LP expiring July 16, 2011, at a monthly rate of \$5,000, plus operating expenses and taxes.
(3)	Pursuant to a one-year lease agreement with Kabili, LLC expiring March 31, 2008, at a monthly rate of \$1,950, plus certain operating expenses.
(4)	Pursuant to a five-year lease agreement with NNN Waterway Plaza, expiring March 31, 2012, at a monthly rate escalating to \$5,574, plus operating expenses and taxes.
(5)	Pursuant to a one-year lease agreement with Kevin Stotz, expiring May 31, 2008, at a monthly rate of \$400, plus certain operating expenses.
(6)	Pursuant to a two year lease agreement with High Associates LTD expiring December 31, 2009, at a monthly rate escalating to \$970, plus certain operating expenses.

(7) Pursuant to a sixty two-month lease agreement with Rickert Property Management, expiring July 31, 2012, a monthly rate escalating to \$3,750, plus operating expenses and taxes.

Item 3. Legal Proceedings

From time to time we may be a party to various legal proceedings arising in the ordinary course of our business. Our management is not aware of any litigation outstanding, threatened or pending as of the date hereof by or against us or our properties which we believe would be material to our financial condition or results of operations.

Item 4. Submission of Matters to a Vote of Security Holders

Not applicable.

PART II

Item 5. Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities

Our common stock (Symbol "XWE") began trading on the Toronto Stock Exchange ("TSX") on November 16, 2006. Prior to this date, there was no established public trading market for our common stock. Our common stock does not currently trade on a U.S. national market.

The following table sets forth the high and low closing prices per share reported on TSX for the periods shown, commencing with the quarter of our initial public offering on November 16, 2006 (in US \$'s):

2007:	High	Low
First quarter	\$1.00	\$0.82
Second quarter	\$1.53	\$0.82
Third quarter	\$1.77	\$1.08
Fourth quarter 2006:	\$1.31	\$0.55
Fourth quarter	\$1.25	\$0.91

On February 29, 2008, the last reported sale price of our common stock on TSX was CAD\$0.75 per share and there were 141 holders of record of our common stock.

We have never declared or paid any cash dividends on our common stock. We currently intend to retain our future earnings, if any, to finance the expansion of our business and do not expect to pay any dividends in the foreseeable future.

Information regarding our equity compensation plans required by this item is incorporated by reference to the information appearing under the caption "Equity Compensation Plan Information" in our definitive Proxy Statement for the 2008 Annual Meeting of Stockholders.

Use of Proceeds from Registered Securities

We registered shares of our common stock in connection with our initial public offering concurrently in Canada under the terms of a Supplemented Prep Prospectus dated November 9, 2006 and in the United States under the Securities Act. Our Registration Statement on Form S-1 (No. 333-136528) in connection with our initial public offering was declared effective by the SEC on November 9, 2006. As of December 31, 2007, we have used approximately \$10 million of the net proceeds of the initial public offering to redeem our long-term debt, to acquire the assets of Energy Gateway, for working capital and to fund operations. The remaining net proceeds of the initial public offering are invested in short-term, highly liquid money market accounts with an original maturity of 90 days or less. There has been no material change in the planned use of proceeds from our initial public offering as described in our final prospectus filed with the SEC pursuant to Rule 424(b).

Recent Sales of Unregistered Securities

None.

Repurchase of Equity Securities None.

- 20 -

Item 6. Selected Consolidated Financial Data

The following table summarizes our consolidated financial data for the periods presented. You should read the following financial information together with the information under "Management's Discussion and Analysis of Financial Condition and Results of Operations" and our consolidated financial statements and the notes to those consolidated financial statements appearing elsewhere in this annual report. The selected consolidated statements of operations data for the fiscal years ended December 31, 2007, 2006 and 2005, and the selected consolidated balance sheet data as of December 31, 2007 and 2006 are derived from the audited consolidated financial statements, which are included elsewhere in this document. The selected consolidated statements of operations data for the years ended December 31, 2004 and 2003, and the consolidated balance sheet data at December 31, 2005 and 2004 are derived from our audited consolidated financial statements not included in this document. The consolidated balance sheet data as of December 31, 2003 is derived from our unaudited consolidated financial statements not included in this document. Historical results are not necessarily indicative of the results to be expected in future periods.

	For the Years Ended December 31,									
		2007		2006	6 2005			2004		2003
Consolidated Statement of Operations Data:					_		_			
Revenue	\$	9,188,265	\$	5,763,098	\$	4,673,987	5	3,191,660	\$	2,474,699
Cost of revenue		2,874,678		1,166,426		648,410		563,972		872,647
Gross profit		6,313,587		4,596,672		4,025,577		2,627,688		1,602,052
Operating expenses:			-							
Sales and marketing		8,598,256		3,227,263		2,649,786		1,814,799		1,781,173
General and administrative		5,858,810		1,862,450		995,703		710,462		557,910
Total operating expenses		14,457,066		5,089,713		3,645,489		2,525,261		2,339,083
Operating income (loss)		(8,143,479)		(493,041)	_	380,088		102,427		(737,031)
Other income (expense), net(1)		563,294		(312,280)		(86,838)		960,524		(180,738)
Income (loss) before income taxes		(7,580,185)		(805,321)		293,250		1,062,951		(917,769)
Income tax (expense) benefit		(1,061,720)		304,228		754,000				
Net income (loss)		(8,641,905)		(501,093)		1,047,250		1,062,951		(917,769)
Accretion of preferred stock issuance										
costs			_	(6,299)	_	(7,199)		(7,199)	_	(7,199)
Net income (loss) available to										
common stockholders	\$	<u>(8,641,905</u>)	\$	(507,392)	<u>\$</u>	<u>1,040,051</u>	\$	1,055,752	\$	<u>(924,968</u>)
Net income (loss) available to common stockholders per share:										
Basic Voting	\$	(0.11)	\$	(0.01)	\$	0.02	\$	0.02	\$	(0.03)
Basic Non-Voting	\$	***	\$		\$	0.02	\$	0.02	\$	(0.03)
Diluted Voting and Non-Voting	\$	(0.11)	\$	(0.01)	\$	0.02	\$	0.02	\$	(0.03)
Weighted average shares outstanding —										
Basic:		70 702 600		46 696 499		22.040.472		22.059.750		20 105 100
Voting Common Stock Non-Voting Common Stock		79,793,590		45,576,477		33,049,472		32,058,759		30,105,188 339,726
—	_	70 702 600	_	45 696 499		6,778,327		2,880,592	_	
Total Common Stock-Basic	_	79,793,590		45,576,477		39,827,799	_	34,939,351	-	30,444,914
Weighted Average Shares Outstanding — Diluted:		79,793,590		45,576,477	_	54,506,566		52,096,206		30,444,914

(1) Other income (expense) for the year ended December 31, 2004 includes a gain of \$1,062,775 from extinguishment of debt.

	As of December 31,									
		2007		2006		2005		2004		2003
Consolidated Balance Sheet Data:								•		
Cash and cash equivalents	\$	7,001,884	\$	17,483,886	\$	1,584,066	\$	49,389	\$	220,796
Working capital		5,323,622		16,639,898		1,372,542		(882,162)		(2,051,277)
Total assets		20,800,565		20,791,381		3,787,842		941,688		934,635
Long-term liabilities		46,222		87,844		1,879,745		150,368		25,565
Series A redeemable convertible				,				-		
preferred stock		_		_		1,501,698		1,494,499		1,487,300
Accumulated deficit		(12,853,272)		(4,211,367)		(3,710,274)		(4,747,524)		(5,820,475)
Total stockholders' equity (deficit)	\$	16,859,799	\$	17,945,002	\$	(938,883)	\$	(2,241,134)	\$	(3,485,026)



Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations

You should read the following discussion and analysis of our financial condition andresults of operations together with our consolidated financial statements and relatednotes appearing elsewhere in this Annual Report on Form 10-K. Some of the information contained inthis discussion and analysis includes forward-looking statements that involve risks and uncertainties. You should review the "Risk Factors" section of this Annual Report for a discussion of important factors that could cause actual results to differ materially from theresults described in or implied by the forward-looking statements contained in thefollowing discussion and analysis.

Overview

World Energy is an energy and environmental commodities brokerage company that has developed online auction platforms, the World Energy Exchange and the World Green Exchange. On the World Energy Exchange, retail energy consumers (commercial, industrial and government, or CIG) and wholesale energy participants (utilities and retail suppliers) in the United States are able to negotiate for the purchase or sale of electricity and other energy resources from competing energy suppliers which have agreed to participate on our auction platform in a given auction. The World Energy Exchange is supplemented with information about market rules, pricing trends, energy consumer usage and load profiles. Our procurement staff uses this auction platform to conduct auctions, analyze results, guide energy consumers through contracting, and track their contracts, sites, accounts and usage history. Although our primary source of revenue is from brokering electricity and natural gas, we adapted our World Energy Exchange auction platform to accommodate the brokering of green power in 2003 (i.e., electricity generated by renewable resources), wholesale electricity in 2004, certain other energy-related products in 2005. In 2007, we created the World Green Exchange based on the World Energy Exchange technology and business process. On the World Green Exchange, buyers and sellers negotiate for the purchase or sale of environmental commodities such as Renewable Energy Certificates, Verified Emissions Reductions and Certified Emissions Reductions.

During 2007, we continued to increase our cost structure to support our growth plans. As of December 31, 2007, we employed 63 full time employees versus 20 as of December 31, 2006. The staffing increases in 2007 were primarily related to the acquisition of substantially all of the assets of EnergyGateway LLC ("EnergyGateway"), which employed eighteen individuals, and the hiring of eleven sales and marketing professionals, eight individuals specifically focused on the green markets and six individuals within our operations and administrative groups. Four of the eleven sales and marketing professionals hired in 2007 are specifically assigned to the wholesale market. These increased employee and related costs were the major factors contributing to our increased cost structure in all functional areas during 2007 as compared to 2006. We expect our operating expenses to rise slightly as we expand our brokerage capabilities in our retail, wholesale and green product lines, including increased sales and marketing efforts and continued expansion of our back office operation. We also expect there to be continued significant compliance costs associated with being a public company reporting in both the United States and Canada.

On June 1, 2007, we acquired substantially all of the assets of EnergyGateway for \$4,951,758 in cash and 5,375,000 of our common shares plus the assumption of certain liabilities. The EnergyGateway operations are included in our financial statements from June 1, 2007. EnergyGateway provided energy procurement and value-added energy services to customers in many major industries in the U.S. and Canada, from large multi-site Fortune 500 industrials to middle-market manufacturing and small commercial operations.

Operations

Revenue

Retail Electricity Transactions

We receive a monthly commission on energy sales contracted through our online auction platform from each energy supplier based on the energy usage generated and transacted between the energy supplier and energy consumer. Our commissions are not based on the retail price for electricity; rather, commissions are based on the energy usage generated and transacted between the energy supplier and energy consumer multiplied by our contractual commission rate. Our contractual commission rate is negotiated with the energy consumer on a procurement-by-procurement basis based on energy consumer specific circumstances, including size of auction, internal effort required to organize and run the respective auction and competitive factors, among others. Once the contractual commission is agreed to with the energy consumer, all energy suppliers participating in the auction agree to that rate. That commission rate remains fixed for the duration of the contractual term regardless of energy usage. Energy consumers provide us with a letter of authority to request their usage history from the local utility. We then use this data to compile a usage profile for that energy consumer that will become the basis for the auction. This data may also be used to estimate revenue on a going forward basis, as noted below. Historically, our revenue and operating results have varied from quarter-to-quarter and are expected to continue to fluctuate in the future. These fluctuations are primarily due to energy usage, particularly electricity, having higher demand in our second and third quarters and lower demand during our fourth and first quarters. In addition, the activity levels on the World Energy Exchange can fluctuate due to a number of factors, including geopolitical events, weather conditions and market prices. To the extent these factors affect the purchasing decisions of energy consumers our future results of operations may be affected.

Contracts between energy suppliers and energy consumers are signed for a variety of term lengths, with a one year contract term being typical for commercial and industrial energy consumers, and government contracts typically having two to three year terms. Our revenue has grown over the last three years through new participants utilizing our World Energy Exchange as well as energy consumers increasing the size or frequency of their transactions on our exchange platform.

We generally do not directly invoice our electricity energy suppliers and, therefore, we report a substantial portion of our receivables as "unbilled." Unbilled accounts receivable represents management's best estimate of energy provided by the energy suppliers to the energy consumers for a specific completed time period at contracted commission rates and is made up of two components. The first component represents energy usage for which we have received actual data from the supplier and/or the utility but for which payment has not been received at the balance sheet date. The majority of our contractual relationships with energy suppliers require them to supply actual usage data to us on a monthly basis and remit payment to us based on that usage. The second component represents energy usage for which we have not received actual data, but for which we have estimated usage.

Retail Natural Gas Transactions

There are two primary fee components to our retail natural gas services - transaction fees and management fees. Transaction fees are billed to and paid by the energy supplier awarded business on the platform. Transaction fees for natural gas and electricity awards are established prior to award and are the same for each supplier. For the majority of our natural gas transactions, we bill the supplier upon the conclusion of the transaction based on the estimated energy volume transacted for the entire award term multiplied by the transaction fee. Management fees are paid by our energy consumers and are generally billed on a monthly basis for services rendered based on terms and conditions included in contractual arrangements. While substantially all of our retail natural gas transactions are accounted in accordance with the retail electricity transaction revenue recognition methodology described above.

Wholesale and Green Transactions

For wholesale energy and environmental commodity transactions, substantially all transaction fees are invoiced upon the conclusion of the auction based on a fixed fee. These revenues are not tied to future energy usage and are recognized upon the completion of the online auction. For reverse auctions where our customers bid for a consumer's business, the fees are paid by the bidder. For forward auctions where a lister is selling energy or credits, the fees are typically paid by the lister. In addition, revenue may not be recognized on certain green transactions until the credits being auctioned have been verified. While substantially all wholesale and green transactions are accounted for in this fashion, a small percentage is accounted for as the wholesale electricity or gas is delivered similar to the retail electricity transaction methodology described above.

Cost of revenue

Cost of revenue consists primarily of:

- salaries, employee benefits and stock-based compensation associated with our auction management services, which are directly
 related to the development and production of the online auction and maintenance of market-related data on our auction platform
 and monthly management fees (our supply desk function);
- amortization of capitalized costs associated with our auction platform and acquired developed technology; and
- · rent, depreciation and other related overhead and facility-related costs.

Sales and marketing

Sales and marketing expenses consist primarily of:

- salaries, employee benefits and stock-based compensation related to sales and marketing personnel;
- third party commission expenses to our channel partners;
- · travel and related expenses;
- · amortization related to customer relationships and contracts;
- · rent, depreciation and other related overhead and facility-related costs; and
- · general marketing costs such as trade shows marketing materials and outsourced services;

General and administrative

General and administrative expenses consist primarily of:

- salaries, employee benefits and stock-based compensation related to general and administrative personnel;
- · accounting, legal, recruiting and other professional fees; and
- · rent, depreciation and other related overhead and facility-related costs.

Interest income (expense), net

Interest income (expense), net consists primarily of:

- interest income in 2007 related to the invested portion of the proceeds from our initial public offering;
- · interest expense in 2007 and 2006 related to capital leases; and
- interest expense in 2006 on our \$2.0 million subordinated note payable to Massachusetts Capital Resource Company, or MCRC, including the accretion of warrants, that was redeemed on December 4, 2006.

Income tax expense

Income tax expense of approximately \$1.1 million for the year ended December 31, 2007 resulted from the application of a full valuation allowance provided against deferred tax assets generated in prior years due to uncertainty regarding the realization of those deferred tax assets, primarily net operating loss carryforwards, in the future.

Results of Operations

The following table sets forth certain items as a percent of revenue for the periods presented:

	For	For the Years Ended December 31,					
	2007	2006	2005				
Revenue	100%	100%	100%				
Cost of revenue	31	20	14				
Gross profit	69	80	86				
Operating expenses:							
Sales and marketing	94	56	57				
General and administrative	64	33	21				
Operating income (loss)	(89)	(9)	8				
Interest income (expense)	7	(5)	(2)				
Income tax benefit (expense)	(12)	5	16				
Net income (loss)	(94)%	<u>(9)%</u>	22%				

Revenue

	For the Ye Decen					
	2007	 2006	Increase			
Revenue	\$ 9,188,265	\$ 5,763,098	\$	3,425,167	59%	

Revenue increased 59% primarily due to increased auction activity in multiple product lines, new customer wins and the acquisition of the assets of EnergyGateway during the year ended December 31, 2007 as compared to the same period in 2006. The revenue increase reflects the growth of our wholesale customer base to 12 in 2007 from 2 in 2006 as well as an increase to 42 channel partners as of December 31, 2007 from 29 as of December 31, 2006. Of those channel partners, 25 had contributed to our revenue by brokering transactions over the exchange during 2007 as compared to 18 during 2006.

Cost of revenue

		For the Years En	ded D	ecember 31	·			
	2	2007			006			
	\$	% of Revenue		\$	% of Revenue	Increase		se
Cost of revenue	\$ 2,874,678	31%	\$	1,166,426	20%	\$	1,708,252	146%

The 146% increase in cost of revenue related to the year ended December 31, 2007 as compared to the same period in 2006 was substantially due to the addition of ten former EnergyGateway employees, an increase in salary and benefit costs associated with nine new supply desk employees and, to a lesser extent, amortization related to developed technology acquired. Cost of revenue as a percent of revenue increased 11% due to the cost increases noted above partially offset by the 59% increase in revenue.

Operating expenses

			For the Years En						
	_	2007			2	006			
		\$	% of Revenue		\$	% of Revenue	_	Increase	
Sales and marketing General and	\$	8,598,256	94%	\$	3,227,263	56%	\$	5,370,993	166%
administrative Total operating	_	5,858,810	64		1,862,450	33		3,996,360	215
expenses	\$	14,457,066	158%	\$	5,089,713	89%	\$	9,367,353	184%

The 166% increase in sales and marketing expense for the year ended December 31, 2007 as compared to the same period in 2006 primarily reflects the addition of eleven new sales and marketing employees, general salary increases and the addition of the former EnergyGateway employees. In addition, the increased costs include amortization related to customer relationships and contracts as well as increases in travel, consulting and marketing expenses. Sales and marketing expense as a percentage of revenue increased 38% due to increased costs discussed above partially offset by the 59% increase in revenue.

The 215% increase in general and administrative expenses related to the year ended December 31, 2007 as compared to the same period in 2006 was primarily due to an increase in legal, audit and other compliance costs, the addition of four corporate employees, the addition of the former EnergyGateway employees and assets and increases in benefits, recruiting, and occupancy related costs. The increase in legal and audit costs were primarily related to increased compliance and infrastructure activities in support of our growth initiatives, including compliance with the Sarbanes-Oxley Act and ongoing public company costs and the acquisition of the assets of EnergyGateway. General and administrative expenses as a percent of revenue increased 31% substantially due to the cost increases noted above, partially offset by the 59% increase in revenue.

Interest income (expense), net

Interest income, net was approximately \$563,000 for the year ended December 31, 2007. Interest expense, net was approximately \$312,000 for the year ended December 31, 2006. Interest income in 2007 was primarily due to interest earned on the invested portion of the proceeds from our initial public offering. Interest expense in 2006 was primarily due to interest paid on our \$2.0 million subordinated note with the MCRC. This note was retired in December 2006.

Income tax provision

We recorded an income tax provision of approximately \$1.1 million for the year ended December 31, 2007, reflecting the application of a full valuation allowance provided against deferred tax assets generated in prior years due to uncertainty regarding the realization of those deferred tax assets, primarily net operating loss carryforwards, in the future. We recorded an income tax benefit of approximately \$304,000 for the year ended December 31, 2006, reflecting an effective federal and



state tax rate of 38% as we determined that it was more likely than not that a portion of our deferred tax assets would be benefited against expected future income.

Net loss

We reported a net loss for the year ended December 31, 2007 of approximately \$8.6 million as compared to a net loss of approximately \$501,000 for the year ended December 31, 2006. The increase in net loss is primarily due to the higher operating expenses discussed above and, to a lesser extent, the income tax provision partially offset by increases in revenue and interest income.

Comparison of the Years Ended December 31, 2006 and 2005

Revenue

	For the Years Ended								
	Decemi	December 31,							
	2006		2005		Increase				
Revenue	\$ 5,763,098	\$	4,673,987	\$	1,089,111	23%			

Revenue increased 23% primarily due to increased energy usage by energy consumers during the year ended December 31, 2006 as compared to the same period in 2005. This increase was primarily due to the addition of new government procurements and the increase to 29 channel partners as of December 31, 2006 from 16 as of December 31, 2005. Of those channel partners, 18 had successfully contributed to our revenue by brokering transactions over the exchange during 2006 as compared to 10 during 2005.

Cost of revenue

	2006			2005				
	\$	% of Revenue	_	\$	% of Revenue		Increa	se
Cost of revenue	\$ 1,166,426	20%	\$	648,410	14%	\$	518,016	80%

The 80% increase in cost of revenue related to the year ended December 31, 2006 as compared to the same period in 2005 was substantially due to an increase in employee and employee-related costs associated with the addition of four new employees and general salary increases. Cost of revenue as a percent of revenue increased 6% primarily due to the cost increases noted above, partially offset by the 23% increase in revenue.

Operating expenses

			For the Years E	÷.				
	_	2006			2	005		
	_	\$	% of Revenue		\$	% of Revenue	 Increase	e
Sales and marketing General and	\$	3,227,263	56%	\$	2,649,786	57%	\$ 577,477	22%
administrative Total operating		1,862,450	33		995,703	21	866,747	87
expenses	\$	5,089,713	89%	\$	3,645,489	78%	\$ 1, 444,224	40%

The 22% increase in sales and marketing expense for the year ended December 31, 2006 as compared to the same period in 2005 was primarily due to an increase in employee and employee-related costs associated with one additional employee, general salary increases and increased commission costs associated with our higher sales activity as well as higher travel expenses. Sales and marketing expense as a percentage of revenue during the period decreased 1% due to the increase in revenue discussed above substantially offset by the increase in employee and employee-related costs.

The 87% increase in general and administrative expenses related to the year ended December 31, 2006 as compared to the same period in 2005 was primarily due to employee-related costs associated with the addition of four new employees, compliance and recruiting costs as we increased staffing in all functional areas. General and administrative expenses as a percent of revenue increased 12% substantially due to the cost increases noted above, partially offset by the 23% increase in revenue.

Other income (expense)

Interest expense increased approximately \$357,000, or 376%, during the year ended December 31, 2006 as compared to the same period in 2005 primarily due to eleven months of interest on the MCRC note in 2006 as compared to two months in 2005. The increase in interest expense was substantially offset by an approximate \$132,000 increase in interest income earned on the invested portion of the proceeds from our initial public offering in November 2006 and, to a lesser extent, proceeds from the MCRC note. In addition, in connection with our initial public offering, we redeemed the MCRC note at face value resulting in non-cash interest expense of approximately \$221,000 due to the accelerated amortization of the balance of unamortized discount related to warrant issuances.



Income tax benefit

We recorded an income tax benefit of approximately \$304,000 for the year ended December 31, 2006 reflecting an expected effective federal and state tax rate of 38% as we determined that it was more likely than not that a portion of our deferred tax assets would be benefited against expected future income. At December 31, 2006, we had gross deferred tax assets of approximately \$1.7 million, against which a valuation allowance of approximately \$639,000 had been applied. Gross deferred tax liabilities of approximately \$266,000 were also applied against the net deferred tax asset. A valuation allowance was provided against the tax benefit generated during this period due to uncertainty regarding the realization of the net operating loss in the future.

We recorded an income tax benefit of \$754,000 for the year ended December 31, 2005 as we determined it was more likely than not that a portion of our deferred tax assets would be benefited against future income.

Net income (loss)

We reported a net loss for the year ended December 31, 2006 of approximately \$501,000 compared to net income of approximately \$1,047,000 in 2005. The net loss in 2006 versus net income in 2005 is primarily due to the increases in operating expenses and net interest expense discussed above, which were partially offset by the 23% increase in revenue for the year ended December 31, 2006 as compared to 2005.

Liquidity and Capital Resources

In November 2005, we received \$2.0 million in exchange for a subordinated note with the MCRC, which bore interest at the rate of 10% per annum, and required quarterly interest payments beginning on December 31, 2005. On December 4, 2006, we exercised our option to redeem the MCRC note without premium as a result of the completion of our initial public offering. In connection with the MCRC note redemption, MCRC exercised warrants to purchase 600,000 shares of our common stock. The remaining warrants to purchase 2,400,000 shares of common stock did not vest and subsequently expired due to the successful completion of our initial public offering on November 16, 2006. The net redemption amount was \$1,795,000, including interest earned through the date of redemption. For the years ended December 31, 2006 and 2005, interest expense related to the subordinated debt was \$437,000 and \$29,444, respectively.

On November 16, 2006, we completed our initial public offering of 27,441,064 shares of common stock at a price of CDN\$1.05 (approximately US\$0.93) per share. The offering was comprised of 20,000,000 shares sold by us for gross proceeds to us of CDN\$21,000,000 and 7,441,064 shares sold by certain selling stockholders for gross proceeds to such stockholders of approximately CDN\$7,800,000.

On December 5, 2006, we issued an additional 3,000,000 shares of common stock in accordance with an over-allotment option granted to the underwriters of the initial public offering at a price of CDN\$1.05 (approximately US\$0.92) per share for gross proceeds to the Company of CDN\$3,150,000. Net proceeds to us from the initial public offering and over-allotment option were approximately \$17.5 million, net of offering costs of approximately \$3.6 million. In addition, we granted the underwriters of the initial public offering, warrants to purchase up to 1,150,000 shares of common stock at the public offering price. The warrants are exercisable from the 1-year anniversary of the warrant issuance (November 16th and December 5th for the over allotment) and expire 18 months from that date (May 15, 2009 and June 4, 2009 for the over allotment). The warrants allow for net share settlement and certain of the holders of these warrants have piggy-back registration rights. We valued the warrants at \$274,129 and recorded it as a component of additional paid-in capital.

At December 31, 2007, we had no commitments for material capital expenditures. We have identified and executed against a number of strategic initiatives that we believe are key components of our future growth, including: expanding our community of channel partners, energy consumers and energy suppliers on the exchange; strengthening and extending our long-term relationships with government agencies; entry into other energy-related markets; wholesale transactions with utilities and emerging green credits market; making strategic acquisitions and growing our sales force. We believe that our current management and financial resources are adequate to pursue these initiatives and to maintain our competitive position in both the short and long term. During 2007 we increased our workforce to 63 from 20 at December 31, 2006. At December 31, 2007, we had 29 professionals in our sales and marketing and account management groups, 22 in our supply desk group and 12 in our general and administrative group. While we expect to hire additional professionals as the need and/or opportunities arise, we do not expect to continue our expansion of the workforce at the same rate as 2007.

- 27 -

Comparison of December 31, 2007 to December 31, 2006

	De	cember 31, 2007	December 31, 2006			Increase/(Decrease)		
Cash and cash equivalents	\$	7,001,884	\$	17,483,886	\$	(10,482,002)	(60)%	
Trade accounts receivable		1,878,233		1,702,798		175,435	10	
Days sales outstanding		56		95		(39)	(41)	
Working capital		5,323,622		16,639,898		(11,316,276)	(68)	
Stockholders' equity		16,859,799		17,945,002		(1,085,203)	(6)	

Cash and cash equivalents decreased 60%, primarily due to the acquisition of the EnergyGateway assets and the pre-tax loss for the year. Trade accounts receivable increased 10% due to the acquisition of the EnergyGateway assets and the 59% increase in revenue, both offset by an improvement in days sales outstanding. Days sales outstanding decreased by 39 days primarily due to the addition of EnergyGateway receivables, which are due net 15, the increase in wholesale transactions, which are typically due net 30, and improved collection rates. Revenue from our energy suppliers representing greater than 10% of our revenue decreased from 51% from four energy suppliers to 32% from two energy suppliers for the year ended December 31, 2007 and 2006, respectively. This decrease was directly related to the addition of the EnergyGateway customers.

Working capital (consisting of current assets less current liabilities) decreased approximately 68%, primarily due to the decrease in cash and cash equivalents resulting from the acquisition of the EnergyGateway assets and funding of the pre-tax loss for the year. Stockholders' equity decreased 6% due to the net loss for the period substantially offset by the fair value of the stock issued as part of the acquisition of EnergyGateway assets.

Cash used in operating activities for the year ended December 31, 2007 was approximately \$4.6 million due primarily to the pre-tax loss for the period. Cash used in investing and financing activities for the year ended December 31, 2007 was approximately \$5.8 million, primarily due to the purchase of the EnergyGateway assets and, to a lesser extent, costs incurred in software development and purchases of property and equipment.

Cash provided by operating activities for the year ended December 31, 2006 was approximately \$259,000, due primarily to increases in accounts payable and accrued expenses substantially offset by an increase in accounts receivable, the net loss for the year and deferred taxes for the period. Cash used in investing activities for the year ended December 31, 2006 was approximately \$138,000, primarily due to the purchases of property and equipment and costs incurred in software development. Cash provided by financing activities for year ended December 31, 2006 was approximately \$138,000, primarily due to the purchases of property and equipment and costs incurred in software development. Cash provided by financing activities for year ended December 31, 2006 was approximately \$15.8 million, primarily due to the net proceeds from our initial public offering and the issuance of warrants partially offset by the retirement of long-term debt.

We have historically funded our operations with cash flow from operations and, when required, the issuance of various debt and equity instruments. We have approximately \$7.0 million of cash and cash equivalents and no long term bank debt as of December 31, 2007. We expect to continue to fund our operations from existing cash resources, operating cash flow and, when required, the issuance of various debt and equity instruments. That notwithstanding, we expect that our cash and cash equivalents on hand at December 31, 2007 will meet our working capital requirements at least to December 31, 2008.

Comparison of December 31, 2006 to December 31, 2005

	De	2006 2006	December 31, 2005			Increase/(Decrease)		
Cash and cash equivalents Trade accounts receivable	\$	17,483,886 1,702,798	\$	1,584,066 1,028,807	\$	15,899,820 673,991	1,004%	
Days sales outstanding		95		75		20	27	
Working capital Stockholders' equity (deficit)		16,639,898 17,945,002		1,372,542 (938,883)		15,267,356 18,883,885	1,112 N/A	

Cash and cash equivalents increased approximately \$15.9 million, or 1,004%, due primarily to net proceeds of \$17.5 million from our initial public offering in late 2006. This increase was partially offset by the redemption of the \$2.0 million MCRC note in December 2006 and the net loss for the year ended December 31, 2006. Trade accounts receivable increased 66%, due to the increase in large government procurements. These procurements tend to have longer payment cycles due to:

- · added complexity and volume;
- the inclusion of various federal, state and local agencies that may be covered by the procurement (e.g. housing and prison authorities);
- · varied and extensive internal approval processes; and

• an additional party (e.g., the local utility) involved in the payment cycle.

To the extent we are awarded any large government procurements in the future, we expect to experience short term increases in our trade accounts receivable. We believe that we have adequate working capital to account for these short term fluctuations and that these accounts will not have a significant short-term impact on our future liquidity. We believe these receivables are collectible and the energy consumers and energy suppliers who are parties to these contracts are all large, credit worthy entities.

Working capital (consisting of current assets less current liabilities) increased approximately \$15.3 million, or 1,112%, primarily due to the increase in cash described above. Stockholders' equity increased due to our initial public offering and, to a lesser extent, the automatic conversion of Series A Preferred stock into common stock concurrent with the offering, and the exercise of warrants. These increases in stockholders' equity were partially offset by the net loss for the year ended December 31, 2006.

Cash provided by operating activities for the year ended December 31, 2006 was approximately \$259,000, due primarily to increases in accounts payable and accrued expenses substantially offset by accounts receivable, the net loss for the year and deferred taxes for the period. Cash used in investing activities for the year ended December 31, 2006 was approximately \$138,000, primarily due to the purchases of property and equipment and costs incurred in software development. Cash provided by financing activities for year ended December 31, 2006 was approximately \$138,000, primarily due to the purchases of property and equipment and costs incurred in software development. Cash provided by financing activities for year ended December 31, 2006 was approximately \$15.8 million, primarily due to the net proceeds from our initial public offering and the issuance of warrants partially offset by the retirement of long-term debt.

Cash provided by operating activities for the year ended December 31, 2005 was approximately \$171,000, primarily resulting from net income for the period partially offset by increases in deferred tax assets and working capital. Cash used in investing activities for the year ended December 31, 2005 was approximately \$80,000 primarily related to costs associated with software development and, to a lesser extent, purchase of property and equipment. Cash provided by financing activities for year ended December 31, 2005 was approximately \$1.4 million primarily due to the MCRC note described above less the repayment of the \$500,000 line of credit.

Contractual Obligations and Other Commercial Commitments

The table below summarizes our gross contractual obligations and other commercial commitments as of December 31, 2007. As of December 31, 2007, we did not have any purchase obligations other than our capital and operating leases.

<u>Contractual Obligations</u>	 Payments Due by Period									
	2008	200	9 and 2010	2011	and 2012		13 and ereafter		Total	
Capital leases Operating leases	\$ 47,223 304,537	\$	47,937 447,656	\$	798 178,603	\$		\$	95,958 930,796	
Total contractual obligations	\$ \$ 351,760	\$	495,593	\$	1 79,401	\$		\$	1,026,754	

Critical Accounting Policies

The preparation of financial statements in accordance with accounting principles generally accepted in the United States of America ("generally accepted accounting principles") requires us to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenue and expenses during the reporting period.

The most judgmental estimates affecting our consolidated financial statements are those relating to revenue recognition and the estimate of actual energy purchased from the energy supplier and end user, or energy consumer, of such energy; software development costs; the fair value of our equity securities prior to our initial public offering when there was no ready market for the purchase and sale of these shares; the valuation of intangible assets and goodwill; impairment of long-lived assets; and estimates of future taxable income as it relates to the realization of our net deferred tax assets. We regularly evaluate our estimates and assumptions based upon historical experience and various other factors that we believe to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities that are not readily apparent from other sources. To the extent actual results differ from those estimates, our future results of operations may be affected. We believe the following critical accounting policies affect our more significant judgments and estimates used in the preparation of our consolidated financial statements filed herewith for a description of our accounting policies.

Revenue Recognition

Retail Electricity Transactions

We receive a monthly commission on energy sales contracted through our online auction platform from each energy supplier based on the energy usage generated and transacted between the energy supplier and energy consumer. Our

commissions are not based on the retail price for electricity; rather, commissions are based on the energy usage generated and transacted between the energy supplier and energy consumer multiplied by our contractual commission rate. Revenue from commissions is recognized as earned on a monthly basis over the life of each contract as energy is consumed, provided there is persuasive evidence of an arrangement, the sales price is fixed or determinable, collection of the related receivable is reasonably assured, and customer acceptance criteria, if any, has been successfully demonstrated.

We record brokerage commissions based on actual usage data obtained from the energy supplier for that accounting period, or to the extent actual usage data is not available, based on the estimated amount of electricity and gas delivered to the energy consumers for that accounting period. We develop our estimates on a quarterly basis based on the following criteria:

- · Payments received prior to the issuance of the financial statements;
- Usage updates from energy suppliers;
- Usage data from utilities;
- · Comparable historical usage data; and
- · Historical variances to previous estimates.

To the extent usage data cannot be obtained, we estimate revenue as follows:

- Historical usage data obtained from the energy consumer in conjunction with the execution of the auction;
- · Geographic/utility usage patterns based on actual data received;
- · Analysis of prior year usage patterns; and
- · Specific review of individual energy supplier/location accounts.

In addition, we perform sensitivity analyses on this estimated data based on overall industry trends including prevailing weather and usage data. Once the actual data is received, we adjust the estimated accounts receivable and revenue to the actual total amount in the period during which the payment is received. Based on management's current capacity to obtain actual energy usage, we currently estimate 4 to 6 weeks of revenue at the end of our accounting period. Differences between estimated and actual revenue have been within management's expectations and have not been material to date.

We generally do not directly invoice our electricity energy suppliers and, therefore, we report a substantial portion of our receivables as "unbilled." Unbilled accounts receivable represents management's best estimate of energy provided by the energy suppliers to the energy consumers for a specific completed time period at contracted commission rates and is made up of two components. The first component represents energy usage for which we have received actual data from the supplier and/or the utility but for which payment has not been received at the balance sheet date. The majority of our contractual relationships with energy suppliers require them to supply actual usage data to us on a monthly basis and remit payment to us based on that usage. The second component represents energy usage for which we have not received actual data, but for which we have estimated usage. Commissions paid in advance by certain energy suppliers are recorded as deferred revenue and amortized to commission revenue on a monthly basis on the energy exchanged that month.

Retail Natural Gas Transactions

There are two primary fee components to our retail natural gas services - transaction fees and management fees. Transaction fees are billed to and paid by the energy supplier awarded business on the platform. Transaction fees for natural gas awards are established prior to award and are the same for each supplier. For the majority of our natural gas transactions, we bill the supplier upon the conclusion of the transaction based on the estimated energy volume transacted for the entire award term multiplied by the transaction fee. Management fees are paid by our energy consumers and are generally billed on a monthly basis for services rendered based on terms and conditions included in contractual arrangements. While substantially all of our retail natural gas transactions are accounted in accordance with this policy, a certain percentage is accounted for as the natural gas is consumed by the customer and recognized as revenue in accordance with the retail electricity transaction revenue recognition methodology described above.

Wholesale and Green Transactions

Substantially all transaction fees are invoiced upon the conclusion of the auction based on a fixed fee. These revenues are not tied to future energy usage and are recognized upon the completion of the online auction. For reverse auctions where our customers bid for a consumer's business, the fees are paid by the bidder. For forward auctions where a lister is selling energy or credits, the fees are typically paid by the lister. In addition, revenue may not be recognized on certain green transactions until the credits being auctioned have been verified. While substantially all wholesale and green transactions are accounted for in this fashion, a small percentage is accounted for as the wholesale electricity or gas is delivered similar to the retail electricity transaction revenue recognition methodology described above.

Channel Partner Commissions

We pay commissions to our channel partners at contractual rates based on monthly energy transactions between energy suppliers and energy consumers. The commission is accrued monthly and charged to sales and marketing expense as revenue is recognized. We pay commissions to our salespeople at contractual commission rates based upon cash collections from our customers.

Revenue Estimation

Our estimates in relation to revenue recognition affect revenue and sales and marketing expense as reflected on our statements of operations, and trade accounts receivable and accrued commissions accounts as reflected on our balance sheets. For any quarterly reporting period, we may not have actual usage data for certain energy suppliers and will need to estimate revenue. We record revenue based on the energy consumers' historical usage profile. At the end of each reporting period, we adjust historical revenue to reflect actual usage for the period. For the year ended December 31, 2007, we estimated usage for approximately 10% of our revenue resulting in an approximate 0.4%, or \$35,179, adjustment. This increase in revenue, which is based on adjusted year to date revenue for the quarter, resulted in an approximate 0.1% reduction in sales and marketing expense related to third party commission expense associated with those revenues. Corresponding adjustments were made to unbilled accounts receivable and accrued commissions, respectively. A 1% difference between this estimate and actual usage would have an approximate \$9,000 effect on our revenue for the year ended December 31, 2007.

Software Development

Certain acquired software and significant enhancements to our software are recorded in accordance with Statement of Position ("SOP") 98-1, "Accounting for Costsof Computer Software Developed or Obtained for Internal Use". Accordingly, internally developed software costs of approximately \$469,000, \$66,000, and \$54,000 related to implementation, coding and configuration have been capitalized in 2007, 2006, and 2005, respectively. We amortize internally developed and purchased software over the estimated useful life of the software (generally three years). Accordingly, during 2007, 2006, and 2005, approximately \$111,000, \$73,000, and \$55,000, respectively, were amortized to cost of revenues resulting in accumulated amortization of approximately \$273,000, \$162,000, and \$89,000 at December 31, 2007, 2006, and 2005, respectively.

Our estimates for capitalization of software development costs affect cost of revenue and capitalized software as reflected on our consolidated statements of operations and on our consolidated balance sheets. During the year ended December 31, 2007, we capitalized approximately \$469,000, or 2.3% of our total assets. Amortization expense related to capitalized software for the year ended December 31, 2007 was approximately 3.8% of cost of revenue. To the extent the carrying amount of the capitalized software costs may not be fully recoverable or that the useful lives of those assets are no longer appropriate, we may need to record an impairment (non-cash) charge and write-off a portion or all of the capitalized software balance on the balance sheet.

Goodwill

We use assumptions in establishing the carrying value and fair value of our goodwill. Goodwill represents the excess of the purchase price over the fair value of identifiable net assets of acquired businesses. We account for goodwill that results from acquired businesses in accordance with Statement of Financial Accounting Standards ("SFAS") No. 142, "Goodwill and Other Intangible Assets" ("SFAS No. 142"). Under SFAS No. 142, goodwill and intangible assets having indefinite lives are not amortized but instead are assigned to reporting units and tested for impairment annually or more frequently if changes in circumstances or the occurrence of events indicate possible impairment.

Pursuant to SFAS No. 142, we perform an annual impairment test for goodwill. The impairment test for goodwill is a two-step process. Step one consists of a comparison of the fair value of a reporting unit with its carrying amount, including the goodwill allocated to each reporting unit. If the carrying amount is in excess of the fair value, step two requires the

comparison of the implied fair value of the reporting unit whereby the carrying amount of the reporting unit's goodwill over the implied fair value of the reporting unit's goodwill will be recorded as an impairment loss. There was no impairment of goodwill in the year ended December 31, 2007. Future events could cause management to conclude that impairment indicators exist and that goodwill of previously acquired businesses is impaired. Any related impairment loss could have a material impact on our financial condition and results of operations.

Intangible Assets

We use assumptions in establishing the carrying value, fair value and estimated lives of our intangible assets. The criteria used for these assumptions include management's estimate of the assets continuing ability to generate positive income from operations and positive cash flow in future periods compared to the carrying value of the asset, as well as the strategic significance of any identifiable intangible asset in our business objectives. If assets are considered impaired, the impairment recognized is the amount by which the carrying value of the assets exceeds the fair value of the assets. Useful lives and related amortization expense are based on our estimate of the period that the assets will generate revenues or otherwise be used by us. Factors that would influence the likelihood of a material change in our reported results include significant changes in the asset's ability to generate positive cash flow, a significant decline in the economic and competitive environment on which the asset depends and significant changes in our strategic business objectives.

Intangible assets consist of customer relationships and contracts, purchased technology and other intangibles, and are stated at cost less accumulated amortization. Intangible assets with a definite life are amortized using the straight-line method over their estimated useful lives, which range from one to ten years.

Impairment of Long-Lived and Intangible Assets

In accordance with SFAS No. 144, "Accounting for the Impairment or Disposal of Long-Lived Assets", we periodically review long-lived assets and intangible assets for impairment whenever events or changes in circumstances indicate that the carrying amount of such assets may not be recoverable or that the useful lives of those assets are no longer appropriate. Recoverability of these assets is determined by comparing the forecasted undiscounted net cash flows of the operation to which the assets relate to the carrying amount. There was no impairment of intangible assets in the year ended December 31, 2007.

Income Taxes

We account for income taxes in accordance with SFAS No. 109, "Accounting for IncomeTaxes" ("SFAS No. 109"). Deferred tax assets and liabilities are determined at the end of each period based on the future tax consequences that can be attributed to net operating loss carryforwards, as well as differences between the financial statement carrying amounts of the existing assets and liabilities and their respective tax basis. Deferred income tax expense or credits are based on changes in the asset or liability from period to period. Valuation allowances are provided if, based on the weight of available evidence, it is more likely than not that some or all of the deferred tax assets will not be realized. The realization of deferred tax assets is dependent upon the generation of future taxable income. In determining the valuation allowance, we consider past performance, expected future taxable income, and qualitative factors which we consider to be appropriate in estimating future taxable income. Our forecast of expected future taxable income is for future periods that can be reasonably estimated. We have historically utilized a three year forecasting window to determine our expected future taxable income. As of December 31, 2006, we had cumulative net income for the three years then ended and the expected future taxable income anticipated from 2007 through 2009 was deemed adequate on a present value basis to benefit the majority of our deferred tax assets (primarily net operating loss carryforwards). During 2007, we accelerated our growth in excess of our initial expectations to address the strategic initiatives outlined at the time of our initial public offering. We grew from 20 employees as of December 31, 2006 to 63 as of December 31, 2007, acquired one of our main competitors in the retail energy market, and developed our wholesale and environmental commodity product lines. These initiatives increased our loss for 2007 resulting in a three-year cumulative loss as of December 31, 2007 and increased our near term operating expenses. A three-year cumulative loss is significant negative evidence in considering whether the deferred tax assets are realizable and generally precludes relying on projections of future taxable income to support the recovery of deferred tax assets. While we expect our future income will be adequate on a present value basis to benefit substantially all of our current deferred tax assets, the recent three-year cumulative loss is given more weight than projected future income when determining the need for a valuation. Therefore, during the fourth quarter of 2007, we recorded a full valuation allowance against its deferred tax assets of approximately \$4.1 million, and maintain a cumulative valuation allowance of approximately \$4.7 million at December 31, 2007.

Our estimates in relation to income taxes affect income tax benefit and deferred tax assets as reflected on our statements of operations and balance sheets, respectively. The deferred tax assets are reduced by a valuation allowance if it is more

likely than not that the tax benefits will not be realized in the near term. As of December 31, 2007, we had deferred tax assets of approximately \$4.7 million against which a full valuation allowance has been established. To the extent we determine that it is more likely than not that we will recover all of our deferred tax assets, it could result in an approximate \$4.7 million non-cash tax benefit.

We adopted Financial Accounting Standards ("FASB") Interpretation No. 48, "Accounting for Uncertainty in Income Taxes" ("FIN 48"), on January 1, 2007. FIN 48 clarifies the accounting for uncertainty in income taxes recognized in an enterprise's financial statements in accordance with SFAS 109. FIN 48 prescribes a two-step process to determine the amount of tax benefit to recognize. First, the tax position must be evaluated to determine the likelihood that it will be sustained upon examination by a tax authority. If the tax position is deemed "more-likely-than-not" to be sustained, the tax position is the largest amount that has a greater than 50 percent likelihood of being realized upon ultimate settlement. If the tax position does not meet the "more-likely-than-not" threshold, then it is not recognized in the financial statements. As of our adoption date of January 1, 2007, and also at December 31, 2007, FIN 48 did not have a material impact on our financial position or results of operations.

We have reviewed the tax positions taken, or to be taken, in our tax returns for all tax years currently open to examination by the taxing authorities in accordance with the recognition and measurement standards of FIN 48. At December 31, 2007, there is no expected material, aggregate tax effect as a result of differences between tax return positions and the benefits recognized in our financial statements.

Share-Based Compensation

In accordance with SFAS No. 123(R), we recognize the compensation cost of share-based awards on a straight-line basis over the requisite service period of the award. In 2007, share-based awards consisted of grants of stock options and restricted stock. The restrictions on the restricted stock lapse over the vesting period. The vesting period of restricted stock is determined by the Board of Directors, and is generally four years for employees.

The per-share weighted-average fair value of stock options granted during the year ended December 31, 2007 and 2006 was \$0.58 and \$0.36, respectively, on the date of grant, using the Black-Scholes option-pricing model with the following weighted-average assumptions and estimated forfeiture rates of 13% in 2007 and 2006, respectively:

Year ended December 31,	Expected Dividend Yield	Risk Interest Rate	Expected Option Life	Expected Volatility
2007		4.66%	4.61 years	54%
2006		4.66%	4.75 years	48%

Prior to January 1, 2006, we accounted for our stock-based awards to employees using the intrinsic value method prescribed in APB 25 and related interpretations. Under the intrinsic value method, compensation expense is measured on the date of grant as the difference between the deemed fair value of the company's common stock and the option exercise price multiplied by the number of options granted. We provided the disclosures as required by SFAS No. 148, "Accountingfor Stock-Based Compensation and Disclosure, an Amendment of FASB StatementNo. 123".

A summary of options outstanding and options exercisable as of December 31, 2007 is as follows:

	0	ptions Outstand	ling		Options Exercisable					
		Weighted				Weighted				
		Average				Average				
Range of		Remaining		Aggregate	Number	Remaining		Aggregate		
Exercise		Contractual		Intrinsic	Of Shares	Contractual		Intrinsic		
Prices	Options	Life		Value	Exercisable	Life		Value		
\$ 0.02 - \$0.23	2,651,029	2.82 Years	\$	2,081,058	2,646,029	2.82 Years	\$	2,077,133		
\$ 0.24 - \$ 1.12	1,612,000	4.78 Years		391,300	670,625	4.87 Years		229,263		
\$ 1.13 - \$ 1.34	2,440,000	6.36 Years			30,000	5.96 Years				
	6,703,029	4.58 Years	\$	2,472,358	3,346,654	3.26 Years	\$	2,306,396		

The aggregate intrinsic value in the table above represents the total intrinsic value of our outstanding options and exercisable options as of December 31, 2007. The aggregate intrinsic value of options exercised during the year ended December 31, 2007 was approximately \$126,000.

A summary of restricted stock activity for the year ended December 31, 2007 is as follows:

	Shares	Weighted Average Grant Price
Outstanding at December 31, 2006		\$
Granted	1,928,500	\$ 1.13
Cancelled	(80,000)	\$ 1.21
Vested	(400,000)	\$ 1.21
Unvested at December 31, 2007	1,448,500	\$ 1.10

Recent Accounting Pronouncements

In December 2007, the Financial Accounting Standards Board ("FASB") issued SFAS No. 141 (revised 2007), "Business Combinations" ("SFAS No. 141(R)"). SFAS No. 141(R) requires an acquiring entity in a business combination to recognize all (and only) the assets acquired and liabilities assumed in the transaction; establishes the acquisition-date fair value as the measurement objective for all assets acquired and liabilities assumed; and requires the acquirer to disclose to investors and other users all of the information they need to evaluate and understand the nature and financial effect of the business combination. SFAS No. 141(R) is effective for fiscal years beginning after December 15, 2008. We expect to adopt SFAS No. 141(R) on January 1, 2009.

In December 2007, the FASB issued SFAS No. 160, "Noncontrolling Interests in Consolidated Financial Statements" ("SFAS No. 160"). SFAS No. 160 clarifies that a noncontrolling or minority interest in a subsidiary is considered an ownership interest and, accordingly, requires all entities to report such interests in subsidiaries as equity in the consolidated financial statements. SFAS No. 160 is effective for fiscal years beginning after December 15, 2008. We expect to adopt SFAS No. 160 on January 1, 2009.

In September 2006, the FASB issued SFAS No. 157, "Fair Value Measurements" ("SFAS No. 157"). This standard defines fair value, establishes a framework for measuring fair value in accounting principles generally accepted in the United States of America, and expands disclosure about fair value measurements. This pronouncement applies to other accounting standards that require or permit fair value measurements. Accordingly, this statement does not require any new fair value measurement. This statement is effective for fiscal years beginning after November 15, 2007, and interim periods within those fiscal years. In December of 2007, the FASB agreed to a one year deferral of SFAS No. 157's fair value measurement requirements for nonfinancial assets and liabilities that are not required or permitted to be measured at fair value on a recurring basis. Management is currently evaluating the requirements of SFAS No. 157, but does not expect it to have a material impact on our 2008 consolidated financial statements.

In February 2007, the FASB issued SFAS No. 159, "The Fair Value Option for Financial Assets and Financial Liabilities" ("SFAS No. 159"). SFAS No. 159 allows entities the option to measure eligible financial instruments at fair value as of specified dates. Such election, which may be applied on an instrument by instrument basis, is typically irrevocable once elected. SFAS No. 159 is effective for fiscal years beginning after November 15, 2007. We adopted SFAS No. 159 on January 1, 2008, but have not yet elected the fair value option for any items permitted under SFAS No. 159.

Seasonality

Our revenue is subject to seasonality and fluctuations during the year primarily as a result of weather conditions and its impact on the demand for energy. Our revenue is generated from the commissions we receive under any given energy contract, which is tied to the energy consumer's consumption of energy. Therefore, revenue from natural gas consumption tends to be strongest during the winter months due to the increase in heating usage, and revenue from electricity consumption tends to be strongest during the summer months due to the increase in air conditioning usage. Our revenue is also subject to fluctuations within any given season, depending on the severity of weather conditions — during a particularly cold winter or an unseasonably warm summer, energy consumption will rise.

Cyclicality

We believe that our business will be cyclical in nature and is tied, in part, to market energy prices which impact transaction volume. When energy prices increase in competitive markets above the price levels of the regulated utilities, energy consumers are less likely to lock-in to higher fixed price contracts in the competitive markets and so they are less likely to use our auction platform. Conversely, when energy prices decrease in competitive markets below the price levels of the regulated utilities, energy consumers are more likely to lock-in to lower fixed price contracts in the competitive markets and so they are more likely to use our auction platform. Although our short term revenue is impacted by usage trends, these cyclical effects will also have longer term implications on our business because we derive future revenue from current auctions.

Item 7A. Ouantitative and Oualitative Disclosures about Market Risk

Market Risk

Market risk is the potential loss arising from adverse changes in market rates and prices, such as foreign currency rates, interest rates, and other relevant market rates or price changes. In the ordinary course of business, we are exposed to market risk resulting from changes in foreign currency exchange rates, and we regularly evaluate our exposure to such changes. Our overall risk management strategy seeks to balance the magnitude of the exposure and the costs and availability of appropriate financial instruments.

Impact of Inflation and Changing Prices

Historically, our business has not been materially impacted by inflation. We provide our service at the inception of the service contract between the energy supplier and energy consumer. Our fee is set as a fixed dollar amount per unit of measure and fluctuates with changes in energy demand over the contract period.

Foreign Currency Fluctuation

Our commission revenue is primarily denominated in U.S. dollars. Therefore, we are not directly affected by foreign exchange fluctuations on our current orders. However, fluctuations in foreign exchange rates do have an effect on energy consumers' access to U.S. dollars and on pricing competition. We have entered into non-U.S. dollar contracts but they have not had a material impact on our operations. We do not believe that foreign exchange fluctuations will materially affect our results of operations.

Item 8. Financial Statements and Supplementary Data

The consolidated financial statements listed in Item 15(a) are incorporated herein by reference and are filed as a part of this report and follow the signature pages to this Annual Report on Form 10-K on page 42.

Item 9. Changes in and Disagreements with Accountants on Accounting and Financial Disclosure

None.

Item 9A. Controls and Procedures

Evaluation of Disclosure Controls and Procedures

As required by Rule 13a-15 under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), the Company carried out an evaluation under the supervision and with the participation of the Company's management, including the Chief Executive Officer and Chief Financial Officer, of the effectiveness of the Company's disclosure controls and procedures as of December 31, 2007. In designing and evaluating the Company's disclosure controls and procedures, the Company and its management recognize that there are inherent limitations to the effectiveness of any system of disclosure controls and procedures, including the possibility of human error and the circumvention or overriding of the controls and procedures. Accordingly, even effective disclosure controls and procedures can only provide reasonable assurance of achieving their desired control objectives. Additionally, in evaluating and implementing possible controls and procedures, the Company's management was required to apply its reasonable judgment. Based upon the required evaluation, the Chief Executive Officer and the Chief Financial Officer concluded that as of December 31, 2007, the Company's disclosure controls and procedures were effective at the reasonable assurance level to ensure that information required to be disclosed by the Company in the reports it files or submits under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the Securities and Exchange Commission's rules and forms.

Internal Control Over Financial Reporting

a) Management's Report on Internal Control Over Financial Reporting

Our management is responsible for establishing and maintaining adequate internal control over financial reporting for the Company, as such term is defined in Rule 13a-15(f) under the Exchange Act. Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate. Our management assessed the effectiveness of our internal control over financial reporting as of December 31, 2007. In making this assessment, the Company's management used the criteria set forth by the Committee of Sponsoring Organizations of the Treadway Commission (COSO) in Internal Control-Integrated Framework.

Management has excluded the assets and business lines of EnergyGateway from its assessment of internal control over financial reporting as of December 31, 2007 because the assets of this entity were acquired by the Company in 2007. The EnergyGateway assets represent 5% of our consolidated total assets and 20% of consolidated revenue as of and for the year ended December 31, 2007.

Based on this assessment, our management believes that, as of December 31, 2007, our internal control over financial reporting is effective based on those criteria.

Our independent registered public accounting firm has issued an audit report on our management's assessment of our internal control over financial reporting. This report appears below.

b) Attestation Report of the Independent Registered Public Accounting Firm

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Board of Directors and Stockholders World Energy Solutions, Inc. Worcester, Massachusetts

We have audited World Energy Solutions, Inc.'s ("the Company") internal control over financial reporting as of December 31, 2007, based on criteria established in *Internal Control—Integrated Framework* issued by the Committee of Sponsoring Organizations of the Treadway Commission (COSO). The Company's management is responsible for maintaining effective internal control over financial reporting and for its assessment of the effectiveness of internal control over financial reporting, included in Part II, Item 9A of this Form 10-K. Our responsibility is to express an opinion on the Company's internal control over financial reporting based on our audit.

We conducted our audit in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether effective internal control over financial reporting was maintained in all material respects. Our audit included obtaining an understanding of internal control over financial reporting, assessing the risk that a material weakness exists, testing and evaluating the design and operating effectiveness of internal control, and performing such other procedures as we considered necessary in the circumstances. We believe that our audit provides a reasonable basis for our opinion.

A company's internal control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with accounting principles generally accepted in the United States of America. A company's internal control over financial reporting includes those policies and procedures that (1) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company; (2) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with accounting principles generally accepted in the United States of America, and that receipts and expenditures of the company are being made only in accordance with authorizations of management and directors of the company; and (3) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use, or disposition of the company's assets that could have a material effect on the financial statements.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

In our opinion, World Energy Solutions, Inc. maintained, in all material respects, effective internal control over financial reporting as of December 31, 2007, based on criteria established in Internal Control-Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway Commission.

We have also audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States), the consolidated balance sheet of World Energy Solutions, Inc. as of December 31, 2007, and the related consolidated statements of operations, stockholders' equity (deficit) and cash flows for the year then ended, and our report dated March 12, 2008, expressed an unqualified opinion on those consolidated financial statements.

As described in Management's Report on Internal Control over Financial Reporting, management excluded Energy Gateway LLC from its assessment of internal control over financial reporting as of December 31, 2007, which is included in the 2007 consolidated financial statements of World Energy Solutions, Inc. and constituted 5% of the consolidated assets and 20% of the consolidated revenue as of and for the year ended December 31, 2007. Our audit of internal control over financial reporting of World Energy Solutions, Inc. also did not include an evaluation of the internal control over financial reporting of Energy Gateway LLC.

/S/ UHY LLP

Boston, Massachusetts March 12, 2008 c) Changes in Internal Control Over Financial Reporting

No change in our internal control over financial reporting occurred during the fiscal quarter ended December 31, 2007 that has materially affected, or is reasonably likely to materially affect, our internal control over financial reporting.

Item 9B. Other Information

Not applicable.

PART III

Item 10. Directors, Executive Officers and Corporate Governance

The information required to be disclosed by this item 10 is hereby incorporated by reference to the Company's definitive proxy statement to be filed by the Company within 120 days of the close of its fiscal year.

We have adopted a code of business conduct and ethics applicable to all of our directors and employees. The code of business conduct and ethics is available on the corporate governance section of "Investor Relations" of our website <u>www.worldenergy.com</u>.

Any waiver of the code of business conduct and ethics for directors or executive officers, or any amendment to the code that applies to directors or executive officers, may only be made by the board of directors. We intend to satisfy the disclosure requirement under Item 5.05 of Form 8-K regarding an amendment to, or waiver from, a provision of this code of ethics by posting such information on our website, at the address and location specified above. To date, no such waivers have been requested or granted.

Item 11. Executive Compensation

The information required to be disclosed by this item 11 is hereby incorporated by reference to the Company's definitive proxy statement to be filed by the Company within 120 days after the close of its fiscal year.

Item 12. Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters

The information required to be disclosed by this item 12 is hereby incorporated by reference to the Company's definitive proxy statement to be filed by the Company within 120 days after the close of its fiscal year.

Item 13. Certain Relationships and Related Transactions and Director Independence

The information required to be disclosed by this item 13 is hereby incorporated by reference to the Company's definitive proxy statement to be filed by the Company within 120 days after the close of its fiscal year.

Item 14. Principal Accountant Fees and Services

The information required to be disclosed by this item 14 is hereby incorporated by reference to the Company's definitive proxy statement to be filed by the Company within 120 days after the close of its fiscal year.

PART IV

Item 15. Exhibits and Financial Statement Schedules

(a)(1) Financial Statements

For a list of the financial information included herein, see "Index to Consolidated Financial Statements" on page 41 of this Annual Report on Form 10-K.

(a)(2) Financial Statements Schedules

All schedules are omitted because they are not applicable or the required information is included in the financial statements or notes thereto.

(a)(3) Exhibits

The list of exhibits filed as a part of this Annual Report on Form 10-K is set forth on the Exhibit Index immediately preceding the exhibits hereto and is incorporated herein by reference.

- 39 -

SIGNATURES

In accordance with Section 13 or 15(d) of the Exchange Act, the registrant caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

WORLD ENERGY SOLUTIONS, INC.

March 13, 2008

By: /s/ Richard Domaleski **Richard** Domaleski Chief Executive Officer

In accordance with the Exchange Act, this report has been signed below by the following persons on behalf of the registrant and in the capacities and on the dates indicated.

Signature	Title	Date
/s/ Richard Domaleski	Chief Executive Officer and Director	March 13, 2008
Richard Domaleski		
/s/ James Parslow	Chief Financial Officer	March 13, 2008
James Parslow		
/s/ Edward Libbey	Chairman of the Board and Director	March 13, 2008
Edward Libbey		
/s/ Patrick Bischoff	Director	March 13, 2008
Patrick Bischoff		
/s/ John Wellard	Director	March 13, 2008
John Wellard		
/s/ Thad Wolfe	Director	March 13, 2008

Thad Wolfe

SUPPLEMENTAL INFORMATION TO BE FURNISHED WITH REPORTS FILED PURSUANT TO SECTION 15(d) OF THE EXCHANGE ACT BY REGISTRANTS WHICH HAVE NOT REGISTERED SECURITIES PURSUANT TO SECTION 12 OF THE EXCHANGE ACT

No annual report to security holders covering the Company's prior fiscal year or proxy materials have been furnished to security holders which have not been furnished to the Securities and Exchange Commission. The Company will furnish to the Securities and Exchange Commission any annual report to security holders covering the Company's prior fiscal year and any proxy materials to be furnished to security holders subsequent to the filing of this annual report.

INDEX TO CONSOLIDATED FINANCIAL STATEMENTS

	Page
REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM	42
CONSOLIDATED FINANCIAL STATEMENTS:	
Consolidated Balance Sheets for World Energy Solutions, Inc. at December 31, 2007 and December 31, 2006 Consolidated Statements of Operations for World Energy Solutions, Inc. for the years ended December 31, 2007, 2006 and	43
2005	44
Consolidated Statements of Stockholders' Equity (Deficit) for World Energy Solutions, Inc. for the years ended December 31, 2007, 2006 and 2005	45
Consolidated Statements of Cash Flows for World Energy Solutions, Inc. for the years ended December 31, 2007, 2006 and 2005	46
Notes to Consolidated Financial Statements	47

- 41 -

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Board of Directors and Stockholders World Energy Solutions, Inc. Worcester, Massachusetts

We have audited the accompanying consolidated balance sheets of World Energy Solutions, Inc. (the "Company") as of December 31, 2007 and 2006, and the related consolidated statements of operations, stockholders' equity (deficit), and cash flows for each of the three years in the period ended December 31, 2007. These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the consolidated financial position of World Energy Solutions, Inc. as of December 31, 2007 and 2006 and the consolidated results of their operations and their cash flows for each of the three years in the period ended December 31, 2007 in conformity with accounting principles generally accepted in the United States of America.

As discussed in Note 2 to the consolidated financial statements, on January 1, 2006, the Company adopted the provisions of Statement of Financial Accounting Standard No. 123(R), Share-Based Payment

We also have audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States), the effectiveness of World Energy Solutions, Inc.'s internal control over financial reporting as of December 31, 2007, based on criteria established in Internal Control—Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway Commission (COSO), and our report dated March 12, 2008 expressed an unqualified opinion on the effectiveness of the Company's internal control over financial reporting.

/S/ UHY LLP

Boston, Massachusetts March 12, 2008

- 42 -

Consolidated Balance Sheets

	Decem	ber 31,
	2007	2006
ASSETS		
C we down do		
Current assets: Cash and cash equivalents	\$ 7,001,884	\$ 17,483,886
Trade accounts receivable, net	1,878,233	1,702,798
Prepaid expenses and other current assets	338,049	211,749
Total current assets	9,218,166	19,398,433
Property and equipment, net	595,950	225,932
Capitalized software, net	463,888	105,296
Intangibles, net	7,316,916	
Goodwill	3,178,701	
Deferred taxes and other assets	26,944	1,061,720
Total assets	<u>\$ 20,800,565</u>	<u>\$ 20,791,381</u>
LIABILITIES AND STOCKHOLDERS' EQUITY		
Current liabilities:		
Accounts payable	\$ 980,488	\$ 848,5 18
Accrued commissions	712,919	1,021,433
Accrued compensation	1,516,418	304,330
Accrued expenses	279,162	232,587
Deferred revenue and customer advances	363,933	295,408
Capital lease obligations	41,624	56,259
Total current liabilities	3,894,544	2,758,535
Capital lease obligations, net of current portion	46,222	87,844
Total liabilities	3,940,766	2,846,379
Commitments (Note 9)		
Stockholders' equity:		
Preferred stock, \$0.0001 par value; 5,000,000 shares authorized, no shares issued or outstanding		_
Common stock, \$0.0001 par value; 150,000,000 shares authorized; 82,399,241 and 76,511,741		
shares issued and outstanding at December 31, 2007 and 2006,		
respectively	8,240	7,651
Additional paid-in capital	29,704,831	22,148,718
Accumulated deficit	(12,853,272)	(4,211,367)
Total stockholders' equity	16,859,799	17,945,002
Total liabilities and stockholders' equity	\$ 20,800,565	\$ 20,791,381

The accompanying notes are an integral part of these consolidated financial statements.

- 43 -

Consolidated Statements of Operations

	Years Ended December 31,				
	2007	2006	2005		
Revenue: Brokerage commissions and transaction fees Management fees Total revenue Cost of revenue Gross profit Operating expenses: Sales and marketing General and administrative	\$ 8,401,791 786,474 9,188,265 2,874,678 6,313,587 8,598,256 5 858 810	\$ 5,762,098 1,000 5,763,098 1,166,426 4,596,672 3,227,263 1,862,450	\$ 4,673,987 4,673,987 648,410 4,025,577 2,649,786 995,703		
Total costs and expenses Operating income (loss) Interest income (expense): Interest income	5,858,810 14,457,066 (8,143,479) 573,395	<u>5,089,713</u> (493,041) 139,617	3,645,489 380,088 8,004		
Interest expense Total interest income (expense), net Income (loss) before income taxes Income tax (expense) benefit Net income (loss) Accretion of preferred stock issuance costs Net income (loss) available to common stockholders	(10,101) 563,294 (7,580,185) (1,061,720) (8,641,905) 	(451,897) (312,280) (805,321) 304,228 (501,093) (6,299) \$ (507,392)	(94,842) (86,838) 293,250 754,000 1,047,250 (7,199) \$ 1,040,051		
Earnings (loss) per share: Net income (loss) per voting common share — basic Net income (loss) per non-voting common share — basic Net income (loss) available to common stockholders — diluted	\$ (0.11) <u>\$</u>	<u>\$ (0.01)</u> <u>\$</u> <u>\$ (0.01</u>)	\$ 0.02 \$ 0.02 \$ 0.02		
Weighted average shares outstanding — basic: Voting common stock Non-voting common stock	79,793,590 	45,576,477 	33,049,472 6,778,327 39,827,799		
Weighted average shares outstanding — diluted	79,793,590	45,576,477	54,506,566		

The accompanying notes are an integral part of these consolidated financial statements.

- 44 -

Consolidated Statements of Stockholders' Equity (Deficit) Years Ended December 31, 2007, 2006, and 2005

		Non-Voting Voting Common Stock Common Stock		Treasury	Stock	Additional		Total	
	Number of	\$0.0001 Par	Number of	S0.0001 Par	Number of	Stated at	Paid-in	Accumulated	Stockholders'
	Shares	Value	Shares	Value	Shares	Cost	Capital	Deficit	Equity (Deficit)
Balance, December 31,									
2004	6,552,135	655	33,724,819	3,372	675,346	(151,953)	2,664,316	(4,757,524)	(2,241,134)
Exercise of stock options	240,000	24	_	_	_	_	5,976		6,000
Accretion of stock	240,000	24	_	_	_				
issuance costs Issuance of	_	_	-	_	-	_	(7,199)	-	(7,199)
warrants in									
connection with							256 200		256,200
long-term debt Net income	_	_	_	_	_	_	256,200	1,047,250	1,047,250
Balance,	i i i i i i i i i i i i i i i i i i i						· · · · · · · · · · · · · · · · · · ·		
December 31, 2005	6,792,135	679	33,724,819	3,372	675,346	(151,953)	2,919,293	(3,710,274)	(938,883)
Exercise of stock	0,752,133	079	33,724,019	3,372	073,340	(131,933)	£,717,£72	(5,11,20)	(200,000)
options	2,190,671	219		-	_	_	56,698	—	56,917
Common stock warrants									
exercised	—		1,070,958	107	—		253,098	-	253,205
Issuance of common stock in									
connection with									
initial public									
offering, net of \$3.6 million of									
offering costs	—		23,000,000	2,300	_	—	17,264,968		17,267,268
Issuance of warrants in									
connection with									
initial public							274,129		274,129
offering Conversion of	_	_		_		_	274,129		214,127
Non-Voting									
Common Stock to Voting Common									
Stock in									
connection with									
initial public offering	(8,982,806)	(898)	8,982,806	898	_	_	_	_	
Conversion of		()	• <i>•</i> ••••••••						
Series A Redeemable									
Convertible									
Preferred Stock in connection with									
initial public									
offering Accretion of stock	—		10,433,504	1,044		-	1 ,506,953	_	1,507,997
issuance costs	_	_		_	_	_	(6,299)		(6,299)
Share-based									
compensation Purchase of treasury				_	_	_	44,261		44,261
stock	—	—		_	25,000	(12,500)	—	—	(12,500)
Retirement of treasury stock	_	_	(700,346)	(70)	(700,346)	164,453	(164,383)	_	_
Net loss								(501,093)	(501,093)
Balance,									
December 31, 2006		_	76,531,741	7,651	_	_	22,148,718	(4,211,367)	17,945,002
Share-based			, 0,2 - 1, 1	7,001					
compensation Issuance of	_	-	—	_			982,190	—	982,190
common stock in									
connection with									
acquisition of EnergyGateway	_		5,375,000	538	_	_	6,522,694	_	6,523,232
Issuance of			-,,				<i>,</i> .		
common stock in connection with									
restricted stock									
grant Reversal of	—	_	400,000	40	-		(40) 48,468	_	48,468
issuance costs	—	—		_	—		-00100		10,100
related to initial									

public offering of common stock Exercise of stock									
options	_	_	112,500	11	—	_	2,801	—	2,812
Net loss	_	_						<u>(8,641,905</u>)	(8,641,905)
Balance,									
December 31,									
2007		<u>\$</u>	82,399,241	<u>\$ 8,240</u>		<u>s </u>	<u>\$ </u>	(12,853,272) \$	16,859,799

The accompanying notes are an integral part of these consolidated financial statements.

Consolidated Statements of Cash Flows

	Years Ended December 31,					
		2007		2006		2005
Cash flows from operating activities:			-	(404 600)		4 0 10 0 00
Net income (loss)	\$	(8,641,905)	\$	(501,093)	\$	1,047,250
Adjustments to reconcile net income (loss) to net cash provided by (used						
in) operating activities:		1 000 407		175 446		105 931
Depreciation and amortization		1,028,497		135,446		105,831
Deferred taxes		1,061,720		(307,720)		(754,000)
Share-based compensation		982,190		44,261		4 200
Accretion of warrants				252,000		4,200
Changes in operating assets and liabilities, net of the effects of						
acquisition: Trade accounts receivable		17,590		(673,991)		(479,903)
Prepaid expenses and other assets				(075,791) (106,798)		(47,451)
		(133,377) 80,491		690,908		(42,041)
Accounts payable Accrued commissions		(308,514)		156,540		496,035
Accrued compensation		1,212,088		193,330		68,507
Accrued expenses		33,762		210,578		38,009
Deferred revenue and customer advances		68,525		165,668		(265,216)
	_		-	259,129		
Net cash provided by (used in) operating activities		(4,598,933)	-	259,129		171,221
Cash flows from investing activities:				/ C E (C E O)		151 205
Costs incurred in software development		(469,171)		(65,652)		(54,375)
Net cash paid in acquisition of EnergyGateway, net of cash acquired		(4,904,358)		(72.0(1)		(05.274)
Purchases of property and equipment	_	(456,095)	-	(72,061)	-	(25,374)
Net cash used in investing activities		(5,829,624)	_	<u>(137,713</u>)	-	(79,749)
Cash flows from financing activities:						
Proceeds from exercise of stock options		2,812		56,917		6,000
Proceeds from exercise of stock warrants		_		253,205		
Proceeds from initial public offering of common stock, net of offering						
costs of \$3.6 million				17,541,397		
Retirement of long-term debt				(2,000,000)		
Principal payments on line of credit				_		(500,000)
Proceeds from the issuance of long-term debt						2,000,000
Principal payments on capital lease obligations		(56,257)		(50,092)		(48,733)
Principal payments on guaranteed return				(10,523)		(14,062)
Purchase of treasury stock				(12,500)		
Net cash provided by (used in) financing activities		(53,445)	_	15,778,404		1,443,205
Net (decrease) increase in cash and cash equivalents		(10,482,002)	_	15,899,820		1,534,677
Cash and cash equivalents, beginning of year	_	17,483,886		1,584,066		49,389
Cash and cash equivalents, end of year	\$	7,001,884	\$	17,483,886	\$	1,584,066
Supplemental Disclosure of Cash Flow Information:						
Net cash received (paid) for interest	\$	612,303	\$	(129,401)	\$	(94,842)
Non-cash investing and financing activities:	<u> </u>	012,000	<u>*</u>		-	<u>(; ;; ; ;)</u> /
Property and equipment acquired through capital lease obligations	¢		¢	12,943	¢	<u>56,205</u>
	\$		3	12,945	<u>\$</u>	50,205
Non-cash investing and financing activities:					•	
Fair value of common stock issued in acquisition of EnergyGateway	\$	6,536,000	<u>\$</u>	<u></u>	<u>\$</u>	
Reversal of accrued expenses related to initial public offering of common						
stock	<u>\$</u>	48,468	<u>\$</u>		\$	
Conversion of series A preferred stock to common stock	\$		\$	1,507,997	\$	
Conversion of non-voting common stock to common stock	<u> </u>		5	898	\$	<u> </u>
-	Ŷ		\$	070	ې	
Issuance of warrants in connection with initial public offering of common	~			074 100	đ	
stock	<u>s</u>		<u>}</u>	274,129	<u>\$</u>	
Issuance of warrants in connection with debt offering	\$		<u>\$</u>		\$	256,200

The accompanying notes are an integral part of these consolidated financial statements.



i

!

Notes to Consolidated Financial Statements

NOTE 1 --- NATURE OF BUSINESS AND BASIS OF PRESENTATION

World Energy Solutions, Inc. ("World Energy" or the "Company") was incorporated in 1999 as a Delaware corporation and is focused in one operating segment – brokering energy and environmental commodities utilizing online auction platforms. The Company commenced operations through an entity named Oceanside Energy, Inc. ("Oceanside"), which was incorporated under the laws of the State of Delaware on September 3, 1996. We incorporated World Energy Solutions, Inc. under the laws of the State of Delaware under the name "World Energy Exchange, Inc." on June 22, 1999 and on October 31, 1999, Oceanside became a wholly-owned subsidiary of World Energy Solutions, Inc.. Oceanside was subsequently dissolved on May 18, 2006. On December 15, 2006 the Company established World Energy Securities Corp., a Massachusetts Securities Corporation, as a wholly-owned subsidiary.

World Energy is an energy and environmental commodities brokerage company that has developed online auction platforms, the World Energy Exchange and the World Green Exchange. On the World Energy Exchange, retail energy consumers (commercial, industrial and government) and wholesale energy participants (utilities and retail suppliers) in the United States are able to negotiate for the purchase or sale of electricity and other energy resources from competing energy suppliers which have agreed to participate on our auction platform in a given auction. Although the Company's primary source of revenue is from brokering electricity and natural gas, the Company adapted its World Energy Exchange auction platform to accommodate the brokering of green power in 2003 (i.e., electricity generated by renewable resources), wholesale electricity in 2004 and certain other energy-related products in 2005. In 2007, the Company created the World Green Exchange based on the World Energy Exchange technology and business process. On the World Green Exchange, buyers and sellers negotiate for the purchase or sale of environmental commodities such as Renewable Energy Certificates, Verified Emissions Reductions and Certified Emissions Reductions.

On November 16, 2006, the Company completed its initial public offering of common stock, including the exercise by the underwriters of an over-allotment option on December 5, 2006, for the sale of 23,000,000 shares of common stock resulting in net proceeds to the Company of approximately \$17.5 million (net of offering costs of approximately \$3.6 million).

On June 1, 2007, the Company acquired substantially all of the assets of EG Partners, LLC, formerly known as EnergyGateway LLC ("EnergyGateway"), for \$4,951,758 in cash and 5,375,000 common shares of World Energy plus the assumption of certain liabilities. The EnergyGateway operations are included in these financial statements from June 1, 2007. See Note 10 for further discussion of this acquisition.

NOTE 2 — SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Principles of Consolidation

The Company's consolidated financial statements include its wholly-owned subsidiary World Energy Securities Corp. All intercompany accounts and transactions have been eliminated in consolidation.

Use of Estimates

The preparation of financial statements in conformity with accounting principles generally accepted in the United States of America ("generally accepted accounting principles") requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Accordingly, actual results could differ from those estimates.

The Company's most judgmental estimates affecting the consolidated financial statements are those relating to revenue recognition and the estimate of actual energy purchased from the energy supplier and end user, or energy consumer, of such energy; software development costs; the fair value of equity securities prior to the Company's initial public offering when there was no ready market for the purchase and sale of these shares; the valuation of intangible assets and goodwill; impairment of long-lived assets; and estimates of future taxable income as it relates to the realization of net deferred tax assets. The Company regularly evaluates its estimates and assumptions based upon historical experience and various other factors that it believes to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities that are not readily apparent from other sources. To the extent actual results differ from those estimates, future results of operations may be affected. The Company believes the following critical accounting policies affect its more significant judgments and estimates used in the preparation of the consolidated financial statements.

Revenue Recognition

The Company generates revenue through commissions on sales of energy usage transacted on the Company's online auction platform. The Company recognizes revenues in accordance with the United States Securities and Exchange Commission's ("SEC") Staff Accounting Bulletin No. 104, *Revenue Recognition* ("SAB 104"). SAB 104 requires that four basic criteria must be met before revenue can be recognized: (1) persuasive evidence of an arrangement exists; (2) the service has been rendered; (3) the fee is fixed or determinable; and (4) collectibility is reasonably assured.

The Company participates in four primary markets: retail electricity, retail natural gas, wholesale energy and environmental commodities. Prior to the June 1, 2007 acquisition of substantially all of the assets of EnergyGateway, the Company primarily received revenue from the brokering of electricity on the World Energy Exchange, utilizing reverse auction technology ("retail electricity transactions"). While EnergyGateway also utilized an online auction format, its primary focus was on the brokering of natural gas and its "post and bid" methodology ("retail natural gas transactions"). In addition, the Company adapted the World Energy Exchange to conduct reverse and forward online auctions for the wholesale electricity and natural gas markets and recently conducted several green credit auctions ("wholesale and green transactions"). The revenue recognition policy and related third party commissions for each of these products is more fully described below.

Retail Electricity Transactions

The Company receives a monthly commission on energy sales contracted through its online auction platform from each energy supplier based on the energy usage generated and transacted between the energy supplier and energy consumer. The Company's commissions are not based on the retail price for electricity; rather, commissions are based on the energy usage generated and transacted between the energy supplier and energy consumer multiplied by the contractual commission rate. Revenue from commissions is recognized as earned on a monthly basis over the life of each contract as energy is consumed, provided there is persuasive evidence of an arrangement, the sales price is fixed or determinable, collection of the related receivable is reasonably assured, and customer acceptance criteria, if any, has been successfully demonstrated.

The Company records brokerage commissions based on actual usage data obtained from the energy supplier for that accounting period, or to the extent actual usage data is not available, based on the estimated amount of electricity delivered to the energy consumers for that accounting period. The Company's estimates are developed on a quarterly basis based on the following criteria:

- Payments received prior to the issuance of the financial statements;
- Usage updates from energy suppliers;
- Usage data from utilities;
- Comparable historical usage data; and
- Historical variances to previous estimates.

To the extent usage data cannot be obtained, revenue is estimated as follows:

- Historical usage data obtained from the energy consumer in conjunction with the execution of the auction;
- Geographic/utility usage patterns based on actual data received;
- Analysis of prior year usage patterns; and
- Specific review of individual energy supplier/location accounts.

In addition, the Company performs sensitivity analyses on this estimated data based on overall industry trends including prevailing weather and usage data. Once the actual data is received, estimated accounts receivable and revenue is adjusted to the actual total amount in the period during which the payment is received. Based on management's current capacity to obtain actual energy usage, the Company currently estimates 4 to 6 weeks of revenue at the end of its accounting period. Differences between estimated and actual revenue have been within management's expectations and have not been material to date.

The Company generally does not directly invoice its electricity energy suppliers and, therefore, reports a substantial portion of its receivables as "unbilled." Unbilled accounts receivable represents management's best estimate of energy provided by the energy suppliers to the energy consumers for a specific completed time period at contracted commission rates and is made up of two components. The first component represents energy usage for which actual data has been received from the supplier and/or the utility but for which payment has not been received at the balance sheet date. The majority of the Company's contractual relationships with energy suppliers require that the suppliers provide the Company with actual usage data on a monthly basis and remit payment based on that usage. The second component represents energy usage for which the Company has not received actual data, but for which it has

estimated usage. Commissions paid in advance by certain energy suppliers are recorded as deferred revenue and amortized to commission revenue on a monthly basis on the energy exchanged that month.

Retail Natural Gas Transactions

There are two primary components related to retail natural gas services - transaction fees and management fees. Transaction fees are billed to and paid by the energy supplier awarded business on the platform. Transaction fees for natural gas awards are established prior to award and are the same for each supplier. For the majority of natural gas transactions, the Company bills the supplier upon the conclusion of the transaction based on the estimated energy volume transacted for the entire award term multiplied by the transaction fee. Management fees are paid by the energy consumers and are generally billed on a monthly basis for services rendered based on terms and conditions included in contractual arrangements. While substantially all retail natural gas transactions are accounted for in accordance with this policy, a certain percentage is accounted for as the natural gas is consumed by the customer and recognized as revenue in accordance with the retail electricity transaction revenue recognition methodology described above.

Wholesale and Green Transactions

Substantially all transaction fees are invoiced upon the conclusion of the auction based on a fixed fee. These revenues are not tied to future energy usage and are recognized upon the completion of the online auction. For reverse auctions where customers bid for a consumer's business, the fees are paid by the bidder. For forward auctions where a lister is selling energy or credits, the fees are typically paid by the lister. In addition, revenue may not be recognized on certain green transactions until the credits being auctioned have been verified. While substantially all wholesale and green transactions are accounted for in this fashion, a small percentage is accounted for as the wholesale electricity or gas is delivered similar to the retail electricity transaction methodology described above.

Third Party Channel Commissions

Commissions are paid to channel partners at contractual rates based on monthly energy transactions between energy suppliers and energy consumers. The commission is accrued monthly and charged to sales and marketing expense as revenue is recognized. Commissions are paid to salespeople at contractual commission rates based upon cash collections from customers.

Revenue Estimation

The Company's estimates in relation to revenue recognition affect revenue and sales and marketing expense as reflected on its statements of operations, and trade accounts receivable and accrued commissions accounts as reflected on its balance sheets. For any quarterly reporting period, the Company may not have actual usage data for certain energy suppliers and will need to estimate revenue. Revenue is recorded based on the energy consumers' historical usage profile. At the end of each reporting period, the Company adjusts historical revenue to reflect actual usage for the period. For the year ended December 31, 2007, usage was estimated for approximately 10% of the Company's revenue resulting in a \$35,179, or 0.4%, adjustment. This increase in revenue resulted in an approximate 0.1% reduction in sales and marketing expense related to third party commissions, respectively. A 1% difference between this estimate and actual usage would have an approximate \$9,000 effect on the Company's revenue for the year ended December 31, 2007.

Software Development

Certain acquired software and significant enhancements to the Company's software are recorded in accordance with Statement of Position ("SOP") 98-1, "Accounting for Costsof Computer Software Developed or Obtained for Internal Use". Accordingly, internally developed software costs of approximately \$469,000, \$66,000, and \$54,000 related to implementation, coding and configuration have been capitalized in 2007, 2006, and 2005, respectively. The Company amortizes internally developed and purchased software over the estimated useful life of the software (generally three years). Accordingly, during 2007, 2006, and 2005, approximately \$111,000, \$73,000, and \$55,000, respectively, were amortized to cost of revenues resulting in accumulated amortization of approximately \$273,000, \$162,000, and \$89,000 at December 31, 2007, 2006, and 2005, respectively. At December 31, 2007, future amortization expense for capitalized internally developed software is as follows:

2008	\$ 188,000
2009	167,000
2010 and thereafter	 109,000
	\$ 464,000

Pre- and post- software implementation and configuration costs have historically been immaterial and charged to cost of revenue as incurred.

Goodwill

The Company uses assumptions in establishing the carrying value and fair value of its goodwill. Goodwill represents the excess of the purchase price over the fair value of identifiable net assets of acquired businesses. The Company accounts for goodwill that results from acquired businesses in accordance with Statement of Financial Accounting Standards ("SFAS") No. 142, "Goodwill and Other Intangible Assets" ("SFAS No. 142"). Under SFAS No. 142, goodwill and intangible assets having indefinite lives are not amortized but instead are assigned to reporting units and tested for impairment annually or more frequently if changes in circumstances or the occurrence of events indicate possible impairment.

Pursuant to SFAS No. 142, the Company performs an annual impairment test for goodwill. The impairment test for goodwill is a two-step process. Step one consists of a comparison of the fair value of a reporting unit with its carrying amount, including the goodwill allocated to each reporting unit. If the carrying amount is in excess of the fair value, step two requires the comparison of the implied fair value of the reporting unit whereby the carrying amount of the reporting unit's goodwill over the implied fair value of the reporting unit's goodwill will be recorded as an impairment loss. There was no impairment of goodwill in the year ended December 31, 2007. Future events could cause management to conclude that impairment indicators exist and that goodwill of previously acquired businesses is impaired. Any related impairment loss could have a material impact on the Company's financial condition and results of operations.

Intangible Assets

The Company uses assumptions in establishing the carrying value, fair value and estimated lives of its intangible assets. The criteria used for these assumptions include management's estimate of the assets continuing ability to generate positive income from operations and positive cash flow in future periods compared to the carrying value of the asset, as well as the strategic significance of any identifiable intangible asset in our business objectives. If assets are considered impaired, the impairment recognized is the amount by which the carrying value of the assets exceeds the fair value of the assets. Useful lives and related amortization expense are based on an estimate of the period that the assets will generate revenues or otherwise be used by the Company. Factors that would influence the likelihood of a material change in the Company's reported results include significant changes in the asset's ability to generate positive cash flow, a significant decline in the economic and competitive environment on which the asset depends and significant changes in the Company's strategic business objectives.

Intangible assets consist of customer relationships and contracts, purchased technology and other intangibles, and are stated at cost less accumulated amortization. Intangible assets with a definite life are amortized using the straight-line method over their estimated useful lives, which range from one to ten years. Accumulated amortization of intangible assets amounted to approximately \$813,000 at December 31, 2007.

Impairment of Long-Lived and Intangible Assets

In accordance with SFAS No. 144, "Accounting for the Impairment or Disposal of Long-Lived Assets", the Company periodically reviews long-lived assets and intangible assets for impairment whenever events or changes in circumstances indicate that the carrying amount of such assets may not be recoverable or that the useful lives of those assets are no longer appropriate. Recoverability of these assets is determined by comparing the forecasted undiscounted net cash flows of the operation to which the assets relate to the carrying amount. There was no impairment of intangible assets in the year ended December 31, 2007.

Cash and Cash Equivalents

The Company considers all highly liquid debt instruments purchased with an original maturity of 90 days or less to be cash equivalents.

Property and Equipment

Property and equipment is stated at cost. Depreciation is provided on a straight-line basis over the estimated useful lives of the related assets or the life of the related lease, whichever is shorter, which range from 3 to 7 years.

Income Taxes

The Company accounts for income taxes in accordance with SFAS No. 109, "Accounting for IncomeTaxes" ("SFAS No. 109"). Deferred tax assets and liabilities are determined at the end of each period based on the future tax consequences that can be attributed to net operating loss carryforwards, as well as differences between the financial statement carrying amounts of the existing assets and liabilities and their respective tax basis. Deferred income tax expense or credits are based on changes in the asset or liability from period to period. Valuation allowances are provided if, based on the weight of available evidence, it is more likely than not that some or all of the deferred tax assets will not be realized. The realization of deferred tax assets is dependent upon the generation of future taxable income. In determining the valuation allowance, the Company considers past performance, expected future taxable income, and qualitative factors which are considered to be appropriate in estimating future taxable income. The Company's forecast of expected future taxable income is for future periods that can be reasonably estimated. The Company has historically utilized a three year forecasting window to determine its expected future taxable income. As of December 31, 2006, the Company had cumulative net income for the three years then ended and the expected future taxable income anticipated from 2007 through 2009 was deemed adequate on a present value basis to benefit the majority of our deferred tax assets (primarily net operating loss carryforwards). During 2007, the Company accelerated its growth in excess of its initial expectations to address the strategic initiatives outlined at the time of our initial public offering. The Company grew from 20 employees as of December 31, 2006 to 63 as of December 31, 2007, acquired one of its main competitors in the retail energy market, and developed its wholesale and environmental commodity product lines. These initiatives increased the Company's loss for 2007 resulting in a three-year cumulative loss as of December 31, 2007 and increased its near term operating expenses. A three-year cumulative loss is significant negative evidence in considering whether the deferred tax assets are realizable and generally precludes relying on projections of future taxable income to support the recovery of deferred tax assets. While the Company expects its future income will be adequate on a present value basis to benefit substantially all of its current deferred tax assets, the recent three-year cumulative loss is given more weight than projected future income when determining the need for a valuation. Therefore, during the fourth quarter of 2007, the Company recorded a full valuation allowance against its deferred tax assets of approximately \$4.1 million, and maintains a cumulative valuation allowance of approximately \$4.7 million at December 31, 2007.

The Company's estimates in relation to income taxes affect income tax expense and deferred tax assets as reflected on its statements of operations and balance sheets, respectively. The deferred tax assets are reduced by a valuation allowance if it is more likely than not that the tax benefits will not be realized in the near term. As of December 31, 2007, the Company had deferred tax assets of approximately \$4.7 million against which a full valuation allowance has been established. To the extent the Company determines that it is more likely than not that it will recover all of its deferred tax assets, it could result in an approximate \$4.7 million non-cash tax benefit.

The Company adopted Financial Accounting Standards ("FASB") Interpretation No. 48, "Accounting for Uncertainty in Income Taxes" ("FIN 48"), on January 1, 2007. FIN 48 clarifies the accounting for uncertainty in income taxes recognized in an enterprise's financial statements in accordance with SFAS 109. FIN 48 prescribes a two-step process to determine the amount of tax benefit to recognize. First, the tax position must be evaluated to determine the likelihood that it will be sustained upon examination by a tax authority. If the tax position is deemed "more-likely-than-not" to be sustained, the tax position is the largest amount that has a greater than 50 percent likelihood of being realized upon ultimate settlement. If the tax position does not meet the "more-likely-than-not" threshold, then it is not recognized in the financial statements. As of its adoption date of January 1, 2007, and also at December 31, 2007, FIN 48 did not have a material impact on our financial position or results of operations.

The Company has reviewed the tax positions taken, or to be taken, in its tax returns for all tax years currently open to examination by the taxing authority in accordance with the recognition and measurement standards of FIN 48. At December 31, 2007, there are no expected material, aggregate tax effect of differences between tax return positions and the benefits recognized in our financial statements.

Advertising Expense

Advertising expense primarily includes promotional expenditures and is expensed as incurred, as such efforts have not met the direct-response criteria required for capitalization. Amounts incurred for advertising expense were not material for the years ended December 31, 2007, 2006, and 2005.

Comprehensive Income

SFAS No. 130, "Reporting Comprehensive Income", establishes standards for reporting and displaying comprehensive income (loss) and its components in financial statements. Comprehensive income (loss)

is defined as the change in stockholders' equity of a business enterprise during a period from transactions and other events and circumstances from non-owner sources. The comprehensive income (loss) for all periods presented does not differ from the reported net income (loss).

Fair Value of Financial Instruments

SFAS No. 107, "Disclosures about Fair Value of Financial Instruments", requires management to disclose the estimated fair value of certain assets and liabilities defined by SFAS 107 as financial instruments. Financial instruments are generally defined as cash, evidence of ownership interest in an entity, or a contractual obligation that both conveys to one entity a right to receive cash or other financial instruments from another entity and imposes on the other entity the obligation to deliver cash or other financial instruments to the first entity. At December 31, 2007, management believes that the carrying value of cash and cash equivalents, receivables and payables approximated fair value because of the short maturity of these financial instruments.

Segment Reporting

SFAS No. 131, "Disclosures about Segments of an Enterprise and RelatedInformation", established standards for reporting information about operating segments in financial statements. Operating segments are defined as components of an enterprise about which separate financial information is available that is evaluated regularly by the chief decision maker in deciding how to allocate resources and in assessing performance. The Company's chief decision maker is the chief executive officer. The Company's chief decision maker reviews the results of operations based on one industry segment: the brokering of energy and environmental commodities utilizing online auction platforms.

Concentration of Credit Risk and Off-Balance Sheet Risk

Financial instruments that potentially expose the Company to concentrations of credit risk consist principally of cash and trade accounts receivable. The Company has no significant off-balance sheet risk such as foreign exchange contracts, option contracts, or other foreign hedging arrangements. The Company places its cash with primarily one institution, which management believes is of high credit quality.

The Company receives commission payments from energy suppliers based on the energy usage transacted between energy consumers and energy suppliers. The Company provides credit in the form of invoiced and unbilled accounts receivable to energy suppliers and energy consumers in the normal course of business. Collateral is not required for trade accounts receivable, but ongoing credit evaluations of energy suppliers and energy consumers are performed. Management provides for an allowance for doubtful accounts on a specifically identified basis, as well as through historical experience applied to an aging of accounts, if necessary. Trade accounts receivable are written off when deemed uncollectible. To date write-offs have not been material.

The following represents revenue and trade accounts receivable from energy suppliers exceeding 10% of the total in each category:

	Revenue for	the year ended	December		of ber 31,
Energy supplier	2007	2006	2005	2007	2006
A	11%	18%	18%	10%	21%
В	5%	13%	10%	7%	12%
C	6%	10%	18%	12%	4%
D	7%	10%	_	10%	4%
Е	6%	6%	11%	2%	1%
F	21%	9%	—	22%	4%

During 2006, supplier A was acquired by another supplier who is not reflected above as its revenues represented less than 10% of total revenue. If these suppliers were combined as of January 1, 2006, revenue attributable to these customers would have been 27% for the year ended December 31, 2006 and unbilled accounts receivable would have represented 46% of total unbilled accounts receivable as of December 31, 2006.

In addition to its direct relationship with energy suppliers, the Company also has direct contractual relationships with energy consumers for the online procurement of certain of their energy needs. These energy consumers are primarily large businesses and government organizations. For the years ended December 31, 2007, 2006 and 2005, two, four and three of these energy consumers accounted for transactions resulting in over 10% individually, respectively, and approximately 30%, 63% and 51% in the aggregate of our revenue, respectively.

Reclassifications

Certain prior year amounts have been reclassified to conform with the current presentation.

Earnings (Loss) Per Share

In March 2004, the Emerging Issues Task Force ("EITF") reached a consensus on Issue No. 03-6, "Participating Securities and the Two -- Class Methodunder FAS 128" ("EITF Issue No. 03-6"). EITF Issue No. 03-6 provides guidance in determining when the two-class method, as defined in SFAS No. 128, "Earnings Per Share", must be utilized in calculating earnings per share. For the year ended December 31, 2005, the Company had determined that its non-voting common stock and Series A redeemable convertible preferred stock (the "Series A Preferred") represented participating securities. The non-voting common had the same privileges and rights of the voting common stock, except for the right to vote. The Series A Preferred participated in dividends, if any, paid by the Company on a proportional basis with the voting common stock. The Series A Preferred and non-voting common automatically converted into voting common stock on a one-for-one basis upon the closing of the Company's initial public offering. EITF Issue No. 03-6 requires the income per share for each class of common stock to be calculated assuming 100% of the Company's earnings are distributed as dividends to each class of common stock based upon their respective dividend rights, even though the Company does not anticipate distributing 100% of its earnings as dividends.

Basic earnings per share for the Company's voting common and non-voting common stock is calculated by dividing net income (loss) allocated to voting common and non-voting common stock by the weighted average number of shares of voting common and non-voting common stock by the weighted average number of shares of voting common and non-voting common stock assumes the conversion of all the Company's non-voting common stock and Series A Preferred therefore no allocation of earnings to non-voting common stock or the Series A Preferred is required.

For the basic earnings per share calculation in 2005, net income was allocated to Series A Preferred shareholders based on dividend rights and then among the Company's two classes of common stock; voting common stock and non-voting common stock based on ownership interests. The allocation among each class of stock was based upon the two-class method. The following table reflects the allocation of net income using this method:

		r the Year Ended cember 31, 2005
Net income available to common shareholders	<u>\$</u>	1,040,051
Allocation of net income for basic:		
Voting common stock	\$	684,354
Non-voting common stock		140,407
Series A Preferred stock		215,290
	\$	1,040,051

On November 16, 2006, the Company completed an initial public offering of its common stock. As a result, all issued and outstanding shares of Series A Preferred Stock and non-voting common stock were automatically converted into voting common stock. As of December 31, 2007 and 2006, the Company only had one issued and outstanding class of stock – common stock. As a result, the basic loss per share for the years ended December 31, 2007 and 2006 is computed by dividing net loss available to common stockholders by the weighted average number of common shares outstanding for the period. The computed loss per share does not assume conversion, exercise, or contingent exercise of securities that would have an anti-dilutive effect on loss per share. The following table reflects the weighted average shares used to calculate basic and diluted earnings per share:

	For the Years Ended December 31,		
	2007	2006	2005
Weighed average number of voting common shares — basic Weighed average number of non-voting common shares — basic	79,793,590	45,576,477	33,049,472 6,778,327
Weighted average number of common and common equivalent shares outstanding — basic Dilutive Shares:	79,793,590	45,576,477	39,827,799
Weighted average number of Series A Preferred shares	_	<u> </u>	10,433,504
Incremental shares under the treasury stock method for outstanding warrants Incremental shares under the treasury stock method for outstanding options			1,191,808 3,053,455
Weighted average number of common and common equivalent shares outstanding — diluted	79,793,590	45,576,477	54,506,566

For the year ended December 31, 2007, 1,048,575, 1,826,375 and 493,718 weighted average shares issuable relative to common stock warrants, common stock options and restricted stock, respectively, were excluded from net loss per share since the inclusion of such shares would be anti-dilutive due to the Company's net loss position. For the year ended December 31, 2006, 9,147,182, 2,251,072 and 2,230,232 weighted average shares issuable relative to preferred stock, common stock warrants and stock options, respectively, were excluded from net loss per share since the inclusion of such shares would be anti-dilutive due to the Company's net loss position. For the year ended December 31, 2006, 9,147,182, 2,251,072 and 2,230,232 weighted average shares ince the inclusion of such shares would be anti-dilutive due to the Company's net loss position. For the year ended December 31, 2007, 2,320,000 shares issuable relative to common stock options had exercise prices that exceeded the average market price of the Company's common stock and were excluded from the calculation of diluted shares since the inclusion of such shares would be anti-dilutive.

The Company did not declare or pay any dividends in 2007, 2006 and 2005.

Share-Based Compensation

At December 31, 2007, the Company had two share-based employee compensation plans, which are more fully described in Note 6. Effective January 1, 2006, the Company accounted for stock-based compensation in accordance with SFAS No. 123(R), "Stock-Based Compensation" ("SFAS No. 123(R)") which is a revision of SFAS No. 123 "Accounting for Stock-Based Compensation" ("SFAS No. 123(R)") which is a revision of SFAS No. 123 "Accounting for Stock-Based Compensation" ("SFAS No. 123(R) superseded Accounting Principles Board (APB) No. 25, "Accounting for Stock Issued to Employees", and amends SFAS No. 95, "Statement of Cash Flows". SFAS No. 123(R) requires all share-based payments to employees, including grants of employee stock options and restricted stock, to be recognized in the income statement based on their fair values. Pro forma disclosure is no longer an alternative.

SFAS No. 123(R) requires nonpublic companies that used the minimum value method in SFAS No. 123 for either recognition or pro forma disclosures to apply SFAS No. 123(R) using the prospective-transition method. As such, the Company will continue to apply APB 25 in future periods to equity awards outstanding at the date of SFAS No. 123(R)'s adoption that were measured using the minimum value method. In accordance with this standard, the prior period pro forma stock information has not been restated. In accordance with SFAS No. 123(R), the Company will recognize the compensation cost of share-based awards on a straight-line basis over the vesting period of the award.

The per-share weighted-average fair value of stock options granted during the year ended December 31, 2007 and 2006 was \$0.58 and \$0.36, respectively, on the date of grant, using the Black-Scholes option-pricing model with the following weighted-average assumptions and estimated forfeiture rates of 13% in 2007 and 2006, respectively:

Year ended December 31.	Expected Dividend Yield	Risk Interest Rate	Expected Option Life	Expected Volatility
	Envidend Tield			
2007		4.66%	4.61 years	54%
2006	—	4.66%	4.75 years	48%

Effective with the adoption of SFAS No. 123(R), the Company elected to use the Black-Scholes option pricing model to determine the weighted average fair value of options granted. As the Company's stock has a limited trading history as of December 31, 2007, the Company determined the volatility for stock options granted in 2007 based on a weighted combination of per share closing prices reported on the Toronto Stock Exchange, and historical and reported data for a peer group of companies that issued options with substantially similar terms. The expected life of options has been determined utilizing the "simplified" method as prescribed by the Securities and Exchange Commission's Staff Accounting Bulletin No. 107, "*Share-Based Payment*". The risk-free interest rate is based on a treasury instrument whose term is consistent with the expected life of the stock options. The Company has not paid and does not anticipate paying cash dividends on its common stock; therefore, the expected dividend yield is assumed to be zero. In addition, SFAS No. 123(R) requires companies to utilize an estimated forfeiture rate when calculating the expense for the period, whereas, SFAS No. 123. As a result, the Company applied estimated forfeiture rates of between 11% and 13%, and 12% and 13% during the years ended December 31, 2007 and 2006, respectively, in determining the expense recorded in the accompanying consolidated statements of operations. The effect on compensation expense for the year ended December 31, 2007 from changes in the forfeiture rate in 2007 of previous years' stock option grants, resulted in an increase in compensation expense of less than \$1,000.

For the year ended December 31, 2007, the Company recorded stock-based compensation expense of approximately \$982,000 in connection with share-based payment awards as follows: approximately \$753,000 in sales and marketing, approximately \$103,000 in cost of revenue and approximately \$126,000 in general and administrative expenses. For the year ended December 31, 2006, the Company recorded stock-based compensation expense of approximately \$44,000 in connection with share-based payment awards as

follows: approximately \$20,000 in sales and marketing, approximately \$2,000 in cost of revenue and approximately \$22,000 in general and administrative expenses. As of December 31, 2007, there was approximately \$3,024,000 of unrecognized compensation expense related share-based awards, including approximately \$1,513,000 related to non-vested stock option awards that is expected to be recognized over a weighted-average period of 3.18 years, and approximately \$1,511,000 related to non-vested restricted stock awards. See Note 6 for a summary of the share-based activity under the Company's stock-based employee compensation plans for the years ended December 31, 2007, 2006 and 2005.

Prior to January 1, 2006, the Company accounted for its stock-based awards to employees using the intrinsic value method prescribed in APB 25 and related interpretations. Under the intrinsic value method, compensation expense is measured on the date of grant as the difference between the deemed fair value of the Company's common stock and the option exercise price multiplied by the number of options granted. Generally, the Company grants stock options with exercise prices equal to the estimated fair value of its common stock; however, to the extent that the deemed fair value of the common stock exceeded the exercise price of stock options granted to employees on the date of grant, the Company recorded stock-based compensation expense ratably over the vesting schedule of the options, generally four years. The fair value of the Company's common stock is determined by the Company's Board of Directors (the "Board").

Prior to November 16, 2006, there had been no public market for the Company's common stock and the fair value for the Company's common stock was estimated by the Board, with input from management as well as from independent appraisals. The Board exercised judgment in determining the estimated fair value of the Company's common stock on the date of grant based on several factors, including the liquidation preferences, dividend rights, and voting control attributable to the Company's then-outstanding convertible preferred stock and, primarily, the likelihood of achieving a liquidity event such as an initial public offering or sale of the Company. In the absence of a public trading market for the Company's common stock, the Board considered objective and subjective factors in determining the fair value of the Company's common stock. The Company believes this to have been a reasonable methodology based upon the Company's internal peer company analyses and based on arms-length transactions, when applicable, involving the Company's common stock supportive of the results produced by this valuation methodology.

For the years ended December 31, 2007 and 2006, options to purchase 3,020,000 and 1,262,000 shares of common stock were granted, respectively, at weighted exercise prices of \$1.24 and \$0.79, respectively. The weighted average fair value of the Company's common stock at option grant date for the years ended December 31, 2007 and 2006 was \$1.17 and \$0.78, respectively. Additionally, 1,928,500 shares of restricted stock were granted for the twelve months ended December 31, 2007.

In preparing for the initial public offering of the Company's common stock, the Company reassessed the valuations of its common stock during 2006, in accordance with the American Institute of Certified Public Accountants ("AICPA's") Practice Aid, *Valuation of Privately-Held-Company Equity Securities Issued as Compensation (the "Practice Aid")*. In May 2006, the company engaged an unrelated third-party valuation specialist to assist management in providing a retrospective valuation report of option grants during the first quarter of 2006 and a prospective valuation of the July 31, 2006 option grants. The third-party valuation specialist valued the Company's common stock (both voting and non-voting common stock) at approximately \$0.30, \$0.39 and \$0.95 as of January 16, 2006, February 10, 2006 and July 31, 2006, respectively, representing the dates when certain stock options were granted.

The Company believes that the valuation methodologies that were used were consistent with the Practice Aid. With the exception of one option to purchase 60,000 shares on March 30, 2006, the Company has concluded that for all options granted to employees during the two years ended December 31, 2005 and the period from January 1, 2006 to November 16, 2006, the fair value of its common stock, for financial reporting purposes, did not exceed the exercise price for those options at the time of grant. The Company granted stock options to purchase 60,000 shares of non-voting common stock, on March 30, 2006, to an employee with an exercise price of \$0.38 per share, which was \$0.12 per share below fair value as subsequently determined by the Board of Directors. Compensation expense related to this grant was deemed to be immaterial. On July 19, 2006 the Company and the option holder agreed to terminate the option grant. On July 31, 2006, the Company granted this employee an option to purchase 72,000 shares of common stock at an exercise price of \$0.95, which the Board determined to be the fair value at that date. No stock-based compensation expense was recorded for the year ended December 31, 2005 as the exercise price of the Company's stock options was equal to the estimated fair value of the Company's common stock on the date of grant.

The Company accounts for transactions in which services are received from non-employees in exchange for equity instruments based on the fair value of such services received or of the equity instruments issued, whichever is more reliably measured, in accordance with SFAS No. 123(R) and EITF 96-18, "Accounting for Equity Instruments That Are Issuedto Other Than Employees for Acquiring, or in Conjunction With Selling, Goods orServices". The Company granted 455,000 shares of restricted common stock during 2007 to non-employees in consideration for services performed. At the time of the grants, these shares of restricted stock had a

fair value of approximately \$523,000, based on the fair value of the Company's common stock at the grant date. The Company recognized a charge of approximately \$488,000 to sales and marketing expense in the statements of operations for the year ended December 31, 2007 related to these grants.

The Company provides the disclosures as required by SFAS No. 148, "Accountingfor Stock-Based Compensation and Disclosure, an Amendment of FASB Statement No.123".

The following table illustrates the assumptions used and the effect on net income if the Company had applied the fair value recognition provisions of SFAS No. 123 to stock-based employee compensation. For purposes of this pro forma disclosure, the Company estimated the fair value of stock options issued to employees using the minimum value option-pricing model. The Company's use of the minimum value model was primarily due to the determination as to its appropriateness, as well as its general acceptance as an option valuation technique for private companies.

The weighted-average assumptions used to calculate the SFAS No. 123 pro forma expense for stock options granted to employees and directors for the year ended December 31, 2005:

Dividend yield	0%
Risk-free interest rate	3.63% to 4.22%
Expected life	5 years
Volatility	0%

Had compensation cost for these awards been determined consistent with SFAS No. 123, the Company's net income would have been as follows:

	 2005
Net income available to common shareholders, as reported	\$ 1,040,051
Deduct: total stock-based employee compensation expense determined under fair value based method for all awards, net of related tax effects	 (7,499)
Net income available to common shareholders, pro forma	\$ 1,032,552
As Reported Per Share Amounts:	
Basic net income per weighted average voting common share	\$ 0.02
Basic net income per weighted average non-voting common share	\$ 0.02
Diluted net income per weighted average common share	\$ 0.02
Pro-forma Per Share Amounts:	
Pro-forma basic net income per weighted average voting common share	\$ 0.02
Pro-forma basic net income per weighted average non-voting common share	\$ 0.02
Pro-forma diluted net income per weighted average common share	\$ 0.02

As stock options vest over several years and additional stock option grants are expected to be made each year, the above proforma disclosures are not necessarily representative of pro-forma effects on results of operations for future periods.

Recent Accounting Pronouncements

In December 2007, the FASB issued SFAS No. 141 (revised 2007), "Business Combinations" ("SFAS No. 141(R)"). SFAS No. 141(R) requires an acquiring entity in a business combination to recognize all (and only) the assets acquired and liabilities assumed in the transaction; establishes the acquisition-date fair value as the measurement objective for all assets acquired and liabilities assumed; and requires the acquirer to disclose to investors and other users all of the information they need to evaluate and understand the nature and financial effect of the business combination. SFAS No. 141(R) is effective for fiscal years beginning after December 15, 2008. The Company expects to adopt SFAS No. 141(R) on January 1, 2009.

In December 2007, the FASB issued SFAS No. 160, "Noncontrolling Interests in Consolidated Financial Statements" ("SFAS No. 160"). SFAS No. 160 clarifies that a noncontrolling or minority interest in a subsidiary is considered an ownership interest and, accordingly, requires all entities to report such interests in subsidiaries as equity in the consolidated financial statements. SFAS No. 160 is effective for fiscal years beginning after December 15, 2008. The Company expects to adopt SFAS No. 160 on January 1, 2009.

In September 2006, the FASB issued SFAS No. 157, "Fair Value Measurements" ("SFAS No. 157"). This standard defines fair value, establishes a framework for measuring fair value in accounting principles generally accepted in the United States of America, and expands disclosure about fair value measurements. This pronouncement applies to other accounting standards that require or permit

fair value measurements. Accordingly, this statement does not require any new fair value measurement. This statement is effective for fiscal years beginning after November 15, 2007, and interim periods within those fiscal years. In December of 2007, the FASB agreed to a one year deferral of SFAS No. 157's fair value measurement requirements for nonfinancial assets and liabilities that are not required or permitted to be measured at fair value on a recurring basis. The Company's management is currently evaluating the requirements of SFAS No. 157, but does not expect it to have a material impact on the Company's 2008 consolidated financial statements.

In February 2007, the FASB issued SFAS No. 159, "The Fair Value Option for Financial Assets and Financial Liabilities." SFAS No. 159 allows entities the option to measure eligible financial instruments at fair value as of specified dates. Such election, which may be applied on an instrument by instrument basis, is typically irrevocable once elected. SFAS No. 159 is effective for fiscal years beginning after November 15, 2007. The Company adopted SFAS No. 159 on January 1, 2008, but has not yet elected the fair value option for any items permitted under SFAS No. 159.

NOTE 3 — TRADE ACCOUNTS RECEIVABLE, NET

The Company generally does not directly invoice energy suppliers and, therefore, reports a significant portion of its receivables as "unbilled." Unbilled accounts receivable represents management's best estimate of energy provided by the energy suppliers to the energy consumers for a specific completed time period at contracted commission rates.

Deservation 21

Bassinkan 21

Trade accounts receivable, net consists of the following:

	December 31,		
	2007	2006	
Unbilled accounts receivable	\$ 1,619,440	\$ 1,698,665	
Billed accounts receivable	263,793	4,133	
	1,883,233	1,702,798	
Allowance for doubtful accounts	(5,000)		
Trade accounts receivable,			
net	<u>\$ 1,878,233</u>	<u>\$ 1,702,798</u>	

NOTE 4 — PROPERTY AND EQUIPMENT, NET

Property and equipment, net consists of the following:

	<u> </u>			<u>, </u>
	_	2007	_	2006
Leasehold improvements	\$	65,451	\$	16,751
Equipment		426,781		225,051
Furniture and fixtures		418,974		194,552
		911,206		436,354
Less accumulated depreciation		(315,256)		(210, 422)
Property and equipment, net	\$	595,950	\$	225,932

Depreciation expense for the years ended December 31, 2007, 2006 and 2005 was \$104,834, \$62,156 and \$51,084, respectively. Property and equipment purchased under capital lease obligations at December 31, 2007 and 2006 was \$271,104 for both years, respectively. Accumulated depreciation for property and equipment purchased under capital lease was \$182,872 and \$137,758 at December 31, 2007 and 2006, respectively.

NOTE 5 — SERIES A REDEEMABLE CONVERTIBLE PREFERRED STOCK

On November 16, 2006, in connection with its initial public offering, the Company amended its Certificate of Incorporation to eliminate all references to the Series A Preferred Stock and authorize 5,000,000 shares of \$0,0001 par value undesignated preferred stock. No shares have been issued as of December 31, 2007 and 2006.

Prior to November 16, 2006, the Company had authorized 15,000,000 shares of \$0.0001 par value Series A Redeemable Convertible Preferred Stock (the "Series A Preferred"). The Series A Preferred had the following terms and conditions:

Dividends — In the event that the Company pays a dividend (other than a dividend payable solely in shares of common stock) on its common stock, the holders of shares of Series A will be entitled to a proportionate share of any such distribution as though it was the holder of the number of shares of common stock into which its share of Series A is convertible as of the date fixed for the determination of the holders of common stock entitled to receive the distribution. Automatic Conversion — Shares of Series A Preferred will automatically convert into common stock at the applicable conversion rate (one share of common stock for one share of Series A Preferred) in the event of: (1) the closing of an underwritten initial public offering with aggregate proceeds to the Company of at least \$10,000,000 and a per share price to the public of at least three times the initial conversion price (a "qualified public offering") or (2) the election of the holders of a majority of the outstanding shares of Series A Preferred.

Optional Conversion — The holders of Series A Preferred have the right to convert their shares, at any time, into shares of voting common stock on a one for one basis.

Liquidation Preference — In the event of liquidation or winding up of the Company, each holder of Series A Preferred shall be entitled to be paid out of the assets of the Company available for distribution to its stockholders in preference to the common stockholders, in an amount equal to the price per share of the Series A Preferred or \$0.146 (the "Liquidation Amount"). After the payment of this amount to holders of Series A Preferred, the common stockholders will be entitled to receive any remaining assets of the Company on a pro rata basis. The following qualify as events that would trigger a liquidation: (1) a merger or consolidation (other than one in which the stockholders, at the time of the merger or consolidation, continue to hold more than 50% of the stock of the surviving entity) or (2) a sale or exclusive license of all or substantially all of the assets or intellectual property of the Company.

Redemption Provisions — All shares of Series A Preferred will be redeemed by the Company, at the Liquidation Amount, at any time after January 31, 2009, if the holders of two-thirds of the outstanding shares of Series A Preferred request, by written notice to the Company, that the shares be redeemed. The shares will be redeemed within 60 days of the Company receiving such written notice.

The Company had classified the Series A Preferred outside of permanent equity due to its redemption feature being outside the control of the Company. The activity within the Series A Preferred shares and redemption value for the years ended December 31, 2007, 2006 and 2005 is as follows:

	Number of Shares	Redemption Value
Balance, December 31, 2004	10,433,504	1,494,499
Accretion of stock issuance costs		7,199
Balance, December 31, 2005	10,433,504	1,501,698
Accretion of stock issuance costs	— —	6,299
Conversion of Series A Preferred into common stock	(10,433,504)	(1,507,997)
Balance, December 31, 2006 and 2007		<u>\$</u>

NOTE 6 — COMMON STOCK

On March 31, 2003, the Board increased the number of shares of \$0.0001 par value common stock from 50,000,000 to 90,000,000, of which 75,000,000 were designated voting common stock ("Voting Common Stock") and 15,000,000 were designated as non-voting common stock ("Non-Voting Common Stock").

On October 16, 2006, the Company amended its Certificate of Incorporation to increase the authorized shares of voting common stock from 75,000,000 to 100,000,000 shares. The total number of shares of all classes of stock which the Company had the authority to issue after this amendment was 130,000,000, consisting of 15,000,000 shares of Non-Voting Common Stock, 100,000,000 shares of Voting Common Stock and 15,000,000 shares of Series A Preferred.

On November 16, 2006, the Company completed its initial public offering of 27,441,064 shares of common stock at a price of CDN\$1.05 (approximately US\$0.93) per share. The offering was comprised of 20,000,000 shares sold by the Company for gross proceeds to the Company of CDN\$21,000,000 and 7,441,064 shares sold by certain selling stockholders for gross proceeds to such stockholders of approximately CDN\$7,800,000. In connection with this offering, the Company amended its Certificate of Incorporation to increase the authorized shares of common stock from 100,000,000 to 150,000,000 shares. The total number of shares of all classes of stock which the Company has the authority to issue after this amendment is 155,000,000, consisting of 150,000,000 shares of common stock and 5,000,000 shares of undesignated preferred stock.

On December 5, 2006, the Company issued an additional 3,000,000 shares of common stock in accordance with an over-allotment option granted to the underwriters of the initial public offering at a price of CDN\$1.05 (approximately US \$0.92) per share for gross proceeds to the Company of CDN\$3,150,000. Net proceeds to the Company from the initial public offering and over-allotment option were approximately \$17.5 million, after deducting the underwriting discount of \$1.3 million and other direct costs of \$2.3 million.

As of December 31, 2007, 2006 and 2005, 82,399,241, 76,511,741 and 33,049,473 shares of Voting Common Stock were issued and outstanding, respectively, while 6,792,135 shares of Non-Voting Common Stock were issued and outstanding as of December 31, 2005. The holders of the Non-Voting Common Stock had the same privileges and rights of the holders of the Voting Common Stock, except for the right to vote. The 8,982,806 shares of Non-Voting Common Stock automatically converted to Voting Common Stock at the closing of the initial public offering of the Company's common stock.

244



On June 1, 2007, the Company issued 5,375,000 common shares of World Energy in connection with the acquisition of substantially all of the assets of EnergyGateway. The shares were valued at approximately \$6,536,000 based on the average closing price on the Toronto Stock Exchange for the two days before and after the announcement date of \$1.22 per share. Approximately 2.175 million of the shares that the Company delivered to EnergyGateway in connection with this transaction (or the proceeds upon any sale thereof) are being held in escrow for 18 months following the closing of the sale to secure various indemnification obligations of EnergyGateway and its members.

Guaranteed Return on Voting Common Stock

In 2002, a certain shareholder purchased shares of Voting Common Stock of the Company and the Company agreed to pay an 8% guaranteed return per year based on the initial investment of \$200,000 with payments to continue until the earlier of a liquid market for the Company's equity or August 31, 2006. In accordance with SFAS 133, "Accounting for Derivatives and Hedging Activities", the Company bifurcated the guaranteed return on the Voting Common Stock at its then deemed fair value of \$66,651 and recorded the guaranteed return as a liability. The payments on the guaranteed return were being amortized to principal and interest, using the interest method, through August 31, 2006. The Company has deemed the difference between the stated value and the fair value of the derivative feature to be immaterial and has not recorded a "mark-to-market" adjustment during the life of the feature.

Fair Value of Common Stock

The Company has granted common stock, common stock warrants, and common stock option awards ("equity awards") to consultants, employees, debt holders, and others since its inception. The Company's determination of the fair value of the underlying common stock is a significant aspect in accounting for these aforementioned equity awards in accordance with generally accepted accounting principles.

During each of the two years in the period ended December 31, 2006, the Company and the Board had determined the fair value of the Series A Preferred, the Voting and Non-Voting Common Stock at each issuance of such equity awards. Prior to the consummation of the Company's initial public offering the fair value of the Voting and Non-Voting Common Stock had been determined by the Board at the measurement date of each of the aforementioned equity awards, based on a variety of different factors including, but not limited to the Company's financial position and historical financial performance, the status of technological developments within the Company, the composition and ability of the current engineering, operations, and management team, an evaluation and benchmark of the Company's competition, the current climate in the marketplace, the illiquid nature of the common stock, arms-length sales and anticipated sales of the Company's capital stock (including preferred stock), the effect of the rights and preferences of preferred shareholders, independent third party appraisals, and the prospects of a liquidity event, among others.

Treasury Stock

On March 14, 2006, the Company purchased 25,000 of its Voting Common Stock from certain shareholders at \$2.00 per share. The excess of the purchase price and the then deemed fair value on the date of the purchase of \$0.50 per share, or \$37,500, has been charged to general and administrative expense. On June 19, 2006, the Board voted to retire all its treasury stock (700,346 shares) which had a cost of \$164,453.

Common Stock Warrants

At December 31, 2007, the Company had outstanding warrants to purchase 2,000,814 shares of common stock at a weighted average exercise price of \$0.54. During 2005, the Company issued warrants to purchase 3,000,000 shares of Voting Common Stock in connection with the subordinated note issued to Massachusetts Capital Resource Company, or MCRC. Of the warrants that were issued in connection with this note, 600,000 were vested upon issuance and the remaining 2,400,000 did not vest and subsequently expired due to the successful consummation of the Company's initial underwritten public offering of its common stock at \$0.40 in connection with the redemption of the subordinated note by the Company.

On November 16, 2006 and December 5, 2006, the Company granted the underwriters of the initial public offering warrants to purchase up to 1,000,000 and 150,000 shares of common stock, respectively at the initial public offering price. The warrants are exercisable at the 1-year anniversary of the warrants issuance (November 16, 2007 and December 5, 2007, respectively) and expire 18 months from the date of the one year anniversary (May 15, 2009 and June 4, 2009, respectively). These warrants allow for net share settlement and certain of the holders of these warrants have piggy-back registration rights. The Company valued the warrants at \$274,129 and recorded them as a component of additional paid-in capital. The following table summarizes the Company's warrant activity:

	Shares	Weighted Average Exercise Price
Warrants outstanding, December 31, 2004	1,321,772	\$ 0.03
Granted	3,000,000	\$ 0.40
Exercised		\$
Canceled/expired		\$ —
Warrants outstanding, December 31, 2005	4,321,772	\$ 0.29
Granted	1,150,000	\$ 0.92
Exercised	(1,070,958)	\$ 0.24
Canceled/expired	(2,400,000)	\$ 0.40
Warrants outstanding, December 31, 2006 and 2007	2,000,814	\$ 0.54

The weighed average remaining contractual life of warrants outstanding is 1.3 years and 2.3 years as of December 31, 2007 and 2006, respectively.

Stock Options

On May 9, 2003, the Company formally approved the adoption of the 2003 Stock Incentive Plan (the "2003 Plan") and authorized 15,000,000 shares for the grant of options to purchase shares of Voting Common Stock. As of December 31, 2007, 4,163,029 shares of common stock were reserved under the 2003 Plan. No further grants are allowed under the 2003 Plan. On November 16, 2006, the Company adopted the 2006 Stock Incentive Plan (the "2006 Plan") and authorized 4,738,161 shares of common stock for grant thereunder. A summary of option activity under the Plans is as follows:

	Shares	A E	eighted verage xercise Price
Outstanding at December 31, 2004	6,664,200	\$	0.03
Granted	440,000	\$	0.24
Canceled	(1,392,500)	\$	0.03
Exercised	(240,000)	\$	0.03
Outstanding at December 31, 2005	5,471,700	\$	0.04
Granted	1,262,000	\$	0.79
Canceled	(60,000)	\$	0.38
Exercised	(2,190,671)	\$	0.03
Outstanding at December 31, 2006	4,483,029	\$	0.26
Granted	3,020,000	\$	1.24
Canceled	(687,500)	\$	0.82
Exercised	(112,500)	\$	0.03
Outstanding at December 31, 2007	6,703,029	\$	0.65

A summary of options outstanding and options exercisable as of December 31, 2007 is as follows:

	Options Outstanding				Options Exercisable			
		Weighted	-			Weighted		
		Average				Average		
Range of		Remaining	I	Aggregate	Number	Remaining	1	Aggregate
Exercise		Contractual		Intrinsic	Of Shares	Contractual		Intrinsic
Prices	Options	<u> </u>		Value	Exercisable	Life	_	Value
\$0.02 - \$0.23	2,651,029	2.82 Years	\$	2,081,058	2,646,029	2.82 Years	\$	2,077,133
\$0.24 - \$1.12	1,612,000	4.78 Years		391,300	670,625	4.87 Years		229,263
\$1.13 - \$1.34	2,440,000	6.36 Years			30,000	5.96 Years		
	6,703,029	4.58 Years	\$	2,472,358	3,346,654	3.26 Years	\$	2,306,396

The aggregate intrinsic value in the table above represents the total intrinsic value of our outstanding options and exercisable options as of December 31, 2007. The aggregate intrinsic value of options exercised during the year ended December 31, 2007 was approximately \$126,000. At December 31, 2007, the weighted average exercise price of options outstanding and exercisable was \$0.64 and \$0.13, respectively. The weighted average fair value of option grants for the years ended December 31, 2007 and 2006 was \$0.58 and \$0.36, respectively.

Restricted Stock

A summary of restricted stock activity under the 2006 Plan for the year ended December 31, 2007 is as follows:

	Shares	Weighted Average Grant Price	
Outstanding at December 31, 2006		<u>s </u>	_
Granted	1,928,500	\$ 1.13	3
Cancelled	(80,000)	\$ 1.21	1
Vested	(400,000)	\$ 1.21	1
Unvested at December 31, 2007	1,448,500	\$ 1.10	0

There were 8,151,529 shares reserved for issuance under these plans, and 349,661 shares available for grant under the 2006 Plan at December 31, 2007.

NOTE 7 - RELATED PARTIES

In 2006, the Company entered into a consulting agreement with a member of the Board to assist the Company with strategic planning. Costs incurred during the year ended December 31, 2007 were approximately \$209,000 and were charged to general and administrative expense. Amounts owed by the Company at December 31, 2007 on behalf of the agreement were \$9,700.

NOTE 8 — INCOME TAXES

The components of the net deferred tax asset are as follows:

	December 31,			
	20)7	2006	
Depreciation and amortization	\$	66,694 \$	22,242	
Commission income	(4	162,079)	(266,310)	
Accruals and reserves	,	753,407	155,263	
Prepaids		(13,428)	20,099	
Net operating loss carryforwards	4,2	254,013	1,769,489	
	4,0	598,607	1,700,783	
Valuation allowance	(4,6	5 98,60 7)	(639,063)	
	S	<u> </u>	1,061,720	

The provision for income taxes is comprised of the following:

	Ye					
	2007		2006			2005
Current income tax expense:						
Federal	\$		\$	—	\$	
State						
				_		
Deferred income tax expense (benefit):						-- -
Federal	823	,703	C.	232,091)		(588,120)
State	238	,017		(72,137)		(165,880)
	1,061	,720	(3	304,228)		(754,000)
Total income tax expense (benefit)	<u>\$ 1,061</u>	,720	<u>\$ (</u>	304,228)	\$	(754,000)

A reconciliation of the Company's federal statutory tax rate to its effective rate is as follows:

	Years	Years Ended December 31,			
	2007	2006	2005		
Income tax at federal statutory rate Increase (decrease) in tax resulting from:	(34.0)%	(34.0)%	34.0%		
State taxes, net of federal benefit	(6.3)%	(6.3)%	6.3%		
Permanent differences	0.7%	2.1%	2.3%		
Change in valuation allowance	53.6%	0.0%	(270.0)%		
	14.0%	(38.1)%	(227.4)%		



As of December 31, 2007, the Company has federal net operating loss carryforwards of approximately \$10,800,000 which begin to expire in 2022, and state net operating loss carryforwards of approximately \$10,300,000, which begin to expire in 2008. A valuation allowance is established, if it is more likely than not, that all or a portion of the deferred tax asset will not be realized.

The Company accounts for income taxes in accordance with SFAS No. 109, "Accounting for IncomeTaxes" ("SFAS No. 109"). Deferred tax assets and liabilities are determined at the end of each period based on the future tax consequences that can be attributed to net operating loss carryforwards, as well as differences between the financial statement carrying amounts of the existing assets and liabilities and their respective tax basis. Deferred income tax expense or credits are based on changes in the asset or liability from period to period. Valuation allowances are provided if, based on the weight of available evidence, it is more likely than not that some or all of the deferred tax assets will not be realized. The realization of deferred tax assets is dependent upon the generation of future taxable income. In determining the valuation allowance, the Company considers past performance, expected future taxable income, and qualitative factors which are considered to be appropriate in estimating future taxable income. The Company's forecast of expected future taxable income is for future periods that can be reasonably estimated. The Company has historically utilized a three year forecasting window to determine its expected future taxable income. As of December 31, 2006, the Company had cumulative net income for the three years then ended and the expected future taxable income anticipated from 2007 through 2009 was deemed adequate on a present value basis to benefit the majority of our deferred tax assets (primarily net operating loss carryforwards). During 2007, the Company accelerated its growth in excess of its initial expectations to address the strategic initiatives outlined at the time of our initial public offering. The Company grew from 20 employees as of December 31, 2006 to 63 as of December 31, 2007, acquired one of its main competitors in the retail energy market, and developed its wholesale and environmental commodity product lines. These initiatives increased the Company's loss for 2007 resulting in a three-year cumulative loss as of December 31, 2007 and increased its operating expenses. A three-year cumulative loss is significant negative evidence in considering whether the deferred tax assets are realizable and generally precludes relying on projections of future taxable income to support the recovery of deferred tax assets. While the Company expects its near term future income will be adequate on a present value basis to benefit substantially all of its current deferred tax assets, the recent three-year cumulative loss is given more weight than projected future income when determining the need for a valuation. Therefore, during the fourth quarter of 2007, the Company recorded a full valuation allowance against its deferred tax assets of approximately \$4.1 million, and maintains a cumulative valuation allowance of approximately \$4.7 million at December 31, 2007.

The amount of the net deferred tax asset considered realizable at December 31,2007 could be reduced or increased in the near term if estimates of future taxable income during the carryforward period change.

The Company has reviewed the tax positions taken, or to be taken, in its tax returns for all tax years currently open to examination by the taxing authorities in accordance with the recognition and measurement standards of FIN 48. At December 31, 2007, there are no expected material, aggregate tax effect of differences between tax return positions and the benefits recognized in the financial statements.

Under the provisions of the Internal Revenue Code, certain substantial changes in the Company's ownership may have limited or may limit in the future the amount of net operating loss carryforwards which could be utilized annually to offset future taxable income and income tax liabilities. The amount of any annual limitation is determined based upon the Company's value prior to an ownership change.

NOTE 9 — COMMITMENTS

Leases

The Company leases certain equipment under capital leases that expire through May 2011 and are collateralized by the related equipment. The Company has accounted for these leases using an incremental borrowing rate of 8%. The Company maintains operating leases for office space in seven locations in the United States, paid in installments due the beginning of each month and that expire through July 2012. Future aggregate minimum payments under capital and operating leases as of December 31, 2007 were as follows:

	Capi Lea		Operating <u>Leases</u>	
2008 2009	\$	47,223	\$	304,537
2009		44,745 3,192		295,381 152,275
2011 2012		798		135,630
Total future minimum lease payments		95,958	\$	42,973 930,796
Less: amounts representing interest		(8,112)		
Present value of future minimum lease payments		87,846		
Less: current portion Capital lease obligation, net of current portion	\$	41,624 46,222		

The accompanying statement of operations for the years ended December 31, 2007, 2006, and 2005 includes \$242,813, \$118,902 and \$78,800 of rent expense, respectively.

Service Agreement

In 2007, the Company entered into a service agreement with an unrelated party for a hosting environment and dedicated server for the Company's online energy procurement software. The terms of the agreement requires quarterly and monthly payments amounting to approximately \$8,400 per quarter. The agreement expires in May 2008.

NOTE 10 - ACQUISITION

On June 1, 2007, the Company acquired substantially all of the assets of EnergyGateway for \$4,951,758 in cash and 5,375,000 common shares of World Energy plus the assumption of certain liabilities. The shares were valued at approximately \$6,536,000 based on the average closing price on the Toronto Stock Exchange for the two days before and after the announcement date of \$1.22 per share. Approximately 2.175 million of the shares that the Company delivered to EnergyGateway in connection with this transaction (or the proceeds upon any sale thereof) are being held in escrow for 18 months following the closing of the sale to secure various indemnification obligations of EnergyGateway and its members. EnergyGateway, located near Columbus, Ohio, was an online broker of natural gas and electricity for commercial and industrial consumers in the United States. It provided its customers with professional energy purchasing services, including access to its proprietary 'post and bid' online procurement system called EGate.

The total purchase price was \$11,440,358, net of cash acquired, and includes related acquisition costs of \$229,500, plus the assumption of certain liabilities of \$138,128. The EnergyGateway operations have been included within our consolidated results from June 1, 2007. The cost of the acquisition was allocated to the assets and liabilities assumed based on estimates of their respective fair values at the date of acquisition resulting in net assets of \$8,261,657. The excess purchase price of \$3,178,701 has been recorded as goodwill which is fully deductible for tax purposes. Management is responsible for the valuation of net assets acquired and considered a number of factors, including valuations and appraisals, when estimating the fair values and estimated useful lives of acquired assets and liabilities. During the third quarter of 2007, the Company refined estimates recorded in the second quarter of 2007 and finalized the valuation of intangible assets through a reduction of \$660,000 in customer relationships resulting in a corresponding increase to goodwill.

The Company has allocated the purchase price for the acquisition as follows:

Current assets	\$ 247,938
Fixed assets	18,755
Other assets	3,092
Intangible assets	
Non-compete agreements	680,000
Customer relationships	4,740,000
Customer contracts	500,000
Supplier agreements	380,000
Developed technology	1,830,000
Goodwill	3,178,701
Current liabilities	 (138,128)
Net assets acquired	\$ 11,440,358

The intangible assets, excluding goodwill, are being amortized on a straight-line basis over their weighted average lives as follows: non-compete agreements -1 to 5 years; customer relationships -10 years; customer contracts -3 years; supplier agreements -5 years; and developed technology -5 years.

The following unaudited pro forma information assumes that the acquisition of EnergyGateway assets had been completed as of January 1, 2006:

	Years Ended December			
	2007			2006
Revenues	\$	10,461,780	\$	8,617,678
Net loss available to common stockholders Loss per share:	\$	(9,448,352)	\$	(1,719,847)
Net loss available to common				
stockholders per share – basic	\$	(0.12)	\$	(0.03)
Net loss available to common				
stockholders per share – diluted	\$	(0.12)	\$	(0.03)
Weighted average number of common and common equivalent shares outstanding – basic		82,017,220		50,951,477
Weighted average number of common and common equivalent shares outstanding – diluted		82,017,220		50,951,477

The pro forma financial information is not necessarily indicative of the results to be expected in the future as a result of the acquisition of EnergyGateway.

NOTE 11 --- 401(k) PLAN

During the fourth quarter of 2006, the Company established the World Energy Solutions, Inc. 401(k) Plan (the "401(k) Plan") under Section 401(k) of the Internal Revenue Code covering all eligible employees. Employees of the Company may participate in the 401(k) Plan after reaching the age of 21. The Company may make discretionary matching contributions as determined from time to time. Employee contributions vest immediately, while Company matching contributions begin to vest after one year service and continue to vest at 20% per year over the next five years. To date, the Company has not made any discretionary contributions to the 401(k) Plan.

- 64 -

EXHIBIT INDEX

<u>Exhibit</u>

Description

- 2.1 Asset Purchase Agreement by and among World Energy Solutions, Inc., EnergyGateway, LLC and the Members of EnergyGateway, LLC dated May 23, 2007 (incorporated by reference to Exhibit 99.1 to our report on Form 8-K filed May 24, 2007 (File No. 333-136528)).
- 3.1 Form of Amended and Restated Certificate of Incorporation of World Energy (incorporated by reference to Exhibit 3.4 to our Registration Statement of Form S-1(File No. 333-136528)).
- 3.2 Form of Amended and Restated By-laws of World Energy (incorporated by reference to Exhibit 3.5 to our Registration Statement of Form S-1(File No. 333-136528)).
- 4.1 Specimen Certificate evidencing shares of common stock (incorporated by reference to Exhibit 4.1 to our Registration Statement of Form S-1(File No. 333-136528)).
- 10.1+ 2003 Stock Incentive Plan (incorporated by reference to Exhibit 10.1 to our Registration Statement of Form S-1(File No. 333-136528)).
- 10.2+ 2006 Stock Incentive Plan (incorporated by reference to Exhibit 10.2 to our Registration Statement of Form S-1(File No. 333-136528)).
- 10.3 Note and Warrant Purchase Agreement, dated November 7, 2005, between World Energy and Massachusetts Capital Resource Company (incorporated by reference to Exhibit 10.3 to our Registration Statement of Form S-1(File No. 333-136528)).
- 10.4 Subordinated Note due 2013, dated November 7, 2005 (incorporated by reference to Exhibit 10.4 to our Registration Statement of Form S-1(File No. 333-136528)).
- 10.5 Voting Common Stock Purchase Warrant, dated November 7, 2005 (incorporated by reference to Exhibit 10.5 to our Registration Statement of Form S-1(File No. 333-136528)).
- 10.6 Form of Common Stock Purchase Warrants (incorporated by reference to Exhibit 10.6 to our Registration Statement of Form S-1(File No. 333-136528)).
- 10.7 Solicitation/Contract/Order for Commercial Items, dated September 28, 2005, between U.S. General Services Administration and World Energy (incorporated by reference to Exhibit 10.7 to our Registration Statement of Form S-1(File No. 333-136528)).
- 10.8 Agreement to Provide Software and Support for a Reverse Energy Auction Procurement to the Maryland Department of General Services, dated March 16, 2006, by and between World Energy and the State of Maryland (incorporated by reference to Exhibit 10.8 to our Registration Statement of Form S-1(File No. 333-136528)).
- 10.9 Contract, dated January 9, 2006, by and between Montgomery County, Maryland and World Energy (incorporated by reference to Exhibit 10.9 to our Registration Statement of Form S-1(File No. 333-136528)).
- 10.10 Emergency Purchase/Interim Agreement, dated March 28, 2006, by and between the Commonwealth of Pennsylvania, Department of General Services and World Energy (incorporated by reference to Exhibit 10.10 to our Registration Statement of Form S-1(File No. 333-136528)).
- 10.11 Professional Services Agreement, dated June 1, 2005, between World Energy and Science Applications International Corporation (incorporated by reference to Exhibit 10.11 to our Registration Statement of Form S-1(File No. 333-136528)).
- 10.12 Escrow Agreement (incorporated by reference to Exhibit 10.12 to our Registration Statement of Form S-1(File No. 333-136528)).
- 10.13+ Offer letter agreement, dated October 1, 2003, between World Energy and Philip V. Adams (incorporated by reference to Exhibit 10.13 to our Registration Statement of Form S-1(File No. 333-136528)).
- 10.14+ Offer letter agreement, dated April 5, 2006, between World Energy and James Parslow (incorporated by reference to Exhibit 10.14 to our Registration Statement of Form S-1(File No. 333-136528)).
- 10.15 Lease, dated September 8, 2004, between Sovereign Bank and World Energy (incorporated by reference to Exhibit 10.15 to our Registration Statement of Form S-1(File No. 333-136528)).

- 10.16 Lease, dated June 21, 2006, between Roosevelt Land, LP and World Energy (incorporated by reference to Exhibit 10.16 to our Registration Statement of Form S-1(File No. 333-136528)).
- 10.17 Form of Warrant being issued to Underwriters (incorporated by reference to Exhibit 10.17 to our Registration Statement of Form S-1(File No. 333-136528)).
- 10.18 Form of Registration Rights Agreement with Underwriters (incorporated by reference to Exhibit 10.18 to our Registration Statement of Form S-1(File No. 333-136528)).
- 21.1* List of Subsidiaries
- 23.1* Consent of UHY LLP, Independent Registered Public Accounting Firm
- 31.1* Certification of the Chief Executive Officer pursuant to Rule 15d-14(a) under the Securities Exchange Act.
- 31.2* Certification of the Chief Financial Officer pursuant to Rule 15d-14(a) under the Securities Exchange Act.

- 65 -

- 32.1* Certification of the Chief Executive Officer pursuant to Rule 15d-14(b) under the Securities Exchange Act.
- 32.2* Certification of the Chief Financial Officer pursuant to Rule 15d-14(b) under the Securities Exchange Act.
 - * Filed herewith
 - + Indicates a management contract or any compensatory plan, contract or arrangement

L

SUBSIDIARIES OF REGISTRANT

Subsidiary

State of Incorporation

World Energy Securities Corp.

Massachusetts

CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Board of Directors and Stockholders of World Energy Solutions, Inc.:

We consent to the incorporation by reference in the Registration Statement on Form S-8 (File No. 333-1140014) and Form S-3 (File No. 333-147301) of our reports dated March 12, 2008, relating to the consolidated financial statements of World Energy Solutions, Inc. (which report expresses an unqualified opinion and includes an explanatory paragraph relating to the Company's change in its method of accounting for share based payments upon the adoption of Statement of Financial Accounting Standard No. 123(R), *"Share-Based Payment"*, effective January 1, 2006) and the effectiveness of World Energy Solutions, Inc.'s internal control over financial reporting dated March 12, 2008, appearing in this Annual Report to Shareholders, which is included in this Annual Report on Form 10-K of World Energy Solutions, Inc., for the year ended December 31, 2007.

/s/ UHY LLP

Boston, Massachusetts March 12, 2008

CERTIFICATION PURSUANT TO §302 OF THE SARBANES-OXLEY ACT OF 2002

I, Richard Domaleski, Chief Executive Officer of World Energy Solutions, Inc. (the "Company"), certify that:

1. I have reviewed this annual report on Form 10-K of the Company.

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report.

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the Company as of, and for, the periods presented in this report.

4. The Company's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the Company and have:

a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the Company, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

c) Evaluated the effectiveness of the Company's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

d) Disclosed in this report any change in the Company's internal control over financial reporting that occurred during the Company's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the Company's internal control over financial reporting.

5. The Company's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the Company's auditors and the audit committee of the Company's board of directors (or persons performing the equivalent functions):

a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the Company's ability to record, process, summarize and report financial information; and

b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the Company's internal control over financial reporting.

Dated:

March 13, 2008

By: /s/ Richard Domaleski

Richard Domaleski Chief Executive Officer

CERTIFICATION PURSUANT TO §302 OF THE SARBANES-OXLEY ACT OF 2002

I, James Parslow, Chief Financial Officer of World Energy Solutions, Inc. (the "Company"), certify that:

1. I have reviewed this annual report on Form 10-K of the Company.

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report.

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the Company as of, and for, the periods presented in this report.

4. The Company's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the Company and have:

a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the Company, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

c) Evaluated the effectiveness of the Company's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

d) Disclosed in this report any change in the Company's internal control over financial reporting that occurred during the Company's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the Company's internal control over financial reporting.

5. The Company's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the Company's auditors and the audit committee of the Company's board of directors (or persons performing the equivalent functions):

a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the Company's ability to record, process, summarize and report financial information; and

b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the Company's internal control over financial reporting.

Dated:

March 13, 2008

By: /s/ James Parslow

James Parslow Chief Financial Officer

CERTIFICATION PURSUANT TO 18 U.S.C. §1350, AS ADOPTED PURSUANT TO §906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the annual report of World Energy Solutions, Inc. (the "Company") on Form 10-K (the "Report") for the year ended December 31, 2007 as filed with the Securities and Exchange Commission on the date hereof, I, Richard Domaleski, Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. §1350, as adopted pursuant to §906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated:

March 13, 2008

By: /s/ Richard Domaleski

Richard Domaleski Chief Executive Officer

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signatures that appear in typed form within the electronic version of this written statement required by Section 906, has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.

CERTIFICATION PURSUANT TO 18 U.S.C. §1350, AS ADOPTED PURSUANT TO §906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the annual report of World Energy Solutions, Inc. (the "Company") on Form 10-K (the "Report") for the year ended December 31, 2007 as filed with the Securities and Exchange Commission on the date hereof, I, James Parslow, Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. §1350, as adopted pursuant to §906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated:

March 13, 2008

By: /s/ James Parslow

James Parslow Chief Financial Officer

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signatures that appear in typed form within the electronic version of this written statement required by Section 906, has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.

Created by 10KWizard www.10KWizard.com

Exhibit C-2 "SEC Filings"

World Energy Solutions, Inc. has attached Form 10-Q and Form 8-K filed with the SEC.

FORM 10-Q

World Energy Solutions, Inc. - WLDE

Filed: November 06, 2008 (period: September 30, 2008)

Quarterly report which provides a continuing view of a company's financial position

Table of Contents

10-Q - FORM 10-Q DATED SEPTEMBER 30, 2008

<u> PART I.</u>

Item 1. Financial Statements (Unaudited)

<u>PART I</u>

<u>ltem 1.</u>	Financial Statements
<u>Item 2.</u>	Management s Discussion and Analysis of Financial Condition and Results of Operations
<u>Item 3.</u>	Quantitative and Qualitative Disclosure about Market Risk
<u>Item 4.</u>	Controls and Procedures

<u>PART II</u>

<u>Item 1.</u>	Legal Proceedings
Item 1A.	Risk Factors
<u>Item 2.</u>	Unregistered Sales of Equity Securities and Use of Proceeds
<u>Item 3,</u>	Defaults Upon Senior Securities
<u>Item 4.</u>	Submission of Matters to a Vote of Security Holders
<u>Item 5.</u>	Other Information
<u>Item 6.</u>	<u>Exhibits</u>
<u>SIGNATUR</u>	ES
<u>EXHIBIT IN</u>	DEX
<u>EX-31.1 (30</u>	2 CERTIFICATION - CEO)

EX-31.2 (302 CERTIFICATION - CFO)

EX-32.1 (906 CERTIFICATION - CEO)

EX-32.2 (906 CERTIFICATION - CFO)

Source: World Energy Solutio, 10-Q, November 06, 2008

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

Quarterly Report Pursuant to Section 13 or 15(d) of the Securities Exchange Act of \mathbb{X} 1934 For the quarterly period ended September 30, 2008; or

Transition Report Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934 For the transition period from to

Commission file number: 333-136528

World Energy Solutions, Inc.

(Exact name of registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation or organization)

446 Main Street Worcester, Massachusetts

(Address of principal executive offices)

(Zip Code)

508-459-8100

(Registrant's telephone number, including area code)

Indicate by check mark whether the registrant: (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the past 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes 🗵 No 🗖

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer or a smaller reporting company. See definition of "large accelerated filer", "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. (Check one):

Large Accelerated Filer

Non-Accelerated Filer (Do not check if a smaller reporting company)

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). No 🗵 Yes 🛛

Indicate the number of shares outstanding of each of the issuer's classes of common stock, as of the latest practicable date: As of November 1, 2008, the registrant had 84,814,728 shares of common stock outstanding.

266

(I.R.S. Employer Identification Number)

01608

Accelerated Filer

Smaller Reporting Company

04-3474959

TABLE OF CONTENTS

PART I.	Financial Information	<u>Page</u>
Item 1.	Financial Statements (Unaudited)	
	Condensed Consolidated Balance Sheets as of September 30, 2008 and December 31, 2007	1
	Condensed Consolidated Statements of Operations for the Three and Nine Months Ended September 30, 2008 and 2007	2
	Condensed Consolidated Statements of Cash Flows for the Nine Months Ended September 30, 2008 and 2007	3
	Notes to Condensed Consolidated Financial Statements	4
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	10
Item 3.	Quantitative and Qualitative Disclosure about Market Risk	21
Item 4.	Controls and Procedures	21

PART II. Other Information

ltem 1.	Legal Proceedings	22
Item 1A.	Risk Factors	22
Item 2.	Unregistered Sales of Equity Securities and Use of Proceeds	22
Item 3.	Defaults Upon Senior Securities	22
Item 4.	Submission of Matters to a Vote of Security Holders	22
Item 5.	Other Information	22
Item 6.	Exhibits	23

PART I FINANCIAL INFORMATION

Item 1. Financial Statements

WORLD ENERGY SOLUTIONS, INC. CONDENSED CONSOLIDATED BALANCE SHEETS

		September 30, 2008		December 31, 2007	
Assets	(Unaudited)				
Current assets					
Cash and cash equivalents	\$	1,756,829	\$	7,001,884	
Trade accounts receivable, net		2,962,684		1,878,233	
Prepaid expenses and other current assets		403,961		338,049	
Total current assets		5,123,474		9,218,166	
Property and equipment, net		524,275		595,950	
Capitalized software, net		587,183		463,888	
Intangibles, net		6,285,637		7,316,916	
Goodwill		3,178,701		3,178,701	
Other assets		27,594		26,944	
Total assets	\$	15,726,864	<u>\$</u>	20,800,565	
Liabilities and Stockholders' Equity					
Current liabilities					
Accounts payable	\$	545,986	S	980,488	
Accrued commissions		669,801		712,919	
Accrued compensation		1,381,742		1,516,418	
Accrued expenses		401,602		279,162	
Deferred revenue and customer advances		851,237		363,933	
Capital lease obligations		41,630		41,624	
Total current liabilities		3,891,998		3,894,544	
Capital lease obligations, net of current portion		14,686		46,222	
Total liabilities	_	3,906,684	_	3,940,766	
Commitments and contingencies					
Stockholders' equity:					
Preferred stock, \$0.0001 par value; 5,000,000 shares authorized; no shares issued or					
outstanding					
Common stock, \$0.0001 par value; 150,000,000 shares authorized; 83,509,840 shares issued and 83,438,825 shares outstanding at September 30, 2008, and 82,399,241 shares issued					
and outstanding at December 31, 2007		8,351		8,240	
Additional paid-in capital		30,499,580		29,704,831	
Accumulated deficit		(18,609,351)		(12,853,272)	
Treasury stock, at cost; 71,015 shares at September 30, 2008 and no shares at December 31, 2007		(78,400)			
Total stockholders' equity		11,820,180		16,859,799	
Total liabilities and stockholders' equity	\$	15,726,864	\$	20,800,565	

See accompanying notes to condensed consolidated financial statements.

WORLD ENERGY SOLUTIONS, INC. CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (Unaudited)

	Three Months Ended September 30,		Nine Months Ended September 30,		
	2008	2007	2008	2007	
Revenue:		·····	·······		
Brokerage commissions and transaction fees	\$ 2,984,110	\$ 2,350,516	\$ 8,184,250	\$ 5,674,775	
Management fees	305,405	324,827	980,477	428,897	
Total revenue	3,289,515	2,675,343	9,164,727	6,103,672	
Cost of revenue	1,113,325	787,517	3,626,237	1,853,660	
Gross profit	2,176,190	1,887,826	5,538,490	4,250,012	
Operating expenses:					
Sales and marketing	2,355,907	3,028,912	7,770,567	5,965,261	
General and administrative	1,041,321	1,876,618	3,562,946	4,274,415	
Total costs and expenses	3,397,228	4,905,530	11,333,513	10,239,676	
Operating loss	(1,221,038)	(3,017,704)	(5,795,023)	(5,989,664)	
Interest income, net	5,218	108,195	38,944	482,438	
Loss before income taxes	(1,215,820)	(2,909,509)	(5,756,079)	(5,507,226)	
Income tax benefit		1,099,125		2,079,836	
Net loss	<u>\$ (1,215,820</u>)	<u>\$ (1,810,384</u>)	\$ (5,756,079)	<u>\$ (3,427,390</u>)	
Loss per share:					
Net loss per common share -basic	\$ (0.01)	\$(0.02) \$	\$ (0.07)	<u>\$</u> (0.04)	
Net loss per common share -diluted	<u>\$ (0.01</u>)	<u>\$ (0.02</u>)	\$ (0.07)	<u>\$ (0.04</u>)	
Weighted average shares outstanding -basic and diluted	83,215,951	81,943,263	82,875,299	78,932,803	

See accompanying notes to condensed consolidated financial statements.

2

WORLD ENERGY SOLUTIONS, INC. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited)

	Nine Months Ended September 30,			
	-	2008		2007
Cash flows from operating activities:	-			
Net loss	\$	(5,756,079)	\$	(3,427,390)
Adjustments to reconcile net loss to net cash used in operating activities:				
Depreciation and amortization		1,313,692		608,213
Deferred taxes				(2,079,836)
Share-based compensation		770,383		779,975
Changes in assets and liabilities, net of the effects of acquisition:				
Trade accounts receivable		(1,084,451)		(389,411)
Prepaid expenses and other assets		(66,562)		(146,725)
Accounts payable		(434,502)		431,502
Accrued commissions		(43,118)		(386,089)
Accrued compensation		(134,676)		844,077
Accrued expenses		122,440		48,465
Deferred revenue and customer advances		487,304		54,398
Net cash used in operating activities	-	(4,825,569)		(3,662,821)
Cash flows from investing activities:				
Cost incurred in software development		(293,329)		(329,995)
Net cash paid in acquisition of EnergyGateway, net of cash acquired		_		(4,904,358)
Purchases of property and equipment	_	(40,704)		(355,272)
Net cash used in investing activities	_	(334,033)	_	(5,589,625)
Cash flows from financing activities:				
Proceeds from exercise of stock options		16,669		—
Proceeds from exercise of stock warrants		7,808		
Purchase of treasury stock		(78,400)		
Principal payments on capital lease obligations		(31,530)		(41,724)
Net cash used in financing activities	_	(85,453)		(41,724)
Net decrease in cash and cash equivalents		(5,245,055)		(9,294,170)
Cash and cash equivalents, beginning of period		7,001,884		17,483,886
Cash and cash equivalents, end of period	5	1,756,829	\$	8,189,716
Supplemental disclosure of cash flow information:	-		<u> </u>	
Net cash received for interest	¢	43,823	c	525,691
	\$	43,023	÷	525,071
Non-cash investing and financing activities:			*	< 200 000
Fair value of common stock issued in acquisition of EnergyGateway, net of issuance costs	<u>\$</u>		\$	6,523,232
Reversal of accrued expenses related to initial public offering of common stock	<u>\$</u>		<u>\$</u>	48,466

See accompanying notes to condensed consolidated financial statements.

3

WORLD ENERGY SOLUTIONS, INC. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

1. Nature of Business and Basis of Presentation

World Energy is an energy and environmental commodities brokerage company that has developed online auction platforms, the World Energy Exchange and the World Green Exchange. On the World Energy Exchange, retail energy consumers (commercial, industrial and government) and wholesale energy participants (utilities and retail suppliers) in the United States are able to negotiate for the purchase or sale of electricity and other energy resources from competing energy suppliers which have agreed to participate on our auction platform in a given auction. Although the Company's primary source of revenue is from brokering electricity and natural gas, the Company adapted its World Energy Exchange auction platform to accommodate the brokering of green power in 2003 (i.e., electricity generated by renewable resources), wholesale electricity in 2004 and certain other energy-related products in 2005. In 2007, the Company created the World Green Exchange based on the World Energy Exchange technology and business process. On the World Green Exchange, buyers and sellers negotiate for the purchase or sale of environmental commodities such as Renewable Energy Certificates, Verified Emissions Reductions, and Certified Emissions Reductions and Regional Greenhouse Gas Initiative allowances.

2. Interim Financial Statements

The accompanying unaudited interim condensed consolidated financial statements have been prepared pursuant to the rules and regulations of the Securities and Exchange Commission regarding interim reporting. Accordingly, they do not include all of the information and footnotes required by accounting principles generally accepted in the United States of America ("generally accepted accounting principles") for complete financial statements and should be read in conjunction with the audited financial statements included in the Company's Annual Report on Form 10-K for the year ended December 31, 2007.

In the opinion of the Company's management, the unaudited interim condensed consolidated financial statements have been prepared on the same basis as the audited consolidated financial statements and include all adjustments consisting of normal recurring adjustments and accruals necessary for the fair presentation of the Company's financial position as of September 30, 2008, the results of its operations for the three and nine months ended September 30, 2008 and 2007 and its cash flows for the nine months ended September 30, 2008 and 2007, respectively. The results of operations for the three and nine months ended September 30, 2008 are not necessarily indicative of the results to be expected for the fiscal year ending December 31, 2008.

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Accordingly, actual results could differ from those estimates.

The Company's most judgmental estimates relate to revenue recognition and the estimate of actual energy purchased from the energy supplier by the end user of such energy; the fair value of the Company's equity securities where there was no ready market for the purchase and sale of those shares prior to our November 16, 2006 initial public offering of common stock; the valuation of intangible assets and goodwill; and estimates of future taxable income as it relates to the realization of the Company's net deferred tax assets. The effect of those estimates could have a material impact on the Company's estimation of commission revenue, accounts receivable, intangible assets, accrued commission expense, income taxes, share-based compensation and earnings per share.

On January 1, 2008, the Company adopted Financial Accounting Standards Board ("FASB") Statement No. 157, Fair Value Measurements ("FAS 157"), which clarifies the definition of fair value, establishes a framework for measuring fair value, and expands the disclosures on fair value measurements. In February 2008, the FASB issued Staff Position 157-2, Effective Date of FASB Statement No. 157 ("FSP 157-2"), that deferred the effective date of FAS 157 for one year for nonfinancial assets and liabilities recorded at fair value on a non-recurring basis. The effect of adoption of FAS 157 for financial assets and liabilities recognized at fair value on a recurring basis did not have a material impact on the Company's financial position and results of operations. The Company is assessing the impact of FAS 157 for nonfinancial assets and liabilities.

4

3. Earnings (Loss) Per Share

As of September 30, 2008 and 2007, the Company only had one issued and outstanding class of stock – common stock. As a result, the basic loss per share for the three and nine months ended September 30, 2008 and 2007, respectively, is computed by dividing net loss for the period by the weighted average number of common shares outstanding during the period.

The computed loss per share does not assume conversion, exercise, or contingent exercise of securities that would have an anti-dilutive effect on loss per share. As the Company was in a net loss position for the reported periods, all common stock equivalents were anti-dilutive. Therefore, the weighted average number of basic and diluted voting common shares outstanding for the three months ended September 30, 2008 and 2007 were 83,215,951 shares and 81,943,263 shares, respectively, and the weighted average number of basic and diluted voting common shares outstanding for the nine months ended September 30, 2008 and 2007 were 83,215,951 shares and 81,943,263 shares, respectively, and the weighted average number of basic and diluted voting common shares outstanding for the nine months ended September 30, 2008 and 2007 were 82,875,299 shares and 78,932,803 shares, respectively.

For the three months ended September 30, 2008, 594,731, 2,621,743 and 175,319 weighted average shares issuable relative to common stock warrants, common stock options and restricted stock, respectively, were excluded from net loss per share since the inclusion of such shares would be anti-dilutive due to the Company's net loss position. For the three months ended September 30, 2007, 754,870 and 1,078,834 and 622,370 weighted average shares issuable relative to common stock warrants, common stock options and restricted stock, respectively, were excluded from net loss per share since the inclusion of such shares would be anti-dilutive due to the Company's net loss per share since the inclusion of such shares would be anti-dilutive due to the Company's net loss per share since the inclusion of such shares would be anti-dilutive due to the Company's net loss position.

For the nine months ended September 30, 2008, 869,162, 2,638,738 and 404,911 weighted average shares issuable relative to common stock warrants, common stock options and restricted stock, respectively, were excluded from net loss per share since the inclusion of such shares would be anti-dilutive due to the Company's net loss position. For the nine months ended September 30, 2007, 666,926, 1,887,340 and 272,538 weighted average shares issuable relative to common stock warrants, common stock options and restricted stock, respectively, were excluded from net loss per share since the inclusion of such shares would be anti-dilutive due to the Company's net loss per share since the inclusion of such shares would be anti-dilutive due to the Company's net loss per share since the inclusion of such shares would be anti-dilutive due to the Company's net loss position.

At September 30, 2008, 1,150,000 shares and 2,258,250 shares issuable relative to common stock warrants and common stock options, respectively, had exercise prices that exceeded the average market price of the Company's common stock during the three months ended September 30, 2008 and were excluded from the calculation of diluted shares since the inclusion of such shares would be anti-dilutive. At September 30, 2008, 598,903 shares of restricted stock were excluded from the calculation of diluted shares during the three months ended September 30, 2008, 598,903 shares of restricted stock were excluded from the calculation of diluted shares during the three months ended September 30, 2008, since the inclusion of such shares would be anti-dilutive.

At September 30, 2008, 1,516,250 shares issuable relative to common stock options had exercise prices that exceeded the average market price of the Company's common stock during the nine months ended September 30, 2008 and were excluded from the calculation of diluted shares since the inclusion of such shares would be anti-dilutive. At September 30, 2007, 2,320,000 shares issuable relative to common stock options had exercise prices that exceeded the average market price of the Company's common stock during the nine months ended September 30, 2007 and were excluded from the calculation of diluted shares since the inclusion of such shares excluded from the calculation of diluted shares since the inclusion of such shares would be anti-dilutive. At September 30, 2008, 428,903 shares of restricted stock were excluded from the calculation of diluted shares during the nine months ended September 30, 2008, 30, 2008, since the inclusion of such shares would be anti-dilutive.

4. Concentration of Credit Risk and Off-Balance Sheet Risk

Financial instruments that potentially expose the Company to concentrations of credit risk consist principally of cash and trade accounts receivable. The Company had no significant off-balance sheet risk such as foreign exchange contracts, open contracts, or other foreign hedging arrangements. The Company places its cash with primarily one institution, which management believes is of high credit quality. As of September 30, 2008, approximately \$1.4 million of our cash and cash equivalents was invested in a highly liquid, U.S. Treasury money market fund.

The Company receives commission payments from energy suppliers based on the energy usage transacted between energy consumers and energy suppliers. The Company provides credit in the form of invoiced and unbilled accounts receivable to energy suppliers in the normal course of business. Collateral is not required for trade accounts receivable, but periodic credit evaluations of energy suppliers are performed. Management provides for an allowance for doubtful accounts on a specifically identified basis, as well as through historical experience applied to an aging of accounts, if necessary. Trade accounts receivable are written off when deemed uncollectible. To date, write-offs have not been material. The following represents revenue and trade accounts receivable from energy suppliers exceeding 10% of the total in each category:

		e three months tember 30,		ne nine months tember <u>30,</u>		Receivable as of ber 30,
Energy Supplier	2008	2007	2008	2007	2008	2007
Α	24%	23%	22%	21%	26%	25%
В	7%	6%	7%	7%	13%	4%
С	9%	11%	7%	13%	14%	16%

In addition to its direct relationship with energy suppliers, the Company also has direct contractual relationships with energy consumers for the online procurement of certain of their energy needs. These energy consumers are primarily large businesses and government organizations. For the three months ended September 30, 2008 and 2007, one and two of these energy consumers accounted for transactions resulting in over 10% individually, and approximately 15% and 26% in the aggregate of our revenue, respectively. For the nine months ended September 30, 2008 and 2007, two of these energy consumers accounted for transactions resulting in over 10% individually and approximately 24% and 33% in the aggregate, respectively.

Statement of Financial Accounting Standards ("SFAS") No. 130, "*Reporting Comprehensive Income*", establishes standards for reporting and displaying comprehensive income (loss) and its components in financial statements. Comprehensive income (loss) is defined as the change in stockholders' equity of a business enterprise during a period from transactions and other events and circumstances from non-owner sources. The comprehensive loss for all periods presented does not differ from the reported net loss.

5. Trade Accounts Receivable, Net

The Company generally does not directly invoice energy suppliers and, therefore, reports a significant portion of its receivables as "unbilled." Unbilled accounts receivable represents management's best estimate of energy provided by the energy suppliers to the energy consumers for a specific completed time period at contracted commission rates.

Trade accounts receivable, net consists of the following:

	September 30, 2008	December 31, 2007	
Unbilled accounts receivable	\$ 2,083,893	\$ 1,619,440	
Billed accounts receivable	883,791	263,793	
	2,967,684	1,883,233	
Allowance for doubtful accounts	(5,000)	(5,000)	
Trade accounts receivable, net	\$ 2,962,684	<u>\$ 1,878,233</u>	

6. Property and Equipment, Net

Property and equipment, net consists of the following:

	September 3 2008	30, December 31, 2007
Leasehold improvements	\$ 65,4	451 \$ 65,451
Equipment	452,3	312 426,781
Furniture and fixtures	434,1	147 418,974
	951,9	910 911,206
Less accumulated depreciation	(427,6	635) (315,256)
Property and equipment, net	<u>\$ 524,2</u>	275 <u>\$ 595,950</u>

Depreciation expense for the three months ended September 30, 2008 and 2007 was \$36,967 and \$28,064, respectively. Depreciation expense for the nine months ended September 30, 2008 and 2007 was \$112,379 and \$68,720, respectively. Property and equipment purchased under capital lease obligations as of September 30, 2008 and December 31, 2007 was \$189,524 and \$271,104, respectively. Accumulated depreciation for property and equipment purchased under capital lease was \$125,944 and \$182,872 at September 30, 2008 and December 31, 2007, respectively.

7. Share-Based Compensation

In accordance with SFAS No. 123(R) "Share-Based Payment", the Company recognizes the compensation cost of share-based awards on a straight-line basis over the requisite service period of the award. For the nine months ended September 30, 2008, share-based awards consisted of grants of restricted stock, and for the nine months ended September 30, 2007, share-based awards consisted of grants of restricted stock and stock options. The restrictions on the restricted stock lapse over the vesting period. The vesting period of restricted stock is determined by the Board of Directors, and is generally four years for employees. The per-share weighted-average fair value of stock options granted during the nine months ended September 30, 2007 was \$0.58 on the date of grant, using the Black-Scholes option-pricing model with the following weighted-average assumptions:

Nine-months ended	Expected	Risk	Expected	Expected
September 30,	Dividend Yield	Interest Rate	Option Life	Volatility
2007		4.66%	4.61 years	54%

For the three and nine months ended September 30, 2008, the Company recorded share-based compensation expense of approximately \$268,000 and \$770,000 in connection with share-based payment awards, respectively. For the three and nine months ended September 30, 2007, the Company recorded share-based compensation expense of approximately \$657,000 and \$780,000 in connection with share-based payment awards, respectively. The approximate total share-based compensation expense included in the condensed consolidated statements of operations for the periods presented is included in the following expense categories:

		Three Mor Septen			Nine Months Ended Septemb 30,		September	
		2008		2007		2008		2007
Cost of revenue	5	43,000	S	22,000	\$	148,000	\$	53,000
Sales and marketing		208,000		581,000		512,000		637,000
General and administrative		17,000	_	54,000		110,000	_	90,000
Total share-based compensation	\$	268,000	\$	657,000	\$	770,000	\$	780,000

The Company has two stock incentive plans: the 2003 Stock Incentive Plan, or the 2003 Plan, and the 2006 Stock Incentive Plan, or the 2006 Plan. As of September 30, 2008, 3,582,300 shares of common stock were reserved under the 2003 Plan. No further grants are allowed under the 2003 Plan. During the second quarter of 2008, stockholders approved an amendment to the 2006 Plan to increase the number of shares of common stock covered by the plan by 4,000,000 shares. As of September 30, 2008, 8,738,161 shares of common stock were reserved under the 2006 Plan representing 1,516,250 outstanding stock options, 2,258,500 shares of restricted stock granted and 4,963,411 shares available for grant. A summary of stock option activity under both plans for the nine months ended September 30, 2008 is as follows:

	Number of Stock Options	Ave	hted rage se Price
Outstanding at December 31, 2007	6,703,029	\$	0.65
Granted		\$	_
Cancelled	(1,023,750)	\$	1.30
Exercised	(580,729)	\$	0.03
Outstanding at September 30, 2008	5,098,550	\$	0.59

A summary of restricted stock activity under the 2006 Plan for the nine months ended September 30, 2008 is as follows:

	Shares	1	Veighted Average rant Price
Outstanding at December 31, 2007	1,448,500	\$	1.10
Granted	774,000	\$	0.88
Cancelled	(364,000)	\$	1.04
Vested	(269,597)	\$	1.17
Unvested at September 30, 2008	1,588,903	\$	1.00

For Company employees, vested restricted stock awards were net-share settled such that the Company withheld shares with value equivalent to employees' minimum statutory obligation for the applicable income and other employment taxes, and remitted the cash to the appropriate taxing authorities. The total shares withheld of 71,015 for the nine months ended September 30, 2008 was based on the value of the restricted stock on their vesting date as determined by the Company's closing stock price. Total payment for employees' tax obligations was approximately \$78,400. These net-share settlements had the effect of share repurchases by the Company as they reduced the number of shares that would have otherwise been issued as a result of the vesting and did not represent an expense to the Company.

A summary of common stock options outstanding and common stock options exercisable as of September 30, 2008 is as follows:

		O	ptions Outstandi	ng		Op	tions Exercisab	le	
Ex	inge of ercise ices	Options	Weighted Average Remaining Contractual Life		Aggregate Intrinsic Value	Number Of Shares Exercisable	Weighted Average Remaining Contractual Life		Aggregate Intrinsic Value
\$	0.02 - \$0.23 0.24 -	2,080,300	2.09 Years	\$	870,477	2,080,300	2.09 Years	\$	870,477
\$	\$1.12 1.13 -	1,502,000	4.35 Years		107,013	946,000	4.18 Years		87,227
\$	\$1.34	<u>1,516,250</u> 5,098,550	5.57 Years 3.79 Years	\$	977,490	493,125 3,519,425	5.51 Years 3.13 Years	5	957,704

The aggregate intrinsic value of options exercised during the nine months ended September 30, 2008 was \$415,718. At September 30, 2008, the weighted average exercise price of common stock options outstanding and exercisable was \$0.59 and \$0.35, respectively.

During the second quarter of 2008, stockholders voted to authorize an amendment the Company's Certificate of Incorporation which would effect a reverse stock split of our common stock in any one of the following ratios: one-for-two, one-for-three, one-for-four, one-for-five, one-for-six, one-for-seven, one-for-eight, one-for-nine and one-for-ten at any time prior to May 21, 2009. The exact ratio and the discretion of whether or when to effect the reverse stock split is at the discretion of the Board of Directors. To date, the Board has not effected a reverse stock split.

8. Credit Arrangement

On September 8, 2008, the Company and its wholly-owned subsidiary, World Energy Securities Corp., entered into a Loan and Security Agreement (the "Agreement") with Silicon Valley Bank ("SVB"). Under the Agreement, SVB has committed to make advances to the Company in an aggregate amount of up to \$3,000,000, subject to availability against certain eligible account receivables, eligible retail backlog and maintenance of financial covenants. The credit facility bears interest at a floating rate per annum based on SVB's prime rate plus three-quarters of one percentage point (0.75%) on advances made against eligible accounts receivable and prime plus one-and-one-half of one percentage point (1.5%) on advances made against eligible retail backlog. These interest rates are subject to change based on the Company's maintenance of an adjusted quick ratio of one-to-one. All unpaid principal and accrued interest is due on September 7, 2009 (the "Maturity Date"). Until the Maturity Date, the Company is only required to pay interest, with each such payment due in arrears on the last calendar day of each month.

There were no outstanding borrowings from the credit facility at September 30, 2008 and the Company was in compliance with its covenants as of September 30, 2008.

9. Acquisition

On June 1, 2007, the Company acquired substantially all of the assets of EnergyGateway LLC, or EnergyGateway, for \$4,951,758 in cash and 5,375,000 common shares of World Energy plus the assumption of certain liabilities. The shares were valued at approximately \$6,536,000 based on the average closing price on the Toronto Stock Exchange for the two days before and after the announcement date of \$1.22 per share. Approximately 2.175 million of the shares that the Company delivered to EnergyGateway in connection with this transaction are, or their cash value, if liquidated is, being held in escrow for 18 months following the closing of the sale to secure various indemnification obligations of EnergyGateway and its members. EnergyGateway, located near Columbus, Ohio, was an online broker of natural gas and electricity for commercial and industrial consumers in the United States. It provided its customers with professional energy purchasing services, including access to its proprietary 'post and bid' online procurement system called EGate.

The total purchase price was \$11,440,358, net of cash acquired, and includes related acquisition costs of \$229,500, plus the assumption of certain liabilities of approximately \$138,128. The EnergyGateway operations have been included within our consolidated results from June 1, 2007. The cost of the acquisition was allocated to the assets and liabilities assumed based on estimates of their respective fair values at the date of acquisition resulting in net assets of \$8,261,657. The excess purchase price of \$3,178,701 has been recorded as goodwill which is fully deductible for tax purposes. Management is responsible for the valuation of net assets acquired and considered a number of factors, including valuations and appraisals, when estimating the fair values and estimated useful lives of acquired assets and liabilities. During the third quarter of 2007, the Company refined estimates recorded in the second quarter of 2007 and finalized the valuation of intangible assets through a reduction of \$660,000 in customer relationships resulting in a corresponding increase to goodwill.

The Company has allocated the purchase price for the acquisition as follows:

Current assets	\$ 247,938
Fixed assets	18,755
Other assets	3,092
Intangible assets	
Non-compete agreements	680,000
Customer relationships	4,740,000
Customer contracts	500,000
Supplier agreements	380,000
Developed technology	1,830,000
Goodwill	3,178,701
Current liabilities	 (138,128)
Net assets acquired	\$ 11,440,358

The intangible assets, excluding goodwill, are being amortized on a straight-line basis over their weighted average lives as follows: non-compete agreements -1 to 5 years; customer relationships -10 years; customer contracts -3 years; supplier agreements -5 years; and developed technology -5 years. Pursuant to SFAS No. 142 "Goodwill and Other Intangible Assets", the Company performs an annual impairment test for goodwill, with any excess of the carrying value of a reporting unit's goodwill over the implied fair value of the goodwill recorded as an impairment loss.

The following unaudited pro forma information assumes that the acquisition of EnergyGateway assets had been completed as of the beginning of 2007:

		ine Months led September 30, 2007
Revenues	\$	7,377,187
Net loss available to common stockholders	\$	(3,934,464)
Loss per share: Net loss available to common stockholders per share – basic Net loss available to common stockholders per share – diluted	\$ \$	(0.05) (0.05)
Weighted average number of common and common equivalent shar outstanding – basic Weighted average number of common and common equivalent shar outstanding – diluted		81,905,789 81,905,789

The pro forma financial information is not necessarily indicative of the results to be expected in the future as a result of the acquisition of EnergyGateway.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

This quarterly report on Form 10-Q including this Item 2, contains forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended, which involve risks and uncertainties. Readers can identify these statements by forward-looking words such as "may," "could," "should," "would," "intend," "will," "expect," "anticipate," "believe," "estimate," "continue" or similar words. The Company's actual results and the timing of certain events may differ significantly from the results and timing discussed in the forward-looking statements. Factors that could cause or contribute to such differences include, but are not limited to, those discussed or referred to in this report and in the "Risk Factors" section of our Annual Report on Form 10-K and any later publicly available filing with the Securities and Exchange Commission. The following discussion and analysis of the Company's financial condition and results of operations should be read in light of those factors and in conjunction with the Company's accompanying consolidated financial statements and notes thereto.

Overview

World Energy is an energy and environmental commodities brokerage company that has developed online auction platforms, the World Energy Exchange and the World Green Exchange. On the World Energy Exchange, retail energy consumers (commercial, industrial and government, or CIG) and wholesale energy participants (utilities and retail suppliers) in the United States are able to negotiate for the purchase or sale of electricity and other energy resources from competing energy suppliers which have agreed to participate on our auction platform in a given auction. The World Energy Exchange is supplemented with information about market rules, pricing trends, energy consumer usage and load profiles. Our procurement staff uses this auction platform to conduct auctions, analyze results, guide energy consumers through contracting, and track their contracts, sites, accounts and usage history. Although our primary source of revenue is from brokering electricity and natural gas, we adapted our World Energy Exchange auction platform to accommodate the brokering of green power in 2003 (i.e., electricity generated by renewable resources), wholesale electricity in 2004 and certain other energy-related products in 2005. In 2007, we created the World Green Exchange based on the World Energy Exchange technology and business process. On the World Green Exchange, buyers and sellers negotiate for the purchase or sale of environmental commodities such as Renewable Energy Certificates, Verified Emissions Reductions, Certified Emissions Reductions and Regional Greenhouse Gas Initiative allowances.

On June 1, 2007, we acquired substantially all of the assets of EnergyGateway LLC, or EnergyGateway, for \$4,951,758 in cash and 5,375,000 of our common shares plus the assumption of certain liabilities. The EnergyGateway operations are included in our financial statements from June 1, 2007. EnergyGateway provided energy procurement and value-added energy services to customers in many major industries in the U.S. and Canada, from large multi-site Fortune 500 industrials to middle-market manufacturing and small commercial operations.

Since our initial public offering (IPO) in November 2006, we have aggressively grown the Company from 20 employees to 56 at September 30, 2007 to 58 as of September 30, 2008. This planned investment has allowed us to pursue our strategic initiatives as outlined in our IPO resulting in a more than 250% growth in revenue. Our cost structure has increased correspondingly. The majority of our infrastructure investment was made during the first 9-months of 2007 and our fixed operating cost structure has remained stable since the fourth quarter of 2007 and declined in Q3 2008. We believe that our fixed operating cost structure will remain at, or decrease from, current levels in the short-term. However, a portion of our operating costs, including channel partner and internal commission structures, are variable in nature and will increase as revenue levels increase. While we do expect there will be changes within our workforce in the future, we believe that we have the necessary resources to pursue our strategic initiatives.

Beginning in the latter half of the first quarter and continuing into the early part of the third quarter, there was a sharp rise in electricity and natural gas prices. We believe that these increases caused some of our customers in our wholesale and retail product lines to delay their energy procurement decisions. We have seen these rapid increases in commodity prices before and, based on our prior experience, we believe that customers who are now on the sidelines will come back to transact on our platform either when they have to renew their contract due to expiration, prices come down to a level at which they are comfortable, or if they believe that the then-present environment represents a new market level. During the latter half of July 2008 we began to observe a contraction in electricity and natural gas prices. We believe this contraction contributed to an increase in procurement activity in both the retail and wholesale product lines in the third quarter as compared to levels in the second quarter.

10

Revenue

Retail Electricity Transactions

We receive a monthly commission on energy sales contracted through our online auction platform from each energy supplier based on the energy usage generated and transacted between the energy supplier and energy consumer. Our commissions are not based on the retail price for electricity; rather, commissions are based on the energy usage generated and transacted between the energy supplier and energy consumer multiplied by our contractual commission rate. Our contractual commission rate is negotiated with the energy consumer on a procurement-by-procurement basis based on energy consumer specific circumstances, including size of auction, internal effort required to organize and run the respective auction and competitive factors, among others. Once the contractual commission is agreed to with the energy consumer, all energy suppliers participating in the auction agree to that rate. That commission rate remains fixed for the duration of the contractual term regardless of energy usage. Energy consumers provide us with a letter of authority to request their usage history from the local utility. We then use this data to compile a usage profile for that energy consumer that will become the basis for the auction. This data may also be used to estimate revenue on a going forward basis, as noted below.

Historically, our revenue and operating results have varied from quarter-to-quarter and are expected to continue to fluctuate in the future. These fluctuations are primarily due to energy usage, particularly electricity, having higher demand in our second and third quarters and lower demand during our fourth and first quarters. In addition, the activity levels on the World Energy Exchange can fluctuate due to a number of factors, including geopolitical events, weather conditions and market prices. To the extent these factors affect the purchasing decisions of energy consumers our future results of operations may be affected.

Contracts between energy suppliers and energy consumers are signed for a variety of term lengths, with a one year contract term being typical for commercial and industrial energy consumers, and government contracts typically having two to three year terms. As a result of recent commodity price fluctuations, we have seen our commercial and industrial customers contracting for multiple year terms. Our revenue has grown over the last three years through new participants utilizing our World Energy Exchange as well as energy consumers increasing the size or frequency of their transactions on our exchange platform.

We generally do not directly invoice our electricity energy suppliers and, therefore, we report a substantial portion of our receivables as "unbilled." Unbilled accounts receivable represents management's best estimate of energy provided by the energy suppliers to the energy consumers for a specific completed time period at contracted commission rates and is made up of two components. The first component represents energy usage for which we have received actual data from the supplier and/or the utility, but for which payment has not been received at the balance sheet date. The majority of our contractual relationships with energy suppliers require them to supply actual usage data to us on a monthly basis and remit payment to us based on that usage. The second component represents energy usage for which we have not received actual data, but for which we have estimated usage.

Retail Natural Gas Transactions

There are two primary fee components to our retail natural gas services - transaction fees and management fees. Transaction fees are billed to and paid by the energy supplier awarded business on the platform. Transaction fees for natural gas and electricity awards are established prior to award and are the same for each supplier. For the majority of our natural gas transactions, we bill the supplier upon the conclusion of the transaction based on the estimated energy volume transacted for the entire award term multiplied by the transaction fee. Management fees are paid by our energy consumers and are generally billed on a monthly basis for services rendered based on terms and conditions included in contractual arrangements. While substantially all of our retail natural gas transactions are accounted for in accordance with this policy, a certain percentage is accounted for as the natural gas is consumed by the customer and recognized as revenue in accordance with the retail electricity transaction revenue recognition methodology described above.

Wholesale and Environmental Commodity Transactions

For wholesale energy and environmental commodity transactions, substantially all transaction fees are invoiced upon the conclusion of the auction based on a fixed fee. These revenues are not tied to future energy usage and are recognized upon the completion of the online auction. For reverse auctions where our customers bid for a consumer's business, the fees are paid by the bidder. For forward auctions where a lister is selling energy or credits, the fees are typically paid by the lister. In addition, revenue may not be recognized on certain green transactions until the credits being auctioned have been verified and/or delivered. While substantially all wholesale and green transactions are accounted for in this fashion, a small percentage of our wholesale revenue is accounted for as electricity or gas is delivered similar to the retail electricity transaction methodology described above.

Cost of revenue

Cost of revenue consists primarily of:

- salaries, employee benefits and stock-based compensation associated with our auction management services, which are directly related to the development and production of the online auction and maintenance of market-related data on our auction platform and monthly management fees (our supply desk function);
- amortization of capitalized costs associated with our auction platform and acquired developed technology; and
- rent, depreciation and other related overhead and facility-related costs.

Sales and marketing

Sales and marketing expenses consist primarily of:

- salaries, employee benefits and stock-based compensation related to sales and marketing personnel;
- third party commission expenses to our channel partners;
- travel and related expenses;
- amortization related to customer relationships and contracts;
- rent, depreciation and other related overhead and facility-related costs; and
- general marketing costs such as trade shows, marketing materials and outsourced services.

General and administrative

General and administrative expenses consist primarily of:

- salaries, employee benefits and stock-based compensation related to general and administrative personnel;
- accounting, legal, recruiting and other professional fees; and
- rent, depreciation and other related overhead and facility-related costs.

Interest income (expense), net

Interest income (expense), net consists primarily of:

- interest income related to the invested portion of the proceeds from our initial public offering; and
- interest expense related to capital leases.

Results of Operations

The following table sets forth certain items as a percent of revenue for the periods presented:

		For the Three Months Ended September 30,		Months aber 30,
	2008	2007	2008	2007
Revenue	100%	100%	100%	100%
Cost of revenue	34	29	40	30
Gross profit	66	71	60	70
Operating expenses:				
Sales and marketing	71	113	84	98
General and administrative	32	70	39	70
Operating loss	(37)	(112)	(63)	(98)
Interest income, net		4		8
Income tax benefit		41		34
Net loss	(37)%	(67)%	(63)%	(56)%

Comparison of the Three Months Ended September 30, 2008 and 2007

Kevenue	For the Three Months Ender	l September 30,	
	2008	2007	Increase
Revenue	<u>\$3,289,515</u>	<u>\$ 2,675,343</u>	<u>\$ 614,172</u> 23%

Revenue increased 23%, primarily due to increased auction activity in all of our product lines and new customer wins. The revenue increase reflects the growth in our wholesale customer base to 38 at September 30, 2008 from 7 at September 30, 2007. In addition, the third quarter of 2008 benefited from increased activity in our retail product line including record commercial and industrial bookings generated by our international sales force, strong renewal rates, an increase to 54 channel partners as of September 30, 2008 from 36 as of September 30, 2007 and large state government procurements run during 2007. Of the 54 channel partners, 30 had contributed to our revenue by brokering transactions over the exchange in 2008 as compared to 23 during 2007.

Cost of revenue

Downard

		For the Three Months September 30,						
	2(D 08	2	007				
	\$	% of Revenue	\$	% of Revenue	Increa	se		
Cost of revenue	\$1,113,325	34%	\$787,517	29%	\$325,808	41%		

The 41% increase in cost of revenue related to the three month period ended September 30, 2008 as compared to the same period in 2007 was substantially due to an increase in salary and benefit costs, costs associated with three additional supply desk employees and, to a lesser extent, increased travel costs and amortization related to developed technology acquired. Cost of revenue as a percent of revenue increased 5% due to the cost increases noted above partially offset by the 23% increase in revenue.

Operating expenses						
	For	the Three Months	Ended Septemb	<u>per 30,</u>		
		:008	2	007		
	\$	% of Revenue	\$	% of Revenue	Decrease	e
Sales and marketing	\$2,355,907	71%	\$3,028,912	113%	\$ (673,005)	(22)%
General and administrative	1,041,321	32	1,876,618	70	(835,297)	(45)
Total operating expenses	\$3,397,228	103%	\$4,905,530	183%	\$(1,508,302)	(31)%

The 22% decrease in sales and marketing expense for the three month period ended September 30, 2008 as compared to the same period in 2007 primarily reflects decreases in share-based compensation, marketing and travel costs. Sales and marketing expense as a percentage of revenue decreased 42% due to the decreased costs discussed above and the 23% increase in revenue.

The 45% decrease in general and administrative expenses for the three month period ended September 30, 2008 as compared to the same period in 2007 was primarily due to decreases in legal, accounting and consulting fees, and a reduction in payroll and payroll related expenses. General and administrative expenses as a percent of revenue decreased 38% substantially due to the cost decreases noted above and the 23% increase in revenue.

Interest income, net

Interest income, net was approximately \$5,000 and \$108,000 for the three months ended September 30, 2008 and 2007, respectively. The decrease in interest income was primarily due to a lower average cash balance during the third quarter of 2008 as compared to the same period last year as we utilized the proceeds from our initial public offering to pursue our strategic initiatives.

Income tax benefit

We did not record an income tax benefit for the three months ended September 30, 2008 as we provided a full valuation allowance against our deferred tax assets due to uncertainty regarding the realization of those deferred tax assets, primarily net operating loss carryforwards, in the future. We recorded an income tax benefit of approximately \$1.1 million for the three months ended September 30, 2007, reflecting an expected federal and state tax rate of 38%.

Net loss

We reported a net loss for the three months ended September 30, 2008 of approximately \$1.2 million compared to a net loss of approximately \$1.8 million for the three months ended September 30, 2007. The decrease in net loss is primarily due to the decrease in operating expenses discussed above and the 23% increase in revenue, both partially offset by decreases in the income tax benefit and interest income, and the decrease in cost of revenue.

Comparison of the Nine Months Ended September 30, 2008 and 2007

Revenue

For the Nine Months Ended September

	 2008		2007		Increase	
Revenue	\$ 9,164,727	\$	6,103,672	\$	3,061,055	50%

Revenue increased 50%, primarily due to increased auction activity in all of our product lines, new customer wins and the inclusion of the EnergyGateway operation for a full nine months during the first nine months of 2008 as compared to four months during the first nine months of 2007. The revenue increase reflects the growth of our wholesale customer base to 38 at September 30, 2008 from 7 at September 30, 2007. In addition, the first nine months of 2008 benefited from increased activity in our retail business including the large state government procurements run during calendar 2007 and an increase to 54 channel partners as of September 30, 2008 from 36 as of September 30, 2007. Of those channel partners, 30 had contributed to our revenue by brokering transactions over the exchange in 2008 as compared to 23 during 2007.

Cost of revenue

		For the Nine Months September 30,					
	2	.008	2	007			
	\$	% of Revenue	\$	% of Revenue	Increa	se	
Cost of revenue	\$3,626,237	40%	\$1,853,660	30%	\$1,772,577	96%	

The 96% increase in cost of revenue related to the nine month period ended September 30, 2008 as compared to the same period in 2007 was substantially due to an increase in salary and benefit costs associated with three additional supply desk employees and the inclusion of the former EnergyGateway employees for a full nine months during 2008 versus four months during the first nine months of 2007 and, to a lesser extent, increased travel costs and amortization related to developed technology acquired. Cost of revenue as a percent of revenue increased 10% due to the cost increases noted above partially offset by the 50% increase in revenue.

Operating expenses

	For	For the Nine Months Ended September 30,					
	2	:008	2	007			
	<u> </u>	% of Revenue	\$	% of Revenue	Increase/(Decrease)		
Sales and marketing	\$ 7,770,567	84%	\$ 5,965,261	98%	\$1,805,306 30%		
General and administrative	3,562,946	39	4,274,415	70	<u>(711,469)</u> (17)		
Total operating expenses	\$11,333,513	123%	\$10,239,676	168%	<u>\$1,093,837</u> 11%		

The 30% increase in sales and marketing expense for the nine month period ended September 30, 2008 as compared to the same period in 2007 primarily reflects general salary increases and the addition of the former EnergyGateway employees, as well as amortization related to customer relationships and contracts. Sales and marketing expense as a percentage of revenue decreased 14% substantially due to the 50% increase in revenue partially offset by the cost increases noted above.

14

The 17% decrease in general and administrative expenses for the nine month period ended September 30, 2008 as compared to the same period in 2007 was primarily due to decreases in compliance related costs and, to a lesser extent, recruiting costs. General and administrative expenses as a percent of revenue decreased 31% substantially due to the 50% increase in revenue and, to a lesser extent, the cost decreases noted above.

Interest income, net

Interest income, net was approximately \$39,000 and \$482,000 for the nine months ended September 30, 2008 and 2007. The decrease in interest income was primarily due to a lower average cash balance during the first nine months of 2008 as compared to the same period last year as we utilized the proceeds from our initial public offering to pursue our strategic initiatives.

Income tax benefit

We did not record an income tax benefit for the nine months ended September 30, 2008 as we provided a full valuation allowance against our deferred tax assets due to uncertainty regarding the realization of those deferred tax assets, primarily net operating loss carryforwards, in the future. We recorded an income tax benefit of approximately \$2.1 million for the nine months ended September 30, 2007, reflecting an expected federal and state tax rate of 38%.

Net loss

We reported a net loss for the nine months ended September 30, 2008 of approximately \$5.8 million compared to a net loss of approximately \$3.4 million for the nine months ended September 30, 2007. The increase in net loss is primarily due to higher cost of revenue and sales and marketing expenses and decreases in interest income and income tax benefit. These increases in net loss were partially offset by the increase in revenue and the decrease in general and administrative expenses.

Liquidity and Capital Resources

At September 30, 2008, we had no commitments for material capital expenditures. We have identified and executed against a number of strategic initiatives that we believe are key components of our future growth, including: expanding our community of channel partners, energy consumers and energy suppliers on our exchanges; strengthening and extending our long-term relationships with government agencies; entering into other energy-related markets including wholesale transactions with utilities and the emerging green credit markets; making strategic acquisitions and growing our sales force. As of September 30, 2008, our workforce numbered 58 reflecting an increase of two from the 56 we employed at September 30, 2007. At September 30, 2008, we had 22 professionals in our sales and marketing and account management groups, 27 in our supply desk group and 9 in our general and administrative group. While we will hire additional professionals as the need and/or opportunity arises, we believe that in the short-term our operating costs will remain at, or decrease from, current levels.

Comparison of September 30, 2008 to December 31, 2007

	Sej	otember 30, 2008	De	cember 31, 2007	 Increase/(Dec	rease)
Cash and cash equivalents	\$	1,756,829	\$	7,001,884	\$ (5,245,055)	(75)%
Trade accounts receivable		2,962,684		1,878,233	1,084,451	58
Days sales outstanding		75		56	19	34
Working capital		1,231,476		5,323,622	(4,092,146)	(77)
Stockholders' equity		11,820,180		16,859,799	(5,039,619)	(30)

Cash and cash equivalents decreased 75%, primarily due to the pre-tax loss for the nine months ended September 30, 2008, an increase in accounts receivable and decreases in accounts payable and accrued expenses. Trade accounts receivable increased 58% primarily due to an increase in day's sales outstanding within our accounts receivable balance. Days sales outstanding (representing account receivable outstanding at September 30, 2008 divided by the average sales per day during the quarter, adjusted for deferred revenue of approximately \$272,000) increased 34% due to the timing of collections, a higher concentration of retail energy sales during the three months ended September 30, 2008 and a greater concentration of auction activity in September 2008 due to the contraction in oil and gas prices in late July and early August. Revenue from our energy suppliers representing greater than 10% of our revenue decreased to 22% from one energy supplier during the nine months ended September 30, 2007. This decrease was directly related to the addition of the EnergyGateway customers and an increase in wholesale transactions.

Working capital (consisting of current assets less current liabilities) decreased 77%, primarily due to the decrease in cash and cash equivalents resulting from the funding of the pre-tax loss for the nine months ended September 30, 2008, and an approximate \$1.1 million increase in accounts receivable discussed above. Stockholders' equity decreased 30% due to the net loss for the period partially offset by share-based compensation and the exercise of stock options and warrants.

Cash used in operating activities for the nine months ended September 30, 2008 and 2007 was approximately \$4.8 million and \$3.7 million, respectively, due primarily to the pre-tax losses in each respective period. Cash used in investing and financing activities for the nine months ended September 30, 2008 was approximately \$419,000 primarily due to costs incurred in software development. Cash used in investing and financing activities for the nine months ended September 30, 2007 was approximately \$5.6 million primarily due to the acquisition of EnergyGateway in June 2007.

We have historically funded our operations with cash flow from operations and the issuance of various debt and equity instruments. We have approximately \$1.8 million of cash and cash equivalents and no bank debt as of September 30, 2008. We expect to continue to fund our operations from existing cash resources, operating cash flow and, when required, the issuance of various debt and equity instruments. That notwithstanding, we believe that our current financial resources are adequate to fund our ongoing operations and pursue our strategic initiatives.

On September 8, 2008, we entered into a Loan and Security Agreement (the "Agreement") with Silicon Valley Bank ("SVB"). Under the Agreement, SVB has committed to make advances to us in an aggregate amount of up to \$3,000,000, subject to availability against certain eligible account receivables, eligible retail backlog and maintenance of financial covenants. The credit facility bears interest at a floating rate per annum based on SVB's prime rate plus three-quarters of one percentage point (0.75%) on advances made against eligible accounts receivable and prime plus one-and-one-half of one percentage point (1.5%) on advances made against eligible retail backlog. These interest rates are subject to change based on our maintenance of an adjusted quick ratio of one-to-one. All unpaid principal and accrued interest is due on September 7, 2009 (the "Maturity Date"). Until the Maturity Date, we are only required to pay interest, with each such payment due in arrears on the last calendar day of each month.

There were no outstanding borrowings from the credit facility and we were in compliance with our bank covenants as of September 30, 2008.

Critical Accounting Policies

The preparation of financial statements in accordance with accounting principles generally accepted in the United States of America ("generally accepted accounting principles") requires us to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenue and expenses during the reporting period.

The most judgmental estimates affecting our consolidated financial statements are those relating to revenue recognition and the estimate of actual energy purchased from the energy supplier and end user, or energy consumer, of such energy; software development costs; the fair value of our equity securities prior to our initial public offering when there was no ready market for the purchase and sale of these shares; the valuation of intangible assets and goodwill; impairment of long-lived assets; and estimates of future taxable income as it relates to the realization of our net deferred tax assets. We regularly evaluate our estimates and assumptions based upon historical experience and various other factors that we believe to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities that are not readily apparent from other sources. To the extent actual results differ from those estimates, our future results of operations may be affected. We believe the following critical accounting policies affect our more significant judgments and estimates used in the preparation of our consolidated financial statements within our Annual Report on Form 10-K as filed on March 13, 2008 for a description of our accounting policies.



Retail Electricity Transactions

We receive a monthly commission on energy sales contracted through our online auction platform from each energy supplier based on the energy usage generated and transacted between the energy supplier and energy consumer. Our commissions are not based on the retail price for electricity; rather, commissions are based on the energy usage generated and transacted between the energy supplier and energy consumer multiplied by our contractual commission rate. Revenue from commissions is recognized as earned on a monthly basis over the life of each contract as energy is consumed, provided there is persuasive evidence of an arrangement, the sales price is fixed or determinable, collection of the related receivable is reasonably assured, and customer acceptance criteria, if any, has been successfully demonstrated.

We record brokerage commissions based on actual usage data obtained from the energy supplier for that accounting period, or to the extent actual usage data is not available, based on the estimated amount of electricity and gas delivered to the energy consumers for that accounting period. We develop our estimates on a quarterly basis based on the following criteria:

- · Payments received prior to the issuance of the financial statements;
- Usage updates from energy suppliers;
- Usage data from utilities;
- Comparable historical usage data; and
- Historical variances to previous estimates.

To the extent usage data cannot be obtained, we estimate revenue as follows:

- Historical usage data obtained from the energy consumer in conjunction with the execution of the auction;
- Geographic/utility usage patterns based on actual data received;
- Analysis of prior year usage patterns; and
- Specific review of individual energy supplier/location accounts.

In addition, we perform sensitivity analyses on this estimated data based on overall industry trends including prevailing weather and usage data. Once the actual data is received, we adjust the estimated accounts receivable and revenue to the actual total amount in the period during which the payment is received. Based on management's current capacity to obtain actual energy usage, we currently estimate four to six weeks of revenue at the end of our accounting period. Differences between estimated and actual revenue have been within management's expectations and have not been material to date.

We generally do not directly invoice our electricity energy suppliers and, therefore, we report a substantial portion of our receivables as "unbilled." Unbilled accounts receivable represents management's best estimate of energy provided by the energy suppliers to the energy consumers for a specific completed time period at contracted commission rates and is made up of two components. The first component represents energy usage for which we have received actual data from the supplier and/or the utility but for which payment has not been received at the balance sheet date. The majority of our contractual relationships with energy suppliers require them to supply actual usage data to us on a monthly basis and remit payment to us based on that usage. The second component represents energy usage for which we have not received actual data, but for which we have estimated usage. Commissions paid in advance by certain energy suppliers are recorded as deferred revenue and amortized to commission revenue on a monthly basis on the energy exchanged that month.

Retail Natural Gas Transactions

There are two primary fee components to our retail natural gas services - transaction fees and management fees. Transaction fees are billed to and paid by the energy supplier awarded business on the platform. Transaction fees for natural gas awards are established prior to award and are the same for each supplier. For the majority of our natural gas transactions, we bill the supplier upon the conclusion of the transaction based on the estimated energy volume transacted for the entire award term multiplied by the transaction fee. Management fees are paid by our energy consumers and are generally billed on a monthly basis for services rendered based on terms and conditions included in contractual arrangements. While substantially all of our retail natural gas transactions are accounted in accordance with this policy, a certain percentage is accounted for as the natural gas is consumed by the customer and recognized as revenue in accordance with the retail electricity transaction revenue recognition methodology described above.

Wholesale and Green Transactions

Substantially all transaction fees are invoiced upon the conclusion of the auction based on a fixed fee. These revenues are not tied to future energy usage and are recognized upon the completion of the online auction. For reverse auctions where our customers bid for a consumer's business, the fees are paid by the bidder. For forward auctions where a lister is selling energy or credits, the fees are typically paid by the lister. In addition, revenue may not be recognized on certain green transactions until the credits being auctioned have been verified and/or delivered. While substantially all wholesale and green transactions are accounted for in this fashion, a small percentage of our wholesale revenue is accounted for as the wholesale electricity or gas is delivered similar to the retail electricity transaction revenue recognition methodology described above.

Channel Partner Commissions

We pay commissions to our channel partners at contractual rates based on monthly energy transactions between energy suppliers and energy consumers. The commission is accrued monthly and charged to sales and marketing expense as revenue is recognized. We pay commissions to our salespeople at contractual commission rates based upon cash collections from our customers.

Revenue Estimation

Our estimates in relation to revenue recognition affect revenue and sales and marketing expense as reflected on our statements of operations, and trade accounts receivable and accrued commissions accounts as reflected on our balance sheets. For any quarterly reporting period, we may not have actual usage data for certain energy suppliers and will need to estimate revenue. We record revenue based on the energy consumers' historical usage profile. At the end of each reporting period, we adjust historical revenue to reflect actual usage for the period. For the nine months ended September 30, 2008, we estimated usage for approximately 11% of our revenue resulting in a negative 0.4%, or approximately \$13,000, adjustment. This decrease in revenue, which is based on adjusted year to date revenue, resulted in an approximate \$8,000 decrease in sales and marketing expense related to third party commission expense associated with those revenues. Corresponding adjustments were made to trade accounts receivable and accrued commissions, respectively. A 1% difference between this estimate and actual usage would have an approximate \$7,000 effect on our revenue for the three months ended September 30, 2008.

Software Development

Certain acquired software and significant enhancements to our software are recorded in accordance with Statement of Position ("SOP") 98-1, "Accounting for Costs of Computer Software Developed or Obtained for Internal Use". Accordingly, internally developed software costs of approximately \$293,000 and \$330,000 related to implementation, coding and configuration have been capitalized during the nine months ended September 30, 2008 and 2007, respectively. We amortize internally developed and purchased software over the estimated useful life of the software (generally three years). During the nine months ended September 30, 2008 and 2007 approximately \$170,000 and \$73,000 were amortized to cost of revenues, respectively. Accumulated amortization was approximately \$443,000 and \$273,000 at September 30, 2008 and December 31, 2007, respectively.

Our estimates for capitalization of software development costs affect cost of revenue and capitalized software as reflected on our consolidated statements of operations and on our consolidated balance sheets. During the nine months ended September 30, 2008, capitalized software costs were 1.9% of our total assets and amortization expense was approximately 4.7% of cost of revenue. To the extent the carrying amount of the capitalized software costs may not be fully recoverable or that the useful lives of those assets are no longer appropriate, we may need to record an impairment (non-cash) charge and write-off a portion or all of the capitalized software balance on the balance sheet.

Goodwill

We use assumptions in establishing the carrying value and fair value of our goodwill. Goodwill represents the excess of the purchase price over the fair value of identifiable net assets of acquired businesses. We account for goodwill that results from acquired businesses in accordance with Statement of Financial Accounting Standards ("SFAS") No. 142, "Goodwill and Other Intangible Assets" ("SFAS No. 142"). Under SFAS No. 142, goodwill and intangible assets having indefinite lives are not amortized but instead are assigned to reporting units and tested for impairment annually or more frequently if changes in circumstances or the occurrence of events indicate possible impairment.

Pursuant to SFAS No. 142, we perform an annual impairment test for goodwill. The impairment test for goodwill is a two-step process. Step one consists of a comparison of the fair value of a reporting unit with its carrying amount, including the goodwill allocated to each reporting unit. If the carrying amount is in excess of the fair value, step two requires the comparison of the implied fair value of the reporting unit's goodwill over the implied fair value of the reporting unit's goodwill will be recorded as an impairment loss. During the first nine months of 2008, no impairment of our goodwill was recorded.

Intangible Assets

We use assumptions in establishing the carrying value, fair value and estimated lives of our intangible assets. The criteria used for these assumptions include management's estimate of the assets continuing ability to generate positive income from operations and positive cash flow in future periods compared to the carrying value of the asset, as well as the strategic significance of any identifiable intangible asset in our business objectives. If assets are considered impaired, the impairment recognized is the amount by which the carrying value of the assets exceeds the fair value of the assets. Useful lives and related amortization expense are based on our estimate of the period that the assets will generate revenues or otherwise be used by us. Factors that would influence the likelihood of a material change in our reported results include significant changes in the asset's ability to generate positive cash flow, a significant decline in the economic and competitive environment on which the asset depends and significant changes in our strategic business objectives.

Intangible assets consist of customer relationships and contracts, purchased technology and other intangibles, and are stated at cost less accumulated amortization. Intangible assets with a definite life are amortized using the straight-line method over their estimated useful lives, which range from one to ten years.

Impairment of Long-Lived and Intangible Assets

In accordance with SFAS No. 144, "Accounting for the Impairment or Disposal of Long-Lived Assets", we periodically review long-lived assets and intangible assets for impairment whenever events or changes in circumstances indicate that the carrying amount of such assets may not be recoverable or that the useful lives of those assets are no longer appropriate. Recoverability of these assets is determined by comparing the forecasted undiscounted net cash flows of the operation to which the assets relate to the carrying amount. During the first nine months of 2008, no impairment of our long-lived assets was recorded as no change in circumstances indicated that the carrying value of the assets was not recoverable.

Income Taxes

We account for income taxes in accordance with SFAS No. 109, "Accounting for Income Taxes" ("SFAS No. 109"). Deferred tax assets and liabilities are determined at the end of each period based on the future tax consequences that can be attributed to net operating loss carryforwards, as well as differences between the financial statement carrying amounts of the existing assets and liabilities and their respective tax basis. Deferred income tax expense or credits are based on changes in the asset or liability from period to period. Valuation allowances are provided if, based on the weight of available evidence, it is more likely than not that some or all of the deferred tax assets will not be realized. The realization of deferred tax assets is dependent upon the generation of future taxable income. In determining the valuation allowance, we consider past performance, expected future taxable income, and qualitative factors which we consider to be appropriate in estimating future taxable income. Our forecast of expected future taxable income is for future periods that can be reasonably estimated. Results that differ materially from current expectations may cause management to change its judgment on future taxable income. These changes, if any, may require us to adjust our existing tax valuation allowance higher or lower than the amount recorded.

Our estimates in relation to income taxes affect income tax benefit and deferred tax assets as reflected on our statements of operations and balance sheets, respectively. The deferred tax assets are reduced by a valuation allowance if it is more likely than not that the tax benefits will not be realized in the near term. As of September 30, 2008, we had deferred tax assets of approximately \$6.9 million against which a full valuation allowance has been established. To the extent we determine that it is more likely than not that we will recover all of our deferred tax assets, it could result in an approximate \$6.9 million non-cash tax benefit.

We adopted Financial Accounting Standards ("FASB") Interpretation No. 48, "Accounting for Uncertainty in Income Taxes" ("FIN 48"), on January 1, 2007. FIN 48 clarifies the accounting for uncertainty in income taxes recognized in an enterprise's financial statements in accordance with SFAS 109. FIN 48 prescribes a two-step process to determine the amount of tax benefit to recognize. First, the tax position must be evaluated to determine the likelihood that it will be sustained upon examination by a tax authority. If the tax position is deemed "more-likely-than-not" to be sustained, the tax position is then assessed to determine the amount of benefit to recognize in the financial statements. The amount of the benefit that may be recognized is the largest amount that has a greater than 50 percent likelihood of being realized upon ultimate settlement. If the tax position does not meet the "more-likely-than-not" threshold, then it is not recognized in the financial statements.

We have reviewed the tax positions taken, or to be taken, in our tax returns for all tax years currently open to examination by the taxing authorities in accordance with the recognition and measurement standards of FIN 48. At September 30, 2008, there is no expected material, aggregate tax effect as a result of differences between tax return positions and the benefits recognized in our financial statements.

Share-Based Compensation

In accordance with SFAS No. 123(R) "Share-Based Payment", the Company recognizes the compensation cost of share-based awards on a straight-line basis over the requisite service period of the award. For the nine months ended September 30, 2008, share-based awards consisted of grants of restricted stock, and for the nine months ended September 30, 2007, share-based awards consisted of grants of restricted stock and stock options. The restrictions on the restricted stock lapse over the vesting period. The vesting period of restricted stock is determined by the Board of Directors, and is generally four years for employees. The per-share weighted-average fair value of stock options granted during the nine months ended September 30, 2007 was \$0.58 on the date of grant, using the Black-Scholes option-pricing model with the following weighted-average assumptions:

Nine-months ended	Expected	Risk	Expected	Expected
September 30,	<u>Dividend</u> Yield	Interest Rate	Option Life	Volatility
2007		4.66%	4.61 years	54%

The Company has two stock incentive plans: the 2003 Stock Incentive Plan, or the 2003 Plan, and the 2006 Stock Incentive Plan, or the 2006 Plan. As of September 30, 2008, 3,582,300 shares of common stock were reserved under the 2003 Plan. No further grants are allowed under the 2003 Plan. As of September 30, 2008, 8,738,161 shares of common stock were reserved under the 2006 Plan representing 1,516,250 outstanding stock options, 2,258,500 shares of restricted stock granted and 4,963,411 shares available for grant. A summary of stock option activity under both plans for the nine months ended September 30, 2008 is as follows:

	Number of Stock Options	Weighted Average Exercise Price		
Outstanding at December 31, 2007	6,703,029	\$	0.65	
Granted		\$		
Cancelled	(1,023,750)	\$	1.30	
Exercised	(580,729)	\$	0.03	
Outstanding at September 30, 2008	5,098,550	\$	0.59	

A summary of restricted stock activity under the 2006 Plan for the nine months ended September 30, 2008 is as follows:

	Weighted Average Shares Grant Price					
Outstanding at December 31, 2007	1,448,500	\$	1.10			
Granted	774,000	\$	0.88			
Cancelled	(364,000)	\$	1.04			
Vested	(269,597)	\$	1.17			
Unvested at September 30, 2008	1,588,903	\$	1.00			

20

A summary of common stock options outstanding and common stock options exercisable as of September 30, 2008 is as follows:

		O	ptions Outstandi		Options Exercisable				
Ex	inge of tercise ices	Options	Weighted Average Remaining Contractual Life	1	Aggregate Intrinsic Value	Number Of Shares Exercisable	Weighted Average Remaining Contractual Life		Aggregate Intrinsic Value
\$	0.02 - \$0.23 0.24 -	2,080,300	2.09 Years	\$	87 0,4 77	2,080,300	2.09 Years	\$	870,477
\$	\$1.12 1.13 -	1,502,000	4.35 Years		107,013	946,000	4.18 Years		87,227
\$	\$1.34	<u>1,516,250</u> <u>5,098,550</u>	5.57 Years 3.79 Years	\$	977,490	<u>493,125</u> <u>3,519,425</u>	5.51 Years 3.13 Years	\$	957,704

The aggregate intrinsic value of options exercised during the nine months ended September 30, 2008 was \$415,718.

Item 3. Quantitative and Qualitative Disclosure about Market Risk

Market Risk

Market risk is the potential loss arising from adverse changes in market rates and prices, such as foreign currency rates, interest rates, and other relevant market rates or price changes. In the ordinary course of business, the Company is exposed to market risk resulting from changes in foreign currency exchange rates, and the Company regularly evaluates its exposure to such changes. The Company's overall risk management strategy seeks to balance the magnitude of the exposure and the costs and availability of appropriate financial instruments.

Impact of Inflation and Changing Prices

Historically, our business has not been materially impacted by inflation. We provide our service at the inception of the service contract between the energy supplier and energy consumer. Our fee is set as a fixed dollar amount per unit of measure and fluctuates with changes in energy demand over the contract period.

Foreign Currency Fluctuation

Our commission revenue is primarily denominated in U.S. dollars. Therefore, we are not directly affected by foreign exchange fluctuations on our current orders. However, fluctuations in foreign exchange rates do have an effect on energy consumers' access to U.S. dollars and on pricing competition. We have entered into non-U.S. dollar contracts but they have not had a material impact on our operations. We do not believe that foreign exchange fluctuations will materially affect our results of operations.

Item 4. Controls and Procedures

Evaluation of Disclosure Controls and Procedures

As required by Rule 13a-15 under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), the Company carried out an evaluation under the supervision and with the participation of the Company's management, including the Chief Executive Officer and Chief Financial Officer, of the effectiveness of the Company's disclosure controls and procedures as of September 30, 2008. In designing and evaluating the Company's disclosure controls and procedures, the Company and its management recognize that there are inherent limitations to the effectiveness of any system of disclosure controls and procedures, including the possibility of human error and the circumvention or overriding of the controls and procedures. Accordingly, even effective disclosure controls and procedures can only provide reasonable assurance of achieving their desired control objectives. Additionally, in evaluating and implementing possible controls and procedures, the Company's management was required to apply its reasonable judgment. Based upon the required evaluation, the Chief Executive Officer and the Chief Financial Officer concluded that as of September 30, 2008, the Company's disclosure controls and procedures were effective at the reasonable assurance level to ensure that information required to be disclosed by the Company in the reports it files or submits under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the Securities and Exchange Commission's rules and forms.

Changes in Internal Control Over Financial Reporting

There was no change in the Company's internal control over financial reporting that occurred during the three months ended September 30, 2008 that has materially affected, or is reasonably likely to materially affect, the Company's internal control over financial reporting.

r.

PART II

OTHER INFORMATION

Item 1. Legal Proceedings

None.

Item 1A. Risk Factors

There have been no material changes to the risk factors previously disclosed in our most recently filed annual report on Form 10-K.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

We registered shares of our common stock in connection with our initial public offering concurrently in Canada under the terms of a Supplemented Prep Prospectus dated November 9, 2006 and in the United States under the Securities Act of 1933, as amended. Our Registration Statement on Form S-1 (No. 333-136528) in connection with our initial public offering was declared effective by the SEC on November 9, 2006.

As of September 30, 2008, we have used approximately \$16 million of the net proceeds of the initial public offering to redeem our long-term debt, to acquire the assets of EnergyGateway, for working capital and to fund operations. The remaining net proceeds of the initial public offering are invested in short-term, highly liquid money market accounts with an original maturity of 90 days or less. There has been no material change in the planned use of proceeds from our initial public offering as described in our final prospectus filed with the SEC pursuant to Rule 424(b).

Item 3. Defaults Upon Senior Securities

None.

Item 4. Submission of Matters to a Vote of Security Holders

None.

Item 5. Other Information

None.



Item 6. Exhibits

- 31.1 Certification of the Chief Executive Officer pursuant to §302 of the Sarbanes-Oxley Act of 2002.
- 31.2 Certification of the Chief Financial Officer pursuant to §302 of the Sarbanes-Oxley Act of 2002.
- 32.1 Certification of the Chief Executive Officer pursuant to 18 U.S.C. §1350, as adopted pursuant to §906 of the Sarbanes-Oxley Act of 2002.
- 32.2 Certification of the Chief Financial Officer pursuant to 18 U.S.C. §1350, as adopted pursuant to §906 of the Sarbanes-Oxley Act of 2002.

23

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

World Energy Solutions, Inc.

Dated: November 6, 2008

By: /s/ Richard Domaleski Richard Domaleski Chief Executive Officer

Dated: November 6, 2008

By: /s/ James Parslow

James Parslow Chief Financial Officer

292

24

EXHIBIT INDEX

<u>Exhibit</u>

Description

- 31.1 Certification of the Chief Executive Officer pursuant to §302 of the Sarbanes-Oxley Act of 2002
- 31.2 Certification of the Chief Financial Officer pursuant to §302 of the Sarbanes-Oxley Act of 2002
- 32.1 Certification of the Chief Executive Officer pursuant to 18 U.S.C. §1350, as adopted pursuant to §906 of the Sarbanes-Oxley Act of 2002
- 32.2 Certification of the Chief Financial Officer pursuant to 18 U.S.C. §1350, as adopted pursuant to §906 of the Sarbanes-Oxley Act of 2002

CERTIFICATION PURSUANT TO §302 OF THE SARBANES-OXLEY ACT OF 2002

I, Richard Domaleski, Chief Executive Officer of World Energy Solutions, Inc. (the "Company"), certify that:

1. I have reviewed this quarterly report on Form 10-Q of the Company.

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report.

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the Company as of, and for, the periods presented in this report.

4. The Company's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the Company and have:

a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the Company, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

c) Evaluated the effectiveness of the Company's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

d) Disclosed in this report any change in the Company's internal control over financial reporting that occurred during the Company's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the Company's internal control over financial reporting.

5. The Company's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the Company's auditors and the audit committee of the Company's board of directors (or persons performing the equivalent functions):

a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the Company's ability to record, process, summarize and report financial information; and

b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the Company's internal control over financial reporting.

Dated: November 6, 2008

By: /s/ Richard Domaleski

Richard Domaleski Chief Executive Officer

CERTIFICATION PURSUANT TO §302 OF THE SARBANES-OXLEY ACT OF 2002

I, James Parslow, Chief Financial Officer of World Energy Solutions, Inc. (the "Company"), certify that:

1. I have reviewed this quarterly report on Form 10-Q of the Company.

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report.

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the Company as of, and for, the periods presented in this report.

4. The Company's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the Company and have:

a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the Company, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

c) Evaluated the effectiveness of the Company's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

d) Disclosed in this report any change in the Company's internal control over financial reporting that occurred during the Company's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the Company's internal control over financial reporting.

5. The Company's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the Company's auditors and the audit committee of the Company's board of directors (or persons performing the equivalent functions):

a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the Company's ability to record, process, summarize and report financial information; and

b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the Company's internal control over financial reporting.

Dated: November 6, 2008

By: /s/ James Parslow

James Parslow Chief Financial Officer

CERTIFICATION PURSUANT TO 18 U.S.C. §1350, AS ADOPTED PURSUANT TO §906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the quarterly report of World Energy Solutions, Inc (the "Company") on Form 10-Q (the "Report") for the period ended September 30, 2008 as filed with the Securities and Exchange Commission on the date hereof, I, Richard Domaleski, Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. §1350, as adopted pursuant to §906 of the Sarbanes-Oxley Act of 2002, that:

1. The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

2. The information contained in the report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated: November 6, 2008

By: /s/ Richard Domaleski

Richard Domaleski Chief Executive Officer

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signatures that appear in typed form within the electronic version of this written statement required by Section 906, has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.

CERTIFICATION PURSUANT TO 18 U.S.C. §1350, AS ADOPTED PURSUANT TO §906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the quarterly report of World Energy Solutions, Inc. (the "Company") on Form 10-Q (the "Report") for the period ended September 30, 2008 as filed with the Securities and Exchange Commission on the date hereof, I, James Parslow, Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. §1350, as adopted pursuant to §906 of the Sarbanes-Oxley Act of 2002, that:

1. The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated: November 6, 2008

By: /s/ James Parslow

James Parslow Chief Financial Officer

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signatures that appear in typed form within the electronic version of this written statement required by Section 906, has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.

Created by 10KWizard www.10KWizard.com

Source: World Energy Solutio, 10-Q, November 06, 2008



FORM 8-K

World Energy Solutions, Inc. - WLDE

Filed: September 08, 2008 (period: September 08, 2008)

Report of unscheduled material events or corporate changes.

Table of Contents

8-K - WORLD ENERGY SOLUTIONS, INC.

- Item 1.01, Entry into a Material Definitive Agreement.
- Item 2.03.Creation of a Direct Financial Obligation or an Obligation under an
Off-Balance Sheet Arrangement of a Registrant.
- Item 9.01. Financial Statements and Exhibits.

SIGNATURE

EXHIBIT INDEX

EX-10.1 (LOAN AND SECURITY AGREEMENT)

EX-99.1 (PRESS RELEASE DATED SEPTEMBER 8)

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): September 8, 2008

World Energy Solutions, Inc.

(Exact Name of Registrant as Specified in Charter)

Delaware (State or Other Jurisdiction of Incorporation 333-136528 (Commission File Number) 04-3474959 (IRS Employer Identification No.)

446 Main Street Worcester, Massachusetts (Address of Principal Executive Offices)

01608 (Zip Code)

Registrant's telephone number, including area code: (508) 459-8100

n/a (Former Name or Former Address, if Changed Since Last Report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Item 1.01. Entry into a Material Definitive Agreement.

On September 8, 2008, World Energy Solutions, Inc. and its wholly-owned subsidiary World Energy Securities Corp. (collectively, the "Company") entered into a Loan and Security Agreement (the "Agreement") with Silicon Valley Bank ("SVB"). Under the Agreement, SVB has committed to make advances to the Company in an aggregate amount of up to \$3,000,000, subject to availability against certain eligible account receivables and eligible retail backlog. The credit facility bears interest at a floating rate per annum based on SVB's prime rate plus three-quarters of one percentage point (0.75%) on advances made against eligible accounts receivable and prime plus one-and-one-half of one percentage point (1.5%) on advances made against eligible retail backlog. These interest rates are subject to change based on the Company's maintenance of an adjusted quick ratio of one-to-one. All unpaid principal and accrued interest is due on September 7, 2009 (the "Maturity Date"). Until the Maturity Date, the Company is only required to pay interest, with each such payment due in arrears on the last calendar day of each month.

The Company intends to use the credit facility solely as working capital and to fund its general business requirements.

The credit facility is secured by a first priority perfected security interest in substantially all of the Company's assets. The Agreement contains certain affirmative and negative covenants, including a tangible net worth covenant. Any failure by the Company to comply with these covenants and any other obligations under the Agreement could result in an event of default which could lead to acceleration of the amounts owed and other remedies.

The foregoing description of the Loan and Security Agreement does not purport to be complete and is qualified in its entirety by reference to the full text of the Loan and Security Agreement attached to this Form 8-K as Exhibit 10.1 and incorporated herein by reference.

Item 2.03. Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant.

Our discussion under Item 1.01 of this Current Report is hereby incorporated by this reference.

Item 9.01. Financial Statements and Exhibits.

(d) Exhibits

See Exhibit Index attached hereto.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

WORLD ENERGY SOLUTIONS, INC.

Date: September 8, 2008

. . I

By: /s/ James Parslow

James Parslow Chief Financial Officer

EXHIBIT INDEX

Exhibit No.	Description
10.1	Loan and Security Agreement with Silicon Valley Bank dated September 8, 2008
99.1	Press Release dated September 8, 2008

- 4 -

.

Source: World Energy Solutio, 8-K, September 08, 2008

EXECUTION COPY

LOAN AND SECURITY AGREEMENT

THIS LOAN AND SECURITY AGREEMENT (this "Agreement") dated as of the Effective Date among SILICON VALLEY BANK, a California corporation with its principal place of business at 3003 Tasman Drive, Santa Clara, California 95054 and with a loan production office located at One Newton Executive Park, Suite 200, 2221 Washington Street, Newton, Massachusetts 02462 ("Bank"), WORLD ENERGY SOLUTIONS, INC., a Delaware corporation with offices located at 446 Main Street, Worcester, Massachusetts 01608, and WORLD ENERGY SECURITIES CORP., a Massachusetts securities corporation with offices located at 446 Main Street, Worcester, Massachusetts 01608 (individually and collectively, jointly and severally, "Borrower"), provides the terms on which Bank shall lend to Borrower and Borrower shall repay Bank. The parties agree as follows:

1 ACCOUNTING AND OTHER TERMS

Accounting terms not defined in this Agreement shall be construed following GAAP. Calculations and determinations must be made following GAAP. Capitalized terms not otherwise defined in this Agreement shall have the meanings set forth in Section 13. All other terms contained in this Agreement, unless otherwise indicated, shall have the meaning provided by the Code to the extent such terms are defined therein.

2 LOAN AND TERMS OF PAYMENT

2.1 Promise to Pay. Borrower hereby unconditionally promises to pay Bank the outstanding principal amount of all Credit Extensions and accrued and unpaid interest thereon as and when due in accordance with this Agreement.

2.1.1 Revolving Advances.

(a) <u>Availability</u>. Subject to the terms and conditions of this Agreement and to deduction of Reserves, Bank will make Advances to Borrower up to the Availability Amount. Amounts borrowed under the Revolving Line may be repaid, and prior to the Revolving Line Maturity Date, reborrowed, subject to the applicable terms and conditions precedent herein.

(b) <u>Termination: Repayment</u>. The Revolving Line terminates on the Revolving Line Maturity Date, when the principal amount of all Advances, the unpaid interest thereon, and all other Obligations relating to the Revolving Line shall be immediately due and payable.

2.1.2 Letters of Credit Sublimit.

As part of the Revolving Line and subject to deduction of Reserves, Bank shall issue or have issued (a) Letters of Credit for Borrower's account. The face amount of outstanding Letters of Credit (including drawn but unreimbursed Letters of Credit and any Letter of Credit Reserve) may not exceed One Million Dollars (\$1,000,000.00) inclusive of Credit Extensions relating to Sections 2.1.3 and 2.1.4. Such aggregate amounts utilized hereunder shall at all times reduce the amount otherwise available for Advances under the Revolving Line. If, on the Revolving Line Maturity Date or after the occurrence and during the continuance of an Event of Default there are any outstanding Letters of Credit, then on such date Borrower shall provide to Bank cash collateral in an amount equal to 105% of the face amount of all such Letters of Credit plus all interest, fees, and costs due or to become due in connection therewith (as estimated by Bank in its good faith business judgment), to secure all of the Obligations relating to said Letters of Credit. All Letters of Credit shall be in form and substance acceptable to Bank in its sole discretion and shall be subject to the terms and conditions of Bank's standard Application and Letter of Credit Agreement (the "Letter of Credit Application"), provided that, to the extent of any conflict, this Agreement shall govern. Borrower agrees to execute any further documentation in connection with the Letters of Credit as Bank may reasonably request. Borrower further agrees to be bound by the regulations and interpretations of the issuer of any Letters of Credit guarantied by Bank and opened for Borrower's account or by Bank's interpretations of any Letter of Credit issued by Bank for Borrower's account, and Borrower understands and agrees that Bank shall not be liable for any error, negligence, or mistake, whether of omission or commission, in following Borrower's instructions or those contained in the Letters of Credit or any modifications, amendments, or supplements thereto, except Bank's gross negligence or willful misconduct.

(b) The obligation of Borrower to immediately reimburse Bank for drawings made under Letters of Credit shall be absolute, unconditional, and irrevocable, and shall be performed strictly in accordance with the terms of this Agreement, such Letters of Credit, and the Letter of Credit Application. Any amounts Bank pays on behalf of Borrower for any Letters of Credit will be treated as Advances under the Revolving Line and will accrue interest at the interest rate applicable to Advances.

(c) Borrower may request that Bank issue a Letter of Credit payable in a Foreign Currency. If a demand for payment is made under any such Letter of Credit, Bank shall treat such demand as an Advance to Borrower of the equivalent of the amount thereof (plus fees and charges in connection therewith such as wire, cable, SWIFT or similar charges) in Dollars at the then-prevailing rate of exchange in San Francisco, California, for sales of the Foreign Currency for transfer to the country issuing such Foreign Currency.

(d) To guard against fluctuations in currency exchange rates, upon the issuance of any Letter of Credit payable in a Foreign Currency, Bank shall create a reserve (the "Letter of Credit Reserve") under the Revolving Line in an amount equal to ten percent (10%) of the face amount of such Letter of Credit. The amount of the Letter of Credit Reserve may be adjusted by Bank from time to time to account for fluctuations in the exchange rate. The availability of funds under the Revolving Line shall be reduced by the amount of such Letter of Credit Reserve for as long as such Letter of Credit remains outstanding.

2.1.3 Foreign Exchange Sublimit. As part of the Revolving Line, Borrower may enter into foreign exchange contracts with Bank under which Borrower commits to purchase from or sell to Bank a specific amount of Foreign Currency (each, a "FX Forward Contract") on a specified date (the "Settlement Date"). FX Forward Contracts shall have a Settlement Date of at least one (1) FX Business Day after the contract date and shall be subject to a reserve of ten percent (10%) of each outstanding FX Forward Contract in a maximum aggregate amount equal to One Hundred Thousand Dollars (\$100,000.00) (the "FX Reserve"). The aggregate amount of FX Forward Contracts at any one time plus Credit Extensions made pursuant to Sections 2.1.2 and 2.1.4 may not exceed ten (10) times the maximum amount of the FX Reserve. Any amounts needed to fully reimburse Bank will be treated as Advances under the Revolving Line and will accrue interest at the interest rate applicable to Advances.

2.1.4 Cash Management Services Sublimit. Borrower may use up to One Million Dollars (\$1,000,000.00) inclusive of Credit Extensions relating to Sections 2.1.2 and 2.1.3 (the "Cash Management Services Sublimit") of the Revolving Line for Bank's cash management services which may include merchant services, direct deposit of payroll, business credit card, and check cashing services identified in Bank's various cash management services agreements (collectively, the "Cash Management Services"). The dollar amount of any Cash Management Services provided under this sublimit will reduce the amount otherwise available under the Revolving Line. Any amounts used or reserved by Borrower for any Cash Management Services will reduce the amount otherwise available for Credit Extensions under the Revolving Line. Any amounts Bank pays on behalf of Borrower for any Cash Management Services will be treated as Advances under the Revolving Line and will accrue interest at the interest rate applicable to Advances.

2.2 Overadvances. If, at any time the sum of (a) the outstanding amount of any Advances (including any amounts used for Cash Management Services) plus (b) the face amount of any outstanding Letters of Credit (including drawn but unreimbursed Letters of Credit and any Letter of Credit Reserve, plus (c) the FX Reserve exceeds the lesser of either the Revolving Line or the Borrowing Base (such excess amount being an "Overadvance"), Borrower shall pay to Bank in cash such Overadvance immediately; <u>provided, however</u>, that if such Overadvance results from Bank's exercising its right to decrease the percentages of the Borrowing Base or to adjust the criteria for Eligible Accounts, Borrower shall have three (3) Business Days from receipt of notice from Bank of such decrease or adjustment to repay such Overadvance.

2.3 Payment of Interest on the Credit Extensions.

(a) <u>Interest Rate; Advances</u>. Subject to Section 2.3(b), (a) the principal amount of the Revolving Line outstanding due to Advances made in respect of Eligible Accounts shall accrue interest at a floating per annum rate equal to the aggregate of the Prime Rate plus one and three-quarters of one percentage point (1.75%), provided, however, during a Streamline Period, the principal amount of the Revolving Line outstanding due to Advances made in respect of Eligible Accounts shall accrue interest at a floating per annum rate equal to the aggregate of the Prime Rate plus three-quarters of one percentage point (0.75%); and (b) the principal amount of the Revolving Line outstanding due to Advances made in respect of Eligible Retail Backlog Accounts shall accrue interest at a floating per annum rate equal to the aggregate of the Prime Rate plus three-quarters of one percentage point (0.75%); and (b) the principal amount of the Revolving Line outstanding due to Advances made in respect of Eligible Retail Backlog Accounts shall accrue interest at a floating per annum rate equal to the aggregate of the Prime Rate plus two and one-quarter of one

- 2 -

percentage point (2.25%), provided, however, during a Streamline Period, the principal amount of the Revolving Line outstanding due to Advances made in respect of Eligible Retail Backlog Accounts shall accrue interest at a floating per annum rate equal to the aggregate of the Prime Rate plus one and one-half of one percentage point (1.50%). Interest on any Credit Extension shall be payable monthly.

(b) <u>Default Rate</u>. Immediately upon the occurrence and during the continuance of an Event of Default, Obligations shall bear interest at a rate per annum which is four percentage points (4.00%) above the rate effective immediately before the occurrence of the Event of Default (the "**Default Rate**"). Payment or acceptance of the increased interest rate provided in this Section 2.3(b) is not a permitted alternative to timely payment and shall not constitute a waiver of any Event of Default or otherwise prejudice or limit any rights or remedies of Bank.

(c) <u>Adjustment to Interest Rate</u>. Changes to the interest rate of any Credit Extension based on changes to the Prime Rate shall be effective on the effective date of any change to the Prime Rate and to the extent of any such change.

elapsed.

(d)

<u>360-Day Year</u>. Interest shall be computed on the basis of a 360-day year for the actual number of days

(e) <u>Debit of Accounts</u>. Bank may debit any of Borrower's deposit accounts, including the Designated Deposit Account, for principal and interest payments or any other amounts Borrower owes Bank when due. These debits shall not constitute a set-off.

(f) <u>Payment; Interest Computation; Float Charge</u>. Interest is payable monthly on the last Business Day of each month. In computing interest on the Obligations, all Payments received after 3:00 p.m. local Boston time on any day shall be deemed received on the next Business Day. Bank shall be entitled to charge Borrower a "float" charge in an amount equal to two (2) Business Days interest, at the interest rate applicable to the Advances, on all Payments received by Bank. Said float charge is not applicable during a Streamline Period, and is not included in interest for purposes of computing Minimum Monthly Interest (if any) under this Agreement. The float charge for each month shall be payable on the last day of the month. Bank shall not, however, be required to credit Borrower's account for the amount of any item of payment which is unsatisfactory to Bank in its good faith business judgment, and Bank may charge Borrower's Designated Deposit Account for the amount of any item of payment which is returned to Bank unpaid.

2.4 Fees. Borrower shall pay to Bank:

(a) <u>Commitment Fee</u>. A fully earned, non-refundable commitment fee of Twenty Thousand Dollars (\$20,000.00), on the Effective Date;

(b) <u>Letter of Credit Fee</u>. Bank's customary fees and expenses for the issuance or renewal of Letters of Credit, upon the issuance or renewal of such Letter of Credit by Bank;

(c) <u>Termination Fee</u>. Subject to the terms of Section 12.1, a termination fee;

(d) <u>Unused Revolving Line Facility Fee</u>. A fee (the "Unused Revolving Line Facility Fee"), which fee shall be paid monthly, in arrears, on the last Business Day of each month, in an amount equal to one-half of one percent (0.50%) per annum of the average unused portion of the Revolving Line, as determined by Bank. Borrower shall not be entitled to any credit, rebate or repayment of any Unused Revolving Line Facility Fee previously earned by Bank pursuant to this Section notwithstanding any termination of the within Agreement, or suspension or termination of Bank's obligation to make loans and advances hereunder; and

(f) <u>Bank Expenses</u>. All Bank Expenses (including reasonable attorneys' fees and expenses for documentation and negotiation of this Agreement) which are the subject of an invoice delivered to Borrower, incurred through and after the Effective Date, when due.

3 CONDITIONS OF LOANS

3.1 Conditions Precedent to Initial Advance. Bank's obligation to make the initial Advance is subject to the condition precedent that Bank shall have received, in form and substance satisfactory to Bank, such

documents, and completion of such other matters, as Bank may reasonably deem necessary or appropriate, including, without limitation:

party;

(a) Borrower shall have delivered duly executed original signatures to the Loan Documents to which it is a

(b) Other than accounts described in Section 6.8(a)(i), Borrower shall have delivered duly executed original signatures to the Control Agreement(s);

(c) Borrower shall have delivered its Operating Documents and a good standing certificate of Borrower certified by the Secretary of State of the applicable state of incorporation or organization of Borrower, dated as of a date no earlier than thirty (30) days prior to the Effective Date;

for Borrower;

(d)

Borrower shall have delivered duly executed original signatures to the completed Borrowing Resolutions

(e) Borrower shall have delivered the Subordination Agreement duly executed by any holder of Subordinated Debt as required by Bank, in favor of Bank;

(f) Bank shall have received certified copies, dated as of a recent date, of financing statement searches, as Bank shall request, accompanied by written evidence (including any UCC termination statements) that the Liens indicated in any such financing statements either constitute Permitted Liens or have been or, in connection with the initial Credit Extension, will be terminated or released;

(g) Borrower shall have delivered the Perfection Certificate(s) executed by Borrower;

(h) Borrower shall have delivered a bailee's/warehouseman's waiver executed by each bailee, if any, of Borrower as required by Bank, in favor of Bank;

(i) Borrower shall have delivered a legal opinion of Borrower's counsel as to authority and enforceability, dated as of the Effective Date together with the duly executed original signatures thereto;

(j) Borrower shall have delivered evidence satisfactory to Bank that the insurance policies required by Section 6.7 hereof are in full force and effect, together with appropriate evidence showing loss payable and/or additional insured clauses or endorsements in favor of Bank;

(k) the completion of the Initial Audit with results satisfactory to Bank in its sole and absolute discretion; and

(1) Borrower shall have paid the fees and Bank Expenses then due as specified in Section 2.4 hereof.

3.2 Conditions Precedent to all Credit Extensions. Bank's obligations to make each Credit Extension, including the initial Credit Extension, is subject to the following:

(a) Subject to Section 6.2(a), timely receipt of an executed Transaction Report;

(b) the representations and warranties in Section 5 shall be true in all material respects on the date of the Transaction Report and on the Funding Date of each Credit Extension; provided, however, that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof; and provided, further that those representations and warranties expressly referring to a specific date shall be true, accurate and complete in all material respects as of such date, and no Default or Event of Default shall have occurred and be continuing or result from the Credit Extension. The Borrower's acceptance of each Credit Extension is Borrower's representation and warranty on that date that the representations and warranties in Section 5 remain true in all material respects; provided, however, that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof; and provided, further that those representations and warranties that already are qualified or modified by materiality in the text thereof; and provided, further that those representations and warranties that already are qualified or modified by materiality in the text thereof; and provided, further that those representations and warranties expressly referring to a specific date shall be true, accurate and complete in all material respects as of such date; and

(c) in Bank's good faith business judgment, there has not been any material impairment in the general affairs, management, results of operation, financial condition or the prospect of repayment of the Obligations, and that Borrower is in general compliance with its most recent business plan presented to and accepted by Bank.

3.3 Covenant to Deliver.

Borrower agrees to deliver to Bank each item required to be delivered to Bank under this Agreement as a condition to any Credit Extension. Borrower expressly agrees that the extension of a Credit Extension prior to the receipt by Bank of any such item shall not constitute a waiver by Bank of Borrower's obligation to deliver such item, and any such extension in the absence of a required item shall be in Bank's sole discretion.

3.4 Procedures for Borrowing. Subject to the prior satisfaction of all other applicable conditions to the making of an Advance set forth in this Agreement, to obtain an Advance (other than Advances under Sections 2.1.2, 2.1.3 or 2.1.4), Borrower shall notify Bank (which notice shall be irrevocable) by electronic mail, facsimile, or telephone by 3:00 p.m. local Boston time on the Funding Date of the Advance. Together with such notification, Borrower must promptly deliver to Bank by electronic mail or facsimile a completed Transaction Report executed by a Responsible Officer or his or her designee. Bank shall credit Advances to the Designated Deposit Account. Bank may make Advances under this Agreement based on instructions from a Responsible Officer or his or her designee or without instructions if the Advances are necessary to meet Obligations which have become due. Bank may rely on any telephone notice given by a person whom Bank believes in its good faith business judgment, to be a Responsible Officer or designee.

4 CREATION OF SECURITY INTEREST

4.1 Grant of Security Interest. Borrower hereby grants Bank, to secure the payment and performance in full of all of the Obligations, a continuing security interest in, and pledges to Bank, the Collateral, wherever located, whether now owned or hereafter acquired or arising, and all proceeds and products thereof. Borrower represents, warrants, and covenants that the security interest granted herein is and shall at all times continue to be a first priority perfected security interest in the Collateral (except for Permitted Liens that may have superior priority to Bank's Lien under this Agreement). If Borrower shall acquire any commercial tort claims in an aggregate amount in excess of Seventy Five Thousand Dollars (\$75,000.00), Borrower shall promptly notify Bank in a writing signed by Borrower of the general details thereof and grant to Bank in such writing a security interest therein and in the proceeds thereof, all upon the terms of this Agreement, with such writing to be in form and substance reasonably satisfactory to Bank.

4.2 Authorization to File Financing Statements. Borrower hereby authorizes Bank to file financing statements, without notice to Borrower, with all appropriate jurisdictions to perfect or protect Bank's interest or rights hereunder, including a notice that any disposition of the Collateral, by either Borrower or any other Person, shall be deemed to violate the rights of Bank under the Code. Without limiting the foregoing, Borrower hereby authorizes Bank to file financing statements which describe the collateral as "all assets" and/or "all personal property" of Borrower or words of similar import.

5 <u>REPRESENTATIONS AND WARRANTIES</u>

Borrower represents and warrants as follows:

5.1 Due Organization and Authorization. Borrower and each of its Subsidiaries, if any, are duly existing and in good standing as Registered Organizations in their respective jurisdictions of formation and are qualified and licensed to do business and are in good standing in any jurisdiction in which the conduct of their business or their ownership of property requires that they be qualified except where the failure to do so could not reasonably be expected to have a material adverse effect on Borrower's business. In connection with this Agreement, Borrower has delivered to Bank a completed certificate substantially in the form provided by Bank, entitled "Perfection Certificate" (the "Perfection Certificate"). Borrower represents and warrants to Bank that (a) Borrower's exact legal name is that indicated on the Perfection Certificate and on the signature page hereof; (b) Borrower is an organization of the type and is organizational identification number or accurately states that Borrower has none; (d) the Perfection Certificate accurately sets forth Borrower's place of business, or, if more than one, its chief executive office as well as Borrower's mailing address (if different than its chief executive

office); (e) Borrower (and each of its predecessors) has not, in the past five (5) years, changed its jurisdiction of formation, organizational structure or type, or any organizational number assigned by its jurisdiction; and (f) all other information set forth on the Perfection Certificate pertaining to Borrower and each of its Subsidiaries is accurate and complete in all material respects (it being understood and agreed that Borrower may from time to time update certain information in the Perfection Certificate after the Effective Date in connection with actions permitted by this Agreement). If Borrower is not a Registered Organization as of the Effective Date but later becomes one, Borrower shall promptly notify Bank of such occurrence and provide Bank with Borrower's organizational identification number.

The execution, delivery and performance by Borrower of the Loan Documents to which it is a party have been duly authorized, and do not conflict with Borrower's organizational documents, nor constitute an event of default under any material agreement by which Borrower is bound. Borrower is not in default under any agreement to which it is a party or by which it is bound in which the default could reasonably be expected to have a material adverse effect on Borrower's business.

5.2 Collateral. Borrower has good title to, has rights in, and the power to transfer each item of Collateral upon which it purports to grant a Lien hereunder, free and clear of any and all Liens except Permitted Liens. Other than as described in Section 6.8(a), Borrower has no deposit accounts other than the deposit accounts with Bank and deposit accounts described in the Perfection Certificate delivered to Bank in connection herewith, or of which Borrower has given Bank notice and taken such actions as are necessary to give Bank a perfected security interest therein.

The Collateral is not in the possession of any third party bailee (such as a warehouse) except as otherwise provided in the Perfection Certificate. None of the components of the Collateral shall be maintained at locations other than as provided in the Perfection Certificate or as permitted pursuant to Section 7.2. In the event that Borrower, after the date hereof, intends to store or otherwise deliver any portion of the Collateral to a bailee, then Borrower will first receive the written consent of Bank and such bailee must execute and deliver a bailee agreement in form and substance satisfactory to Bank in its sole discretion.

All Inventory is in all material respects of good and marketable quality, free from material defects.

Borrower is the sole owner of its intellectual property, except for non-exclusive licenses granted to its customers in the ordinary course of business. To the best of Borrower's actual knowledge, each patent is valid and enforceable and no part of the intellectual property has been judged invalid or unenforceable, in whole or in part, and to the best of Borrower's knowledge, no claim has been made that any part of the Intellectual Property violates the rights of any third party.

Except as noted on the Perfection Certificate, Borrower is not a party to, nor is bound by, any material license or other agreement with respect to which Borrower is the licensee that prohibits or otherwise restricts Borrower from granting a security interest in Borrower's interest in such license or agreement or any other property. Borrower shall provide written notice to Bank within ten (10) days of entering or becoming bound by any such license or agreement (other than over-the-counter software that is commercially available to the public) which is reasonably likely to have a material impact on Borrower's business or financial condition. Borrower shall take such steps as Bank requests to obtain the consent of, or waiver by, any person whose consent or waiver is necessary for all such licenses or contract rights to be deemed "Collateral" and for Bank to have a security interest in it that might otherwise be restricted or prohibited by law or by the terms of any such license or agreement (such consent or authorization may include a licensor's agreement to a contingent assignment of the license to Bank if Bank determines that is necessary in its good faith judgment), whether now existing or entered into in the future.

5.3 Accounts Receivable.

(a) For each Account with respect to which Advances are requested, on the date each Advance is requested and made, such Account shall meet the Minimum Eligibility Requirements set forth in Section 13 below.

(b) All statements made and all unpaid balances appearing in all invoices, instruments and other documents evidencing the Accounts are and shall be true and correct to the best of Borrower's actual knowledge, and all such invoices, instruments and other documents, and all of Borrower's Books are genuine and in all respects what they purport to be. All sales and other transactions underlying or giving rise to each Account shall comply in all material respects with all applicable laws and governmental rules and regulations. Borrower has no

- 6 -

actual knowledge of any actual or imminent Insolvency Proceeding of any Account Debtor whose accounts are an Eligible Account in any Borrowing Base Certificate. To the best of Borrower's actual knowledge, all signatures and endorsements on all documents, instruments, and agreements relating to all Accounts are genuine, and all such documents, instruments and agreements are legally enforceable in accordance with their terms.

5.4 Litigation. Except as disclosed to the Bank in writing pursuant to Section 6.2 or as listed in the Perfection Certificate, there are no actions or proceedings pending or, to the knowledge of the Responsible Officers, threatened in writing by or against Borrower or any of its Subsidiaries involving more than Fifty Thousand Dollars (\$50,000.00).

5.5 No Material Deviation in Financial Statements. All consolidated financial statements for Borrower and any of its Subsidiaries delivered to Bank fairly present in all material respects Borrower's consolidated financial condition and Borrower's consolidated results of operations in accordance with GAAP. There has not been any material deterioration in Borrower's consolidated financial condition since the date of the most recent financial statements submitted to Bank.

5.6 Solvency. The fair salable value of Borrower's assets (including goodwill minus disposition costs) exceeds the fair value of its liabilities; Borrower is not left with unreasonably small capital after the transactions in this Agreement; and Borrower is able to pay its debts (including trade debts) as they mature.

5.7 Regulatory Compliance. Borrower is not an "investment company" or a company "controlled" by an "investment company" under the Investment Company Act. Borrower is not engaged as one of its important activities in extending credit for margin stock (under Regulations T and U of the Federal Reserve Board of Governors). Borrower has complied in all material respects with the Federal Fair Labor Standards Act. Borrower has not violated any laws, ordinances or rules, the violation of which could reasonably be expected to have a material adverse effect on its business. None of Borrower's or any of its Subsidiaries' properties or assets has been used by Borrower or any Subsidiary or, to the best of Borrower's knowledge, by previous Persons, in disposing, producing, storing, treating, or transporting any hazardous substance other than legally which use could reasonably be expected to cause a Material Adverse Change. Borrower and each of its Subsidiaries have obtained all consents, approvals and authorizations of, made all declarations or filings with, and given all notices to, all government authorities that are necessary to continue its business as currently conducted.

5.8 Subsidiaries; Investments. Borrower does not own any stock, partnership interest or other equity securities except for Permitted Investments.

Tax Returns and Payments; Pension Contributions. Borrower has timely filed all required tax returns and 5.9 reports (except such returns or reports related to taxes as may be due or owing in an amount not to exceed Twenty Five Thousand Dollars (\$25,000.00) in the aggregate), and Borrower and its Subsidiaries, if any, have timely paid all foreign, federal, state and local taxes, assessments, deposits and contributions owed by Borrower (except such returns or reports related to taxes as may be due or owing in an amount not to exceed Twenty Five Thousand Dollars (\$25,000.00) in the aggregate). Borrower may defer payment of any contested taxes, provided that Borrower (a) in good faith contests its obligation to pay the taxes by appropriate proceedings promptly and diligently instituted and conducted, (b) notifies Bank in writing of the commencement of, and any material development in, the proceedings, (c) posts bonds or takes any other steps required to prevent the governmental authority levying such contested taxes from obtaining a Lien upon any of the Collateral that is other than a "Permitted Lien". Borrower is unaware of any claims or adjustments proposed for any of Borrower's prior tax years which could result in additional taxes becoming due and payable by Borrower. Borrower has paid all amounts necessary to fund all present pension, profit sharing and deferred compensation plans in accordance with their terms, and Borrower has not withdrawn from participation in, and has not permitted partial or complete termination of, or permitted the occurrence of any other event with respect to, any such plan which could reasonably be expected to result in any liability of Borrower, including any liability to the Pension Benefit Guaranty Corporation or its successors or any other governmental agency.

5.10 Use of Proceeds. Borrower shall use the proceeds of the Credit Extensions solely as working capital and to fund its general business requirements, and not for personal, family, household or agricultural purposes.

5.11 Full Disclosure. No written representation, warranty or other statement of Borrower in any certificate or written statement given to Bank, as of the date such representations, warranties, or other statements were made, taken together with all such written certificates and written statements given to Bank, contains any

-7-

untrue statement of a material fact or omits to state a material fact necessary to make the statements contained in the certificates or statements not misleading (it being recognized by Bank that the projections and forecasts provided by Borrower in good faith and based upon reasonable assumptions are not viewed as facts and that actual results during the period or periods covered by such projections and forecasts may differ from the projected or forecasted results).

6 <u>AFFIRMATIVE COVENANTS</u>

Borrower shall do all of the following:

6.1 Government Compliance. Maintain its and all its Subsidiaries' legal existence and good standing in their respective jurisdictions of formation and maintain qualification in each jurisdiction in which the failure to so qualify would reasonably be expected to have a material adverse effect on Borrower's business or operations. Borrower shall comply, and have each Subsidiary comply, with all laws, ordinances and regulations to which it is subject, the noncompliance with which could have a material adverse effect on Borrower's business.

6.2 Financial Statements, Reports, Certificates.

(a) Borrower shall provide Bank with the following:

(i) weekly, whenever there are any outstanding Credit Extensions, and upon each request for a Credit Extension, a Transaction Report;

(ii) whenever there are any outstanding Credit Extensions, within twenty (20) days after the end of each month, (A) monthly accounts receivable agings, aged by invoice date, (B) monthly accounts payable agings, aged by invoice date, and outstanding or held check registers, if any, (C) monthly unbilled accounts receivable agings (aged by revenue date), Deferred Revenue report and general ledger, and (D) a schedule of expected collections;

(iii) as soon as available, and in any event within thirty (30) days after the end of each month, monthly unaudited financial statements, unless Borrower has filed financial information with the Securities and Exchange Commission on form 10-Q or 10-K within any given month, in which case Borrower's compliance with Section 6.2(b) of this Agreement shall satisfy this subsection (iii);

(iv) within thirty (30) days after the end of each month a monthly Compliance Certificate signed by a Responsible Officer, certifying that as of the end of such month, Borrower was in full compliance with all of the terms and conditions of this Agreement, and setting forth calculations showing compliance with the financial covenants set forth in this Agreement and such other information as Bank shall reasonably request, including, without limitation, a statement that at the end of such month there were no held checks;

(v) within thirty (30) days after approval by Borrower's board of directors, (A) annual operating budgets (including income statements, balance sheets and cash flow statements, by month) for the upcoming fiscal year of Borrower, and (B) annual financial projections for the following fiscal year (on a quarterly basis), together with any related business forecasts delivered to Borrower's board of directors in the preparation of such annual financial projections; and

(vi) as soon as available, and in any event within one hundred fifty (150) days following the end of Borrower's fiscal year, annual financial statements certified by, and with an unqualified opinion of, independent certified public accountants acceptable to Bank.

Notwithstanding the foregoing, during a Streamline Period, provided no Event of Default has occurred and is continuing, Borrower shall not be required to provide Bank with the Transaction Report required pursuant to clause (a)(i) above; <u>provided</u>, <u>however</u>, that during such Streamline Period, Borrower shall provide Bank, within twenty (20) days after the end of each month in which there were any outstanding Credit Extensions, a duly completed Borrowing Base Certificate signed by a Responsible Officer.

(b) In the event that Borrower is or becomes subject to the reporting requirements under the Securities Exchange Act of 1934, as amended, within five (5) days after filing, all reports on Form 10-K, 10-Q and 8-K filed with the Securities and Exchange Commission or a link thereto on Borrower's or another website on the Internet.

(c) Prompt written notice of (i) any material change in the composition of the intellectual property, (ii) the registration of any copyright, including any subsequent ownership right of Borrower in or to any copyright, patent or trademark not previously disclosed to Bank, or (iii) Borrower's knowledge of an event that has an actual material adverse effect on the value of the intellectual property.

6.3 Accounts Receivable.

(a) <u>Schedules and Documents Relating to Accounts</u>. Borrower shall deliver to Bank transaction reports and schedules of collections, as provided in Section 6.2, on Bank's standard forms; provided, however, that Borrower's failure to execute and deliver the same shall not affect or limit Bank's Lien and other rights in all of Borrower's Accounts, nor shall Bank's failure to advance or lend against a specific Account affect or limit Bank's Lien and other rights therein. If requested by Bank, and to the extent commercially reasonable, Borrower shall furnish Bank with copies (or, at Bank's request, originals) of all contracts, orders, invoices, and other similar documents, and all shipping instructions, delivery receipts, bills of lading, and other evidence of delivery, for any goods the sale or disposition of which gave rise to such Accounts. In addition, Borrower shall deliver to Bank, on its reasonable request and to the extent commercially reasonable, the originals of all instruments, chattel paper, security agreements, guarantees and other documents and property evidencing or securing any Accounts, in the same form as received, with all necessary endorsements, and copies of all credit memos.

(b) <u>Disputes</u>. Borrower shall promptly notify Bank of all material disputes or claims relating to Accounts in an aggregate amount in excess of Fifty Thousand Dollars (\$50,000.00) at any time. Borrower may forgive (completely or partially), compromise, or settle any Account for less than payment in full, or agree to do any of the foregoing so long as (i) Borrower does so in good faith, in a commercially reasonable manner, in the ordinary course of business, in arm's-length transactions, and reports the same to Bank in the regular reports provided to Bank; (ii) no Default or Event of Default has occurred and is continuing; and (iii) after taking into account all such discounts, settlements and forgiveness, the total outstanding Advances will not exceed the Availability Amount.

(c) <u>Collection of Accounts</u>. Borrower shall have the right to collect all Accounts, unless and until a Default or an Event of Default has occurred and is continuing. All payments on, and proceeds of, Accounts shall be deposited directly by the applicable Account Debtor into a lockbox account, or such other "blocked account" as Bank may specify, pursuant to a blocked account agreement in form and substance satisfactory to Bank in its sole discretion. Whether or not an Event of Default has occurred and is continuing, Borrower shall hold all payments on, and proceeds of, Accounts in trust for Bank, and Borrower shall promptly deliver all such payments and proceeds to Bank in their original form, duly endorsed, to be applied to the Obligations pursuant to the terms of Section 9.4 hereof; provided, however, during a Streamline Period, such payments shall be deposited in Borrower's Designated Deposit Account.

(d) <u>Returns</u>. Provided no Event of Default has occurred and is continuing, if any Account Debtor returns any Inventory to Borrower, Borrower shall promptly (i) determine the reason for such return, (ii) issue a credit memorandum to the Account Debtor in the appropriate amount, and (iii) provide a copy of such credit memorandum to Bank, upon request from Bank. In the event any attempted return occurs after the occurrence and during the continuance of any Event of Default, Borrower shall hold the returned Inventory in trust for Bank, and immediately notify Bank of the return of the Inventory.

(e) <u>Verification</u>. Bank may, from time to time, verify directly with the respective Account Debtors the validity, amount and other matters relating to the Accounts, either in the name of Borrower or Bank or such other name as Bank may choose.

(f) <u>No Liability</u>. Bank shall not be responsible or liable for any shortage or discrepancy in, damage to, or loss or destruction of, any goods, the sale or other disposition of which gives rise to an Account, or for any error, act, omission, or delay of any kind occurring in the settlement, failure to settle, collection or failure to collect any Account, or for settling any Account in good faith for less than the full amount thereof, nor shall Bank be deemed to be responsible for any of Borrower's obligations under any contract or agreement giving rise to an Account. Nothing herein shall, however, relieve Bank from liability for its own gross negligence or willful misconduct.

6.4 Remittance of Proceeds. Except as otherwise provided in Section 6.3(c), deliver, in kind, all proceeds arising from the disposition of any Collateral to Bank in the original form in which received by Borrower not later than the following Business Day after receipt by Borrower, to be applied to the Obligations pursuant to the terms of Section 9.4 hereof; provided that, if no Default or Event of Default has occurred and is continuing,

Borrower shall not be obligated to remit to Bank the proceeds of the sale of worn out or obsolete Equipment disposed of by Borrower in good faith in an arm's length transaction for an aggregate purchase price of Fifty Thousand Dollars (\$50,000.00) or less (for all such transactions in any fiscal year). Borrower agrees that it will not commingle proceeds of Collateral with any of Borrower's other funds or property, but will hold such proceeds separate and apart from such other funds and property and in an express trust for Bank. Nothing in this Section limits the restrictions on disposition of Collateral set forth elsewhere in this Agreement.

6.5 Taxes; Pensions. Make, and cause each of its Subsidiaries, if any, to make, timely payment of all foreign, federal, state and local taxes or assessments (other than taxes and assessment which Borrower or its Subsidiaries are contesting pursuant to the terms of Section 5.9 hereof), and shall deliver to Bank, on demand, appropriate certificates attesting to such payments, and pay all amounts necessary to fund all present pension, profit sharing and deferred compensation plans in accordance with their terms.

6.6 Access to Collateral; Books and Records. At reasonable times, on one (1) Business Day's notice (provided no notice is required if an Event of Default has occurred and is continuing), Bank, or its agents, shall have the right on a semi-annual basis (or more frequently if an Event of Default has occurred and is continuing), to inspect the Collateral and the right to andit and copy Borrower's Books. The foregoing inspections and audits shall be at Borrower's expense, and the charge therefor shall be \$750 per person per day (or such higher amount as shall represent Bank's then-current standard charge for the same), plus reasonable out-of-pocket expenses. In the event Borrower and Bank schedule an audit more than ten (10) days in advance, and Borrower cancels or seeks to reschedules the audit with less than ten (10) days written notice to Bank, then (without limiting any of Bank's rights or remedies), Borrower shall pay Bank a fee of \$1,000 plus any out-of-pocket expenses incurred by Bank to compensate Bank for the anticipated costs and expenses of the cancellation or rescheduling.

6.7 Insurance. Keep its business and the Collateral insured for risks and in amounts standard for companies in Borrower's industry and location and as Bank may reasonably request. Insurance policies shall be in a form, with companies, and in amounts that are satisfactory to Bank. All property policies shall have a lender's loss payable endorsement showing Bank as the sole lender loss payee and waive subrogation against Bank, and all liability policies shall show, or have endorsements showing. Bank as an additional insured. All property and liability policies (other than directors and officers liability policies) (or the loss payable and additional insured endorsements) shall provide that the insurer must give Bank at least thirty (30) days notice before canceling, or declining to renew its policy. At Bank's request, Borrower shall deliver certified copies of policies and evidence of all premium payments. Proceeds payable under any policy shall, at Bank's option, be payable to Bank on account of the Obligations. Notwithstanding the foregoing, (a) so long as no Event of Default has occurred and is continuing, Borrower shall have the option of applying the proceeds of any casualty policy up to Fifty Thousand Dollars (\$50,000.00), in the aggregate, toward the replacement or repair of destroyed or damaged property; provided that any such replaced or repaired property (i) shall be of equal or like value as the replaced or repaired Collateral and (ii) shall be deemed Collateral in which Bank has been granted a first priority security interest, and (b) after the occurrence and during the continuance of an Event of Default, all proceeds payable under such casualty policy shall, at the option of Bank, be payable to Bank on account of the Obligations. If Borrower fails to obtain insurance as required under this Section 6.7 or to pay any amount or furnish any required proof of payment to third persons and Bank, Bank may make all or part of such payment or obtain such insurance policies required in this Section 6.7, and take any action under the policies Bank deems prudent.

6.8 Operating Accounts.

(a) Subject to the following, maintain its and its Subsidiaries', if any, primary depository, operating accounts and securities accounts with Bank and Bank's affiliates with all excess funds maintained at or invested through Bank or an affiliate of Bank:

(i) Borrower is permitted to maintain accounts nos. 000000708266184 and 000002330740974 (collectively, the "JP Morgan Chase Bank Accounts"), provided that, whenever the aggregate balance in the JP Morgan Chase Bank Accounts exceeds Fifty Thousand Dollars (\$50,000), such excess amounts in such JP Morgan Chase Bank Accounts shall promptly be transferred to Bank for deposit into such account as Bank shall specify.

(b) Provide Bank five (5) days prior-written notice before establishing any Collateral Account at or with any bank or financial institution other than Bank or its Affiliates. In addition, for each Collateral Account that Borrower at any time maintains, other than the JP Morgan Chase Bank Accounts, Borrower shall cause

- 10 -

the applicable bank or financial institution (other than Bank) at or with which any Collateral Account is maintained to execute and deliver a Control Agreement or other appropriate instrument with respect to such Collateral Account to perfect Bank's Lien in such Collateral Account in accordance with the terms hereunder. The provisions of the previous sentence shall not apply to deposit accounts exclusively used for payroll, payroll taxes and other employee wage and benefit payments to or for the benefit of Borrower's employees and identified to Bank by Borrower as such.

6.9 Financial Covenants.

Borrower shall maintain at all times, to be tested as of the last day of each month, unless otherwise noted, on a consolidated basis with respect to Borrower and its Subsidiaries:

(a) <u>Tangible Net Worth</u>. A Tangible Net Worth of at least Five Hundred Thousand Dollars (\$500,000.00).

6.10 Protection of Intellectual Property Rights. Borrower shall use commercially reasonable efforts to: (a) protect, defend and maintain the validity and enforceability of its intellectual property; (b) promptly advise Bank in writing of material infringements of its intellectual property; and (c) not allow any intellectual property material to Borrower's business to be abandoned, forfeited or dedicated to the public without Bank's written consent.

6.11 Litigation Cooperation. From the date hereof and continuing through the termination of this Agreement, make available to Bank, at reasonable times and intervals, without expense to Bank, Borrower and its officers, employees and agents and Borrower's Books and records, to the extent that Bank may deem them reasonably necessary to prosecute or defend any third-party (for clarity, other than Borrower or its Affiliates) suit or proceeding instituted by or against Bank with respect to any Collateral or relating to Borrower.

6.12 Further Assurances. Borrower shall execute any further instruments and take further action as Bank reasonably requests to perfect or continue Bank's Lien in the Collateral or to effect the purposes of this Agreement.

7 <u>NEGATIVE COVENANTS</u>

Borrower shall not do any of the following without Bank's prior written consent:

7.1 Dispositions. Convey, sell, lease, transfer or otherwise dispose of (collectively, "Transfer"), or permit any of its Subsidiaries to Transfer, all or any part of its business or property, except for Transfers of (a) of Inventory in the ordinary course of business; (b) of worn out or obsolete Equipment; (c) in connection with Permitted Liens and Permitted Investments; and (d) of non-exclusive licenses for the use of the property of Borrower or its Subsidiaries in the ordinary course of business, and licenses that could not result in a legal transfer of title of the licensed property but that may be exclusive in respects other than territory and that may be exclusive as to territory only as to discreet geographical areas outside of the United States. Borrower shall not enter into an agreement with any Person other than Bank which restricts the subsequent granting of a security interest in Borrower's intellectual property.

7.2 Changes in Business, Management, Ownership, Control, or Business Locations. (a) Engage in or permit any of its Subsidiaries, if any, to engage in any business other than the businesses currently engaged in by Borrower and such Subsidiary, as applicable, or reasonably related thereto; (b) liquidate or dissolve; or (c) (i) have a change in any member of the Senior Management Group or (ii) enter into any transaction or series of related transactions in which the stockholders of Borrower who were not stockholders immediately prior to the first such transaction or related series of such transactions (other than by the sale of Borrower's equity securities in a public offering or to venture capital investors so long as Borrower identifies to Bank the venture capital investors prior to the closing of the transaction). Borrower shall not, without at least thirty (30) days prior written notice to Bank: (1) add any new offices or business locations, including warehouses (unless such new offices or business locations contain less than Ten Thousand Dollars (\$10,000.00) in Borrower's assets or property), (2) change its jurisdiction of organization, (3) change its organizational structure or type, (4) change its legal name, or (5) change any organizational number (if any) assigned by its jurisdiction of organization.

- 11 -

7.3 Mergers or Acquisitions. Merge or consolidate, or permit any of its Subsidiaries to merge or consolidate, with any other Person, or acquire, or permit any of its Subsidiaries to acquire, all or substantially all of the capital stock or property of another Person. A Subsidiary may merge or consolidate into another Subsidiary or into Borrower.

7.4 Indebtedness. Create, incur, assume, or be liable for any Indebtedness, or permit any Subsidiary to do so, other than Permitted Indebtedness.

7.5 Encumbrance. Create, incur, or allow any Lien on any of its property, or assign or convey any right to receive income, including the sale of any Accounts, or permit any of its Subsidiaries to do so, except for Permitted Liens, permit any Collateral not to be subject to the first priority security interest granted herein, or enter into any agreement, document, instrument or other arrangement (except with or in favor of Bank) with any Person which directly or indirectly prohibits or has the effect of prohibiting Borrower or any Subsidiary from assigning, mortgaging, pledging, granting a security interest in or upon, or encumbering any of Borrower's or any Subsidiary's intellectual property, except as is otherwise permitted in Section 7.1 hereof and the definition of "Permitted Lien" herein.

7.6 Maintenance of Collateral Accounts. Maintain any Collateral Account except pursuant to the terms of Section 6.8(b) hereof.

7.7 Investments; Distributions. (a) Directly or indirectly make any Investment other than Permitted Investments, or permit any of its Subsidiaries to do so; or (b) pay any dividends or make any distribution or payment or redeem, retire or purchase any capital stock provided that (i) Borrower may convert any of its convertible securities into other securities pursuant to the terms of such convertible securities or otherwise in exchange thereof, (ii) Borrower may pay dividends solely in common stock; and (iii) Borrower may repurchase the stock of existing and former employees or consultants pursuant to board approved stock repurchase agreements so long as (X) an Event of Default does not exist at the time of such repurchase and would not exist after giving effect to such repurchase, and (Y) such repurchase does not exceed Two Hundred Fifty Thousand Dollars (\$250,000.00) in the aggregate in any fiscal year.

7.8 Transactions with Affiliates. Directly or indirectly enter into or permit to exist any material transaction with any Affiliate of Borrower, except for transactions that are in the ordinary course of Borrower's business, upon fair and reasonable terms that are no less favorable to Borrower than would be obtained in an arm's length transaction with a non-affiliated Person.

7.9 Subordinated Debt. (a) Make or permit any payment on any Subordinated Debt, except under the terms of the subordination, intercreditor, or other similar agreement to which such Subordinated Debt is subject, or (b) amend any provision in any document relating to the Subordinated Debt which would increase the amount thereof or adversely affect the subordination thereof to Obligations owed to Bank.

7.10 Compliance. Become an "investment company" or a company controlled by an "investment company", under the Investment Company Act of 1940 or undertake as one of its important activities extending credit to purchase or carry margin stock (as defined in Regulation U of the Board of Governors of the Federal Reserve System), or use the proceeds of any Credit Extension for that purpose; fail to meet the minimum funding requirements of ERISA, permit a Reportable Event or Prohibited Transaction, as defined in ERISA, to occur; fail to comply with the Federal Fair Labor Standards Act or violate any other law or regulation, if the violation could reasonably be expected to have a material adverse effect on Borrower's business, or permit any of its Subsidiaries to do so; withdraw or permit any Subsidiary to withdraw from participation in, permit partial or complete termination of, or permit the occurrence of any other event with respect to, any present pension, profit sharing and deferred compensation plan which could reasonably be expected to result in any liability of Borrower, including any liability to the Pension Benefit Guaranty Corporation or its successors or any other governmental agency.

8 <u>EVENTS OF DEFAULT</u>

Any one of the following shall constitute an event of default (an "Event of Default") under this Agreement:

8.1 Payment Default. Borrower fails to (a) make any payment of principal or interest on any Credit Extension on its due date, or (b) pay any other Obligations within three (3) Business Days after such Obligations are

- 12 -

due and payable. During the cure period, the failure to cure the payment default is not an Event of Default (but no Credit Extension will be made during the cure period);

8.2 Covenant Default.

(a) Borrower fails or neglects to perform any obligation in Sections 6.2, 6.4, 6.5, 6.6, 6.7, 6.8, 6.9, 6.10, or 6.11, or violates any covenant in Section 7; or

(b) Borrower fails or neglects to perform, keep, or observe any other term, provision, condition, covenant or agreement contained in this Agreement, any Loan Documents, and as to any default (other than those specified in this Section 8) under such other term, provision, condition, covenant or agreement that can be cured, has failed to cure the default within ten (10) days after the occurrence thereof; provided, however, that if the default cannot by its nature be cured within the ten (10) day period or cannot after diligent attempts by Borrower be cured within such ten (10) day period, and such default is likely to be cured within a reasonable time, then Borrower shall have an additional period (which shall not in any case exceed thirty (30) days) to attempt to cure such default, and within such reasonable time period the failure to cure the default shall not be deemed an Event of Default (but no Credit Extensions shall be made during such cure period). Grace periods provided under this section shall not apply, among other things, to financial covenants or any other covenants set forth in subsection (a) above;

8.3 Material Adverse Change. A Material Adverse Change occurs;

8.4 Attachment. (a) Any material portion of Borrower's assets is attached, seized, levied on, or comes into possession of a trustee or receiver and the attachment, seizure or levy is not removed in ten (10) days; (b) the service of process upon Bank (or Bank's Affiliate) seeking to attach, by trustee or similar process, any funds of Borrower, or of any entity under control of Borrower (including a Subsidiary) on deposit with Bank (or bank's Affiliate) in an aggregate amount in excess of Fifty Thousand Dollars (\$50,000.00); (c) Borrower is enjoined, restrained, or prevented by court order from conducting a material part of its business; (d) a judgment or other claim in excess of Fifty Thousand Dollars (\$50,000.00) becomes a Lien on any of Borrower's assets and is not stayed or satisfied within ten (10) days; or (e) a notice of lien, levy, or assessment is filed against any of Borrower's assets in an aggregate amount in excess of Fifty Thousand Dollars (\$50,000.00) at any time by any government agency and not paid, discharged or stayed within ten (10) days after Borrower receives notice. These are not Events of Default if stayed or if a bond is posted pending contest by Borrower (but no Credit Extensions shall be made during the cure period);

8.5 Insolvency. (a) Borrower is unable to pay its debts (including trade debts) as they become due or otherwise becomes insolvent; (b) Borrower begins an Insolvency Proceeding; or (c) an Insolvency Proceeding is begun against Borrower and not dismissed or stayed within thirty (30) days (but no Credit Extensions shall be made while of any of the conditions described in clause (a) exist and/or until any Insolvency Proceeding is dismissed);

8.6 Other Agreements. There is a default in any agreement to which Borrower is a party with a third party or parties resulting in a right by such third party or parties, whether or not exercised, to accelerate the maturity of any Indebtedness in an amount in excess of Fifty Thousand Dollars (\$50,000.00) or that could have a material adverse effect on Borrower's business;

8.7 Judgments. A judgment or judgments for the payment of money in an amount, individually or in the aggregate, of at least Fifty Thousand Dollars (\$50,000.00) (not covered by independent third-party insurance) shall be rendered against Borrower and shall remain unsatisfied, unvacated and unstayed for a period of ten (10) days after the entry thereof (provided that no Credit Extensions will be made prior to the satisfaction or stay of such judgment);

8.8 Misrepresentations. Borrower or any Person acting for Borrower makes any representation, warranty, or other statement now or later in this Agreement, any Loan Document or in any writing delivered to Bank or to induce Bank to enter into this Agreement or any Loan Document, and such representation, warranty, or other statement is incorrect in any material respect when made; or

8.9 Subordinated Debt. A default or breach occurs under any agreement between Borrower and any creditor of Borrower that signed a Subordination Agreement, intercreditor, or other similar agreement with Bank, or any creditor that has signed such an agreement with Bank breaches any terms of such agreement.

- 13 -

9 BANK'S RIGHTS AND REMEDIES

9.1 Rights and Remedies. While an Event of Default occurs and continues Bank may, without notice or demand, do any or all of the following:

(a) declare all Obligations immediately due and payable (but if an Event of Default described in Section 8.5 occurs all Obligations are immediately due and payable without any action by Bank);

(b) stop advancing money or extending credit for Borrower's benefit under this Agreement or under any other. agreement between Borrower and Bank;

(c) demand that Borrower (i) deposits cash with Bank in an amount equal to the aggregate amount of any Letters of Credit remaining undrawn, as collateral security for the repayment of any future drawings under such Letters of Credit, and Borrower shall forthwith deposit and pay such amounts, and (ii) pay in advance all Letter of Credit fees scheduled to be paid or payable over the remaining term of any Letters of Credit;

(d) terminate any FX Forward Contracts;

(e) settle or adjust disputes and claims directly with Account Debtors for amounts on terms and in any order that Bank considers advisable, notify any Person owing Borrower money of Bank's security interest in such funds, and verify the amount of such account;

(f) make any payments and do any acts it considers necessary or reasonable to protect the Collateral and/or its security interest in the Collateral. Borrower shall assemble the Collateral if Bank requests and make it available as Bank designates. Bank may enter premises where the Collateral is located, take and maintain possession of any part of the Collateral, and pay, purchase, contest, or compromise any Lien which appears to be prior or superior to its security interest and pay all expenses incurred. Borrower grants Bank a license to enter and occupy any of its premises, without charge, to exercise any of Bank's rights or remedies;

(g) apply to the Obligations any (i) balances and deposits of Borrower it holds, or (ii) any amount held by Bank owing to or for the credit or the account of Borrower;

(h) ship, reclaim, recover, store, finish, maintain, repair, prepare for sale, advertise for sale, and sell the Collateral. Bank is hereby granted a non-exclusive, royalty-free license or other right to use, without charge, Borrower's labels, patents, copyrights, mask works, rights of use of any name, trade secrets, trade names, trademarks, service marks, and advertising matter, or any similar property as it pertains to the Collateral, in completing production of, advertising for sale, and selling any Collateral and, in connection with Bank's exercise of its rights under this Section, Borrower's rights under all licenses and all franchise agreements inure to Bank's benefit;

(i) place a "hold" on any account maintained with Bank and/or deliver a notice of exclusive control, any entitlement order, or other directions or instructions pursuant to any Control Agreement or similar agreements providing control of any Collateral;

(j) demand and receive possession of Borrower's Books; and

(k) exercise all rights and remedies available to Bank under the Loan Documents or at law or equity, including all remedies provided under the Code (including disposal of the Collateral pursuant to the terms thereof).

9.2 Power of Attorney. Borrower hereby irrevocably appoints Bank as its lawful attorney-in-fact, exercisable upon the occurrence and during the continuance of an Event of Default, to: (a) endorse Borrower's name on any checks or other forms of payment or security; (b) sign Borrower's name on any invoice or bill of lading for any Account or drafts against Account Debtors; (c) settle and adjust disputes and claims about the Accounts directly with Account Debtors, for amounts and on terms Bank determines reasonable; (d) make, settle, and adjust all claims under Borrower's insurance policies; (e) pay, contest or settle any Lien, charge, encumbrance, security interest, and adverse claim in or to the Collateral, or any judgment based thereon, or otherwise take any action to terminate or discharge the same; and (f) transfer the Collateral into the name of Bank or a third party as the Code permits. Borrower hereby appoints Bank as its lawful attorney-in-fact to sign Borrower's name on any documents necessary to perfect or continue the perfection of any security interest in the Collateral regardless of whether an Event of Default has occurred until all Obligations have been satisfied in full and Bank is under no further obligation to make Credit Extensions hereunder. Bank's foregoing appointment as Borrower's attorney in fact, and all of Bank's rights and powers, coupled with an interest, are irrevocable until all Obligations have been fully repaid and performed and Bank's obligation to provide Credit Extensions terminates.

9.3 Protective Payments. If Borrower fails to obtain the insurance called for by Section 6.7 or fails to pay any premium thereon or fails to pay any other amount which Borrower is obligated to pay under this Agreement or any other Loan Document, Bank may obtain such insurance or make such payment, and all amounts so paid by Bank are Bank Expenses and immediately due and payable, bearing interest at the then highest applicable rate charged by Bank, and secured by the Collateral. Bank will make reasonable efforts to provide Borrower with notice of Bank obtaining such insurance at the time it is obtained or within a reasonable time thereafter. No payments by Bank are deemed an agreement to make similar payments in the future or Bank's waiver of any Event of Default.

9.4 Application of Payments and Proceeds. Unless an Event of Default has occurred and is continuing, Bank shall apply any funds in its possession, whether from Borrower account balances, payments, or proceeds realized as the result of any collection of Accounts or other disposition of the Collateral, first, to Bank Expenses, including without limitation, the reasonable costs, expenses, liabilities, obligations and attorneys' fees incurred by Bank in the exercise of its rights under this Agreement; second, to the interest due upon any of the Obligations; and third, to the principal of the Obligations and any applicable fees and other charges, in such order as Bank shall determine in its sole discretion. Any surplus shall be paid to Borrower or other Persons legally entitled thereto; Borrower shall remain liable to Bank for any deficiency. If an Event of Default has occurred and is continuing, Bank may apply any funds in its possession, whether from Borrower account balances, payments, proceeds realized as the result of any collection of Accounts or other disposition of the Collateral, or otherwise, to the Obligations in such order as Bank shall determine in its sole discretion. Any surplus shall be paid to Borrower shall remain liable to Bank for any deficiency. If an Event of Default has occurred and is continuing, Bank may apply any funds in its possession, whether from Borrower account balances, payments, proceeds realized as the result of any collection of Accounts or other disposition of the Collateral, or otherwise, to the Obligations in such order as Bank shall determine in its sole discretion. Any surplus shall be paid to Borrower shall remain liable to Bank for any deficiency. If Bank, in its good faith business judgment, directly or indirectly enters into a deferred payment or other credit transaction with any purchaser at any sale of Collateral, Bank shall have the option, exercisable at any time, of either reducing the Obligations by the principal amount of the purchase price or deferring the reducti

9.5 Bank's Liability for Collateral. So long as Bank complies with reasonable banking practices regarding the safekeeping of the Collateral in the possession or under the control of Bank, Bank shall not be liable or responsible for: (a) the safekeeping of the Collateral; (b) any loss or damage to the Collateral; (c) any diminution in the value of the Collateral; or (d) any act or default of any carrier, warchouseman, bailee, or other Person. Borrower bears all risk of loss, damage or destruction of the Collateral.

9.6 No Walver; Remedies Cumulative. Bank's failure, at any time or times, to require strict performance by Borrower of any provision of this Agreement or any other Loan Document shall not waive, affect, or diminish any right of Bank thereafter to demand strict performance and compliance herewith or therewith. No waiver hereunder shall be effective unless signed by Bank and then is only effective for the specific instance and purpose for which it is given. Bank's rights and remedies under this Agreement and the other Loan Documents are cumulative. Bank has all rights and remedies provided under the Code, by law, or in equity. Bank's exercise of one right or remedy is not an election, and Bank's waiver of any Event of Default is not a continuing waiver. Bank's delay in exercising any remedy is not a waiver, election, or acquiescence.

9.7 Demand Waiver. Borrower waives demand, notice of default or dishonor, notice of payment and nonpayment, notice of any default, nonpayment at maturity, release, compromise, settlement, extension, or renewal of accounts, documents, instruments, chattel paper, and guarantees held by Bank on which Borrower is liable.

10 <u>NOTICES</u>

All notices, consents, requests, approvals, demands, or other communication (collectively, "Communication"), other than Advance requests made pursuant to Section 3.4, by any party to this Agreement or any other Loan Document must be in writing and be delivered or sent by facsimile at the addresses or facsimile numbers listed below. Bank or Borrower may change its notice address by giving the other party written notice thereof. Each such Communication shall be deemed to have been validly served, given, or delivered: (a) upon the earlier of actual receipt and three (3) Business Days after deposit in the U.S. mail, registered or certified mail, return receipt requested, with proper postage prepaid; (b) upon transmission, when sent by facsimile transmission (with such facsimile promptly confirmed by delivery of a copy by personal delivery or United States mail as otherwise provided in this Section 10); (c) one (1) Business Day after deposit with a reputable overnight courier with all charges prepaid; or (d) when delivered, if hand-delivered by messenger, all of which shall be addressed to the party to be notified and sent to the address or facsimile number indicated below. Advance requests made pursuant to Section 3.4 must be in writing and may be in the form of electronic mail, delivered to Bank by Borrower at the e-mail address of Bank provided below and shall be deemed to have been validly served, given, or delivered when sent (with such electronic mail promptly confirmed by delivery of a copy by personal delivery or United States mail as otherwise provided in this Section 10). Bank or Borrower may change its address, facsimile number, or electronic mail address by giving the other party written notice thereof in accordance with the terms of this Section 10.

If to Borrower:	World Energy Solutions, Inc. World Energy Securities Corp. c/o World Energy Solutions, Inc. 446 Main Street Worcester, Massachusetts 01608 Attn: Jim Parslow, CFO Fax: (508) 459-8101 Email: jparslow@worldenergy.com
with a copy to:	Wilmer Cutler Pickering Hale and Dorr LLP 60 State Street Boston, Massachusetts Attn: Mr. Mitchel Appelbaum Fax: (617) 526-5000 Email: mitchel.appelbaum@wilmerhale.com
If to Bank:	Silicon Valley Bank One Newton Executive Park, Suite 200 2221 Washington Street, Newton, MA 02462 Attn: Mr. David Rodriguez Fax: (617) 969-4395 Email: drodriguez@svb.com
with a copy to:	Riemer & Braunstein LLP Three Center Plaza Boston, Massachusetts 02108 Attn: Charles W. Stavros, Esquire Fax: (617) 880-3456 Email: cstavros@riemerlaw.com

11 CHOICE OF LAW, VENUE AND JURY TRIAL WAIVER

Massachusetts law governs the Loan Documents without regard to principles of conflicts of law. Borrower and Bank each submit to the exclusive jurisdiction of the State and Federal courts in Massachusetts; provided, however, that nothing in this Agreement shall be deemed to operate to preclude Bank from bringing suit or taking other legal action in any other jurisdiction to realize on the Collateral or any other security for the Obligations, or to enforce a judgment or other court order in favor of Bank. Borrower expressly submits and consents in advance to such jurisdiction in any action or suit commenced in any such court, and Borrower hereby waives any objection that it may have based upon lack of personal jurisdiction, improper venue, or forum non conveniens and hereby consents to the granting of such legal or equitable relief as is deemed appropriate by such court. Borrower hereby waives personal service of the summons, complaints, and other process issued in such action or suit and agrees that service of such summons, complaints, and other process may be made by registered or certified mail addressed to Borrower at the address set forth in Section 10 of this Agreement and that service so made shall be deemed completed upon the earlier to occur of Borrower's actual receipt thereof or three (3) days after deposit in the U.S. mails, proper postage prepaid. NOTWITHSTANDING ANYTHING TO THE CONTRARY SET FORTH HEREINABOVE,

- 16 -

BANK SHALL SPECIFICALLY HAVE THE RIGHT TO BRING ANY ACTION OR PROCEEDING AGAINST BORROWER OR ITS PROPERTY IN THE COURTS OF ANY OTHER JURISDICTION WHICH BANK DEEMS NECESSARY OR APPROPRIATE IN ORDER TO REALIZE ON THE COLLATERAL OR TO OTHERWISE ENFORCE BANK'S RIGHTS AGAINST BORROWER OR ITS PROPERTY.

TO THE EXTENT PERMITTED BY APPLICABLE LAW, BORROWER AND BANK EACH WAIVE THEIR RIGHT TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION ARISING OUT OF OR BASED UPON THIS AGREEMENT, THE LOAN DOCUMENTS OR ANY CONTEMPLATED TRANSACTION, INCLUDING CONTRACT, TORT, BREACH OF DUTY AND ALL OTHER CLAIMS. THIS WAIVER IS A MATERIAL INDUCEMENT FOR BOTH PARTIES TO ENTER INTO THIS AGREEMENT. EACH PARTY HAS REVIEWED THIS WAIVER WITH ITS COUNSEL.

12 <u>GENERAL PROVISIONS</u>

12.1 Termination Prior to Maturity Date. This Agreement may be terminated prior to the Revolving Line Maturity Date by Borrower, effective three (3) Business Days after written notice of termination is received by Bank or if Bank's obligation to fund Credit Extensions terminates pursuant to the terms of Section 2.1.1(b). Notwithstanding any such termination, Bank's lien and security interest in the Collateral shall continue until Borrower fully satisfies its Obligations. If such termination is at Borrower's election or at Bank's election due to the occurrence and continuance of an Event of Default, Borrower shall pay to Bank, in addition to the payment of any other expenses or fees then-owing, a termination fee in an amount equal to Fifteen Thousand Dollars (\$15,000.00) provided that no termination fee shall be charged if the credit facility hereunder is replaced with a new facility from another division of Silicon Valley Bank. Upon payment in full of the Obligations and at such time as Bank's obligation to make Credit Extensions has terminated, Bank shall release its liens and security interests in the Collateral and all rights therein shall revert to Borrower.

12.2 Successors and Assigns. This Agreement binds and is for the benefit of the successors and permitted assigns of each party. Borrower may not assign this Agreement or any rights or obligations under it without Bank's prior written consent (which may be granted or withheld in Bank's discretion). Bank has the right, without the consent of or notice to Borrower, to sell, transfer, negotiate, or grant participation in all or any part of, or any interest in, Bank's obligations, rights, and benefits under this Agreement and the other Loan Documents.

12.3 Indemnification. Borrower agrees to indemnify, defend and hold Bank and its directors, officers, employees, agents, attorneys, or any other Person affiliated with or representing Bank harmless against: (a) all obligations, demands, claims, and liabilities (collectively, "Claims") asserted by any other party in connection with the transactions contemplated by the Loan Documents; and (b) all losses or Bank Expenses incurred, or paid by Bank from, following, or arising from transactions between Bank and Borrower (including reasonable attorneys' fees and expenses), except for Claims and/or losses directly caused by Bank's gross negligence or willful misconduct.

12.4 Time of Essence. Time is of the essence for the performance of all Obligations in this Agreement.

12.5 Severability of Provisions. Each provision of this Agreement is severable from every other provision in determining the enforceability of any provision.

12.6 Amendments in Writing; Integration. All amendments to this Agreement must be in writing signed by both Bank and Borrower. This Agreement and the Loan Documents represent the entire agreement about this subject matter and supersede prior negotiations or agreements. All prior agreements, understandings, representations, warranties, and negotiations between the parties about the subject matter of this Agreement and the Loan Documents merge into this Agreement and the Loan Documents.

12.7 Counterparts. This Agreement may be executed in any number of counterparts and by different parties on separate counterparts, each of which, when executed and delivered, are an original, and all taken together, constitute one Agreement.

12.8 Survival. All covenants, representations and warranties made in this Agreement continue in full force until this Agreement has terminated pursuant to its terms and all Obligations (other than inchoate indemnity obligations and any other obligations which, by their terms, are to survive the termination of this Agreement) have

- 17 -

been satisfied. The obligation of Borrower in Section 12.3 to indemnify Bank shall survive until the statute of limitations with respect to such claim or cause of action shall have run.

12.9 Confidentiality. In handling any confidential information, Bank shall exercise the same degree of care that it exercises for its own proprietary information, but disclosure of information may be made: (a) to Bank's Subsidiaries or Affiliates; (b) to prospective transferees or purchasers of any interest in the Credit Extensions (provided, however, Bank shall use commercially reasonable efforts to obtain such prospective transferee's or purchaser's agreement to the terms of this provision); (c) as required by law, regulation, subpoena, or other order; (d) to Bank's regulators or as otherwise required in connection with Bank's examination or audit; and (e) as Bank considers appropriate in exercising remedies under this Agreement. Confidential information does not include information that either: (i) is in the public domain or in Bank's possession when disclosed to Bank, or becomes part of the public domain after disclosure to Bank; or (ii) is disclosed to Bank by a third party, if Bank does not know that the third party is prohibited from disclosing the information.

12.10 Attorneys' Fees, Costs and Expenses. In any action or proceeding between Borrower and Bank arising out of or relating to the Loan Documents, Bank shall be entitled to recover its reasonable attorneys' fees and other costs and expenses incurred, in addition to any other relief to which it may be entitled.

Borrower Liability. Either Borrower may, acting singly, request Credit Extensions hereunder. Each Borrower 12.11 hereby appoints the other as agent for the other for all purposes hereunder, including with respect to requesting Credit Extensions hereunder. Each Borrower hereunder, jointly and severally, shall be obligated to repay all Credit Extensions made hereunder, regardless of which Borrower actually receives said Advance, as if each Borrower hereunder directly received all Credit Extensions. Each Borrower waives any suretyship defenses available to it under the Code or any other applicable law. Each Borrower waives any right to require Bank to: (i) proceed against any Borrower or any other person; (ii) proceed against or exhaust any security; or (iii) pursue any other remedy. Bank may exercise or not exercise any right or remedy it has against any Borrower or any security it holds (including the right to foreclose by judicial or non-judicial sale) without affecting any Borrower's liability. Notwithstanding any other provision of this Agreement or other related document, each Borrower irrevocably waives all rights that it may have at law or in equity (including, without limitation, any law subrogating Borrower to the rights of Bank under this Agreement) to seek contribution, indemnification or any other form of reimbursement from any other Borrower, or any other Person now or hereafter primarily or secondarily liable for any of the Obligations, for any payment made by Borrower with respect to the Obligations in connection with this Agreement or otherwise and all rights that it might have to benefit from, or to participate in, any security for the Obligations as a result of any payment made by Borrower with respect to the Obligations in connection with this Agreement or otherwise. Any agreement providing for indemnification, reimbursement or any other arrangement prohibited under this Section shall be null and void. If any payment is made to a Borrower in contravention of this Section, such Borrower shall hold such payment in trust for Bank and such payment shall be promptly delivered to Bank for application to the Obligations, whether matured or unmatured.

12.12 Right of Set Off. Borrower hereby grants to Bank, a lien, security interest and right of set off as security for all Obligations to Bank, whether now existing or hereafter arising upon and against all deposits, credits, collateral and property, now or hereafter in the possession, custody, safekeeping or control of Bank or any entity under the control of Bank (including a Bank subsidiary) or in transit to any of them. At any time after the occurrence and during the continuance of an Event of Default, without demand or notice, Bank may set off the same or any part thereof and apply the same to any liability or obligation of Borrower even though unmatured and regardless of the adequacy of any other collateral securing the Obligations. ANY AND ALL RIGHTS TO REQUIRE BANK TO EXERCISE ITS RIGHTS OR REMEDIES WITH RESPECT TO ANY OTHER COLLATERAL WHICH SECURES THE OBLIGATIONS, PRIOR TO EXERCISING ITS RIGHT OF SETOFF WITH RESPECT TO SUCH DEPOSITS, CREDITS OR OTHER PROPERTY OF BORROWER ARE HEREBY KNOWINGLY, VOLUNTARILY AND IRREVOCABLY WAIVED.

13 <u>DEFINITIONS</u>

13.1 Definitions. As used in this Agreement, the following terms have the following meanings:

"Account" is any "account" as defined in the Code with such additions to such term as may hereafter be made, and includes, without limitation, all accounts receivable and other sums owing to Borrower.

- 18 -

"Account Debtor" is any "account debtor" as defined in the Code with such additions to such term as may hereafter be made.

"Advance" or "Advances" means an advance (or advances) under the Revolving Line.

"Affiliate" of any Person is a Person that owns or controls directly or indirectly the Person, any Person that controls or is controlled by or is under common control with the Person, and each of that Person's senior executive officers, directors, partners and, for any Person that is a limited liability company, that Person's managers and members.

"Agreement" is defined in the preamble hereof.

"Adjusted Quick Ratio" is, on any date of determination, the ratio of (a) (i) Borrower's unrestricted cash at Bank, <u>plus</u> (ii) Eligible Accounts, to (b) (i) all Indebtedness of Borrower owed to Bank, <u>plus</u> (ii) without duplication, the aggregate amount of Borrower's Total Liabilities that mature within one (1) year, <u>minus</u> (ii) Deferred Revenue.

"Availability Amount" is (a) the lesser of (i) the Revolving Line or (ii) the Borrowing Base minus (b) the amount of all outstanding Letters of Credit (including drawn but unreimbursed Letters of Credit plus an amount equal to the Letter of Credit Reserves), minus (c) the FX Reserve, and minus (d) the outstanding principal balance of any Advances (including any amounts used for Cash Management Services).

"Bank" is defined in the preamble hereof.

"Bank Expenses" are all audit fees and reasonable expenses, costs, and expenses (including reasonable attorneys' fees and expenses) for preparing, negotiating, administering, defending and enforcing the Loan Documents (including, without limitation, those incurred in connection with appeals or Insolvency Proceedings) or otherwise incurred with respect to Borrower.

"Borrower" is defined in the preamble hereof.

"Borrower's Books" are all Borrower's books and records including ledgers, federal and state tax returns, records regarding Borrower's assets or liabilities, the Collateral, business operations or financial condition, and all computer programs or storage or any equipment containing such information.

"Borrowing Base" is (a) eighty percent (80.0%) of Eligible Accounts, plus (b) twenty-five percent (25%) of Eligible Retail Backlog Accounts, capped at an amount not to exceed twenty-five percent (25%) of clause (a) plus clause (b), in each case as determined by Bank from Borrower's most recent Borrowing Base Certificate; provided, however, that Bank may decrease the foregoing percentages in its good faith business judgment based on events, conditions, contingencies, or risks which, as determined by Bank, may adversely affect the value of the Collateral.

"Borrowing Base Certificate" is that certain certificate included within each Transaction Report.

"Borrowing Resolutions" are, with respect to any Person, those resolutions adopted by such Person's Board of Directors or other appropriate body and delivered by such Person to Bank approving the Loan Documents to which such Person is a party and the transactions contemplated thereby, together with a certificate executed by its secretary on behalf of such Person certifying that (a) such Person has the authority to execute, deliver, and perform its obligations under each of the Loan Documents to which it is a party, (b) that attached as Exhibit A to such certificate is a true, correct, and complete copy of the resolutions then in full force and effect authorizing and ratifying the execution, delivery, and performance by such Person of the Loan Documents to which it is a party, (c) the name(s) of the Person(s) authorized to execute the Loan Documents on behalf of such Person, together with a sample of the true signature(s) of such Person(s), and (d) that Bank may conclusively rely on such certificate unless and until such Person shall have delivered to Bank a further certificate canceling or amending such prior certificate.

"Business Day" is any day that is not a Saturday, Sunday or a day on which Bank is closed.

"Cash Equivalents" means (a) marketable direct obligations issued or unconditionally guaranteed by the United States or any agency or any State thereof having maturities of not more than one (1) year from the date of acquisition; (b) commercial paper maturing no more than one (1) year after its creation and having the highest rating from either Standard & Poor's Ratings Group or Moody's Investors Service, Inc., (c) Bank's certificates of deposit issued maturing no more than one (1) year after issue; and (d) money market funds at least ninety-five percent (95%) of the assets of which constitute Cash Equivalents of the kinds described in clauses (a) through (c) of this definition.

"Cash Management Services" is defined in Section 2.1.4.

"Cash Management Services Sublimit" is defined in Section 2.1.4.

"Code" is the Uniform Commercial Code, as the same may, from time to time, be enacted and in effect in the Commonwealth of Massachusetts; provided, that, to the extent that the Code is used to define any term herein or in any Loan Document and such term is defined differently in different Articles or Divisions of the Code, the definition of such term contained in Article or Division 9 shall govern; provided further, that in the event that, by reason of mandatory provisions of law, any or all of the attachment, perfection, or priority of, or remedies with respect to, Bank's Lien on any Collateral is governed by the Uniform Commercial Code in effect in a jurisdiction other than the Commonwealth of Massachusetts, the term "Code" shall mean the Uniform Commercial Code as enacted and in effect in such other jurisdiction solely for purposes on the provisions thereof relating to such attachment, perfection, priority, or remedies and for purposes of definitions relating to such provisions.

"Collateral" is any and all properties, rights and assets of Borrower described on Exhibit A.

"Collateral Account" is any Deposit Account, Securities Account, or Commodity Account.

"Commodity Account" is any "commodity account" as defined in the Code with such additions to such term as may hereafter be made.

"Communication" is defined in Section 10.

"Compliance Certificate" is that certain certificate in the form attached hereto as Exhibit B.

"Contingent Obligation" is, for any Person, any direct or indirect liability, contingent or not, of that Person for (a) any indebtedness, lease, dividend, letter of credit or other obligation of another such as an obligation directly or indirectly guaranteed, endorsed, co-made, discounted or sold with recourse by that Person, or for which that Person is directly or indirectly liable; (b) any obligations for undrawn letters of credit for the account of that Person; and (c) all obligations from any interest rate, currency or commodity swap agreement, interest rate cap or collar agreement, or other agreement or arrangement designated to protect a Person against fluctuation in interest rates, currency exchange rates or commodity prices; but "Contingent Obligation" does not include endorsements in the ordinary course of business. The amount of a Contingent Obligation is the stated or determined amount of the primary obligation for which the Contingent Obligation is made or, if not determinable, the maximum reasonably anticipated liability for it determined by the Person in good faith; but the amount may not exceed the maximum of the obligations under any guarantee or other support arrangement.

"Control Agreement" is any control agreement entered into among the depository institution at which Borrower maintains a Deposit Account or the securities intermediary or commodity intermediary at which Borrower maintains a Securities Account or a Commodity Account, Borrower, and Bank pursuant to which Bank obtains control (within the meaning of the Code) over such Deposit Account, Securities Account, or Commodity Account.

"Credit Extension" is any Advance, Letter of Credit, FX Forward Contract, amount utilized for Cash Management Services, or any other extension of credit by Bank for Borrower's benefit.

"Default" means any event which with notice or passage of time or both, would constitute an Event of Default.

"Default Rate" is defined in Section 2.3(b).

"Deferred Revenue" is all amounts received or invoiced in advance of performance under contracts and not yet recognized as revenue.

"Deposit Account" is any "deposit account" as defined in the Code with such additions to such term as may hereafter be made.

"Designated Deposit Account" is Borrower's deposit account, account number ______, maintained with Bank.

"Dollars," "dollars" and "S" each mean lawful money of the United States.

"Effective Date" is the date Bank executes this Agreement and as indicated on the signature page hereof.

"Eligible Accounts" are Accounts which arise in the ordinary course of Borrower's business that meet all Borrower's representations and warranties in Section 5.3. Bank reserves the right at any time and from time to time after the Effective Date upon notice to Borrower, to adjust any of the criteria set forth below and to establish new criteria in its good faith business judgment. Without limiting the fact that the determination of which Accounts are eligible for borrowing is a matter of Bank's good faith judgment, the following ("Minimum Eligibility Requirements") are the minimum requirements for an Account to be an Eligible Account. Unless Bank agrees otherwise in writing, Eligible Accounts shall not include:

(a) Accounts that the Account Debtor has not paid within ninety (90) days of invoice date;

(b) Accounts owing from an Account Debtor, fifty percent (50%) or more of whose Accounts have not been paid within ninety (90) days of invoice date;

(c) Credit balances over ninety (90) days from invoice date;

(d) Accounts owing from an Account Debtor, including Affiliates, whose total obligations to Borrower exceed twenty-five (25%) of all Accounts, for the amounts that exceed that percentage, unless Bank approves in writing;

(e) Represent progress billings, or be due under a fulfillment or requirements contract;

(f) Accounts owing from an Account Debtor which does not have its principal place of business in the United States or Canada;

(g) Accounts owing from the United States or any department, agency, or instrumentality thereof except for Accounts of the United States if Borrower has assigned its payment rights to Bank and the assignment has been acknowledged under the Federal Assignment of Claims Act of 1940, as amended;

(h) Accounts owing from an Account Debtor to the extent that Borrower is indebted or obligated in any manner to the Account Debtor (as creditor, lessor, supplier or otherwise - sometimes called "contra" accounts, accounts payable, customer deposits or credit accounts), with the exception of customary credits, adjustments and/or discounts given to an Account Debtor by Borrower in the ordinary course of its business;

(i) Accounts for demonstration or promotional equipment, or in which goods are consigned, or sold on a "sale guaranteed", "sale or return", "sale on approval", "bill and hold", or other terms if Account Debtor's payment may be conditional;

(j) Accounts for which the Account Debtor is Borrower's Affiliate, officer, employee, or agent;

(k) Accounts in which the Account Debtor disputes liability or makes any claim (but only up to the disputed or claimed amount), or if the Account Debtor is subject to an Insolvency Proceeding, or becomes insolvent, or goes out of business;

(1) Accounts owing from an Account Debtor with respect to which Borrower has received Deferred Revenue (but only to the extent of such Deferred Revenue);

(m) Accounts for which Bank in its good faith business judgment determines collection to be doubtful; and

(n) other Accounts Bank deems ineligible in the exercise of its good faith business judgment.

- 21 -

"Eligible Retail Backlog Accounts" are Accounts that (i) are otherwise Eligible Accounts and (ii) are contractually due to be recognized as revenue within ninety (90) days and are the result of completed auctions.

"Equipment" is all "equipment" as defined in the Code with such additions to such term as may hereafter be made, and includes without limitation all machinery, fixtures, goods, vehicles (including motor vehicles and trailers), and any interest in any of the foregoing.

"ERISA" is the Employment Retirement Income Security Act of 1974, and its regulations.

"Event of Default" is defined in Section 8.

"Funding Date" is any date on which a Credit Extension is made to or on account of Borrower which shall be a Business Day.

"FX Business Day" is any day when (a) Bank's Foreign Exchange Department is conducting its normal business and (b) the Foreign Currency being purchased or sold by Borrower is available to Bank from the entity from which Bank shall buy or sell such Foreign Currency.

"FX Forward Contract" is defined in Section 2.1.3.

"FX Reserve" is defined in Section 2.1.3.

"GAAP" is generally accepted accounting principles set forth in the opinions and pronouncements of the Accounting Principles Board of the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or in such other statements by such other Person as may be approved by a significant segment of the accounting profession, which are applicable to the circumstances as of the date of determination.

"General Intangibles" is all "general intangibles" as defined in the Code in effect on the date hereof with such additions to such term as may hereafter be made, and includes without limitation, all copyright rights, copyright applications, copyright registrations and like protections in each work of authorship and derivative work, whether published or unpublished, any patents, trademarks, service marks and, to the extent permitted under applicable law, any applications therefor, whether registered or not, any trade secret rights, including any rights to unpatented inventions, payment intangibles, royalties, contract rights, goodwill, franchise agreements, purchase orders, customer lists, route lists, telephone numbers, domain names, claims, income and other tax refunds, security and other deposits, options to purchase or sell real or personal property, rights in all litigation presently or hereafter pending (whether in contract, tort or otherwise), insurance policies (including without limitation key man, property damage, and business interruption insurance), payments of insurance and rights to payment of any kind.

"Indebtedness" is (a) indebtedness for borrowed money or the deferred price of property or services, such as reimbursement and other obligations for surety bonds and letters of credit, (b) obligations evidenced by notes, bonds, debentures or similar instruments, (c) capital lease obligations, and (d) Contingent Obligations.

"Initial Audit" is Bank's initial inspection of the Collateral and Borrower's Books.

"Insolvency Proceeding" is any proceeding by or against any Person under the United States Bankruptcy Code, or any other bankruptcy or insolvency law, including assignments for the benefit of creditors, compositions, extensions generally with its creditors, or proceedings seeking reorganization, arrangement, or other relief.

"Inventory" is all "inventory" as defined in the Code in effect on the date hereof with such additions to such term as may hereafter be made, and includes without limitation all merchandise, raw materials, parts, supplies, packing and shipping materials, work in process and finished products, including without limitation such inventory as is temporarily out of Borrower's custody or possession or in transit and including any returned goods and any documents of title representing any of the above.

"Investment" is any beneficial ownership interest in any Person (including stock, partnership interest or other securities), and any loan, advance or capital contribution to any Person.

- 22 -

"Letter of Credit" means a standby letter of credit issued by Bank or another institution based upon an application, guarantee, indemnity or similar agreement on the part of Bank as set forth in Section 2.1.2.

"Letter of Credit Application" is defined in Section 2.1.2(a).

"Letter of Credit Reserve" has the meaning set forth in Section 2.1.2(d).

"Lien" is a mortgage, lien, deed of trust, charge, pledge, security interest or other encumbrance.

"Loan Documents" are, collectively, this Agreement, the Perfection Certificate, the Subordination Agreement, if any, any note, or notes or guaranties executed by Borrower, or any other present or future agreement executed by Borrower, if any, and/or for the benefit of Bank in connection with this Agreement, each as amended, restated, or otherwise modified.

"Material Adverse Change" is (a) a material impairment in the perfection or priority of Bank's Lien in the Collateral or in the value of such Collateral; (b) a material adverse change in the business, operations, or condition (financial or otherwise) of Borrower; (c) a material impairment of the prospect of repayment of any portion of the Obligations or (d) Bank determines, based upon information available to it and in its reasonable judgment, that there is a reasonable likelihood that Borrower shall fail to comply with one or more of the financial covenants in Section 6 during the next succeeding financial reporting period.

"Minimum Eligibility Requirements" is defined in the defined term "Eligible Accounts".

"Obligations" are Borrower's obligation to pay when due any debts, principal, interest, Bank Expenses and other amounts Borrower owes Bank now or later, whether under this Agreement, the Loan Documents, including, without limitation, all obligations relating to letters of credit, cash management services, and foreign exchange contracts, if any, and including interest accruing after Insolvency Proceedings begin and debts, liabilities, or obligations of Borrower assigned to Bank, and the performance of Borrower's duties under the Loan Documents.

"Operating Documents" are, for any Person, such Person's formation documents, as certified with the Secretary of State of such Person's state of formation on a date that is no earlier than 30 days prior to the Effective Date, and, (a) if such Person is a corporation, its bylaws in current form, (b) if such Person is a limited liability company, its limited liability company agreement (or similar agreement), and (c) if such Person is a partnership, its partnership agreement (or similar agreement), each of the foregoing with all current amendments or modifications thereto.

"Payment" means all checks, wire transfers and other items of payment received by Bank (including proceeds of Accounts and payment of all the Obligations in full) for credit to Borrower's outstanding Credit Extensions or, if the balance of the Credit Extensions has been reduced to zero, for credit to its Deposit Accounts.

"Perfection Certificate" is defined in Section 5.1.

"Permitted Indebtedness" is:

- (a) Borrower's Indebtedness to Bank under this Agreement and the other Loan Documents;
- (b) Indebtedness existing on the Effective Date and shown on the Perfection Certificate;
- (c) Subordinated Debt, if any;
- (d) unsecured Indebtedness to trade creditors incurred in the ordinary course of business; any
- (e) Indebtedness incurred as a result of endorsing negotiable instruments received in the ordinary course of business;
- (f) Indebtedness secured by Permitted Liens;

(g) Indebtedness not otherwise permitted by Section 7.4 in an amount not to exceed Fifty Thousand Dollars (\$50,000) in the aggregate outstanding at any time; and

(h) extensions, refinancings, modifications, amendments and restatements of any items of Permitted Indebtedness (a) through (g) above, provided that the principal amount thereof is not increased or the terms thereof are not modified to impose more burdensome terms upon Borrower or its Subsidiary, as the case may be.

"Permitted Investments" are:

(a) Investments shown on the Perfection Certificate and existing on the Effective Date;

(b) Cash Equivalents;

(c) Investments consisting of the endorsement of negotiable instruments for deposit or collection or similar transactions in the ordinary course of Borrower's business;

(d) Investments consisting of (i) travel advances and employee relocation loans and other employee loans and advances in the ordinary course of business, and (ii) loans to employees, officers or directors relating to the purchase of equity securities of Borrower or its Subsidiaries pursuant to employee stock purchase plans or agreements approved by Borrower's Board of Directors;

(e) Investments (including debt obligations) received in connection with the bankruptcy or reorganization of customers or suppliers and in settlement of delinquent obligations of, and other disputes with, customers or suppliers arising in the ordinary course of business; and

(f) Investments consisting of notes receivable of, or prepaid royalties and other credit extensions, to customers and suppliers who are not Affiliates, in the ordinary course of business; provided that this paragraph (f) shall not apply to Investments of Borrower in any Subsidiary;

"Permitted Liens", whether now existing or hereafter arising, are:

(a) Liens existing on the Effective Date and shown on the Perfection Certificate or arising under this Agreement and the other Loan Documents;

(b) Liens for taxes, fees, assessments or other government charges or levies, either not delinquent or being contested in good faith and for which Borrower maintains adequate reserves on its Books, if they have no priority over any of Bank's Liens;

(c) purchase money Liens (i) on Equipment acquired or held by Borrower incurred for financing the acquisition of the Equipment securing no more than One Hundred Thousand Dollars (\$100,000.00) in the aggregate amount outstanding, or (ii) existing on Equipment when acquired, <u>if</u> the Lien is confined to the property and improvements and the proceeds of the Equipment;

(d) Liens of carriers, warehousemen, suppliers, or other Persons that are possessory in nature arising in the ordinary course of business, so long as such Liens attach only to Inventory, that are not delinquent, that remain payable without penalty, or that are being contested in good faith and by appropriate proceedings that have the effect of preventing the forfeiture or sale of the property subject to such Liens;

(e) Liens to secure payment of workers' compensation, employment insurance, old age pensions, social security, and other like obligations incurred in the ordinary course of business (other than Liens imposed by ERISA);

(f) Liens incurred in the extension, renewal or refinancing of the indebtedness secured by Liens described in (a) through (c), but any extension, renewal or replacement Lien must be limited to the property encumbered by the existing Lien and the principal amount of the indebtedness may not increase;

(g) leases or subleases of real property granted in the ordinary course of business, and leases, subleases, non-exclusive licenses or sublicenses of property (other than real property or intellectual property) granted in the ordinary course of Borrower's business, if the leases, subleases, licenses and sublicenses do not prohibit granting Bank a security interest;

- 24 -

(h) non-exclusive licenses of intellectual property granted to third parties in the ordinary course of business;

(i) Liens arising from judgments, decrees or attachments in circumstances not constituting an Event of Default under Section 8.4 or 8.7;

(j) Liens securing Permitted Indebtedness;

(k) Liens in favor of any landlord of Borrower solely to secure the payment of rent by Borrower; and

(k) Liens in favor of other financial institutions arising in connection with Borrower's deposit and/or securities accounts held at such institutions.

"Person" is any individual, sole proprietorship, partnership, limited liability company, joint venture, company, trust, unincorporated organization, association, corporation, institution, public benefit corporation, firm, joint stock company, estate, entity or government agency.

"Prime Rate" is Bank's most recently announced "prime rate," even if it is not Bank's lowest rate.

"Registered Organization" is any "registered organization" as defined in the Code with such additions to such term as may hereafter be made.

"Reserves" means, as of any date of determination, such amounts as Bank may from time to time establish and revise in good faith reducing the amount of Advances, Letters of Credit and other financial accommodations which would otherwise be available to Borrower under the lending formulas: (a) to reflect events, conditions, contingencies or risks which, as determined by Bank in good faith, do or may affect (i) the Collateral or any other property which is security for the Obligations or its value (including without limitation any increase in delinquencies of Accounts), (ii) the assets or business of Borrower or any guarantor, or (iii) the security interests and other rights of Bank in the Collateral (including the enforceability, perfection and priority thereof); or (b) to reflect Bank's good faith belief that any collateral report or financial information furnished by or on behalf of Borrower or any guarantor to Bank is or may have been incomplete, inaccurate or misleading in any material respect; or (c) in respect of any state of facts which Bank determines in good faith constitutes an Event of Default or may, with notice or passage of time or both, constitute an Event of Default.

"Responsible Officer" is any of the Chief Executive Officer, President, Chief Financial Officer and Controller of Borrower.

"Revolving Line" is an Advance or Advances in an aggregate amount of up to Three Million Dollars (\$3,000,000.00) outstanding at any time.

"Revolving Line Maturity Date" is September 7, 2009.

"Securities Account" is any "securities account" as defined in the Code with such additions to such term as may hereafter be made.

"Senior Management Group" is the group of individuals, exclusive of others, who, as of the Effective Date, hold any of the following positions at Borrower: Chief Executive Officer, Chief Operating Officer, and Chief Financial Officer.

"Settlement Date" is defined in Section 2.1.3.

"Streamline Period" is any date of measurement in which Borrower's Adjusted Quick Ratio is greater than or equal to 1.0 to 1.0.

"Subordinated Debt" is indebtedness incurred by Borrower subordinated to all of Borrower's now or hereafter indebtedness to Bank (pursuant to a subordination, intercreditor, or other similar agreement in form and substance satisfactory to Bank entered into between Bank and the other creditor), on terms acceptable to Bank.

- 25 -

"Subordination Agreement" is any agreement, in form and substance acceptable to Bank in its sole discretion, as required by Bank in its sole discretion, subordinating Subordinated Debt to the Bank.

"Subsidiary" means, with respect to any Person, any Person of which more than fifty percent (50%) of the voting stock or other equity interests is owned or controlled, directly or indirectly, by such Person or one or more Affiliates of such Person.

"Tangible Net Worth" is, on any date, the consolidated total assets of Borrower and its Subsidiaries <u>minus</u> (a) any amounts attributable to (i) goodwill, (ii) intangible items including unamortized debt discount and expense, patents, trade and service marks and names, copyrights and research and development expenses except prepaid expenses, (iii) notes, accounts receivable and other obligations owing to Borrower from its officers or other Affiliates, and (iv) reserves not already deducted from assets, <u>minus</u> (b) Total Liabilities <u>plus</u> (c) Subordinated Debt.

"Total Liabilities" is on any day, obligations that should, under GAAP, be classified as liabilities on Borrower's consolidated balance sheet, including all Indebtedness, and current portion of Subordinated Debt permitted by Bank to be paid by Borrower, but excluding all other Subordinated Debt.

"Transaction Report" is the Bank's standard reporting package provided by Bank to Borrower.

"Transfer" is defined in Section 7.1.

"Unused Revolving Line Facility Fee" is defined in Section 2.4(d).

[Signature page follows.]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed as a sealed instrument under the laws of the Commonwealth of Massachusetts as of the Effective Date.

BORROWER:

WORLD ENERGY SOLUTIONS, INC.

By_____

Name:_____

Title:_____

WORLD ENERGY SECURITIES CORP.

By_____

Name:_____

Title:_____

BANK:

SILICON VALLEY BANK

By_____

Name:_____

Title:_____

Effective Date: September ____, 2008

[Signature page to Loan and Security Agreement]

Source: World Energy Solutio, 8-K, September 08, 2008

EXHIBIT A

The Collateral consists of all of Borrower's right, title and interest in and to the following personal property:

All goods, Accounts (including health-care receivables), Equipment, Inventory, contract rights or rights to payment of money, leases, license agreements, franchise agreements, General Intangibles (except as provided below), commercial tort claims, documents, instruments (including any promissory notes), chattel paper (whether tangible or electronic), cash, deposit accounts, fixtures, letters of credit rights (whether or not the letter of credit is evidenced by a writing), securities, and all other investment property, supporting obligations, and financial assets, whether now owned or hereafter acquired, wherever located; and

all Borrower's Books relating to the foregoing, and any and all claims, rights and interests in any of the above and all substitutions for, additions, attachments, accessories, accessions and improvements to and replacements, products, proceeds and insurance proceeds of any or all of the foregoing.

Notwithstanding the foregoing, the Collateral does not include any of the following, whether now owned or hereafter acquired: any copyright rights, copyright applications, copyright registrations and like protections in each work of authorship and derivative work, whether published or unpublished, any patents, patent applications and like protections, including improvements, divisions, continuations, renewals, reissues, extensions, and continuations-in-part of the same, trademarks, service marks and, to the extent permitted under applicable law, any applications therefor, whether registered or not, and the goodwill of the business of Borrower connected with and symbolized thereby, know-how, operating manuals, trade secret rights, rights to unpatented inventions, and any claims for damage by way of any past, present, or future infringement of any of the foregoing, and any personal property in which a purchase money creditor has or obtains a purchase money security interest, but only (a) to the extent such purchase money creditor prohibits the Borrower from further encumbering such personal property; and (b) for as long as such purchase money security interest exists in such personal property; provided, however, the Collateral shall include all Accounts, license and royalty fees and other revenues, proceeds, or income arising out of or relating to any of the foregoing.

Pursuant to the terms of a certain negative pledge arrangement with Bank, Borrower has agreed not to encumber any of its copyright rights, copyright applications, copyright registrations and like protections in each work of authorship and derivative work, whether published or unpublished, any patents, patent applications and like protections, including improvements, divisions, continuations, renewals, reissues, extensions, and continuations-in-part of the same, trademarks, service marks and, to the extent permitted under applicable law, any applications therefor, whether registered or not, and the goodwill of the business of Borrower connected with and symbolized thereby, know-how, operating manuals, trade secret rights, rights to unpatented inventions, and any claims for damage by way of any past, present, or future infringement of any of the foregoing, without Bank's prior written consent.

EXHIBIT B

COMPLIANCE CERTIFICATE

Date:

TO: SILICON VALLEY BANK FROM: WORLD ENERGY SOLUTIONS, INC. AND WORLD ENERGY SECURITIES CORP.

The undersigned authorized officers of World Energy Solutions, Inc., and World Energy Securities Corp. (individually and collectively, jointly and severally, "Borrower"), solely in their capacities as officers of their respective entities, certify that under the terms and conditions of the Loan and Security Agreement between Borrower and Bank (the "Agreement"), (1) Borrower is in with all required covenants except as noted below, (2) there are no complete compliance for the period ending Events of Default, (3) all representations and warranties in the Agreement are true and correct in all material respects on this date except as noted below; provided, however, that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof; and provided, further that those representations and warranties expressly referring to a specific date shall be true, accurate and complete in all material respects as of such date, (4) Borrower, and each of its Subsidiaries, has timely filed all required tax returns and reports, and Borrower has timely paid all foreign, federal, state and local taxes, assessments, deposits and contributions owed by Borrower except as otherwise permitted pursuant to the terms of Section 5.9 of the Agreement, and (5) no Liens have been levied or claims made against Borrower or any of its Subsidiaries, if any, relating to unpaid employee payroll or benefits of which Borrower has not previously provided written notification to Bank. Attached are the required documents supporting the certification. The undersigned solely in their capacities as officers of their respective entities, certify that these are prepared in accordance with generally GAAP consistently applied from one period to the next except as explained in an accompanying letter or footnotes. The undersigned solely in their capacities as officers of their respective entities, acknowledge that no borrowings may be requested at any time or date of determination that Borrower is not in compliance with any of the terms of the Agreement, and that compliance is determined not just at the date this certificate is delivered. Capitalized terms used but not otherwise defined herein shall have the meanings given them in the Agreement.

Please indicate compliance status by circling Yes/No under "Complies" column.

Reporting Covenant	Required	<u>Complies</u>		
Monthly financial statements with Compliance Certificate	Monthly within 30 days	Yes No		
Annual financial statement (CPA Audited) + CC	FYE within 150 days	Yes No		
10-Q, 10-K and 8-K	Within 5 days after filing with SEC	Yes No		
A/R & A/P Agings, Deferred Revenue report, and schedule of expected collections	Monthly within 20 days	Yes No		
Borrowing Base Certificate	Monthly within 20 Days during Streamline Period	Yes No		
Transaction Report	Weekly when not Streamline Period	Yes No		
Board-approved projections	Within 30 days of approval	Yes No		
Financial Covenant	Required Actual	Complies		
Maintain, to be tested monthly:		e tra sugar e day e		
Tangible Net Worth	\$500,000 \$	Yes No		

The following financial covenant analyses and information set forth in Schedule 1 attached hereto are true and accurate as of the date of this Certificate.

The following are the exceptions with respect to the certification above: (If no exceptions exist, state "No exceptions to note.")

WORLD ENERGY SOLUTIONS, INC.	BANK USE ONLY				
By:	Received by:				
Name:		RIZED SIGNER			
Title:	Date:				
WORLD ENERGY SECURITIES CORP.	Verified:				
Den		RIZED SIGNER			
By:	Date:				
Name:					
Title:	Compliance Status:	Yes No			

Schedule 1 to Compliance Certificate

Financial Covenants of Borrower

Dated: _		
I. '	Tangible Net Worth (Section 6.9(a))	
Required	ð: \$500,000.00	
Actual: A.	Aggregate value of consolidated total assets of Borrower and its Subsidiaries	\$
B.	1. Aggregate value of goodwill of Borrower and its Subsidiaries	\$
	2. Aggregate value of intangible assets of Borrower and its Subsidiaries	\$
	3. Aggregate value of notes, accounts receivables, and other obligations owing to Borrower from its officers or other Affiliates	\$
	4. Aggregate value of any reserves not already deducted from assets	\$
	5. Total Liabilities	\$
С.	Net Tangible net Worth (line A minus lines B.1 through B.5)	\$
D.	Subordinated Debt	\$
E.	Tangible Net Worth (line C_plus line D)	\$
Is line E	equal to or greater than \$500,000.00?	
	No, not in compliance Yes, in compliance	

.



World Energy Solutions secures US\$3 million credit facility with Silicon Valley Bank

Worcester, MA – September 8, 2008 – World Energy Solutions, Inc. (TSX: XWE) (the "Company") today announced that it has entered into a Loan and Security Agreement (the "Agreement") with Silicon Valley Bank ("SVB"). Under the Agreement, SVB has committed to make advances to the Company in an aggregate amount of up to \$3,000,000 subject to availability against certain eligible account receivables and eligible retail backlog.

Jim Parslow, Chief Financial Officer of the Company, said, "We are pleased that SVB provided this credit facility, which gives us increased financial strength and flexibility as we grow the business. As we reported in our second quarter earnings, we continue to believe we have the financial resources to achieve our strategic growth initiatives, and expect further decreases in cash used in operations this year based on strong revenue growth in multiple product lines and a stable fixed cost base. We have continued to experience a rise in auction activity during the third quarter as commodity prices have declined significantly after the dramatic run-up from February through June."

About Silicon Valley Bank

Silicon Valley Bank is the premier commercial bank for companies in the technology, life science, private equity and premium wine industries. SVB provides a comprehensive suite of financing solutions, treasury management, corporate investment and international banking services to its clients worldwide. Through its focus on specialized markets and extensive knowledge of the people and business issues driving them, Silicon Valley Bank provides a level of service and partnership that measurably impacts its clients' success. Founded in 1983 and headquartered in Santa Clara, Calif., the company serves clients around the world through 27 U.S. offices and five international operations. Silicon Valley Bank is a member of global financial services firm SVB Financial Group, with SVB Analytics, SVB Capital, SVB Global and SVB Private Client Services. More information on the company can be found at http://www.svb.com.

About World Energy

World Energy operates leading online exchanges for energy and environmental commodities. Our proven approach provides market intelligence, promotes liquidity, and creates price transparency for all market participants, enabling our customers to transact with confidence and to seek the best possible price. To date, the company has transacted more than 45 billion kWh of electricity, 1 billion kWh of green power and Renewable Energy Certificates (RECs) and over 100 billion Dth of natural gas. For more information, please visit www.worldenergy.com.

This press release contains forward-looking statements that are subject to risks and uncertainties that could cause actual results to differ from those indicated in the forward-looking statements. Such risks and uncertainties include, but are not limited to: our revenue is dependent on actual future energy purchases pursuant to completed procurements; the demand for our services is affected by changes in regulated prices or cyclicality or volatility in competitive market prices for energy; we depend on a small number of key energy consumers, suppliers and channel partners; there are factors outside our control that affect transaction volume in the electricity market; and there are other factors identified in our Annual Report on Form 10-K and subsequent reports filed with the Securities and Exchange Commission.

For additional information, contact:

Jim Parslow World Energy Solutions, Inc. (508) 459-8100 jparslow@worldenergy.com

Craig Armitage The Equicom Group (416) 815-0700 x278 carmitage@equicomgroup.com Created by 10KWizard www.10KWizard.com

.

ð

Exhibit C-3 "Financial Statements"

World Energy Solutions, Inc. has attached the most recent two years of audited financial statements which are for year ending 2006 and 2007.

Consolidated Balance Sheets

		December 31.		
		2007		2006
ASSETS				
Current assets:				
Cash and cash equivalents	\$	7,001,884	\$	17,483,886
Trade accounts receivable, net		1,878,233		1,702,798
Prepaid expenses and other current assets		<u>338,049</u>		<u>211.749</u>
Total current assets		9,218,166		19,3 98,4 33
Property and equipment, net		595,950		225,932
Capitalized software, net		463,888		105,296
Intangibles, net		7,316,916		
Goodwill		3,178,701		_
Deferred taxes and other assets		26,944		1.061.720
Total assets	<u>s</u> :	<u>20,800,565</u>	<u>\$</u>	20,791,381
LIABILITIES AND STOCKHOLDERS' EQUIT	TY			
Current liabilities:				
Accounts payable	\$	980,488	\$	848,518
Accrued commissions		712,919		1,021,433
Accrued compensation		1,516,418		304,330
Accrued expenses		279,162		232,587
Deferred revenue and customer advances		363,933		295,408
Capital lease obligations		41,624		56,259
Total current liabilities		3,894,544		2,758,535
Capital lease obligations, net of current portion		46.222		87,844
Total liabilities		3,940,766		2,846,379
Commitments (Note 9)				
Stockholders' equity:				
Preferred stock, \$0.0001 par value; 5,000,000 shares authorized, no shares issued or outstanding				
Common stock, \$0,0001 par value; 150,000,000 shares authorized; 82,399,241 and 76,511,741 shares issued and outstanding at December 31, 2007 and 2006, respectively		8,240		7,651
Additional paid-in capital		29,704,831		22,148,718
Accumulated deficit		12.853.272)	_	(4,211,367)
Total stockholders' equity		16.859.799	_	17.945.002
Total liabilities and stockholders' equity	\$	20,800,565	\$	20.791.381
		<u>_</u>	-	

The accompanying notes are an integral part of these consolidated financial statements.

Consolidated Statements of Operations

	Years Ended December 31,			
	2007	2006	2005	
Revenue:				
Brokerage commissions and transaction fees	\$ 8,401,79 1	\$ 5,762,098	\$ 4,673,987	
Management fees	786,474	1.000		
Total revenue	9,188,265	5,763,098	4,673,987	
Cost of revenue	2.874.678	<u>1.166,426</u>	648,410	
Gross profit	6,313,587	4.596.672	4,025,577	
Operating expenses:				
Sales and marketing	8,598,256	3,227,263	2,649,786	
General and administrative	<u> </u>	1.862,450	995,703	
Total costs and expenses	14,457,066	5.089.713	3,645,489	
Operating income (loss)	(8,143,479)	<u>(493,041)</u>	380,088	
Interest income (expense):				
Interest income	573,395	139,617	8,004	
Interest expense	(10,101)	<u>(451,897)</u>	<u>(94,842</u>)	
Total interest income (expense), net	563,294	(312,280)	<u>(86,838</u>)	
Income (loss) before income taxes	(7,580,185)	(805,321)	293,250	
Income tax (expense) benefit	(1,061,720)	304,228	754,000	
Net income (loss)	(8,641,905)	(501,093)	1,047,250	
Accretion of preferred stock issuance costs		<u>(6,299</u>)	<u>(7,199</u>)	
Net income (loss) available to common stockholders.	<u>\$ (8.641.905)</u>	<u>\$ (507,392)</u>	<u>\$ 1.040,051</u>	
Earnings (loss) per share:				
Net income (loss) per voting common share — basic.	<u>\$ (0.11)</u>	<u>\$ (0.01)</u>	<u>\$ 0.02</u>	
Net income (loss) per non-voting common share - basic	<u>\$</u>	<u>\$</u>	<u>\$ 0.02</u>	
Net income (loss) available to common stockholders — diluted	<u>\$ (0.11)</u>	<u>\$ (0.01)</u>	<u>\$ 0.02</u>	
Weighted average shares outstanding basic:				
Voting common stock	79,793,590	45,576,477	33,049,472	
Non-voting common stock			6,778,327	
	79,793,590	45,576,477	<u> </u>	
Weighted average shares outstanding diluted	<u> </u>	45.576,477	54.506.566	

The accompanying notes are an integral part of these consolidated financial statements.

ł

Consolidated Statements of Stockholders' Equity (Deficit) Years Ended December 31, 2007, 2006, and 2005

		Non-Voting Common Stock		Voting Common Stock		ry Stock	Additional		Tetal
	Number of	<u>s0.0001</u>	<u> </u>	\$0,0001	Number of	Stated at	Paid-in	Accumulated	Stockholders'
	Shares	Par Value	Shares	<u>Per Value</u>	Shares	Cost	Canital	Deficit	Equity (Deficit)
Balance, December 31, 2004	6,552,135	655	33,724,819	3,372	675,346	(151,953)	2,664,316	(4,757,524)	(2,241,134)
Exercise of stock options	240,000	24		-		_	5,976	_	6,000
Accretion of stock issuance costs	_	_	-	_	-	—	(7,199)		(7,199)
Issuance of warrants in connection with long- term debt	_	_	_	_		_	256,200		256,200
Net income								1.047.250	1.047.250
Balance, December 31, 2005	6,792,135	679	33,724,819	3,372	675,346	(151,953)	2,919,293	(3,710,274)	(938,883)
Exercise of stock options	2,190,671	219	_	—	_		56,698	_	56,917
Common stock warrants exercised	_	_	1,070,958	107	-	-	253,098	-	253,205
Issuance of common stock in connection with initial public offering, net of \$3.6 million of							10 04 040		17.267.268
offering costs			23,000,000	2,300	_	_	17,264.968	_	17,207,206
Issuance of warrants in connection with initial public offering	_	-	_	_	_	_	274,129	-	274,129
Conversion of Non-Voting Common Stock to Voting Common Stock in connection with initial public offering	(8,982,806)	(898)	8 ,982 ,806	8 98	_	_	_	-	_
Conversion of Series A Redeemable Convertible Preferred Stock in connection with initial public offering		_	10,433,504	1,044	_	_	1.506.953		1,507,997
Accretion of stock issuance costs	_	_			_	—	(6,299)	_	(6,299)
Share-based compensation	_	_	_	_	_		44.261	_	44,261
Purchase of treasury stock	-	<u> </u>	-	_	25,000	(12,500)		_	(12,500)
Retirement of treasury stock			(700.346)	(70)	(700,346)	164,453	(164,383)	_	_
Net loss		_	(100,540)	(/•)	(/001010)		·····	(501.093)	(501,093)
Balance, December 31, 2006			76,511,741	7,651			22,148,718	(4,211,367)	17,945,002
Share-based compensation	_	_	-		_	_	982,190		982,190
Issuance of common stock in connection with acquisition of EnergyGateway	_	-	5,375,000	538	—	-	6,522,694	-	6,523,232
Issuance of common stock in connection with restricted stock grant	_	_	400,000	40	_	_	(40)	-	_
Reversal of issuance costs related to initial public offering of common stock		_		_	-	_	48,468	_	48,468
Exercise of stock options	_	_	112,500	11	_	_	2.801	_	2,812
Net loss		<u>~</u> ~						(8.641.905)	(8.641.905)
Balance, December 31, 2007		<u>s</u>	<u>82.399.241</u>	<u>\$ 8.240</u>		<u>s </u>	<u>\$29,704,831</u>	<u>\$(12,853,272</u>)	<u>\$ 16,859,799</u>

.

ł.

The accompanying notes are an integral part of these consolidated financial statements.

Consolidated Statements of Cash Flows

	Years		
	2007	2006	2005
Cash flows from operating activities:			
Net income (loss)	\$ (8,641,905)	\$ (501,093)	\$ 1,047,250
Adjustments to reconcile net income (loss) to net cash provided by			
(used in) operating activities:			
Depreciation and amortization	1,028,497	135,446	105,831
Deferred taxes	1,061,720	(307,720)	(754,000)
Share-based compensation	982,190	44,261	
Accretion of warrants	_	252,000	4,200
Changes in operating assets and liabilities, net of the effects of acquisition:			
Trade accounts receivable	17,5 9 0	(673,991)	(479,903)
Prepaid expenses and other assets	(133,377)	(106,798)	(47,451)
Accounts payable	80,491	690,908	(42,041)
Accrued commissions	(308,514)	156,540	496,035
Accrued compensation	1,212,088	193,330	68,507
Accrued expenses	33,762	210,578	38,009
Deferred revenue and customer advances	68,525	<u> </u>	(265.216)
Net cash provided by (used in) operating activities	<u>(4,598,933)</u>	259,129	<u> </u>
Cash flows from investing activities:			
Costs incurred in software development	(469,171)	(65,652)	(54,375)
Net cash paid in acquisition of EnergyGateway, net of cash			
acquired	(4,904,358)		
Purchases of property and equipment	(456,095)	(72,061)	(25,374)
Net cash used in investing activities	(5,829,624)	<u>(137,713</u>)	<u>(79,749</u>)
Cash flows from financing activities:			
Proceeds from exercise of stock options	2,812	56,917	6,000
Proceeds from exercise of stock warrants	—	253,205	
Proceeds from initial public offering of common stock, net of			
offering costs of \$3.6 million	—	17,541,397	_
Retirement of long-term debt	—	(2,000,000)	
Principal payments on line of credit			(500,000)
Proceeds from the issuance of long-term debt			2,000,000
Principal payments on capital lease obligations	(56,257)	(50,092)	(48,733)
Principal payments on guaranteed return	—	(10,523)	(14,062)
Purchase of treasury stock		(12.500)	
Net cash provided by (used in) financing activities	(53,445)	<u>15,778,404</u>	1,443,205
Net (decrease) increase in cash and cash equivalents	(10,482,002)	15,899,820	1,534,677
Cash and cash equivalents, beginning of year	17.483.886	1.584.066	49,389
Cash and cash equivalents, end of year	<u>\$ 7.001.884</u>	<u>\$ 17,483,886</u>	<u>\$ 1,584.066</u>
Supplemental Disclosure of Cash Flow Information:			
Net cash received (paid) for interest	<u>\$ 612,303</u>	<u>\$ (129,401)</u>	<u>\$ (94,842)</u>
Non-cash investing and financing activities:			
Property and equipment acquired through capital lease obligations	<u>\$</u>	<u>\$ 12,943</u>	<u>\$ 56,205</u>
Non-cash investing and financing activities:	<u> </u>		
Fair value of common stock issued in acquisition of EnergyGateway	<u>\$ 6,536,000</u>	<u>\$</u>	<u>s</u>
Reversal of accrued expenses related to initial public offering			
of common stock	<u>\$ 48,468</u>	<u>\$</u>	<u>s </u>
Conversion of series A preferred stock to common stock	<u>\$</u>	<u>\$ 1.507,997</u>	<u>s </u>
Conversion of non-voting common stock to common stock	<u>\$</u>	<u>\$898</u>	<u>\$</u>
Issuance of warrants in connection with initial public offering of common		× ×=· · · •	*
stock	<u>\$</u>	<u>\$ 274,129</u>	<u>\$</u>
Issuance of warrants in connection with debt offering	<u>s </u>	<u>s </u>	<u>\$256,200</u>
The accompanying notes are an integral part of these consolidated financial sta	atements.		

Exhibit C-4 "Financial Arrangements"

World Energy Solutions, Inc. is an online auction provider and energy broker. World Energy Solutions does not take title to the electricity or natural gas. We conduct an auction that matches the end user to the energy supplier. As a result, World Energy Solutions does not have bank commitments, lines-of-credit, guarantees, or any other financial arrangement to perform its service obligations in Ohio.

Exhibit C-5 "Forecasted Financial Statements"

As a public company World Energy Solutions, Inc. can not disclose two years of forecasted financial statements.

Exhibit C-6 "Credit Rating"

World Energy Solutions, Inc. is not rated by any of the large rating agencies other than Dun & Bradstreet. The Dun and Bradstreet report on World Energy Solutions, Inc. is attached in Exhibit C-7.

World Energy Solutions' D&B DUNS Number is: 01-114-1343

Exhibit C-7 "Credit Report"

.

See the enclosed Dun & Bradstreet report for World Energy Solutions, Inc.



ATTN:ANDREW THOMAS

Report Printed:December 08, 2008

Live Report : WORLD ENERGY SOLUTIONS, INC

D-U-N-S® Number: 01-114-1343

Trade Names: WORLD ENERGY SOLUTIONS - ECHOICENET Endorsement/Billing Reference: athomas@worldenergy.com

D&B Address

Address 446 Main St Ste 14 Worcester,MA - 01608 Phone 508 459-8100 Fax Location Type Headquarters Web www.worldenergy.com Added to Portfolio:03/20/2007

Company Summary

Score Bar

PAYDEX®	0 72
Commercial Credit Score Class	3
Financial Stress Class	1
Credit Limit - D&B Conservative	\$25,000.00
D&B Rating	1 R2

D&B 3-month PAYDEX®

3-month D&B PAYDEX®: 80 0

(Lowest Risk:100; Highest Risk:1) When weighted by dottar amount, Payments to suppliers average Within terms

D&B Company Overview

This is a headquarters location

Branch(es) or Division(s) exist Y

Chief Executive	RICHARD DOMALESKI, CEO
Year Started	1995
Employees	24 (UNDETERMINED Here)
SIC	7389
Line of business	Business services
NAICS	561499
History Status	CLEAR

D&B 12-month PAYDEX®

......

12-month D&B PAYDEX®: 72

(Lowest Risk:100; Highest Risk:1) When weighted by dollar amount, Payments to suppliers average 12 days beyond terms

Public Filings

The following data includes both open and closed filings found in D&B's database on this company.

Record Type	Number of Records	Most Recent Fillng Date
Bankruptcies	0	
Judgments	0	•
Liens	0	
Suits	0	
UCC's	4	09/08/08

The public record items contained herein may have been paid, terminated, vacated or released prior to todays date.

Financial Stress Score Class

Financial Stress Score Class: 1

(Lowest Risk:1; Highest Risk:5) Low risk of severe financial stress over the next 12 months.

Commercial Credit Score Class

Commercial Credit Score Class: 3

346

Commercial Credit Score Class:

(Lowest Risk:1; Highest Risk:5) Medium risk of severe payment delinquency over next 12 months.

Corporate Linkage	· · · · · · · · · · · · · · · · · · ·	andara mangana mangana sa sa na sa	and and a start of the second s
Branches (Domestic)			
Company	، میکند. <u>بوسم</u> ر بر میکند. ب	City, State	D-U-N-S@ NUMBER
WORLD ENERGY SOLUTIC	DNS, INC	DUBLIN, Ohio	80-285-8675
redictive Scores			
Credit Capacity Sumn	nary		
This credit rating was assign " D&B Rating Key".	ed because of D&B's assessment of the o	ompany's financial ratios and its (cash flow. For more information, see the
D&B Rating : 1R2		Number of employees: 1R Composite credit appraisa	t indicates 10 or more employees
do not contain a current finar	ncial statement. In 1R and 2R Ratings, the je and other important factors. 2 is the hig	total number of employees for the a 2, 3, or 4 creditworthiness indic	 a business. They are assigned to business files that ator is based on analysis by D&B of public filings, a company not supplying D&B with current
Below is an overview of the 10-1999	e company's rating history since 08-	Number of Employees Total:	24 (UNDETERMINED here)
D&B Rating 1R2 1R3 1R2 1R3 1R2 -	Date Applied 11-13-2007 12-05-2006 11-15-2006 04-19-2005 08-09-2000 08-10-1999		
Payment Activity: Average High Credit: Highest Credit: Total Highest Credit:	(based on 25 experiences) \$2,952 \$25,000 \$59,350		
D&B Credit Limit Rec	ommendation		
Conservative credit Limit Aggressive credit Limit:	\$25. \$45,		
Risk category for this busi	ness : LOV	N	
This recommended Credit Lir Risk is assessed using D&Bs	nit is based on the company profile and o scoring methodology and is one factor us	n profiles of other companies wit ed to create the recommended i	h similarities in size, industry, and credit usage. mits. See Help for details.
Financial Stress Class	Summary		
The Financial Stress Class Su obtaining relief from crediton	immary Model predicts the likelihood of a sunder state/federal law over the next 12	firm ceasing business without pa 2 months. Scores were calculated	ying all creditors in full, or reorganization or using a statistically valid model derived from

D&Bs extensive data files. The Financial Stress Class of 1 for this company shows that firms with this classification had a failure rate of 1.2% (120 per 10,000), which is lower than the average of businesses in D & B's database

Financial Stress Class : 1

(Lowest Risk:1; Highest Risk:5) Low risk of severe financial stress, such as bankruptcy, over the next 12 months.

Incidence of Financial Stress:

- Average of Businesses In D&Bs database: 2.60 % (260 per 10000)
- Financial Stress National Percentile : 57 (Highest Risk: 1; Lowest Risk: 100)
- ٠ Financial Stress Score : 1388 (Highest Risk: 1001; Lowest Risk: 1875)

The Financial Stress Class of this business is based on the following factors:

- No record of open suit(s), lien(s), or judgment(s) in the D & B files.
- . 28% of trade dollars indicate slow payment(s) are present.
- Payment experiences exist for this firm which are greater than 60 days past due.
- Control age or date entered in D & B files indicates higher risk.

Notes:

- The Financial Stress Class indicates that this firm shares some of the same business and financial characteristics of other companies with this classification. It does not mean the firm will necessarily experience financial stress.
- The Incidence of Financial Stress shows the percentage of firms in a given Class that discontinued operations over the past year with loss to creditors. The Incidence of Financial Stress - National Average represents the national failure rate and is provided for comparative purposes.
- The Financial Stress National Percentile reflects the relative ranking of a company among all scorable companies in D&Bs file. The Financial Stress Score offers a more precise measure of the level of risk than the Class and Percentile. It is especially helpful to customers using
- a scorecard approach to determining overall business performance.
- All Financial Stress Class, Percentile, Score and Incidence statistics are based on sample data from 2004

Norms	National %
This Business	57
Region: NORTHEAST	50
Industry: BUSINESS, LEGAL AND ENGINEERING SERVICES	52
Employee range: 20-99	80
Years in Business: 11-25	52

This Business has a Financial Stress Percentile that shows:

Lower risk than other companies in the same region.

Lower risk than other companies in the same industry.

Higher risk than other companies in the same employee size range.

Lower risk than other companies with a comparable number of years in business.

Credit Score Class Summary

The Credit Score class predicts the likelihood of a firm paying in a severely delinquent manner (90+ Days Past Terms) over the next twelve months. It was calculated using statistically valid models and the most recent payment information in D&Bs files. The Credit Score class of 3 for this company shows that 14.3% of firms with this classification paid one or more bills severely delinquent, which is lower

than the average of businesses in D & B's database. Credit Score Class : 3

(Lowest Risk:1; Highest Risk:5) Moderate risk of severe payment delinguency over next 12 months.

Incidence of Delinquent Payment

- Among Companies with this Classification: 14.30 %
- Average compared to businesses in D&B's database: 20.10 %
- Credit Score Percentile : 58 (Highest Risk: 1; Lowest Risk: 100) Credit Score : 474 (Highest Risk: 101; Lowest Risk: 670)

The Credit Score Class of this business is based on the following factors:

- 28% of trade doltars indicate slow payment(s) are present.
- No record of open suit(s), lien(s), or judgment(s) in the D & B files.
- Payment experiences exist for this firm which are greater than 60 days past due.

Notes:

- The Credit Score Class indicates that this firm shares some of the same business and payment characteristics of other companies with this classification. It does not mean the firm will necessarily experience delinquency.
- The Incidence of Delinguent Payment is the percentage of companies with this classification that were reported 90 days past due or more by creditors. The calculation of this value is based on an inquiry weighted sample.
- The Percentile ranks this firm relative to other businesses. For example, a firm in the 80th percentile has a lower risk of paying in a severely delinquent manner than 79% of all scorable companies in D&Bs files.
- The Credit Score offers a more precise measure of the level of risk than the Class and Percentile. It is especially helpful to customers using a scorecard approach to determining overall business performance.
- All Credit Class, Percentile, Score and Incidence statistics are based on sample data from 2004

Norms	National %
This Business	58
Region: NORTHEAST	50
Industry: BUSINESS, LEGAL AND ENGINEERING SERVICES	49

Employee range:20-9969Years in Business:11-2564

This business has a Credit Score Percentile that shows:

Lower risk than other companies in the same region.

Lower risk than other companies in the same industry.

Higher risk than other companies in the same employee size range.

Higher risk than other companies with a comparable number of years in business.

Trade Payments

D&B PAYDEX®

The D&B PAYDEX is a unique, dollar weighted indicator of payment performance based on payment experiences as reported to D&B by trader references. Learn more about the D&B PAYDEX Score

Timeliness of historical payments for this company.

Current PAYDEX is 72	Equal to 12 days beyond terms (Pays more slowly than the average for its industry of 3 days beyond terms)
Industry Median is 78	Equal to 3 days beyond terms
Payment Trend currently is 种	Unchanged, compared to payments three months ago

Indications of slowness can be the result of dispute over merchandise, skipped invoices etc. Accounts are sometimes placed for collection even though the existence or amount of the debt is disputed.

Total payment Experiences in D&Bs File	25
Payments Within Terms (not dollar weighted)	85 %
Trade Experiences with Slow or Negative Payments(%)	12.00%
Total Placed For Collection	0
Average High Credit	\$2,952
Largest High Credit	\$25,000
Highest Now Owing	\$10,000
Highest Past Due	\$500

12-Month D& PAYDEX® : 720

(Lowest Risk:100; Highest Risk:1) Based on payments collected over last 12 months.

When weighted by dollar amount, payments to suppliers average 12 days beyond terms

3-Month D& PAYDEX® : 800

(Lowest Risk:100; Highest Risk:1) Based on payments collected over last 3 months.

When weighted by dollar amount, payments to suppliers average within terms

D&B PAYDEX® Comparison

Current Year

PAYDEX® of this Business compared to the Primary Industry from each of the last four quarters. The Primary Industry is Business services , based on SIC code 7389.

Shows the trend in D&B PAYDEX scoring over the past 12 months.

	1/082	/083	/084	/085	/086	/087	/088	/089	/0810)/0811	L/0813	2/08
This Business	69	70	70	70	71	70	71	71	71	71	72	72
Industry Quartiles												
Upper			80	-		80			80			
Median			77			77			78	•		
Lower			69	•		68			68			

• Current PAYDEX for this Business is 72, or equal to 12 days beyond terms

The 12-month high is 72, or equal to 12 DAYS BEYOND terms

The 12-month low is 69, or equal to 16 DAYS BEYOND terms

Previous Year

Shows PAYDEX of this Business compared to the Primary Industry from each of the last four quarters. The Primary Industry is Business services , based on SIC code 7389.

Previous Year	12/06 03/07 06/07 09/07 Q4'06Q1'07 Q2'07 Q3'07					
This Business Industry Quartiles	UN	60	60	66		
Upper	60	80	80	80		
Median	77	77	77	77		
Lower	69	69	69	68		

- Based on payments collected over the last 4 quarters. Current PAYDEX for this Business is 72, or equal to 12 days beyond terms
 - ٠ The present industry median Score is 78, or equal to 3 days beyond terms .
 - Industry upper quartile represents the performance of the payers in the 75th percentile
 - Industry lower quartile represents the performance of the payers in the 25th percentile

Payment Habits

For all payment experiences within a given amount of credit extended, shows the percent that this Business paid within terms. Provides number of experiences to calculate the percentage, and the total credit value of the credit extended.

\$ Credit Extended	# Payment Experiences	\$ Total Dollar Amount	% of Payments Within Terms
Over 100,000	0	\$0	0%
50,000-100,000	0	SO	0%
15,000-49,999	2	\$40,000	62%
5,000-14,999	1	\$10,000	100%
1,000-4,999	4	\$5,500	82%
Under 1,000	13	\$3,550	92%

Based on payments collected over last 12 months.

For all Payment experiences reflect how bills are met in relation to the terms granted. In some instances, payment beyond terms can be the result of disputes over merchandise, skipped involces etc.

Payment Summary

There are 25 payment experience(s) in D&Bs file for the most recent 12 months, with 14 experience(s) reported during the last three month period. The highest Now Owes on file is \$10,000. The highest Past Due on file is \$500 Below is an overview of the company's dollar-weighted payments, segmented by its suppliers' primary industries:

.....

	Total Revd (#)	Total Dollar Amts (\$)	Largest High Credit (\$)	Within Terms (%)	<3'	Days Slow 1 31-60 61-90 90> (%)
Top Industries						
Nonclassified	8	14,250	10,000	98	0	020
Telephone communictns	2	200	100	75	25	000
Ret-direct selling	1	25,000	25,000	100	0	000
Misc equipment rental	1	15,000	15,000	0	10	0000
Executive office	1	1,000	1,000	100	0	000
Radiotelephone commun	1	1,000	1,000	0	0	0 0 100
Misc general gov't	1	750	750	100	Û	000
Short-trm bush credit	1	750	750	100	0	000
Data processing svcs	1	500	500	100	Û	000
Photocopying service	1	250	250	100	0	000
Ret mail-order house	1	100	100	100	Ó	000
Misc business credit	1	250		100	0	000
Misc business service	1	0	0	0	0	000
Other payment categories						
Cash experiences	4	300	100			
Payment record unknown	0	0	0			
Unfavorable comments	0	0	0			
Placed for collections:	-	-	-			
With D&B	0	Û	0			
Other	ŏ	N/Ā	-			
Total in D&Bs file	25	59,350	25,000			

Accounts are sometimes placed for collection even though the existence or amount of the debt is disputed.

Indications of slowness can be result of dispute over merchandise, skipper invoices etc.

Payment Details

Now Owes (\$)

Past Due (\$)

11/08	Ppt		250	0	1 mo
	Ppt	25,000	250	0N30	1 mo
	Ppt	2,500	0	0	2-3 mos
	Ppl	750	750	0	1 mo
	Ppt-Slow 30	100	100	50	1 mo
	Ppt-Slow 90	500	0	0	6-12 mos
10/08	Ppt	10,000	10,000	0	1 mo
	Ppt	1,000	100	0	1 mo
	Ppt	100	100	N30	1 mo
	Ppt	100	50	0	1 mo
	(011) Cash own option	100		Cash account	1 mo
	(012) Cash own option .	100		Cash account	1 mo
09/0B	Ppt	250	250		1 mo
	(014) Cash own option .	50		Cash account	2-3 mos
08/08	Ppt	500	100	0	1 mo
	(016)	50		Cash account	1 mo
07/08	Ppt	100	0	· 0	
	Ppt	50	0	0	
	Ppt	50	0	0	4-5 mos
03/08	(020) Satisfactory .	750			1 mo
01/08	Slow 30	15,000	500	500	
11/07	Ppt	1,000			1 mo
	Slow 90+	1,000	0	0	6-12 mos
07/07	Ppt	50	0	0	6-12 mos
	Ppt	0	0		1 mo

Payments Detail Key: 30 or more days beyond terms

Payment experiences reflect how bills are met in relation to the terms granted. In some instances payment beyond terms can be the result of disputes over merchandise, skipped invoices etc. Each experience shown is from a separate supplier. Updated trade experiences replace those previously reported.

.....

Public Filings

Summary

The following data includes	both open and clo	sed filings found in D&B's database on this company.
Record Type	# of Records	Most Recent Filing Date
Bankruptcy Proceedings	0	
Judgments	0	-
Liens	0	-
Sults	0	•
UCCs	4	09/08/08

The following Public Filing data is for information purposes only and is not the official record. Certified copies can only be obtained from the official source.

UCC Filings

Collateral Type Sec. Party Debtor Filing No. Filed With	All Assels Original SILICON VALLEY BANK, SANTA CLARA, CA WORLD ENERGY SECURITIES CORP. 200868174350 SECRETARY OF STATE/UCC DIVISION, BOSTON, MA	
Date Filed Latest Info Received	2008-09-08 10/17/08	
Collateral Type Sec. Party Debtor Filing No. Filed With	All Assets Original SILICON VALLEY BANK, SANTA CLARA, CA WORLD ENERGY SOLUTIONS, INC. 2008 3018593 SECRETARY OF STATEAUCC DIVISION, DOVER, DE	
Date Filed Latest Info Received	2008-09-08 10/09/08	
Collaterai Type Sec. Party Debtor	Leased Fixtures and proceeds Original GENERAL ELECTRIC CAPITAL CORPORATION, DANBURY, CT WORLD ENERGY SOLUTIONS INC.	

Filed With	4324054 8 SECRETARY OF STATE/UCC DIVISION, DOVER, DE
Date Filed Latest Info Received	2004-11-12 12/22/04
Collateral	Leased Equipment
Туре	
Sec. Party Debtor	GENERAL ELECTRIC CAPITAL CORPORATION, BANNOCKBURN, IL WORLD ENERGY SOLUTIONS, INC.
Filling No. Filled With	5060796 1 SECRETARY OF STATE/JCC DIVISION, DOVER, DE
Date Filed Latest Info Received	200 5- 02-24 03/17/05
The public record items contained h	erein may have been paid, terminated, vacated or released prior to today's date.
Government Activity	
Activity summary	
Borrower (Dir/Guar)	NO
Administrative Debt	NO
Contractor	YES
Grantee Party excluded from federal progran	n(s) NO
rany excluded from lederal program	
Possible candidate for socio-aco	
Labour Surplus Area Small Business	N/A YES (2008)
8(A) firm	N/A
	nent Activity section are as reported to Dun & Bradstreet by the federal government and other sources.
pecial Events Special Events	
pecial Events Special Events	
pecial Events Special Events 11/08/2008 -EARNINGS UPDATE According to published reports. corr	: parative operating results for the 9 months ended September 30, 2008: Revenue of \$9,164,727, Net Income
pecial Events Special Events 11/08/2008 -EARNINGS UPDATE According to published reports, com (\$5,756,079); compared to Revenue co/coccepts.	: nparative operating results for the 9 months ended September 30, 2008: Revenue of \$9,164,727, Net Income of s of \$6,103,672, Net Income of (\$3,427,390) for the comparable period in the prior year.
Special Events Special Events 11/08/2008 -EARNINGS UPDATE According to published reports, com (\$5,756,079); compared to Revenue 09/09/2008 -LINE OF CREDIT : According to published reports, Woi Under the Agreement, SVB has con certain eligible account receivables	the sof \$6,103,672, Net Income of (\$3,427,390) for the comparable period in the prior year. The Energy Solutions, Inc. announced that it has entered into a Loan and Security Agreement with Silicon Valle mitted to make advances to the Company in an aggregate amount of up to \$3,000,000 subject to availability and eligible retail backlog.
pecial Events Special Events 11/08/2008 -EARNINGS UPDATE According to published reports, com (\$5,756,079); compared to Revenue 09/09/2008 -LINE OF CREDIT : According to published reports, Woi Under the Agreement, SVB has con certain eligible account receivables 08/09/2008 -EARNINGS UPDATE	the parative operating results for the 9 months ended September 30, 2008: Revenue of \$9,164,727, Net Income of a of \$6,103,672, Net Income of (\$3,427,390) for the comparable period in the prior year. Ind Energy Solutions, Inc. announced that it has entered into a Loan and Security Agreement with Silicon Valle mmitted to make advances to the Company in an aggregate amount of up to \$3,000,000 subject to availability and eligible retail backlog.
Special Events Special Events 11/09/2009 -EARNINGS UPDATE According to published reports, com (\$5,756,079); compared to Revenue 09/09/2008 -LINE OF CREDIT : According to published reports, Wol Under the Agreement, SVB has con certain eligible account receivables 08/09/2008 -EARNINGS UPDATE According to published reports, com (\$4,540,259); compared to Revenue	The parative operating results for the 9 months ended September 30, 2008: Revenue of \$9,164,727, Net Income of a of \$6,103,672, Net Income of (\$3,427,390) for the comparable period in the prior year. The Energy Solutions, Inc. announced that it has entered into a Loan and Security Agreement with Silicon Valle mmitted to make advances to the Company in an aggregate amount of up to \$3,000,000 subject to availability and eligible retail backlog.
Special Events Special Events 11/09/2008 -EARNINGS UPDATE According to published reports, com (\$5,756,079); compared to Revenue 09/09/2008 -LINE OF CREDIT : According to published reports, Wol Under the Agreement, SVB has con certain eligible account receivables 08/09/2008 -EARNINGS UPDATE According to published reports, com (\$4,540,259); compared to Revenue 05/10/2008 -EARNINGS UPDATE	 i: inparative operating results for the 9 months ended September 30, 2008: Revenue of \$9,164,727, Net Income of \$6,103,672, Net Income of (\$3,427,390) for the comparable period in the prior year. rid Energy Solutions, Inc. announced that it has entered into a Loan and Security Agreement with Silicon Valle numlited to make advances to the Company in an aggregate amount of up to \$3,000,000 subject to availability and eligible retail backlog. : inparative operating results for the 6 months ended June 30, 2008: Revenue of \$5,875,212, Net Income of e of \$3,426,329, Net Income of (\$1,617,006) for the comparable period in the prior year.
pecial Events Special Events 11/09/2008 -EARNINGS UPDATE According to published reports, com (\$5,756,079); compared to Revenue 09/09/2008 -LINE OF CREDIT : According to published reports, Woi Under the Agreement, SVB has com certain eligible account receivables 08/09/2008 -EARNINGS UPDATE According to published reports, com (\$4,540,259); compared to Revenue 05/10/2008 -EARNINGS UPDATE According to published reports, com	i:
pecial Events Special Events 11/09/2009 -EARNINGS UPDATE According to published reports, com (\$5,756,079); compared to Revenue 09/09/2008 -LINE OF CREDIT : According to published reports, Wol Under the Agreement, SVB has con- certain eligible account receivables 08/09/2008 -EARNINGS UPDATE According to published reports, com (\$4,540,259); compared to Revenue 05/10/2008 -EARNINGS UPDATE According to published reports, com (\$2,159,942); compared to Revenue 03/13/2008 -EARNINGS UPDATE	i:
Special Events Special Events 11/09/2009 -EARNINGS UPDATE According to published reports, com (\$5,756,079); compared to Revenue 09/09/2008 -LINE OF CREDIT : According to published reports, Woi Under the Agreement, SVB has con certain eligible account receivables 08/09/2008 -EARNINGS UPDATE According to published reports, com (\$4,540,259); compared to Revenue 05/10/2009 -EARNINGS UPDATE According to published reports, com (\$2,159,942); compared to Revenue 03/13/2008 -EARNINGS UPDATE According to published reports, com (\$3,641,905); compared to Revenue (\$3,641,905); compared to Revenue	i:: nparative operating results for the 9 months ended September 30, 2008: Revenue of \$9,164,727, Net Income of of \$6,103,672, Net Income of (\$3,427,390) for the comparable period in the prior year. rid Energy Solutions, Inc. announced that it has entered into a Loan and Security Agreement with Silicon Valle minited to make advances to the Company in an aggregate amount of up to \$3,000,000 subject to availability and eligible retail backlog. parative operating results for the 6 months ended June 30, 2008: Revenue of \$5,875,212, Net income of e of \$3,426,329, Net income of (\$1,617,006) for the comparable period in the prior year. parative operating results for the 3 months ended March 31, 2008: Revenue of \$3,114,517, Net income of e of \$1,501,137, Net Income of (\$439,436) for the comparable period in the prior year. parative operating results for the year ended December 31, 2007: Revenue of \$9,188,265, Net Income of e of \$5,763,098, Net Income of (\$501,093) for the comparable period in the prior year.
pecial Events Special Events 11/09/2009 -EARNINGS UPDATE According to published reports, com (\$5,756,079); compared to Revenue 09/09/2008 -LINE OF CREDIT : According to published reports, Wol Under the Agreement, SVB has con- certain eligible account receivables 08/09/2008 -EARNINGS UPDATE According to published reports, com (\$4,540,259); compared to Revenue 05/10/2008 -EARNINGS UPDATE According to published reports, com (\$2,159,942); compared to Revenue 03/13/2008 -EARNINGS UPDATE According to published reports, com (\$2,159,942); compared to Revenue 11/13/2007 -EARNINGS UPDATE	It is a set of \$1,000 in the set of \$2,164,727. Net Income of \$2,164,727. Net Income of \$2,164,727. Net Income of \$3,103,672, Net Income of \$3,427,390) for the comparable period in the prior year. It defines the the company in an aggregate amount of up to \$3,000,000 subject to availability and eligible retail backlog. It is a parative operating results for the 6 months ended June 30, 2008: Revenue of \$5,875,212, Net Income of e of \$3,428,329, Net Income of \$1,617,006) for the comparable period in the prior year. It is a parative operating results for the 3 months ended June 30, 2008: Revenue of \$5,875,212, Net Income of e of \$3,428,329, Net Income of \$1,617,006) for the comparable period in the prior year. It is a parative operating results for the 3 months ended March 31, 2008: Revenue of \$3,114,517, Net Income of e of \$1,501,137, Net Income of \$439,436) for the comparable period in the prior year. It is a parative operating results for the 3 months ended March 31, 2008: Revenue of \$3,114,517, Net Income of e of \$1,501,137, Net Income of \$3,438) for the comparable period in the prior year. It is a parative operating results for the 3 months ended March 31, 2008: Revenue of \$3,114,517, Net Income of e of \$1,501,137, Net Income of (\$439,436) for the comparable period in the prior year. It is a parative operating results for the 3 months ended December 31, 2007: Revenue of \$9,188,265, Net Income of e of \$5,763,098, Net Income of (\$501,093) for the comparable period in the prior year. It is a parative operating results for the year ended December 31, 2007: Revenue of \$9,188,265, Net Income of e of \$5,763,098, Net Income of \$1,000,003) for the comparable period in the prior year. It is a parative operating results for the year ended December 31, 2007: Revenue of \$9,188,265, Net Income of e of \$5,763,098, Net Income of \$1,000,003) for the comparable period in the prior year. It is a parative operating results for the year ended December 31, 2007: Revenue of \$9,188,265, Net Income of e of \$5,763,098, Net Income of \$
pecial Events Special Events 11/09/2009 -EARNINGS UPDATE According to published reports, com (\$5,756,079); compared to Revenue 09/09/2008 -LINE OF CREDIT : According to published reports, Wol Under the Agreement, SVB has con certain eligible account receivables 08/09/2008 -EARNINGS UPDATE According to published reports, com (\$4,540,259); compared to Revenue 05/10/2008 -EARNINGS UPDATE According to published reports, com (\$2,159,942); compared to Revenue 03/13/2008 -EARNINGS UPDATE According to published reports, com (\$8,641,905); compared to Revenue 11/13/2007 -EARNINGS UPDATE	It is imparative operating results for the 9 months ended September 30, 2008: Revenue of \$9,164,727, Net Income of so of \$6,103,672, Net Income of (\$3,427,390) for the comparable period in the prior year. Ind Energy Solutions, Inc. announced that it has entered into a Loan and Security Agreement with Silicon Valle minited to make advances to the Company in an aggregate amount of up to \$3,000,000 subject to availability and eligible retail backlog. Imparative operating results for the 6 months ended June 30, 2008: Revenue of \$5,875,212, Net Income of e of \$3,426,329, Net Income of (\$1,617,006) for the comparable period in the prior year. Imparative operating results for the 3 months ended March 31, 2008: Revenue of \$3,114,517, Net Income of e of \$1,501,137, Net Income of (\$4,39,436) for the comparable period in the prior year. Imparative operating results for the year ended December 31, 2007: Revenue of \$9,168,265, Net Income of e of \$5,763,098, Net Income of (\$501,093) for the comparable period in the prior year. Imparative operating results for the 9 months ended September 30, 2007: Revenue of \$9,168,265, Net Income of e of \$5,763,098, Net Income of (\$501,093) for the comparable period in the prior year. Imparative operating results for the 9 months ended September 30, 2007: Revenue of \$9,168,265, Net Income of e of \$5,763,098, Net Income of (\$501,093) for the comparable period in the prior year. Imparative operating results for the 9 months ended September 30, 2007: Revenue of \$9,168,265, Net Income of e of \$5,763,098, Net Income of (\$207,451) for the comparable period in the prior year. Imparative operating results for the 9 months ended September 30, 2007: Revenue of \$6,103,672, Net Income of e of \$4,166,763, Net Income of (\$207,451) for the comparable period in the prior year. Imparative operating results for the 9 months ended September 30, 2007: Revenue of \$6,103,672, Net Income of e of \$4,166,763, Net Income of (\$207,451) for the comparable period in the prior year. Imparative operating resul
pecial Events Special Events 11/09/2009 -EARNINGS UPDATE According to published reports, com (\$5,756,079); compared to Revenue 09/09/2008 -LINE OF CREDIT : According to published reports, Wol Under the Agreement, SVB has con- certain eligible account receivables 08/09/2008 -EARNINGS UPDATE According to published reports, com (\$4,540,259); compared to Revenue 05/10/2008 -EARNINGS UPDATE According to published reports, com (\$2,159,942); compared to Revenue 03/13/2008 -EARNINGS UPDATE According to published reports, com (\$8,641,905); compared to Revenue 11/13/2007 -EARNINGS UPDATE According to published reports, com (\$3,427,390); compared to Revenue 10/11/2007 -OFFICER CHANGE :	i: parative operating results for the 9 months ended September 30, 2008: Revenue of \$9,164,727. Net Income of a of \$6,103,672, Net Income of (\$3,427,390) for the comparable period in the prior year. rid Energy Solutions, Inc. announced that it has entered into a Loan and Security Agreement with Silicon Valle minited to make advances to the Company in an aggregate amount of up to \$3,000,000 subject to availability and eligible retail backlog. : parative operating results for the 6 months ended June 30, 2008: Revenue of \$5,875,212, Net Income of e of \$3,426,329, Net Income of (\$1,617,006) for the comparable period in the prior year. : parative operating results for the 3 months ended March 31, 2008: Revenue of \$5,875,212, Net Income of e of \$1,501,137, Net Income of (\$439,436) for the comparable period in the prior year. : parative operating results for the year ended December 31, 2007: Revenue of \$9,189,285, Net Income of e of \$5,763,098, Net Income of (\$501,093) for the comparable period in the prior year. : parative operating results for the 9 months ended September 30, 2007: Revenue of \$9,189,285, Net Income of e of \$5,763,098, Net Income of (\$501,093) for the comparable period in the prior year. : parative operating results for the 9 months ended September 30, 2007: Revenue of \$9,108,285, Net Income of e of \$5,763,098, Net Income of (\$201,093) for the comparable period in the prior year. :

7

35Z

According to published reports, comparative operating results for the 6 months ended June 30, 2007: Revenue of \$3,428,329, Nat Income of (\$1,617,006); compared to Revenue of \$2,479,770, Net Income of (\$219,812) for the comparable period in the prior year.

07/18/2007 -MERGER/ACQUISITION :

According to published reports on May 25, 2007, World Energy Solutions, Inc. (Dublin, OH) announced that it has completed the previously announced acquisition of privately held EnergyGateway, LLC (Worcester, MA). Under the terms of the final agreement, World Energy purchased substantially all of the assets of EnergyGateway for US\$4.95 million in cash and 5.375 million common shares of World Energy plus the assumption of certain liabilities.

-OFFICER CHANGE :

According to published reports, World Energy Solutions, Inc. announced that Tyler Wooddell is now a Senior Vice President at World Energy Solutions.

-ANNOUNCED MERGER/ACQUISITION :

According to published reports on May 22, 2007, World Energy Solutions, Inc. (Worcester, MA), announced that it has entered into a definitive agreement to acquire privately held EnergyGateway, LLC (Dublin, OH). The acquisition is expected to provide customers of World Energy Exchange a bigger marketplace, an expanded online toolset, and a deeper pool of market expertise.

Under the terms of the agreement, World Energy will acquire substantially all of the assets of EnergyGateway for US\$5 million in cash and 5.375 million common shares of World Energy plus the assumption of certain liabilities. The transaction is expected to close on or about May 31, 2007, subject to customary closing conditions.

-ANNOUNCED MERGER/ACQUISITION :

According to published reports on April 17, 2007, World Energy Solutions, Inc. (Worcester, MA) announced that it has entered into a non-binding letter of intent with respect to a potential acquisition candidate. The transaction is subject to a number of conditions including the parties negotiating and entering into a definitive agreement. There can be no assurance that a definitive agreement will be entered into or that the transaction will be consummated.

-ANNOUNCED PURCHASE OF ASSET :

According to published reports on March 26, 2007, World Energy Solutions, Inc. (Worcester, MA) announced that it has entered a letter of intent to purchase up to 20% of the issued and outstanding memberships interests of Sol-Gel Solutions, LLC (Gainesville, FL). The company will be given an option to purchase 50% of the issued and outstanding memberships interests of Sol-Gel.

History and Operations

Company Overview

Company Name: Doing Business As :

Street Address:

Phone: URL: History Present management control WORLD ENERGY SOLUTIONS, INC WORLD ENERGY SOLUTIONS, ECHOICENET 446 Main SI Ste 14 Worcester, 01608 508 459-8100 http://www.worldenergy.com Is clear 13 years

History

Officer(s):

The following information was reported: 10/11/2007

RICHARD DOMALESKI, CEO EDWARD LIBBEY, PHD, CHM PHILIP ADAMS, PRES-COO JAMES F. PARSLOW, CFO-TREASURER TYLER WOODDELL, SVP

DIRECTOR(S) :

THE OFFICER(S)

On July 18, 2007 source(s) contacted confirmed the existence of this corporation. Dun & Bradstreet makes a regular search for corporate details and will provide the information as it becomes available.

Business started 1995 by as a partnership. 87% of capital stock is owned by officers, 13% of capital stock is owned by outside investors.

STOCK OWNERSHIP :

On November 16, 2006, the Company completed its initial public offering of common stock, including the exercise by the underwriters of an over-allotment option, for the sale of 23,000,000 shares of common stock resulting in net proceeds to the Company of approximately \$17.5 million (net of offering costs of approximately \$3.6 million. Symbol : (TSX: XWE).

As of March 31, 2007, the Company only had one issued and outstanding class of stock common stock. As a result, the basic loss per share for the three months ended March 31, 2007 is computed by dividing net loss for the period by the weighted average number of common shares outstanding during the period.

The executive officers and directors and entities affiliated with them, in the aggregate, beneficially own approximately 35.8% of the company's outstanding common stock. In particular, Richard Domaleski, president and Chief Executive Officer, beneficially owns approximately 25.0% of outstanding common stock.

RECENT EVENT:

On June 4, 2007, World Energy Solutions, Inc, Worcestor, MA announced that it has completed the previously announced acquisition of EnergyGateway, LLC, Dublin, OH. The operations of EnergyGateway will be merged into World Energy. World Energy will continue to operate an office at the Dublin, OH facility. The management and employees were retained through this acquisition. **OFFICERS BACKGROUND :**

RICHARD DOMALESKI born 1969. 1995-present active here.

E T LIBBEY PHD born 1950. 1995-present active here.

PHILIP ADAMS born 1958, 2001-present active here. 1999-2001 worked as Chief marketing officer in Exchange Application at Boston MA. 1997-1999 worked in Pega Systems as Sr. VP Marketing and Product Management, 1994-1997 Worked as Chief Marketing Office in Corporate Software. JAMES F. PARSLOW. Parslow was CFO with Spire Corp., Bedford, MA. He also spent eight years in public accounting at Arthur Andersen LLP prior to joining Thermo Electron. Parslow holds an MBA from Bentley College and has significant management experience in financial planning and control, financing, acquisitions, divestitures and Securities and Exchange Commission reporting. TYLER WOODDELL. Formerly President and CEO of EnergyGateway.

Business Registration *****

CORPORATE AND BUSINESS REGISTRATIONS PROVIDED BY MANAGEMENT OR OTHER SOURCE The Corporate Details provided below may have been submitted by the management of the subject business and may not have been verified with the government agency which records such data.

Registered Name:	Oceanside Energy, Inc.
Business type:	CORPORATION
Corporation type:	PROFIT
Date incorporated:	Sep 03 1996
State of incorporation:	DELAWARE
Filing date:	Sep 03 1996
Where filed:	SECRETARY OF STATE/CORPORATIONS DIVISION, DOVER, DE

Operations

10/11/2007	Operates business services, specializing in auction, appraisal and exchange services (100%).
Description:	Terms vary depending on services. Sells to commercial concerns. Territory : United States & Canada.
Employees: Facilities: Branches:	Nonseasonal. 24 which includes officer(s), UNDETERMINED employed here. Leases 2,500 sq. ft. in building. This business has multiple branches; detailed branch/division information is available in D & B's linkage or family tree.

SIC & NAICS

SIC:

Based on information in our file, D& has assigned this company an extended 8-digit SIC. D&'s use of 8-digit SICs enables us to be more specific about a company's operations than if we use the standard 4-digit code. The 4-digit SIC numbers link to the description on the Occupational Safety & Health Administration (OSHA) Web site. Links open in a new browser window. 7389 1400 Auction, appraisal, and exchange services NAICS:

561499 All Other Business Support Services

Financial Statements

Company Financial: D&B

The second se D&B currently has no financial information on file for this company. You can ask D&B to make a personalized request to this company on your behalf to obtain its latest financial information by clicking the Request Financial Statements button below.

Additional Financial Data

This report was updated utilizing information the company has filed with the Securities Exchange Commission.

Request Financial Statements

Key Business Ratios

D & B has been unable to obtain sufficient financial information from this company to calculate business ratios. Our check of edditional outside sources also found no information available on its financial performance.

To help you in this instance, ratios for other firms in the same industry are provided below to support your analysis of this business.

Based on this Number of Establishments

	Industry Norms Bas	ed On 19 Establishments	
	This Business	Industry Median	Industry Quartile
Profitability		-	
Return on Sales	UN	6.2	UN
Return on Net Worth	UN	13.4	UN
Short-Term Solvency			
Current Ratio	UN	1.9	UN
Quick Ratio	UN	1.1	UN
Efficiency			
Assets/Sales	UN	121.3	UN
Sales / Net Working Capital	UN	2.2	UN
Utilization			
Total Liabilities / Net Worth	UN	101.6	UN

UN = Unavailable

View Snapshots

View Snapshots

This information may not be reproduced in whole or in part by any means of reproduction.

and a set along them & Branking

19

Exhibit C-8 "Bankruptcy Information"

World Energy Solutions, Inc. has not filed for protections from creditors, reorganization or any other form of bankruptcy filings in the two most recent years preceding the applications.

Exhibit C-9 "Merger Information"

World Energy Solutions, Inc. (Worcester, MA) acquired privately held EnergyGateway, LLC (Dublin, OH) on June 4, 2007. The acquisition provided customers of World Energy Exchange a bigger marketplace, an expanded online toolset, and a deeper pool of market expertise.

Exhibit D-1 "Operations"

World Energy Solutions plans:

World Energy is an energy brokerage company that has developed an online reverse auction platform, the World Energy Exchange. We bring energy suppliers and commercial, industrial and government energy customers together in our virtual marketplace, often with the assistance of channel partners. Although our core competency is brokering electricity, we have adapted our World Energy Exchange platform to accommodate the brokering of natural gas, green power and energy-related financial instruments, in response to customer demand. The World Energy Exchange is part of a complete end-to-end online energy procurement process that includes managing deals and tasks, providing market intelligence, developing RFPs, conducting auctions, analyzing results, assisting customers through the contracting process, and tracking customers' energy contracts and usage history.

Our end-to-end online energy procurement process is attractive to channel partners as it provides them with a business automation platform to enhance their growth, profitability and customer satisfaction. Channel partners such as energy service companies, demand side consultants and manufacturers, ABCs and strategic sourcing companies are an integral component of our business as these firms increasingly utilize our World Energy Exchange to enhance their service offerings to their customers. By accessing our market intelligence, auction and automated platform, channel partners significantly contribute to the Company's trading activity and procurement volume.

World Energy Solutions experience:

- World Energy Solutions does not trade or ever take title or ownership of natural gas, electricity or other energy commodity. World Energy Solutions is an energy broker.
- World Energy Solutions is the market share leader in North America, having brokered over 40 billion kWh of electricity and 700 million kWh of green power and Renewable Energy Certificates (RECs) and one trillion cubic feet of natural gas.
- World Energy Solutions runs a procurement exchange that is **supplier-neutral**. Our mission is to get the consumer the lowest price for their energy needs.
- World Energy Solutions provides a toll free number (1-800-578-0718) for all customers inquires.
- World Energy Solutions will not be billing retail customers so has not included a sample bill. World Energy Solutions' fee is paid for directly by the supplier.

Exhibit D-2 "Operations Expertise"

See the enclosed Excel spreadsheet. This spreadsheet summarizes World Energy Solutions, Inc. experience in natural gas procurement.

ŀ

i

1

0.000.600	Dih	ца С	£	Ē	£	ŧ	£	튭	튭	Ę	Dth		đ	튭	ā	£	ŧ	Ē	튭	ŧ	₫	Ē	튐	ų Š	ą	튑	£	ð	ŧ	튭	0th
Oight o	24,527	14,570	133,198	237,628	340,092	267,480	73,551	37,004	100,136	81,499	199,042	133,326	96,028	450,560	4,153	2,408	53,900	9,594	43,352	133,603	20,310	49,241	100,978	34,800	189,968	355,096	278,711	5,188	227,100	61,680	32,766
S engle Datas	20	ME	CA	GA	GA	Ş	¥	교		£	£	z	¥	£	£	₹	≥	₅		-	2	2	2	Ž	Ż	Z	8	5	IM	ž	λ
Ada at	B Washington Gas Light Company	Northern Utilities Natural Gas	San Diego Gas and Electric Co. / Sempra		Attanta Gas Light Company	3 Columbia Gas of Kentucky Inc	Delta Natural Gas Company, Inc.	TECO Peoples Gas	Ameren CIPS	Duke Energy-Ohlo	2 Duke Energy-Ohio	Citizens Gas and Coke Utility	Dominion East Ohio	Columbia Gas of Ohio Inc.	Columbia Gas of Ohio Inc.	Dominion East Ohio	Detroit Edison	Detroit Edison	Ameren IP	Nicor Incorporated	Peoples Gas Light and Coke Company	Peoples Gas Light and Coke Company	Peoples Gas Light and Coke Company	SEMCD ENERGY	SENCO ENERGY	Vectren Corporation	VECTREN (Formerty Dayton Power & Light)	VECTREN (Formerly Dayton Power & Light)	Wisconsin Gas Company	2 Consolitiated Edison Company of New York	2 Company of New York
A Property	8	-	£	37	e S	e E			5	Ċ	2	2	2	εu	4		-	2	-	-	10	-	2		~	-	-	-	-	~	α -
August in the set	НЕИРУ НИВ		HENRY HUB																					-							
$\exp_{T_{i}}(x_{i}) \exp_{T_{i}}(x_{i})$	Indexed	Fixed	fndexed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Rxed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fbrad	Fixed	Fixed
Lowean Sainte	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired
Steff to Arrit	11/22/2002	10/20/2003	12/1/2002	8/2/2004	8/2/2004	8/2/2004	8/2/2004	8/2/2004	8/2/2004	8/2/2004	8/2/2004	8/2/2004	8/2/2004	8/2/2004	8/2/2004	8/2/2004	8/2/2004	8/2/2004	8/2/2004	B/2/2004	8/2/2004	6/2/2004	8/2/2004	8/3/2004	8/3/2004	8/3/2004	8/3/2004	8/3/2004	8/3/2004	8/3/2004	8/3/2004
al ann aite Aite na c	12	24	24	24	54	14 1	4	54	24	14	12	24	e	24	ल –	10	13	14	4	4	च	ৰ	<u>1</u> 2	.t.t.	12	12	13	21 F	24	24	7
lla over Eri. Dise	10/31/2003	11/30/2005	12/30/2004	8/31/2006	3002/12/9	10/31/2005	10/31/2005	8/31/2006	B/31/2006	10/31/2005	8/31/2005	8/31/2006	12/31/2004	8/31/2006	12/31/2004	12/31/2004	10/31/2005	10/31/2005	12/31/2004	12/31/2004	12/31/2004	12/31/2004	8/31/2005	10/31/2005	3/30/2002	9/30/2005	10/31/2005	9002/06/8	9/30/2006	11/30/2006	11/30/2006
liter der die Volume Rocht	11/1/2002	12/1/2003	1/2/2003	9/1/2004	9/1/2004	9/1/2004	9/1/2004	9/1/2004	P/1/2004	9/1/2004	9/1/2004	9/1/2004	10/1/2004	9/1/2004	10/1/2004	10/1/2004	10/1/2004	9/1/2004	9/1/2004	9/1/2004	8/1/2004	9/1/2004	9/1/2004	10/1/2004	10/1/2004	10/1/2004	10/1/2004	10/1/2004	10/1/2004	12/1/2004	(2)(2004
tak ar	×	×	×	×	×	×	×	×	x	x	×	×	×	x	×	X	×	×	×	×	×	×	×	x	x	×	×	×	x	X	×
9 4 AZ	Retail	Retail	Retail	Retai	Retai	Retail	Retail	Retail	Retaíl	Retail	Relail	Retail	Retail	Retail	Retall	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retali	Retail	Rotal
17.00 M	1003	1004			1010	1011	1012						1018		_			1023								1031					1036

Page 1 of 18 Report Date: 10/2/2008 3:24:19 PM

Page 2 of 18 Report Date: 10/2/2008 3:24:19 PM

			<u> </u>			1	r		<u>۱</u>			1			1	- 1	<u> </u>					_					- -1
ate d'an	Dit	÷,	ŧ	Ð	Ър С	Ę	둼	臣	₫	£	f	£	Ê	튭	ŧ	ā	Ē	Ê	đ	ā	ā	튭	đ				튭
al antiques	23,602	19,019	419,144	303,882	307,306	511,128	40,982	453,098	130,854	316,46 2	169,560	22,179	163,386	174,006	113,776	130,484	555,606	887'81	106,862	669,314	266,378	154,754	10,366	249,626	280,034	258,240	97,838
ára ca Star	λ	λ	γY	Å	¥	٨V	QW	Ŵ	QW	QW	QW	٨٨	Q	A	Ħ	٨v	PA	PA	PA	ΡA	ΡA	A	PA	٨٨	۲d	×۸	Ī
12 A.	3 National Fuel Gas Distribution Corp.	l National Fuel Gas Distribution Corp.	3 National Fuel Gas Distribution Corp.	1 National Grid - Niagara Mohawk	1 New York State Electric & Gas Corp.	3 Mountaineer Gas	l Baltimore Gas & Electric Company (BGE)	li Baltimore Gas & Electric Company (BGE)	2 Baitimore Gas & Electric Company (BGE)	5 Washington Gas & Light Company	Washington Gas & Light Company	25 Washington Gas Light Company	53 Washington Gas & Lighi Company	Columbia Gas of Maryland Inc.	Delmarva Power Companyo	Hope Gas Incorporated	3 Equitable Gas Company PA	National Fuel Gas Distribution Corporati	I National Fuel Gas Distribution Corporati	HUGI Penn Natural Gas - PG Energy	I Exeton (PECO Energy Company)	PPL Utfilles	I UGI Utilities, Inc.	Virginia Natural Gas	duci Utilities, Inc.	I Cotumbia Gas of Virginia Inc.	i Detroit Edison
t tra Autour N	.,						1			<u>a</u> ,		52	ŝ	-	-	-		-	÷				-		-		
Prent Sec.																											
ad Victoria II.	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed
fubrite m Siblue	Expired	Expired	Expired	Expired	Expired	Expired	Epired	Expired	Eppired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Equired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired
Seri kernen Lätta	B/3/2004	8/3/2004	B/3/2004	8/3/2004	B/3/2004	8/3/2004	8/3/2004	8/3/2004	8/3/2004	8/4/2004	8/4/2004	8/4/2004	8/4/2004	8/4/2004	8/4/2004	8/4/2004	8/4/2004	8/4/2004	8/4/2004	8/4/2004	8/4/2004	8/4/2004	8/4/2004	8/4/2004	8/4/2004	8/5/2004	8/4/2004
Turi ngel Mangreg	24	5	24	24	24	24	24	24	24	54	5	12	12	4 3	54	24	24	24	24	24	24	24	24	24	24	9 0	14
Pre- verve hru. Distr	11/30/2006	11/30/2005	11/30/2006	1/31/2007	1/31/2006	1/31/2007	1/31/2007	1/31/2007	10/31/2006	10/31/2006	10/31/2005	10/31/2006	10/31/2005	1/31/2007	1/31/2007	1/31/2007	10/31/2006	10/31/2006	1/31/2002	10/31/2006	10/31/2006	1/31/2007	1/31/2007	10/31/2006	1/31/2007	10/31/2008	10/31/2005
Do ware Bland and C	12/1/2004	12/1/2004	12/1/2004	2/1/2005	2/1/2004	2/1/2005	2/1/2005	2/1/2005	11/1/2004	11/1/2004	11/1/2004	11/1/2004	11/1/2004	2/1/2005	2/1/2005	2/1/2005	11/1/2004	11/1/2004	2/1/2005	11/1/2004	11/1/2004	2/1/2005	2/1/2005	11/1/2004	2/1/2005	11/1/2004	9/1/2004
and a straight of the	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	x	x	×	×	×	×	×	x	×	×	×
1	Retail	Retail	Retail	Retaìl	Retail	Retail	Retail	Retall	Retail	Hetail	Retail	Hetait	Retail	Retail	Retail	Retail	Retail	Retail	Retall	Retali	Retail	Retail	Retall	Retail	Retail	Retall	Retall
the second second	1037	1038	1039	1040	1041	1042	1043	1044	1045	1045	1047										1057	r -			1081		1083

361

3 of 18 Report Date: 10/2/2008 3:24:19 PM	
Page	

astra ve	ti Li	ц П О	ŧ	ц.	ŧ	ŧ	Ħ	ŧ		튤	훕	đ	ŧ	ŧ	ŧ	ŧ	ŧ	ŧ	£	튱	€	L O	s	ŝ	튭
2010-0117	212,810	358,842	253,842	104,454	500,326	123,916	206,937	86,883	115,452	90,232	70,044	249.418	308,122	195,278	570,230	3,635	609,434	12,585	128,704	129,474	96.5 46	136,070	1,048,044	266,326	36,766
f e anto - 1 Side	WN	X	хı	тх	ð	8	8	8	8	WA	WA	WA	CA	CA	CA	ک	đ	క	¥	g	¥	ž	ż	N	g
and the second second	Public Service Company of New Mexico	5 Atmos Energy	Atmos Energy	Atmos Energy	Oklahoma Natural Gas Company	Atmos Energy Corporation (Greeley Gas)	Xcel Energy/Public Service Co. Colorado	5 Xcel Energy/Public Service Co. Colorado	Xcel Energy/Public Service Co. Colorado	Avista Utilities	Cascade Natural Gas Company	Puget Sound	1 Long Beach Gas	Southern California Gas Co- The Gas Comp	San Diago Ges and Electric Co. / Sempre	Pacific Gas and Electric Company (PGE)	Pacific Gas and Electric Company (PGE)	Pacific Gas and Electric Company (PGE)	Unisource (formly.Citizens Arizona Gas)	2 Laciede Gas Company	Southern Union Company (SUG)	l Central Hudson Gas & Electric Corp.	Company of New York	d Public Service Electric and Gas Company	Washington Gas Light Company
atel Accession	-	1,			4	~	6	5	2	-	-	-	-		~			-					64	2	8
seller de la																									
in group lights	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Ficed	Fixed	Fixed	Fixed	Fixad	Fixed	Fixed	Dex II.	Fixed	Fixed	Fixed	Fixed	pexta	Fixed	Fixed	Fixed
Dord acr Sizina	Expired	Expired	Expired	Expired	Expired	Expired	Expired		Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired
Settion of L		8/25/2004			8/25/2004	8/26/2004	8/26/2004		8/26/2004	8/31/2004		8/31/2004		8/31/2004			8/31/2004	8/31/2004							9/27/2004
Due tam Versio	24	4	24	24	24	মূ	<u>ы</u> .	5	5	24	24	24	24 1	*	54	Ŧ	24	5	9 N	12	54	24	6	24	4 4 2
É Cara y Cont Cara	10/31/2006	10/31/2005	10/31/2006	9/30/2006	10/30/2006	10/31/2005	10/31/2005	10/31/2005	10/31/2005	3/31/2007	11/30/2006	11/30/2006	11/30/2006	11/30/2006	11/30/2006	10/31/2005	11/30/2006	10/31/2005	11/30/2008	11/30/2005	10/31/2006	10/31/2006	11/30/2006	11/30/2006	10/31/2006
$D=0=\sqrt{2} N_{\rm eff}^{\rm eff}$	11/1/2004	11/1/2004	11/1/2004	10/1/2004	11/1/2004	11/1/2004	11/1/2004	11/1/2004	11/1/2004	4/1/2005	12/1/2004	12/1/2004	12/1/2004	12/1/2004	12/1/2004	12/1/2004	12/1/2004	12/1/2004	12/1/2004	12/1/2004	11/1/2004	11/1/2004	12/1/2004	12/1/2004	11/1/2004
Cubuch en	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×
3081-20	Retail	Retail	Retail	Retail	Wholesale	Retai!	Retail	Retal	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retal	Retail	Retail	Retail	Retail	Retail	Retal	Retail	Retall	Retail
10 - 1 - 20 1	1064	1			1068	1069	1070		1072	1073		1075		101	1078	1079	1080	1081	1082	1083	· · · · ·	Г			8801-

маг * с	ЪС	ŧ	Đ	E	Dth	Dth	Oth	Oth	D <u>t</u> h	Ð	둼	ŧ	Dih	ĥ	ŧ	튑	튭	Ę.	а С	ц; О	ŧ	ŧ		ŧ	đ	Ð	\$ <u></u>
Guarta.	32,279	134,445	11,955	1,627,465	213,191	610,200	57,500	206,177	136,700	106,000	222,000	1,005,125	1,461,820	290,849	440.628	29,652	53,271	16,802	53,634	63,963	83,799	32,829		16,868	67,671	28,344	2,868
Su ware 2 tale	н Н	ΡA	ΡA	sc	GA	GA	GA	FL	Ъ	хı	×	ž	DC	ЧV	r.	ž	MA	ME	MA	ΓN	Ν	PA A		æ	NΥ	٨Y	À
24-32	2 NUI City Gas Company of FL	Exelon (PECO Energy Company)	Exalon (PECO Energy Company)	Transco	Enbridge Marketing	Textcan Industrial Enargy Marketing	Texican Industrial Energy Marketing	TECO Peoples Gas	Infinite Energy, Inc Florica/917914	Atmos Energy	Atmos Energy	Coral Energy Resources, LP	Washington Gas Light Company	Arkanses Oklahoma Gas Corporation (AOG)	South Jersey Gas Company (SJG)	Keyspen Energy - Energy North	Boston Gas Company	Northern Utilities Natural Gas	NSTAR Electric and Gas Corporation	2 New Jersey Natural Gas (NJNG)	Consolidated Edison Company of New York	PECO Energy		NSTAR Electric and Gas Corporation	Orange and Rockland Utilities, inc.	Company of New York	Company of New York
Tota Autorite	.01	-	-	T		L I		÷	1	-	•	ł	1	F	16	-		-	¢,	8	~		0	-	8	F	-
xon, fooda _t	HENRY HUB	HENRY HUB	HENRY HUB	Transco Stalion 30, 45 & 65	IFERC SNG	IFERC SNG	IFERC SNG	NYMEX	NYMEX	WAHA .	WAHA	WAHA		centérpoin T east	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX			
$\mathcal{W} \mathcal{S}_{\pm}(\tilde{0}, \tilde{0}) \sim_{\Xi}$	Indexed	Indexed	Indexed	Indexed	Indexed	pexepul	Indexed	Indexed	Indexed	Indexed	Indexed	Indexed	Fixed	pexequ	Indexed	Indexed	hdexed	Indexed	Indexed	Indexed	Indexed	Indexed	Indexed	Indexed	Fbed	Fixed	Fixed
Concrust Marus	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Éxpired	Expired	Expired	Expired	Active	Active	Active	Active	Active	Active	Active	Cancelled	Active	Expired	Expired	Expired
Sertioner - Joine	6/2/2004	4/27/2004	4/27/2004	3/2/2005	3/2/2005	3/2/2005	3/2/2005	3/2/2006	3/2/2006	3/2/2005	3/2/2005	3/2/2005	7/31/2003	6/30/2005	11/2/2005	9/30/2005	9/30/2005	9/30/2006	9/30/2006	9/27/2005	9/27/2005	9/27/2006	9/30/2005	9/30/2005	2/24/2006	2/1/2002	2/1/2002
Contract Vendes	2	8	ន	<u>8</u>	ş	5	<u>2</u>	<u>0</u>	12	12	<u>e</u>	12	24	en	2	8	8	т. С	36	æ	8	8	8	8 N	<u>ې</u>	<u>6</u>	12
10 Parkana Ala	8/1/2005	5/1/2008	5/1/2006	4/1/2006	4/1/2006	4/1/2006	4/1/2006	4/1/2006	4/1/2006	4/1/2006	4/1/2008	4/1/2006	10/31/2005	7/31/2006	11/30/2007	8/30/2008	8002/08/6	9/30/2008	8002/08/8	9/30/2008	8002/06/8	9/30/2008	8002/08/8	9/30/2008	3/31/2007	3/1/2003	3/1/2006
an were San t	8/1/2004	6/1/2004	6/1/2004	4/1/2005	4/1/2005	4/1/2005	4/1/2005	4/1/2005	4/1/2005	4/1/2005	4/1/2005	4/1/2005	11/1/2003	8/1/2005	12/1/2005	10/1/2005	10/1/2005	12/1/2006	10/1/2005	10/1/2005	10/1/2005	10/1/2005	10/1/2005	10/1/2005	4/1/2006	3/1/2002	3/1/2002
2000 - 100 -	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×
n and the number	Retail	Retail	Retâl	Retail	Retail	Retail	Retail	Helai	Retail	Retail	Retail	Retali	Retail	Wholesale	Retail	Retail	Retail	Relati	Retail	Retail	Retai	Retail	Retail	Hetal	Retail	Retait	Hetail
	6Z 11	1180	1181	1217	1218	1219	1220	1221		1		1225	1307	1320	1394	1396		1399	1400	1401	1402			1405	1416	1429	1430

Page 4 of 18 Report Date: 10/2/2008 3:24:19 PM

Page 5 of 18 Report Date: 10/2/2008 3:24:19 PM

NOT NE	Ę	ŧ	흄	튭	£	ŧ	Ē	£	£	ŧ	£	ŧ	Ē	ŧ	튭	ŧ	ŧ		흄	ŧ	ŝ	튭	툼	đ	÷	ŧ	튭
Carte 6	638,256	B.723	131,718	4,626,384	136,200	488,232	268,272	2,811,744	3,435,372	1,125,912	9,180	666'6	106.996	8,618	237,939	142,749	61,612		5,100	22,452	804,000	175,715	22,867	107.713	508,868	167,360	173,925
	٨	λ	۲Y	8	ž	Å	٨	Å	Ş	5	٨N	ž	ž	AM	z	¥	¥		λ		임 임	8	۲Ň	g	z	M	W
50 U.S	Central Hudson Gas & Electric Corp.	New York State Electric & Gas Corp.	Keyspan Energy Delivery, NY	Xoel Energy/Public Service Co. Colorado	Consolidated Edison Company of New York	National Fuel Gas Distrbution Corp.	Bath Electric, Gas & Water Systems	Consolidated Edison Company of New York	Washington Gas & Light Company	Pacfile Gas and Electric Company (PGE)	Consolidated Edison Company of New York	Consolidated Edison Company of New York	Consolidated Ectson Company of New York	NSTAR Electric and Gas Corporation	Valley Cities (Valley Energy)	UGI Penn Natural Gas - PG Energy	Rockwood Water and Gas		Consolidated Edison Company of New York	Consolidated Edison Company of New York	Washington Gas Light Company	Washington Gas & Light Company			Veotren Corporation	Detroit Edison	SEMCO ENERGY
TOTA ACTINE	L	-	-	1	-	-	-	-		-	-	-	.		-	1	4	0	-	स्व	#	-	24	42	-	4	CN
																	-	NYMEX									
	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Indexed	Fixed	. pexia	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed
	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Cancelled	Expired	Expired	Expired	Expired	Ëxpired	Expired	Expired	Expired	Expired
	10/1/2002		9/1/2002	10/2/2002	4/1/2002	11/1/2002	11/1/2002	2002/1/11	10/1/2003	11/1/2003	2/1/2002	12/1/2001	1/1/2002	11/24/2003	10/1/2003	10/1/2003	3/1/2006		2/1/2002		5/9/2006	5/9/2006	5/9/2006	5/9/2006	5/10/2006	5/10/2006	5/10/2008
	24	÷	58	4 0	<u>19</u>	54	т С	2	ŭ	Ň	œ	ö	<u>م</u>	5	\$	17	12	12	<u>21</u>	54	<u>₽</u>	₽	<u>e</u>	<u>R</u>	24	2	<u>5</u>
	11/1/2004	4/1/2003	3/1/2005	11/1/2004	5/1/2002	12/1/2004	12/1/2004	12/1/2004	11/1/2004	12/4/2004	8/1/2003	1/1/2003	2/1/2003	12/24/2004	5/1/2005	4/1/2005	5/1/2007	6/1/2007	3/1/2003	7/1/2008	<u>6/1/2007</u>	6/1/2007	6/1/2007	6/1/2007	6/1/2008	6/1/2007	6/1/2007
	11/1/2002	3/1/2002	10/1/2002	11/2/2002	5/1/2001	12/2/2002	12/1/2002	12/1/2002	11/1/2003	12/1/2003	3/1/2002	1/1/2002	2/1/2002	12/24/2005	11/1/2003	11/1/2003	5/1/2006	6/1/2008	3/1/2002	7/1/2006	6/1/2008	6/1/2006	6/1/2008	6/1/2006	6/1/2006	6/1/2006	6/1/2006
	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×
	Relail	Relail	Retail	Hetail	Retail	Retall	Retall	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retall	Reteil	Retail	Retail	Retail	Retail	Retail	Retail	Retali	Retail	Retall	Retail
i shawr I	1434		1456	1462	1463	1465	1466	1467	1984 1984	1468	1472	1497	1505	1513	1517	1518	1571	1579		1607	1623	1624	1625	1626		1	1629

_

Page 6 of 18 Report Date: 10/2/20	08 3:24:19 PM	
Page 6 of 18 Repor	1 Date: 10/2/20	
	^D age 6 of 18 Repor	

ACT NE	훕	造	- BO	ŧ	Ъср.	튭	å	ŧ	튭	튭	튭	ŧ	ŝ	Ę	Ę	ŧ	ā	튭	ŧ	ŧ	Ē	튪		Dth	đ	튭	ŧ	튭	ŧ	ź	ŧ	£	ŧ	튭
G. and a	302,D16	74,229	269,622	108,335	410,330	35,482	3,203	62,794	624,635	6,167	144,064	34,477	14,850	39,114	12,208	478,988	120,429	198,609	141,060	109,446	394,564	88.572		123,956	238,418	57,976	123,040	57,508	128,350	47,943	87.078	90,708	252,872	32,637
tativalar 10 Amerika	ъ	-	¥}	QM	8	¥	5	ÿ	ř	£	ۍ لا	 2	ξ	<u>ک</u>	Ł	Z	٨d	PA	Ł	¥.	ΡA	₹d		ĸ	£	ř	X	5	₹	z	¥	¥	g	æ
	Z Dominion East Ohio	1 Feoples Gas Light and Coke Company	1 Alliant Energy	ZLaclede Gas Company	3 Xoel Energy/Public Service Co. Colorado	1 SEMCO ENERGY	1 Columbia Gas of Ohio Inc.	B Washington Gas & Light Company	7 Atmos Energy	1 Dominion East Ohio	1 Columbia Gas of Kentucky Inc	1 Atmos Energy	1 Abros Energy	1 Atmos Energy	1 Atmos Energy	16 New Jersey Natural Gas (NJNG)	1 Equilable Gas Company PA	3 Dominion Peoples	2 Columbia Gas of Pennsytvania Inc.	3 National Fuel Gas Distribution Corporati	3 PPL Utilities	2 UGI Utilities, Inc.	0	Z Atmos Energy	1 VECTREN (Formerly Dayton Power & Light)	4 Atmos Energy	1 Amos Energy	4 Pacific Gas and Electric Company (PGE)	2 Duke Energy-Ohlo	4 Public Service Electric and Gas Company	5 Public Service Electric and Gas Company	1 Duke Energy-Ohio	35 Attanta Gas Light Company	1 TECO Peoples Gas
Ser Courtes																																		
No. of the American											NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX						NYMEX	NYMEX	NYMEX										
and the right	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Indexed	Indexed	Indexed	hdexed	Indexed	Indexed	Fixed	Fixed	Fixed	Fixed	Fixed	bexebra	Indexed	Indexed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed
Operation Setemation	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired			Expired	Expired	Γ	Expired	Expired		Expired	Expired	Expired	Expired	Active		bed	Active	Expired		Expired	Expired	Expired	Expired	Expired		Expirad	Expired
êrin çente iz ⊒de	5/10/2006	5/10/2006	5/10/2006	5/10/2006	5/11/2008	5/11/2008	5/11/2006	5/11/2006	5/11/2006	5/16/2006	4/13/2006	7/10/2006	7/10/2006	7/10/2006	7/10/2006	6/27/2005	9/6/2008	9/6/2008	9/6/2006	3/6/2006	9/6/2006	9/8/2008	9/6/2006	9/6/2006	5/10/2008	5/16/2008	5/16/2006	5/31/2006	8/7/2006	6/13/2006	8/13/2008	7/18/2006	5/24/2006	5/24/2006
Comaet Minute	12	12 12	24	12	24	12	12	ស្	12	<u>č</u> e	12	12	12	12	<u>8</u>	24	5	12	<u>5</u>	12	54	12	24	24	12	24	24	24	a	N.	24	12	24	₽
Bet Jeny Lee Meri	6/1/2007	6/1/2007	6/1/2008	6/1/2007	6/1/2008	6/1/2007	6/1/2007	6/1/2007	6/1/2007	6/1/2007	6/1/2007	8/1/2007	8/1/2007	8/1/2007	8/1/2007	12/1/2008	10/31/2007	11/1/2007	11/1/2007	11/1/2007	11/1/2008	11/1/2007	12/1/2008	12/1/2008	7/1/2007	7/1/2008	7/1/2008	7/1/2008	8/1/2008	8/1/2007	8/1/2008	8/1/2007	8002/1/8	9/1/2007
Tara were Sheet	6/1/2006	6/1/2006	6/1/2006	6/1/2006	6/1/2006	6/1/2006	6/1/2006	6/1/2006	6/1/2006	6/1/2006	6/1/2006	8/1/2006	8/1/2008	8/1/2006	8/1/2006	12/1/2006	1/1/2007	11/1/2006	11/1/2006	11/1/2006	11/1/2008	11/1/2006	12/1/2006	12/1/2006	7/1/2008	7/1/2006	7/1/2006	7/1/2006	B/1/2006	B/1/2006	8/1/2006	8/1/2006	B/1/2006	9/1/2006
and the second second	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	x	×	×	×	×	×	×	x	×	x	x	×	×	×	×	X	×	×
e en Sec	Retail	Retail	Retail	Hetai	Hetail	Retail	Retail	Retail	Retail	Relail	Retail	Retell	Retail	Retail	Retail	Retali	Retal	Retail	Retall	Retail	Retail	Retail	Retail	Retal	Relait	Retail	Retail	Tetail	Retail	Retail	Retail	Rotali	Retail	Hetaii
2010-121 2010-121	1630 F		1632 F		_	1635 F		1637 F				Г	1	_							1682 F						1688 F						1694	1695

ŧ 돌 툼 튭 튭 £ 튭 튭튭 튐툼 हिहि 튭 ₫ E)£ f Ē 튭튭 튭 틆 ā 튭 튭 튭 266,242 83,916 44,551 608,154 367.322 442,918 107,942 50,246 223,285 17,246 66,237 212,110 96,558 191,759 192,288 269,838 125,944 142,791 23,740 725,676 518,466 263,096 208,039 13, 794 4,720 222,204 45, 132 \$ 8 ¥ 2 A g ≿ z = ਤਿ MA 55 ž ¥≩ ⊒ ≤ ĕĕ ₹ 8 a ¥ ≿ -1 1 1 Aumos Ertergy 4 Oktahoma Natural Gas Company 1 Public Service Company of New Mexico Company) 3 UGI Pern Natural Gas -PG Energy 1 Virginia Natural Gas Columbia Gas of Ohio Inc. 16 Yankee Gas 2 Southern Connecticut Gas Wisconsin Gas Company Bay State Gas Company 9 Aunos Energy 1 Cheyenne Light Fuel & Power Co 1 Aviste Utilities 2 Washington Gas Light Company 3 Peoples Gas Light and Coke Company 1 Peoples Gas Light and Coke Company 1 Niccr Incorporated 1 MidAmerican Energy 23 Bay State Gas Company 8 Xcel Energy/Public Service Co. Colorado 1 Exelon (PECO Energy Citizens Gas and Coke Unitry Atmos Energy Corporation (Greeley 3 Atlanta Gas Light Company 3 Columbia Gas of Kentucky Inc 1 Detta Natural Gas Ameren CIPS Company, Inc. Company Gas) 8 pext∤_l Fixed ixed pex!) Xed Dex: Pex Fixed Fixed pex: pexia Fixed ixed Expired Expired Expired Active Expired Expired Expired Expired Expired 5/23/2006 Expired 5/23/2006 Expired Expired Expired Expired Expired 5/31/2006 Active 5/31/2006 Active Active 6/1/2006 Active Active Active Active 7/18/2006 Active 5/16/2008 Active Active Active 5/25/2006 / 5/24/2006 5/25/2006 5/25/2006 5/25/2006 5/25/2006 5/24/2006 5/25/2006 5/31/2008 8/1/2008 6/1/2006 6/7/2008 5/23/2006 5/23/2006 5/25/2006 5/24/2006 5/24/2006 5/25/2006 5/31/2006 តី 4 Ņ 12 <u>ы</u> <u>N</u> 2 24 <u>N</u> 2 2 42 ŝ 4 2 4 4 8 24 24 র 11/1/2008 10/1/2008 11/1/2008 11/1/2008 11/1/2008 9/1/2008 9/1/2007 9/1/2007 10/1/2007 10/1/2007 10/1/2008 4/1/2009 11/1/2007 11/1/2007 11/1/2008 9/1/2008 9/1/2007 10/1/2007 11/1/2008 11/1/2008 9/1/2008 10/1/2007 9/1/2008 11/1/2007 11/1/2006 10/1/2006 10/1/2006 10/1/2006 4/1/2007 11/1/2006 11/1/2006 11/1/2006 11/1/2006 11/1/2008 10/1/2006 11/1/2006 11/1/2006 9/1/2006 9/1/2006 10/1/2006 9/1/2006 10/1/2006 9/1/2006 9/1/2006 8/1/2006 9/1/2006 11/1/2006 Retail Retail 1722 Retail Retail Retai Retail Retail Retail: Retail Hets Retail Retai Retail Retail Retail Retail Retail 1705 1706 1710 1715 1700 1696 1698 1699 1702 1718 1719 07<u>7</u> 1703 1708 1712 1714 1715 1704 1709 1713 1697 Ę 2 1711

Page 7 of 18 Report Date: 10/2/2008 3:24:19 PM

~
ã.
e.
÷.
8
ő
8
S
8
÷.
9
ő
Æ
ā
æ
÷
5
õ
8
a.
-

240° 1072	튭	ĥ	ŧ	fo		f	₫	ŧ	ŧ	£	Ē	â	đ	₫	£	£	£	튭	ŧ	₫	f	£	đ	£	£	ŧ
a da ante	22,690	23,262	521,104	109,934	268,644	119,970	600,148	311,725	290,398	9,804	47,972	143,776	594,662	283,564	109,040	65,254	803,860	196,096	246,198	23,740	402,964	1,113,830	36,662	149,162	282,812	467,460
Survice - hite	¥	PA	٩٩	DW W	٨٧	ž	QN	₽	8	8	ž	¥	Ś	8	A2	WA	S	¥ Ŭ	WA	ž	ž	λ	PA	۲d	٧d	Ŵ
CTIN.	MidAmerican Energy Company	National Fuel Gas Distribution Corporati	Equitable Gas Company PA	Baltimore Gas & Electric Company (BGE)	Columbia Gas of Virginia Inc.	Central Hudson Gas & Electric Corp.	Washington Gas & Light Company	Intermountain Gas Company	Xcel Energy/Public Service Co. Colorado	Xcel Energy/Public Service Co. Colorado	ZConsolidated Edison Company of New York	Public Service Electric and Gas Company	Paolfic Gas and Electric Company (PGE)	Long Beach Gas	Unisource (formly Citizens Arizona Gas)	Cascade Natural Gas Company	San Diego Gas and Electric Co. / Sempra	Southern Cattornia Gas Co- The Gas Comp	Puget Sound	National Fuel Gas Distribution Corp.	National Fuel Gas Distribution Corp.	Consolidated Edison Company of New York	National Fuel Gas Distribution Corporati	PPL Utilihes	UĜI UMINes, Inc.	Mountaineer Gas
$\sum_{i=1}^{n} \frac{1}{2} $	L		e	CJ	-	-	4	-	R	1	8	2	Ŧ	Ē	÷	-	0	-	-		0	4	*	-	÷	Ø
Kindowa Star Daller																										
Recent 7 MA	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed
Transaction of the second of t	Active	Active	Active	Active	Active	Active	Active	Expired	Active	Expired	Active	Expired	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active
Fichto Spar-	6/6/2006	6/7/2006	6/7/2006	6772006	6/7/2006	6/13/2006	6/13/2006	6/13/2006	6/20/2006	6/20/2006	5/18/2006	5/17/2006	5/17/2006	5/31/2006	5/31/2006	6/6/2006	6/6/2006	6/6/2006	6/6/2006	6/13/2006	6/13/2006	6/14/2006	5/24/2006	5/24/2006	5/24/2006	8/1/2006
lar tart Vennes	24	24	24	54 74	2	7 7	24	12	24	2	24	12	54	24	7 5	R.	Ň	24	24	24	24	54	47		2	1 1
the second from the second s	11/1/2008	11/1/2008	11/1/2008	11/1/2008	11/1/2008	11/1/2008	11/1/2008	11/1/2007	11/1/2008	11/1/2007	12/1/2008	12/1/2007	12/1/2008	12/1/2008	12/1/2008	12/1/2008	12/1/2008	12/1/2008	12/1/2008	12/1/2008	12/1/2008	12/1/2008	2/1/2009	2/1/2008	8/1/2008	2/1/2009
ી, જુવાયું કે, જ	11/1/2006	11/1/2006	11/1/2006	11/1/2006	11/1/2006	11/1/2006	11/1/2006	11/1/2008	11/1/2006	9002/1/11	12/1/2006	12/1/2006	12/1/2006	12/1/2006	12/1/2006	12/1/2005	12/1/2006	12/1/2008	12/1/2006	12/1/2006	12/1/2006	12/1/2006	2/1/2007	2/1/2007	2/1/2007	2/1/2007
. Turtur en	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×
8	Retait	Retail	Retail	Retail	Retail	Retail	Retai	Retai	Heta!	Hetail	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retal	Retail	Hetail	Retali	Retail	Retail
1	1723	1724		1726		1728	1729	1730				1734	1735	1736	1	Г			1741	1			· · · · ·			1748

(12) × 42	ŧ	ŧ.	ŧ	튭	£	Ē	đ	ŧ	đ	ŧ	Ð	ŧ	Ē	臣	튪	£	ŧ	ŧ	£	튭	ę	f	흄	đ	€	ŧ	f	đđ
Summy -	420,934	164,458	114,802	111,480	271,424	330,856	402,830	287,552	36,482	8,167	173,925	302,016	3,203	47,943	167,360	238,418	223,285	624,635	102.638	50,246	107,713	261,326	108,335	1,608,000	4,720	191,759	311,725	9,804
S.1.401 - 3 1.4.46	ЯW	αw	8	^^	٨	۲V	VA	2	¥	н	MI	ę	н	Z	¥	£	 ₹	Ϋ́	8	_	QW	QW	Q	2	_		₽	8
At 11	l Baltimore Gas & Electric Company (BGE)	l Columbia Gas of Maryland Inc.	I Detmarva Power Company⊡	Hope Gas Incorporated	New York State Electric & Gas Corp.	National Grid - Niagare Mohawk	Roanoke Gas Company	Public Service Electric and Gas Company	SEMCO ENERGY	Dominion East Ohio	SEMCO ENERGY	Dominion East Ohio	Columbia Gas of Ohio Inc.	Public Service Electric and Gas Company	Detroit Edison	VECTREN (Formerly Dayton Power & Light)	Columbia Gas of Ohio Inc.	Atmos Energy	Atmos Energy Corporation (Greekey Gas)	Ameren CIPS	42 Washington Gas & Light Company	l Washington Gas & Light Company	Ladede Gas Company	i Washington Gas Light Company	Peoples Gas Light and Coke Company	Peoples Gas Light and Coke Company	Intermountain Gas Company	Xcel Energy/Public Service Co. Colorado
14 TU A	L	-			1	4	-	N		-	2	8	F	4	4		~		-	5	42	-		•	-	сл	-	
Pvic i Li Pdrx												-																
KALLUD.C	Fixed	Fixed	Fixed	Fixed	Fixed	Debri H	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed .	f besd	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	pexid	Fixed	Fixed	Fixed	Fixed
Compar Statur	Active	Active	Active	Active	Active	Active	Active	Active	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Expired	Active	Active	Active	Active	Active
Serren en Jato	6/1/2006	6/1/2006	6/1/2006	6/1/2006	6/6/2006	E/14/2006	9/13/2006	9/27/2006	9/27/2006	9/27/2006	9/27/2008	9/27/2006	9/27/2006		9/27/2006	9/27/2006	9/27/2006	9/27/2006	9/28/2006	9/28/2006	9/28/2006	8/28/2008	9/28/2006	9/28/2006	9/28/2006	9/28/2006	9/28/2006	9/29/2006
المراجع من المراجع المراجع من المراجع	54	54	24	24	24	24	24	24	মূ	12	12	12	12	<u>N</u>	한	2	<u>e</u>	12	F	臣		5	<u>a</u>	24	<u>1</u>	12	12	<u>8</u>
an An Air an Air an	600Z/1/Z	5007/1/2	2/1/2009	2/1/2009	2/1/2009	2/1/2009	10/1/2008	12/1/2009	6/1/2008	6/1/2008	6/1/2008	6/1/2008	6/1/2008	8/1/2008	6/1/2008	7/1/2008	9/1/2008	5/1/2008	4/1/2008	9/1/2008	6/1/2008	6/1/2008	6/1/2008	60/1/5008	11/1/2008	11/1/2008	11/1/2008	11/1/2008
ਦੀ '' ਹ 1012 ਨਿਹੋਲ ਨੂਪ	2/1/2002	2/1/2007	2/1/2007	2/1/2007	2/1/2007	2002/1/2	10/1/2006	12/1/2007	6/1/2007	6/1/2007	6/1/2007	6/1/2007	2007/1/9	8/1/2007	6/1/2007	7/1/2007	6/1/2002	6/1/2007	8/1/2007	9/1/2007	6/1/2007	6/1/2007	6/1/2007	6/1/2007	11/1/2007	11/1/2007	2002/1/11	11/1/2007
in state i	×	×	×	×	×	×	×	×	x	×	x	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	x	×
$S_{\rm eff} = 1S_{\rm eff}^{\rm eff}$	Retail	Retail	Retail	Retail	Retail	Hetail	Retail	Retail	Retail	Retail	Retail	Retail	Retai	Rotail	Rotail	Retair	Retail	Retail	Retail	Retal	Retait	Retal	Retail	Retail	Retail	Retall	Retail	Retail
the state	1749	1750			1753	1754	1755	1769		1761						1	1768	1769	r.				1774		1776	1777		1779

Page 9 of 18 Report Date: 10/2/2008 3:24:19 PM

$N^{1+1}N$	Ę	ŧ	đ	튭	튭	튭	Ē	€	ŧ	Ē	튭	튭	ħ	ŝ	ā	ā	â	튭	ŧ	훔	ŧ	훕	ā	똜	Ē	â	튭	ŧ	통	ę
S arong	6,195	104,645	14,425	9 9 9	88,200	100,503	19,120	25,970	53,934	66,522	90,013	3,050	65,043	60,675	24,707	66,307	20,879	113,710	33,725	64,530	35,500	93,672	97,022	327,972	4,981	23,428	369,900	1,220,000	1,220,000	106,500
Sile vite. Alger	īz	ΡA	МА	Ī	₹		M	NO M	KS	ð	8	£	ъ В	Ю	۲d	NC	NO	No O	Ş	کر ا	ĸ	¥I	٨	¥	¥	PA	×	ž	ř	
A H HA	26 South Jersey Gas Company (SJG)	1 Columbia Gas of Pennsylvania Inc.	5 Keyspan Energy Delivery (National Grid)	9 Consumers Energy Company	7 SEMCO ENERGY	3 Nicor Incorporated	t WE Energies	I Ameren Corporation	I Kansas Gas Service	1 Oktahoma Gas and Elecctric	1 Xoal Energy/Public Service Co. Colorado	I Columbia Gas of Ohlo Inc.	I Ohio Gas Company	1 VECTREN (Formerly Dayton Power & Light)	i PPt. Utilitias	Pleamont Netural Gas	z Enbridge Inc.	2 Union Gas	i Atmos Energy	1 Columbia Gas of Kentucky Inc	1 Louisville Gas and Electric	MidAmerican Energy Company	3 Orange and Rockland Utilities, Inc.	1 Columbia Gas of Kentucky Inc	1 Columbia Gas of Kentucky Inc	2 National Fuel Gas Distribution Corporati	1 Almos Energy	1 Austin Energy	1 Austin Energy	1 Wholesale
na tanan Barta	Q			-					-			·		-											-			-		
C. Cr. J. Catz	NYMEX	INYMEX		NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	XEMYYY	NYMEX			NYMEX	NYMEX	NYMEX	HENRY FUB		NYMEX	NYMEX		NYMEX	HOUSTON SHIP CHANNEL	WAHA	NYMEX
ant dur c	Indexed	Indexed	Fixed	Indexed	Indexed	Indexed	Indexed	Indexed	Indexed	Indexed	Indexed	Indexed	Indexed	Indexed	Indexed	Indexed	indexed	bexec	Indexed	Indexed	Indexed	Indexed	Fixed	indexed	Inclexed	Fixed	indexed	Indexed	Indexed	Fixed
Ant theor Stella	Expired	Active	Expired	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Canoelied	Expired	Active	Explred	Expired	Expired
$S_{1,2}(u_1) \in a$	11/2/2005	12/7/2006	12/11/2008	12/6/2006	12/6/2006	12/6/2006	12/6/2006				12/7/2008	12/7/2006	12/7/2006			12/8/2006			12/8/2006			2		5/11/2007	5/15/2007		Γ	6/23/2007	1	11
Domaid Vorme	23	F	2	œ-	φ	12	16	12	12	12	12	14	12	12	11	12	R F	12	ę,		11	12	18	54	12	ю	12	*	Ŧ	F
$\frac{1}{1+\frac{1}{2}} \sum_{i=1}^{n} \frac{1}{2} \sum_{i=1}^$	2007/08/11	10/1/2008	7/31/2007	10/1/2008	10/1/2008	10/1/2008	10/1/2008	10/1/2008	10/1/2008	10/1/2008	10/1/2008	10/1/2008	10/1/2008	10/1/2008	10/1/2008	10/1/2008	11/1/2008	11/1/2008	10/1/2008	10/1/2008	10/1/2008	10/1/2008	9/30/2008	5/31/2009	5/31/2008	12/31/2007	9/30/2008	1002/05/6	9/30/2007	4/30/2008
Tan meru Shur Di me	1/1/2006	11/1/2007	1/1/2007	4/1/2008	4/1/2008	10/1/2007	6/1/2007	10/1/2007	10/1/2007	10/1/2007	10/1/2007	8/1/2007	10/1/2007	10/1/2007	11/1/2007	10/1/2007	11/1/2007	11/1/2007	10/1/2007	11/1/2007	11/1/2007	10/1/2007	4/1/2007	6/1/2007	6/1/2007	2/1/2007	10/1/2007	6/1/2007	6/1/2007	10/1/2007
	×	×	×	×	×	×	×	×	×	×	×	×	×	x	×	×	×	×	×	×	×	×	×	×	×	×	×	×.	×	×
diam tan	Retall	Retail	Retail	Retail	Retail	Retail	Hetail	Retail	Retail	Ratai	Retail	Retall	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retal	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Wholesale	Wholesale	Wholesele
1997 - 1997 1997 - 1997	1783	1845	1846		1848			1851								1869			1962								1961			1972

Page 10 cf 18 Report Date: 10/2/2008 3:24:19 PM

£1	Ē	цн D	Dth	Dth	Ē	£	ŧ	튭	Dth	Dth	Oth	D	£	ŧ	Ð	둰	ŧ	Ð	đ	G	£a ⊡	fi D	đ	Ē	£	튭	ĥ	Ē	₫
Article -	266,156	91,704	354,264	6,562	192,224	236,333	404,388	16,432	17,638	26,458	122,060	83,378	26,471	27,598	167,281	201,177	63,957	159,834	113,575	40,083	8,934	3,444	583,946	136,800	2,240	51,262	527,131	24'812	210,089
3102 CO 3102 CO 3100 C	٨٨	PA	РA	A	λN	E	MA	QW	AV A	4	QМ	wv	<i></i>	MA	PA	Å	٩A	PA	ΡA	7	20	VA	MA	MA	MA	λN	A	٩	I=
$\tilde{E}_{0} = 0.05$	Consolidated Edison Company of New York	UGI Penn Natural Gas - PG Energy	2 Exelon (PECO Energy Company)	PPL Utilities	National Fuel Gas Distribution Corp.	Nicor Incorporaled	23 Bay State Gas Company	2 Washington Gas & Light Company	23 Washington Gas Light Company	TECO Peoples Gas	2 Washington Gas & Light Company	Mountaimeer Gas	Peoples Gas Light and Colos Company	Bay State Gas Company	Equitable Gas Company PA	3 Dominion Peoples	UGI Ulilities, Inc.	Columbia Gas of Permsylvania Inc.	National Fuel Gas Distribution Corporati	TECO Peoples Gas	Washington Gas Light Company	Washington Gas Light Company	16 Bay State Gas Company	Bay State Gas Company	Keyspan Energy Delivery (National Grid)	Consolidated Edison Company of New York	MidAmerican Energy Company	Michemenicen Energy Company	MidAmerican Energy
- 201 100 100 100 100 100	4	N	2	-	-	0	53	RI	53	-	2	i.	<u>, </u>	-		ю.	2	2	3	1	m	12	15	N	0	-	ю 	F	N
NAL DUT 4	ИҮМЕХ	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX									NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX							NYMEX	NYMEX	NYMEX
truk i button	Indexed	Indexed			Indexed	Indexed	Fixed	Fixed	Fixed	Fbæd	peoq.4	Fbœd	Ficed	Fixed	Indexed	Indexed				Indexed	Fixed	Fixed	pexits	Fixed	Fixed	Fixed	indexed	indexed	bexebul
lis viruat Natura	Cancelled	Active		Γ	Active	Expired	Active	Expired	Expired	Expired	Expired	Active	Expired	Active	Active	Active	Active	Active		Expired	Active		Adive	Active	Expired	Active	Active	Active	Active
میں بردہ میں۔ داروالیہ	6/8/2007	5/25/2007			6/12/2007	6/12/2007	6/12/2007	6/12/2007	6/12/2007	6/12/2007	6/12/2007	6/12/2007	6/12/2007	6/12/2007	6/26/2007	6/26/2007	6/26/2007	6/26/2007	6/26/2007	7/17/2007	7/17/2007	7/17/2007	2002/21/2	7/17/2007	7/30/2007	8/16/2007	B/16/2007	B/16/2007	B/(6/2007
n i krajnovanov Poženovajske	27	18	<u>8</u>	<u>8</u>	2	<u>ਦ</u> ਾ	2	<u>છ</u>	2	ñ	<u>5</u>	8	~	त्र	<u>e</u>	5	₹Ì	ম	₽	2	24	24	Ω.	4	ल	RI F	19	÷	2
in (Service a con- tailer	9/30/2009	12/31/2008	12/31/2008	12/31/2008	6/30/2009	6/30/2008	6002/06/6	5/31/2008	5/31/2008	8/31/2008	5/31/2008	1/31/2009	12/31/2007	6/30/2003	10/31/2008	10/31/2008	10/31/2008	10/31/2008	10/31/2008	8/31/2008	8/31/2009	B/31/2009	9/30/2008	9/30/2008	10/31/2007	9/30/2008	5/31/2009	5/31/2009	5/31/2009
Among Astron	7/1/2007	7/1/2007	7/1/2007	7/1/2007	7/1/2007	7/1/2007	10/1/2007	6/1/2007	8/1/2007	B/1/2007	6/1/2007	8/1/2007	6/1/2007	10/1/2007	11/1/2007	11/1/2007	11/1/2007	11/1/2002	11/1/2007	9/1/2007	9/1/2007	9/1/2007	10/1/2007	10/1/2007	8/1/2007	10/1/2007	11/1/2007	2/1/2008	11/1/2007
ann an th	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×
2.12 B 2.2	Retail	Retail	Hetail	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retai	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Reta	Retall	Retail	Retail
and the second	1973	1974 1		1976		1	1979	1981	1962	÷			1986	1987		Ł	<u>8</u>				2007		2012	2013	2019	2031	2032	r	2034

Page 11 of 18 Report Date: 10/2/2008 5:24:19 PM

19 - 11 H.	법	훕		f	đ	đ	ŧ	ŧ	ŧ	ŧ	튭	ŧ	Ê	Ē	튭	튭	튭	튭	훕	đ	Ē	ŧ	ŧ	Ē	Ē	Ē	Ē	튭	ŧ	₫	ŧ	ŧ	튭	Ð	Ē	ź	£	Oth
Summer by	12,357	87,078	796,62	91,175	97,622	155,000	382,898	12,394	145,000	150,000	460,000	155,000	155,000	155.000	155,000	140,000	156,000	1,825,000	66,885	123.960	41,190	61,400	24.475	171,394	29,383	91,205	75,467	44,690	B1,946	26,346	55,218	31,340	70,340	1,825,000	580,000	155,000	150,000	155,000
Folger 1 Pages	٣d	£	Q	Q	S	CA	ĩ	MA	8	5	ř	ð	5	\$	5	5	S	Ğ	W	5	2	ð	ş	ş	z	 <i>≈</i>	£	₹	8	M	ÿ	<u>ح</u>	≿	5	5	3	5	Q
1.1P	1 PPL Unitries	1 Duke Energy-Ohio	0 Washington Gas & Light Company	6 Washington Gas & Light Company	I Piedmont Natural Gas	1 Wholesale	2 South Jersey Gas Company (SJG)	2 Keyspan Energy Delivery (National Grid)	1 Wholesale	1 Wholesale	1 Wholesale	1 Wholesale	Wholesale	1 Wholesale	l Wholesale	i Whotesele	Wholesale	Whotesale	Consumers Energy	SEMCO ENERGY	Kansas Gas Service	1 Ottahoma Gas and Elemento	Ameren Corporation	e Missouri Gas Energy	1 Northern Indiana Public Service Company	Micor Incorporated	I Chio Gas Company	I VECTREN (Formerly Dayton Power & Light)	I Xcel Energy/Public Service Co. Colorado	2 PPL Utilities	Pleamont Natural Gas	I Almos Energy	l Columbia Gas of Kentucky Inc	I Wholesale	Wholesele	I Wholesale	Wholesele	(Wholesale
lanu - nu Miritan			÷				42						•					-	30																			-
500 A 0 A	NYMEX				NYMEX														NAMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX	NYMEX					
1906 - U. ³ 906	Indexed	Fixed	Fixed	Fixed	Indexed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Indexed	pexepul	Indexed	Indexed	pexepul	Indexed	Indexed	Indexed	Indexed	Indexed	Indexed	Indexed	Indexed	bexebul	Indexect	Fixed	Hyed	Hxed	Fixed	Fixed
The Number	Active	Active	Active	Active	Active	Expired	Active	Expired	Expired	Expired	Expired	Expired	Expired	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Expired	Expired	Expired	Expired
Sum ent	8/17/2007	7/10/2007	8/28/2007	8/28/2007	10/19/2007	11//8/2007	11/9/2007	10/26/2007	11/19/2007	11/19/2007	12/5/2007	12/5/2007			12/5/2007	12/5/2007		12/14/2007	12/14/2007	12/14/2007	1	12/14/2007	12/14/2007	12/14/2007	Γ.			12/12/2007					12/12/2007	1	12/20/2007		1 1	12/20/2007
Go et Marine	15	Å.	12	5 	মু	-	54	ന	-	-	8	1	1	-	1	1	1		মু	12		10	মূ				12	12	15		12		15	12	-		Ŧ	-
line very fan e Goeta	12/31/2008	12/31/2008	12/31/2008	12/31/2008	11/30/2008	7/31/2008	12/31/2009	7/31/2008	2/29/2008	6/30/2008	B/31/2008	7/31/2008	8/31/2009	12/31/2008	1/31/2009	2/28/2009	8/31/2009	12/31/2010	9/30/2009	8/30/2008	8/30/2008	8/30/2008	9/30/2009	9/30/2006	9/30/2006	8/30/2008	9/30/2009	9/30/2009	8/30/2008	8/30/2008	8/30/2008	9/30/2009	8/30/2008	12/31/2010	2/29/2009	3/31/2008	6/30/2008	7/31/2008
AUST - ANT	10/1/2007	8002/1/1	8002/1/1	1/1/2008	12/1/2007	11/1/2007	1/1/2008	11/1/2007	11/1/2007	11/1/2007	12/1/2007	12/1/2007	12/1/2007	12/1/2007	12/1/2007	12/1/2007	12/1/2007	12/1/2007	10/1/2008	10/1/2008	10/1/2008	10/1/2008	10/1/2008	10/1/2008	10/1/2008	10/1/2008	10/1/2008	10/1/2008	10/1/2008	10/1/2008	10/1/2008	10/1/2006	10/1/2008	12/1/2007	12/1/2007	12/1/2007	12/1/2007	12/1/2007
1811 A	x	×	×	×	×	×	x	x	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	x	x	×	×	×	×	×	×	×	×	×	x	×	×
Bust mas	Retail	Retail	Retait	Retail	Hetai	Wholesale	Retail	Retai	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	eale	Retal	Helail	Retail	Retall	Retail	Retail	Retait	Retail	Retail	Retail	Retail	Retal	Retal	Retail	Retail	Wholesale	Wholesale	Wholesate	Wholesale	Wholesale
100 million	2035					2083			2095								- 1	2126		Т		2130	2131	[2148	-

Page 12 of 18 Report Date: 10/2/2008 3:24:19 PM

NO. 40	튭	4 0	튑	둽	Ē	£	ŧ	ā	튭	튭	튭	臣	ā	£	đ	튭	đ	ŧ	흂	đ	뮲	휸	ā	뜅	đ	휸	ŧ	ŧ	đ	£	Ē	£	ŧ	Ź	튭		Ē	ŧ
Cuair 14	1,825,000	B1,825	310,000	300'000	1,825,000	1,825,000	1,825,000	1,625,000	1,825,000	155,000	460,000	310,000	1,825,000	1,825,000	280,000	1,825,000	1,825,000	2,750,000	1,825,000	1,825,000	19,637	133,547	232,006	10,386,000	249,476	35,000	63,095	9,120,000	239,118	1,825,000	167.735	1,826,802	2,386,274	1,825,000	1,070,000	000'006	755,000	3,246,500
5 - 1.5 - 1. S.R.R.	CA	۶A	S	с¥	сA	S	3	Ş	сA	5	сA	CA	CA	e S	ð	ð	CA	ð	cA	cA	٨d	¥	z	₹	Z	ž	ÅA	8	Ļ	5	5	CT	СТ	Ś	5	CA	8	8
and the	Wholesale	National Fuel Gas Distribution Corporati	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Wholesele	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	National Fuel Gas Distribution Corporati	5 DTE - MichCon	Northern Indiana Public Service Company	Columbia Gas of Ohio Inc.	SCINew Jersey Natural Gas (NJNG)	Keyspan Energy Delivery, NY	CenterPoint Energy ARKLA	Xcal Energy	Nicor Incorporated	Wholesele	Connecticut Natural Gas	Southern Connecticut Gas Co	Yankee Gas	Wholesale	Wholesale	W/holesale	Xoel Energy	Xcel Energy
ALL AND	1	2	-	1	1	1	•	-	-	-	÷	Ŧ	Ŧ	-	F	÷	-	÷	Ŧ	÷	Z	K)	ন্য	Ŧ	8	8	-	Ŧ	8	-	, N	463	377	1	1	1		Ē
A CHARLES AND A	NYMEX	NYMEX																				NYMEX	NYMEX		NYMEX	NYMEX	NYMEX	NNG Ventura	NYMEX	NYMEX		NYMEX	NYMEX				NYMEX	NYMEX
and the second	Fixed	paxepu	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Indexed	pexepu	Fixed	Indexed	Indexed	Indexed	Indexed	indexed	Fixed	Fixed	Indexed	Indexed	Fixed	Fixed	Fixed	pexepu)	Indexed
Domesic	Active	Active	Expired	Expired				Active	Active	Active	Active	Active			Active	Active	Active			Active	Expired	Active	Active	Active	Active	Active	Active	Active 1	Active	Active	Active							Active
Settian ent	1/3/2008	12/14/2007	1/8/2008	1/8/2008	1/8/2008	1/10/2008		1/17/2008	1/17/2008	1/17/2008	1/17/2008				1/24/2008		2/22/2008	2/22/2008		2/22/2008		2/20/2008	2/20/2008	3/10/2008	2/28/2008	3/12/2008	3/26/2008	4/11/2008	4/16/2008	5/2/2008	5/19/2008		5/21/2008	5/29/2008	5/29/2008			5/29/2008
اليامة إيران فيلومات ا	12	=	F	*	12	12	27 7	12	얻	-	e	-	12	5 2	-	5	12	a	12	¢1 ↓		ભ્ર	0	-	24	2	\$	₽	12	P	8	8	24	12	7	9	- 12	w.
Oekveru Eric Daar	12/31/2010	12/31/2008	5/31/2008	6/30/2008	12/31/2010	12/31/2010	12/31/2011	12/31/2010	12/31/2011	1/31/2009	6002/08/6	10/31/2009	12/31/2010	12/31/2011	2/28/2009	12/31/2010	12/31/2011	12/31/2008	12/31/2010	12/31/2011	1/31/2008	4/30/2009	10/30/2009	3/31/2009	3/31/2010	4/30/2009	4/30/2009	3/31/2009	6/30/2009	12/31/2011	10/31/2010	6/30/2011	0102/06/9	12/31/2011	10/31/2010	3/31/2011	3/31/2009	3/31/2009
lietweit Stor Date	1/1/2008	8002/1/2	1/1/2008	8002/1/1	1/1/2008	8002/1/1	1/1/2008	1/1/2008	1/1/2008	1/1/2008	1/1/2008	1/1/2008	8002/1/1	1/1/2008	1/1/2008	2/1/2008	2/1/2008	2/1/2008	2/1/2008	2/1/2008	8002/1/1	4/1/2008	4/1/2008	3/1/2008	4/1/2008	5/1/2008	5/1/2008	4/1/2008	7/1/2008	£/1/2008	6/1/2008	7/1/2008	8002/1/2	5/1/2008	5/1/2008	5/1/2008	5/1/2008	5/1/2008
Cast H	×	×	×	×	×	×	×	×	×	×	×	x	x	×	×	×	×	×	×	×	x	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×
53.0 × 51.0	Wholesale	Retail	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Whotesale	Wholesale		Wholesale	Wholesale	Wholessle	Wholesale	Wholesale	Wholesale	salə	Revail	Retail	Retail	Wholesele	Hetail	Retail	Retail	Wholesale	Retail	Wholesale	Retail	Retail	Retail	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale
ard vit	2150	2152	2153	2154		-		2163	2164			_		2169 \		1			ľ		2202		N N N N	2212		2218		5236		5244					2276		1	

Page 13 of 18 Report Date: 10/2/2008 3:24:19 PM

.

NOT AR	Ę	Бţ	다. 다.	đ	ā	£	튭	Ē	튭	Ъ.	튭	ца О	튭	튭	Dah	СШ,	Ē	Ð	튭	Ē	-	B	£	Ē	đ	4	-ta D	튭	ŧ	튭	đ
0.4.4.5	1,800,000	1,125,000	156,183	1.825.000	1,070,000	310,000	2,265,000	1,825,000	1,070,000	905,000	1,825,000	1,070,000	450,000	310,000	310,000	310,000	93,105	30,836	55,944	121,196	14,508	1,800,000	000'006	450.000	450,000	450,000	000'006	1,230,000	1,230,000	1,230,000	36,231
1,673 1,072	Q	8	MA	CA CA	CA	W	8	S	CA C	сA	CA	CA	¥0	₹ S	CA	сA	Đ	£	X	¥	СA	8	8	8	8	8	8	8	8	8	Z
2.12	Xcel Energy	Xcel Energy	Keyspan Energy Delivery (National Grid)	Wholesale	Wholesale	Consumers Energy Company	Xoal Energy	Wholesale	Wholesale	Wholesale	Wholesale	Wholesele	Wholesale	Wholesale	Wholesale	Wholesele	VECTREN (Formerly Dayton Power & Light)	Duke Energy-Ohio	Atmos Energy	Atmos Energy	Pacific Gas and Electric Company (PGE)	Xcel Energy/Public Service Co. Cotorado	Xoal Energy/Public Service Co. Colorado	Xoal Energy/Public Service Co. Colorado	Xoal Energy/Public Service Co. Colorado	Xcet Energy/Public Service Co. Colorado	Xoef Energy/Public Service Co. Colorado	Xcel Energy/Public Service Co. Colorado	Xcel Energy/Public Service Co. Colorado	Xcel Energy	Public Service Electric and Gas Company
118 Anit a constant	-	-		1	•		1	1	-	1	I	1	-	F	1	1	1	2	1	1	4	ţ	-	-	-	-	-	÷	-	-	4
Kaling and the	NYMEX	NYMEX				IFERC Trunkline LA	NYMEX	NYMEX	NYMEX	NYMEX																		IFERC CIG - Rocky Mountains	IFERC CIG - Rocky Mountains	IFERC CIG - Rodky Mountains	
$\mathcal{T}W_{i}=\{i,j\}, i\in \{n,j\}$	Indexed	Indexed	Fixed	Fixed	Fixed	Indexed	Indexed	Fixed	Fixed	Fixed	pex¦⊰	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Indexed	Indexed	Indexed	Fixed
agenes.	Active	Active	Active	Active	Active	Expired	Active	Active	Active	Active	Active	Active	Active	Expired	Active	Active	Active	Active	Active	Active	Active	Active	Adive	Active	Active	Active	Active	Active	Active	Active	Active
ácmo nem Jara	5/29/2008	5/29/2008	6/28/2008	5/28/2008	5/28/2008	6/11/2008	6/16/2008	6/18/2008	6/18/2008	6/18/2008	6/26/2008	6/26/2008	6/26/2008	6/26/2008	6/26/2008	6/26/2008	5/18/2008	7/1/2008	5/5/2008	6/5/2008	6/5/2008	7/3/2008	7/3/2008	7/3/2008	7/3/2008	7/3/2008	7/3/2008	6/24/2008	6/24/2008	6/24/2008	7/11/2008
a jaran (ja Bilan ja	6	ന	98	21	ĸ	-	цŢ	12	~	Ø	₹ Į	Ł	es.	1	-	-	9	9	24	24	٥	e	n	ю	æ	en -	(C)	•	•	4	12
i at the rig Free Lotae	2/28/2009	2/28/2009	9/30/2011	12/31/2011	10/31/2010	7/31/2008	3/31/2009	12/31/2011	10/31/2010	6/30/2011	12/31/2011	10/31/2010	2/28/2010	B/31/2008	10/31/2008	12/31/2008	12/31/2008	1/31/2009	6/30/2010	6/30/2010	12/31/2008	2/28/2009	2/28/2000	2/28/2009	2/28/2009	2/28/2009	2/28/2009	10/31/2008	10/31/2008	10/31/2008	7/31/2009
i van gewonen en de ser	5/1/2008	5/1/2008	10/1/2008	5/1/2008	5/1/2008	6/1/2008	6/1/2008	6/1/2008	6/1/2008	6/1/2008	6/1/2008	6/1/2008	6/1/2008	6/1/2008	6/1/2008	6/1/2008	7/1/2008	8/1/2008	7/1/2008	7/1/2008	7/1/2006	7/1/2008	7/1/2008	7/1/2008	7/1/2008	7/1/2008	7/1/2008	6/1/2008	6/1/2008	5/1/2008	8/1/2008
.∺	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	x
Dus reut	Wholesale	Wholesale	Retall	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale		-	Wholesale				Retall	Hetail	Retail		Retail	Wholesale	Wholesala	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Retell
1914 - 1914 1914 - 1914 1914 - 1914		2281		2295	2296	2306	2307	T	1													2350	2351	1	2353	1	2365	2356	2357	2358	2360

Page 14 of 18 Report Date: 10/2/2008 3:24:19 PM

ā 튭튭 ₫ £ £ £ 튪튭 튭 튪 ₫ සි සි සි å S ₫ E 55 튭 퉁 £ 훕 튪 ₹ ã đ 튭 ā 172,484 71,752 1,070,000 1,825,000 1,825,000 300,511 20,750 19,037 53,170 69,958 173, 443 583,946 1,070,000 259,973 26,297 198,572 145,399 200,972 136, BOO 171,472 11,629 169.636 283,310 1,825,000 62,084 209,935 123,088 87,192 37,038 354,022 391,138 555555 5 5 5 ş F 8 Z ¥ ¥ žΣŞ 9 ₹ ≚ ຟ ອິ GA ₹ <u>⊣</u> | -F ٨N Ā ⊵ 2 Dominion East Ohio 1 Dominion East Ohio 2 Columbia Gas of Ohio Inc. Coke Company Columbia Gas of Chio Inc. Company It Washington Gas & Light Company I Atmos Energy Corporation (Greeky Gas) Columbia Gas of Ohio Inc. Visconsin Gas Company 15 Bay State Gas Company Bay State Gas Company 1 VECTREN (Formerly Dayton Power & Light) 2 Citizens Gas and Coke Utility Roanoke Gas Company 5 Public Service Electric and Gas Company Ameren CIPS Peoples Gas Light and 2 Lactede Gas Company Washington Gas Light 1 Vectren Corporation 1 TECO Peoples Gas 5 Atlanta Gas Light Company Attenta Gas Light Company Contumbia Gas of Kentucky Inc I Delfa Natural Gas I SEMCO ENERGY SEMCO ENERGY Company, Inc. Wholesale Wholesale Wholesale Wholesale Wholesale Fixed Fixed Fixed Fixed Fixed Fixed **P** Fixed Fixed Fixed Fixed **Dex**i, (DBX) Fixed Dex: Fixed Pex!ixed **Dexi** Fixed Fixed Fixed X **Xex Dex** Fixed Active 7/31/2008 Active Active Aclive Active 7/31/2008 7/31/2008 7/31/2008 7/9/2008 7/23/2008 7/23/2008 7/31/2008 7/23/2008 7/23/2008 7/23/2008 7/11/2008 7/17/2008 7/23/2008 7/31/2008 7/31/2008 7/9/2008 7/23/2008 7/23/2008 7/31/2008 7/23/2008 7/23/2008 7/23/2008 7/31/2008 7/17/2008 7/23/2008 7/23/2008 7/23/2008 র জ 2 4 2 2 শ্ৰ হ ŝ ö Ņ 2 24 న 12 2 4 2 2 24 N. 2 10/31/2010 12/31/2010 12/31/2010 12/31/2011 12/31/2009 8/31/2010 9/30/2009 9/30/2010 9/30/2009 9/30/2009 8/31/2009 8/31/2010 10/1/2010 10/1/2009 10/1/2010 10/1/2010 10/1/2010 7/31/2010 9/30/2008 9/30/2009 8/31/2010 8/31/2009 8/31/2010 10/1/2010 8/31/2010 8/31/2010 10/1/2010 10/1/2010 9/30/2009 9/1/2009 8/31/2009 10/1/2008 10/1/2008 7/1/2006 7/1/2006 7/1/2006 7/1/2006 7/1/2008 10/1/2008 10/1/2008 10/1/2008 9/1/2008 9/1/2008 10/1/2008 10/1/2008 10/1/2008 10/1/2008 10/1/2008 10/1/2008 9/1/2008 9/1/2008 10/1/2008 10/1/2008 10/1/2008 9/1/2008 9/1/2008 0/1/2008 9/1/2008 9/1/2008 8/1/2008 Wholesale Wholesale Wholesale Wholesale Wholesale 2383 Retail 2384 Retail Retail Retail Retail 2390 Retall 2391 Retall Retal Retail Retail Retail Retail Retail 230B Retall Retail Retail Retail Retail 2397 Retail Retail 2377 |Retail Retail Retail Refeil Retail 2368 2369 2370 2371 2372 2373 2374 2376 2376 2361 2366 2367 2375 2399 2394 2363 2364 2376 2360 2302 2388 2392 2393 2381

Page 15 of 18 Report Date: 10/2/2008 3:24:19 PM

No. 12. 1	Pch Pth	ы Осн	f	£	튭	HE C	튭	툼	ā		ŝ	ф.	đ	£	흄	튤	£	L	흄	Ê	훕	đ	튭		Ē	Ē		흄	â	L	훕	훕
A CHARTER	258,874	85,202	611,230	117,016	290,372	77,397	530,919	79,200	1,825,000	1,825,000	1,825,000	15,049	20,374	60,524	22,771	70,000	25,000,000	1,830,000	1,825,000	1,510,000	1,800,000	800,008	1,800,008		50,640	136,782	243,340	34,778	39. 912	129,901	467,720	104,370
tearna a t Naistean	IM	۶	Ϋ́	Ϋ́	A.	2	<u>_</u>	8	5	Ş	5	ΡA	Ł	A	ş	£	£	ð	ð	8	8	8	8		Q	R	Ā	WA	P	ΥM	Q	ΡA
-141 - 1	3 Detroit Edison	1 Detroit Edison	Atmos Energy	Atmos Energy	Alliant Energy	i Peoples Gas Light and Coke Company	NIcor Incorporated	Xcel Energy/Public Service Co. Colorado	Wholesale	Wholesale	Wholesale	Equitable Gas Company PA	6 Dominion Peoples	Exelon (PECO Energy Company)	2 CenterPoint Energy Minnegasco	VECTREN (Formerly Dayton Power & Light)	Dominion East Ohio	Wholesale	Wholesate	Xcal Energy/Public Service Co. Colorado	Xosi Energy/Public Service Co. Colorado	Xoal Energy/Public Service Co. Colorado	Xoal Energy/Public Service Co. Colorado		6 Washington Gas & Light Company	PPL Utilities	UGI Utilities, Inc.	Cascade Natural Gas Company	Duke Energy-Ohio	Priget Sound	Washington Gas & Light Company	National Fuei Gas Dishibution Corrorati
Particular Aux on the			¢	63	-	5	35	-	-	-	-	2	8	~~ ·	2	F		1	-	÷	•	-	-	o	9			-			4	-
Solo School						NGI Chicago Citygate	NGI Chicago Citygate					NYMEX	NYMEX	нүмех						IFERC CIG - Rocky Mountains												
100 S. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1.	Fixed	Fixed	Fixed	Fixed	Fixed	Indexed	Indexed	Fixed	Pexid	Fixed	Fixed	Indexed	Indexed	Indexed	Fixed	Fixed	Fixed	Fixed	Fixed	Indexed	Fixed	Fbed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed
Comport Street	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Active	Draft	Active	Active	Active	Active	Active	Active	Active	Active
Soft in 1993 Anti-	7/31/2008	7/31/2008	7/31/2008	7/31/2008	7/31/2008	7/30/2008	7/30/2008	8/6/2008	7/31/2008	7/31/2008	7/31/2008	6/13/2008	8/13/2008	8/13/2008	8/21/2008	8/20/2008	7/22/2008	8/21/2008	8/21/2008	8/21/2008	8/21/2008	8/21/2008	8/21/2008	8/27/2008	9/9/2008	9/9/2008	9/9/2008	9/9/2008	9/9/2008	9/9/2008	9/9/2008	8/4/2008
Dornau Vaarbs	54	24	21	24	24	50	50	5	P	12	12	12	12	10 10	12		7	12	रा म	ज	ल	ਲ 	ro M	24	24	24	न्द्र	<u>ल</u>	12	12	24	24
D RUNG - 1 Print	10/1/2010	10/1/2010	10/1/2009	10/1/2010	10/1/2010	5/1/2010	5/1/2010	9/30/2010	12/31/2012	12/31/2010	12/31/2011	6002/06/6	8/30/2008	9/30/2009	9/30/2008	3/31/2010	3/31/2008	12/31/2012	12/31/2008	3/31/2009	5/28/2009	6002/82/2	2/28/2009	8/30/2010	11/30/2010	1/31/2011	1/31/2011	11/30/2009	12/31/2009	11/30/2009	10/31/2010	1/31/2011
For exercy North	10/1/2008	10/1/2008	10/1/2006	10/1/2008	10/1/2008	6/1/2008	9/1/2008	10/1/2008	2/1/2008	7/1/2008	7/1/2008	10/1/2008	8002/1/01	10/1/2008	10/1/2008	8002/1/9	7/1/2008	8/1/2008	8/1/2008	800Z/L/8	8002/1/8	8002/1/8	8/1/2008	10/1/2008	12/1/2008	2/1/2009	2/1/2009	12/1/2008	1/1/2008	12/1/2008	11/1/2008	6002/1/2
Cartan en	×	×	×	×	×	×	×	×	×		x	×	×	x	x	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×
Bran 125	Retail	Retail	Retail	Relai	Retail	Retail	Retail	Hetail	Wholesele	Wholesale	Wholesale	Retail	Retail	Ratail	Retal	Wholesale	Wholesele	Wholesele	Wholesale	Wholesale	Wholesale	Wholesale	Wholesale	Retall	Retail	Retail	Retail	Retail	Retail	Retal	Retail	Retail
		2400			2403			2408		2419			2437					2467						2475		2480			2483			2486

Page 15 of 18 Report Date: 10/2/2008 3:24:19 PM

ge 17 of 18 Report Date: 10/2/2008 3:24:19 PM

anti-anti-anti-anti-anti-anti-anti-anti-		Detection of the	Jerbere Biul Datu	Durthaer 57. tens	Sarimme Lite	J'ryn r <u>ai</u> t Prtettar	ecolul de la est	an u e jara 1967	na. As anns	5 40 2	Sum an Sum a	25 1 lw	
		÷	12/31/2009	t T	9/4/2008	Active	Fixed		~	VECTREN (Formerly Dayton Power & Light)	НО	213,752	Ē
		÷	1/31/2011	24	9/4/2008	Active	Fixed		-	Delmarva Power Company⊡	DE	117,840	ŧ
		7	1/31/2011	24	9/4/2008	Active	Fixed		-	Mountaineer Gas	Ŵ	104,682	₽
		1	/31/2011	24	9/4/2008	Active	Fixed		3	Mountaineer Gas	Ŵ	496,698	듐
		2	1/31/2010	12	9/4/2008	Adive	Fixed		-	Hope Gas Incorporated	Ŵ	65.337	đ
5002/1.C		2	1/3//2011	75 	900ZM;B	Acrive	Fixed		6 7	Bathimore Gas & Éfectric Company (BGE)	- CIW	538, 134	Ψ
X 2/1/2009 1/5		*	1/31/2011	24	9/4/2008	Active	Fixed		-	Columbia Gas of Maryland Inc.	Q	156,746	Ъ Ц
X 11/1/2008 10/		0	0/31/2011	38		Active	Indexed	NYMEX	en 	Keyspan Energy Delivery, NY	Ż	701,763	đ
X 11/1/2008 34	-	ð	3/31/2010	17	9/18/2008	Active	Indexed	NYMEX	60	Nitoor Incorporated	Ŧ	172,747	Dật
Wholesale X 8/1/2008 12/		12	12/31/2012	ដ	8/23/2008	Active	Fixed		Ŧ	Wholesate	8	1,830,000	Ę
		2	10/31/2009	10	[]	Active		NYMEX	8	PPL Utilities	٩d	9,347	D#P
X 11/1/2008 10/		10/	10/31/2009	12		Active		NYMEX	1	Dominion Peoples	ΡA	8,266	Dth D
		0	31/2009	12	8/23/2008	Active	pexepul	NYMEX	-	National Fuel Gas Distribution Corporati	PA	4,819	Ę
X 1/1/2010 12	┝	ğ	12/31/2010	ត្តា	8/28/2008	Active	Indexed	NYMEX	65	Netional Grid	H	1,098,725	đ
12/1/2008		11/2	10/2010	24	9/11/2008		Fixed		1	Public Service Electric and Gas Company	Ñ	162,520	Dth
		114	30/2010	54	10/8/2008	Active	El xed		1	Keyspan Energy Delivery. NY	٨٨	2,098	둼
		*	7/31/2011	54			Fixed		-	Washington Gas & Light Company	MD	56.470	Dth
54 6005/1/9 X		3	5/31/2014	8	10/8/2008	Active	Fixed		•	Washington Gas Light Company	20	5,195,000	Ę.
<u> </u>	<u> </u>	4	4/1/2010	<u>N</u>	10/9/2008	Active	Fixed			Avista Ulitties	WA	51,376	ŧ
		=	11/30/2009	12	9/26/2008	Active	Indexed	NYMEX	4	New Jersey Natural Gas (NJNG)	ī	25,821	ŧ
		11	11/30/2010	54	9/26/2008	Active	Indexed	NYMEX	9	6NUI Elizabethtown Gas-	ĥ	61,036	¥
		F	11/30/2009	12	9/26/2008	Active	Indexed	NYMEX	22	Public Service Electric and Gas Company	ſN	113,658	Ē
		12	12/31/2012	12	10/13/2008	Active	Fixed		-	Wholesale	CA	1,830,000	ц.
		10	10/31/2009	12	10/7/2008	Active	Fixed		2	NSTAR Electric and Gas Corporation	MA	62,995	đ
X 10/1/2008		÷	2/31/2010	1	10/31/2008	Active	Fixed		1	Wholesale	CA	155.000	сці О
		ത	1102/06/	e	10/31/2008	Active	Fixed		-	Wholesale	сA	460,000	đ
-	-	÷	10/31/2011		10/31/2008	Active	Fixed		-	Wholesale	3	155,000	4 0
X 10/1/2008		Ē	12/31/2011		10/31/2008	Active	Fixed			Wholesale	Ş	155,000	튭
		L I	12/31/2012	12	10/31/2008	Active	bex!≍		1	Wholesale	CA	1,830,000	u o
6002/1/6 X			9/1/2010	ţ.	11/5/2008	Active	Fixed		-	Columbia Gas of Ohio Inc.	но	157,022	£ 0
		-	12/31/2011	24	11/5/2008	Active	Fixed		-	VECTREN (Formerty Deyton Power & Light)	융	422,280	
3 8002/1/01 X	ŀ		9/30/2011	24	11/6/2008	Active	Fixed		0	2 SEMCO ENERGY	IW	357,500	ŝ

	Dth	Oth	Đ	Ę	Ē	ā	훕	Dth	Dth	다. D	Ę.	Сф Оф	튭	Dật Dât	đ	튭	Ē	Сф.
	115,002	300,222	129,901	34,778	114,742	64,170	171,790	100,468	97,867	298,470	49,565	94,921	25,797	91,626	10,387	419,870	225,724	1,830,000
serence and and and a	IE I	Ю	WA	WA	¥	CA	8	8	но	٨٨	MT	AA	Я	Ā	Ā	λ	ΡĐ	CA
	1 Ameren CiPS	2 Dominion East Ohio	1 Puget Sound	1 Cascade Natural Gas Company	1 Unisource (formly.Citizens Arizona Gas)	4 Pacific Gas and Electric Company (PGE)	1 Xcel Energy/Public Service Co. Colorado	1 Xcel Energy/Public Service Co. Colorado	1 Duke Energy-Chio	1 Bath Electric, Gas & Weter Systems	1 Northwestern Energy	1 CenterPoint Energy ARKLA	1 TECO Peoples Gas	1 MidAmerican Energy Company	1 MidAmentcan Energy Company	3 Columbia Gas of Kentucky Inc	2 Atlanta Gas Light Company	1 Whotesale
vo. du																		
ma, farter	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed	Hxed	Fixed	Fixed	Fixed	Fixed	Fixed	Fixed
	Active	Active	Active	Active	Active	Active	Active	Active	Adive	Active	Active	Draft	Active	Active	Active	Active	Active	Adtive
	11/6/2008	11/6/2008	11/6/2008	11/6/2008	10/23/2008 Active	10/23/2008 Active	10/23/2008	10/23/2008	10/23/2008	10/23/2008 Adive	10/23/2008	11/17/2008	11/18/2008	11/19/2008 Active	11/19/2008	11/19/2008	11/19/2008	11/26/2008
therman No mer l	24	12	42	د ا ا	24	24	<u>R</u>	24	5	96	N.	21	12	<u>8</u>	<u>N</u>	র	54	<u>5</u>
	8/31/2011	8/30/2010	11/30/2010	11/30/2010	11/30/2010	12/31/2010	5/31/2010	5/31/2011	12/31/2010	12/31/2011	12/31/2009	4/30/2010	8/31/2010	10/31/2010	10/31/2010	8/31/2011	1/31/2011	12/31/2012
	601/2008	10/1/2009	12/1/2009	12/1/2009	12/1/2008	1/1/2009	6/1/2009	6/1/2008	1/1/2010	1/1/2009	1/1/2009	12/1/2008	8/1/2008	11/1/2009	11/1/2009	6002/1/6	2/1/2008	11/1/2008
	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×
	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retail	Retail		Retail		Retail	Retail	Retail	Retail	Retail	Wholesale
	2583 F	2584 F	2585 F	2586 F	2587 F	2588 F	2580	2590	2691		2693 F	2610 F	2611 F	2612	2613 F	2614	2615	2619

Page 18 of 18 Report Date: 10/2/2008 3:24:19 PM



Exhibit D-3 "Key Technical Personnel"

Daniel P. Pugh, CEP – Dan Pugh is Vice President of Supply Management for World Energy's commercial and industrial business. He is responsible for all new and existing supplier relationships, customer-specific energy purchasing and daily operational processes to manage all supply contracts. Dan has 15 years of experience in the energy industry, providing natural gas marketing and management services for large end users. Dan's experience includes major account procurement, contracting, transportation, risk management and storage arbitrage. Dan was an Energy Trader for AEP Energy Services for two years and prior to that was Market Coordinator with Enron Energy Services. Dan graduated Magna Cum Laude with a Bachelor's degree in Business Administration from The Ohio State University. (614) 790-0400 dpugh@worldenergy.com

Dave Laipple – Dave Laipple is Vice President of Sales and Operations for World Energy's commercial and industrial business. He is responsible for leading World Energy's national sales efforts, as well as targeting and developing strategic partnerships. Dave has more than 24 years of sales and sales management experience and has spent the past 15 years in the energy industry. Prior to joining World Energy, Dave served in several regional and national sales leadership roles with Enron and more recently served as VP of Sales for Accent Energy. Dave holds a BA in Business Administration from Capital University and an MBA from the University of Dayton. (614) 790-0400 diaipple@worldenergy.com

Sean Zurbrick, CEP – Sean Zurbrick is Supply Director within World Energy's Supply Management group. He is responsible for supplier offer coordination along with daily procurement and operational processes for consumers. In addition, Sean has a working knowledge of database structure and programming. Prior to World Energy, Sean spent two years at FSG Energy Services and 5 years at Clinton Energy Services as a Senior Natural Gas Analyst. In these positions he was responsible for procurement, scheduling, tariff analysis, purchasing and trading pipeline capacity, least cost routing analysis and storage and capacity management. Sean holds a Bachelor's degree from The Ohio State University. (614) 790-0400 szubrick@worldenergy.com

Jack J Shumaker, CEP – Jack Shumaker is a Supply Director within World Energy's Supply Management group and has nearly a decade of experience in supply-side energy management services. He began his career with Clinton Energy, a division of Enron Energy Services. At Clinton he was responsible for various purchasing and operational processes. Later he moved to Ashland Energy (Delta) where he was responsible for managing the purchasing and operational processes for numerous large industrial energy users across the U.S. In early 2004 Jack left Delta Energy to join EnergyGateway, which was acquired by World Energy in 2007. Jack earned a Bachelor's degree from The Ohio State University and his MBA from Franklin University. (614) 790-0400 jshumaker@worldenergy.com

Bill Yeager, CEP – Bill Yeager is a Supply Director within World Energy's Supply Management group and has more than 22 years experience in the energy industry, with focused knowledge on risk management, procurement, pipeline and utility logistics, tariffs and customer service. At World Energy, Bill's responsibilities include daily procurement, supplier offer coordination and operational processes for consumers. Bill works with customers to explain the effects of energy prices and the ways to get out in front of the risks associated with purchasing energy. Prior to World Energy, Yeager worked for FirstEnergy Solutions American Electric Power Energy Services Enron Energy Services, and Ohio Gas Marketing. Bill holds a Bachelor of Science in Agricultural Economics from The Ohio State University. (614) 790-0400 byeager@worldenergy.com

Roberto Martinez, CFA – Rob Martinez is a Supply Director within World Energy's Supply Management group. He has over ten years experience in the energy industry. He began his career with Enron Corporation in 1998. Rob was in the Analyst and Associate programs and held positions in various groups including Enron Capital and Trade and Enron Energy Services Power Structuring. In early 2002, Rob joined the Mergers and Acquisitions group at Marathon Oil Corporation. Rob worked on buying and selling upstream oil and gas assets and companies. In late 2005, Rob joined Accent Energy as Power and Gas Structuring Manager. Rob priced and structured Mid-Market Power and Gas deals. Rob joined World Energy in December of 2008. Rob holds a Bachelor of Business Administration degree from the University of Notre Dame Mendoza College of Business.

(614) 790-0400 rmartinez@worldenergy.com

Rob T. Barkley – Rob Barkley is Vice President of Strategic Accounts and Customer Relations. He is responsible for developing and maintaining relationships with large strategic energy consumers throughout the country. He is a veteran of the industry with over 20 years of experience in the retail energy business. As a pioneer in the deregulated natural gas business, Rob started his career with Yankee Resources, Inc. followed by Access Energy where he was a principal. Rob led sales teams in both commercial and industrial natural gas markets. In 1992, Access Energy was sold to Enron when Rob joined its Energy Services group and created integrated solutions for large energy users. Rob left Enron in 1999 and pursued sales opportunities with technology firms before returning to the energy industry at World Energy. Rob holds a Bachelor's degree from University of Michigan and a MBA from The Ohio State University.

(614) 790-0400 rbarkley@worldenergy.com